Forests, Trees and the Eradication of Poverty: Potential and Limitations

A Global Assessment Report

Editors: Daniel C. Miller, Stephanie Mansourian and Christoph Wildburger







IUFRO World Series Vol. 39



Forests, Trees and the Eradication of Poverty: Potential and Limitations

A Global Assessment Report

Editors: Daniel C. Miller, Stephanie Mansourian and Christoph Wildburger

Funding support for this publication was provided by the German Federal Ministry for Economic Cooperation and Development, the Ministry for Foreign Affairs of Finland, the United States Forest Service and the Austrian Federal Ministry of Agriculture, Regions and Tourism. The views expressed within this publication do not necessarily reflect the official policy of the governments represented by these institutions/agencies or the institutions to whom the authors are affiliated.

Publisher:

International Union of Forest Research Organizations (IUFRO)

Recommended catalogue entry:

Daniel C. Miller, Stephanie Mansourian & Christoph Wildburger (eds.), 2020. Forests, Trees and the Eradication of Poverty: Potential and Limitations. A Global Assessment Report. IUFRO World Series Volume 39. Vienna. 240 p.

ISBN 978-3-903345-06-5 ISSN 1016-3263

Published by: International Union of Forest Research Organizations (IUFRO)

Available from:

IUFRO Headquarters Secretariat Marxergasse 2 1030 Vienna Austria

Tel +43 (0)1 877 01 51-0 **E-mail** office@iufro.org **www.iufro.org**

Language editor: Stephanie Mansourian Layout: Eugénie Hadinoto Cover photographs: Jennifer Zavaleta Cheek, Nelson Grima, Reem Hajjar Printed in Austria by Eigner Druck, Tullner Straße 311, 3040 Neulengbach, Austria

Preface

Since its establishment in the year 2007, the Global Forest Expert Panels (GFEP) initiative of the Collaborative Partnership on Forests (CPF) has been effectively linking scientific knowledge with political decision-making on forests. GFEP responds directly to key forest-related policy questions by consolidating available scientific knowledge and expertise on these questions at a global level. It provides decision-makers with the most relevant, objective and accurate information, and thus makes an essential contribution to increasing the quality and effectiveness of international forest governance.

This report, titled "Forests, Trees and the Eradication of Poverty: Potential and Limitations", presents the results of the seventh global scientific assessment undertaken so far within the framework of GFEP. All assessment reports are prepared by internationally recognised scientists from a variety of biophysical and social science disciplines. The publications are presented to stakeholders across relevant international policy fora. In this way, GFEP supports a more coherent policy dialogue about the role of forests in addressing the broader environmental, social and economic challenges reflected in the United Nations Sustainable Development Goals (SDGs).

Poverty is one of the greatest challenges facing humanity. Globally, one out of every 10 people lives in extreme poverty. Poverty eradication has therefore found a place at the top of the United Nations 2030 Agenda for Sustainable Development. "End poverty in all its forms everywhere" is the first Goal of this Agenda, which is supported by all 193 UN member states. The international community is now stepping up efforts to achieve this goal, especially in response to the severe setback caused by the Covid-19 pandemic. The zoonotic nature of the Covid-19 virus has also illustrated the urgency to reduce human pressure on nature.

One way to relieve this pressure and alleviate poverty is to recognize and further optimize the critical role of forests and trees as allies in the fight against poverty. This report consolidates available scientific evidence on the wide range of contributions which forests and trees outside forests make to curbing poverty, and on the effectiveness of diverse forest management policies, programs, technologies and strategies. It does so based on an understanding of poverty not only in terms of monetary values, but also as an obstacle that keeps people from attaining a certain level of well-being and participating fully in society.

Poor and vulnerable people often depend on the use of natural resources and, in many regions, they are able to harness forest goods and services to manage and mitigate risk, especially in the face of crises. It is therefore essential to take into account the role that forests play in development in general, and in achieving poverty eradication, in particular. Scientific reports like the one in hand are important tools for supporting policy-makers and stakeholders in their ambition to ensure sustainable development and to advance the implementation of the United Nations 2030 Agenda.

I would like to thank the Chair of the Global Forest Expert Panel on Forests and Poverty, Daniel C. Miller, GFEP Coordinator Christoph Wildburger, GFEP Editor Stephanie Mansourian, and the GFEP Team for their excellent work in guiding the assessment process and in leading the development of this publication. It is my sincere hope that those with a responsibility for implementing the SDGs at all levels will find this report, and its accompanying policy brief, a useful source of information and inspiration.

Alexander Buck IUFRO Executive Director

Acknowledgements

This publication is the product of the collaborative work of scientific experts within the framework of the Global Forest Expert Panel (GFEP) assessment on Forests and Poverty, who served in different capacities as panel members and authors. We express our sincere gratitude to all of them:

Arun Agrawal, Jennifer Alix-Garcia, Maria Brockhaus, Sarah E. Brown, James T. Erbaugh, Dikshya Devkota, Mónica Gabay, Reem Hajjar, Karl Hughes, David Humphreys, Markus Ihalainen, Pamela A. Jagger, Gillian Kabwe, Judith Kamoto, P.P. Krishnapriya, Patricia M. Kristjanson, Patrice Bigombe Logo, Samuel Mawutor, Blaise-Pascal Ntirumenyerwa Mihigo, Doris N. Mutta, Peter Newton, Johan A. Oldekop, Symphorien Ongolo, Pablo Pacheco, Nitin D. Rai, Laura Vang Rasmussen, Onja H. Razafindratsima, Casey Ryan, Laura Aileen Sauls, George Schoneveld, Priya Shyamsundar, Erin O. Sills, Conghe Song, Dietmar Stoian, Kira A. Sullivan-Wiley, Terry Sunderland, Joleen Timko, Jianmin Xiao. Jennifer Zavaleta Cheek and Yue Zhao.

Without their efforts and commitments, the preparation of this publication would not have been possible. We are also grateful to the institutions and organisations to which the authors are affiliated for enabling them to contribute their expertise to this assessment. At the same time, we wish to note that the views expressed within this publication do not necessarily reflect the official policy of these institutions and organisations.

We acknowledge, and also sincerely thank, the reviewers of the report, whose comments have greatly improved the quality of this publication: Sofia Ahlroth, William L. Ascher, David Conte, Julia P. G. Jones, Thais Linhares-Juvenal, Adrian Martin, Alexandre Meybeck, Sheona Shackleton, Carsten Smith-Hall, Andrew Taber and Bhaskar Vira. We furthermore thank Anneli Cers, Carly Hopkings and Katia Nakamura for their support in preparation of the report.

We also gratefully acknowledge the generous financial and in-kind support provided by the German Federal Ministry for Economic Cooperation and Development, the Ministry for Foreign Affairs of Finland, the United States Forest Service and the Austrian Federal Ministry of Agriculture, Regions and Tourism.

Our special thanks go to the IUFRO Secretariat for providing indispensable administrative and technical support. Furthermore, we would like to thank the member organisations of the Collaborative Partnership on Forests (CPF) for providing their in-kind contributions and overall guidance to the assessment. We are particularly grateful to the Food and Agriculture Organization of the United Nations (Rome, Italy), the University of Michigan (Ann Arbor, USA) and the Center for International Forestry Research (CIFOR) and World Agroforestry (ICRAF) (Nairobi, Kenya) for hosting expert meetings.

Daniel C. Miller Chair

Christoph Wildburger GFEP Coordinator

Stephanie Mansourian Content Editor

Contents

Pre	eface
Ac	knowledgements
Lis	t of Acronyms and Abbreviations 10
1.	Introduction: Forests, Trees and Poverty
2.	Key Concepts for Understanding Forest-Poverty Dynamics
3.	Forest-Poverty Dynamics: Current State of Knowledge
4.	Contextual Factors Shaping Forest-Poverty Dynamics
5.	Levers for Alleviating Poverty in Forests and Tree-Based Systems
6.	Global Forces of Change: Implications for Alleviating Poverty 177 and Sustaining Forests 177 Coordinating lead author: Priya Shyamsundar 120 Lead authors: Jennifer Zavaleta Cheek, Laura Vang Rasmussen, Daniel C. Miller and Johan A. Oldekop 101 Contributing authors: Laura Aileen Sauls, Kira A. Sullivan-Wiley, James T. Erbaugh and P.P. Krishnapriya
7.	Conclusion: Forests, Trees and the Eradication of Poverty
Ap	pendices to the report

List of Acronyms and Abbreviations

ACOFOP	Association of Forest Communities
	of Petén
AI	Artificial Intelligence
CBD	UN Convention on Biological Diversity
CCB	Climate, Community and Biodiversity
CCP	Company-Community Partnership
CFE	Community Forest Enterprise
CFM	Community Forest Management
CFUG	Community Forest User Group
CGIAR	Consultative Group on International
	Agricultural Research
CIFOR	Center for International Forestry
	Research
CIS	Commonwealth of Independent States
CLIP	Community Life Improvement
	Programme
CPF	Collaborative Partnership on Forests
DHS	Demographic and Health Survey
DRC	Democratic Republic of the Congo
ECOSOC	United Nations Economic and Social
	Council
EID	Emerging Infectious Disease
ES	Ecosystem Service
EU	European Union
EUTR	European Union Timber Regulation
FAO	Food and Agriculture Organization of the
	United Nations
FLEGT	Forest Law Enforcement, Governance and
	Trade
FLR	Forest Landscape Restoration
FLU	Forest and other Land Use
FMNR	Farmer Managed Natural
	Regeneration
FOE	Friends Of the Earth
FPIC	Free, Prior and Informed Consent
FPO	Forest Producer Organisation
FSC	Forest Stewardship Council
FTA	Forests, Trees and Agroforestry
GDP	Gross Domestic Product
GEF	Global Environment Facility
GFEP	Global Forest Expert Panel
GIIN	Global Impact Investing Network
GIS	Geographic Information System
GNI	Gross National Income
ha	Hectare
HIC	High-Income Country
HIV	Human Immunodeficiency Virus
HLPE	High Level Panel of Experts on Food
	Security and Nutrition

IFRI	International Forest Resources
	and Institutions
Int\$	International dollar
IFAD	International Fund for Agricultural
	Development
ILO	International Labour Organization
IPBES	Intergovernmental Panel on Biodiversity
	and Ecosystem Services
IPCC	Intergovernmental Panel on Climate
	Change
ITTO	International Tropical Timber
	Organization
IUCN	International Union for Conservation
	of Nature
IUFRO	International Union of Forest
	Research Organizations
LAPSSET	Lamu Port and Lamu-Southern
L/11 00L 1	Sudan-Ethionia Transport Corridor
ICF	Locally Controlled Forestry
IFR	Log Export Ban
	Low and Middle Income Country
ICMC	Living Standards Massurement
LOWIO	
	Surveys
LOLOCI	Land Ose, Land-Ose Change,
N 617 A	Millennium Freewotern Assessment
MLA	Millennium Ecosystem Assessment
MPI	Multiannensional Poverty Index
MRL	Mountain-River-Lake
	Metric formes
NGMM	National Commission for Museums
	and Monuments
NGO	Non-Governmental Organization
NTFP	Non-Timber Forest Product
ODA	Overseas Development Assistance
OECD	Organisation for Economic
	Cooperation and Development
PA	Protected Area
PEFC	Programme for the Endorsement of Forest
	Certification
PEN	Poverty Environment Network
PES	Payments for Ecosystem Services
PPF	Plan Piloto Forestal (Forestry Pilot Plan)
PPP	Purchasing Power Parity
PRIME	Productivity, Rights, complementary
	Investments, Markets and Ecosystem
	services
PROFOR	Program on Forests
RAFI	The Rural Advancement Foundation

International

RCT	Randomised Control Trial
REDD+	Reducing Emissions from Deforestation
	and forest Degradation, and fostering
	conservation, sustainable management of
	forests and enhancement of forest carbon
	stocks
RRI	Rights and Resources Initiative
RSPO	Roundtable on Sustainable Palm Oil
SAN	Sustainable Agriculture Network
SDG	Sustainable Development Goal
SER	Society for Ecological Restoration
SLCP	Sloping Land Conversion Programme
SME	Small and Medium Enterprise
SMFE	Small and Medium Forest Enterprise
SVLK	Sistem Verificasi Legalitas Kayu
	(Indonesia's national timber legality
	assurance system)
SWIFT	Survey of Well-being via Instant and
	Frequent Tracking
TNC	The Nature Conservancy
UAV	Unmanned Aerial Vehicle
UN	United Nations
UNCCD	United Nations Convention to
	Combat Desertification
UNDP	United Nations Development
	Programme
UNEP	United Nations Environment
	Programme
UNFCCC	United Nations Framework
	Convention on Climate Change
UNFF	United Nations Forum on Forests
USD	United States Dollars
VCS	Voluntary Carbon Standard
VPA	Voluntary Partnership Agreement
WEF	World Economic Forum
WHO	World Health Organization
WTTC	World Travel & Tourism Council
WWII	Second World War
WWF	World Wide Fund for Nature
ZDC	Zero Deforestation Commitment



Chapter 1

Introduction: Forests, Trees and Poverty

Coordinating Lead Author: Daniel C. Miller Lead Authors: Jennifer Zavaleta Cheek, Peter Newton, Christoph Wildburger, Terry Sunderland and Pamela A. Jagger

TABLE OF CONTENTS

1.1. Problem Statement: Can forests and tree-based systems contribute to poverty alleviation?	14
1.2. Conceptualising Forests, Trees and Poverty	17
1.3. Policy Context for this Report	19
1.4. Assessment Objectives, Scope and Approach	23
1.5. An Overview of Forest-Poverty Linkages	23
1.6. Structure of this Report	25
1.7. References	27

1.1 Problem Statement: Can forests and tree-based systems contribute to poverty alleviation?

Poverty¹ is one of the greatest challenges facing humanity. The poorest around the world often go hungry, lack adequate shelter, suffer from poor health, live with the threat of violence, and worry about their immediate future and that of their families. Despite major reductions in recent decades (McArthur and Rasmussen, 2018; World Bank, 2018), poverty remains stubbornly persistent. Globally, one out of every 10 people lives in extreme poverty, defined as less than 1.90 international dollars (Int\$)² per day (World Bank, 2019). About two-thirds of the world's population still lives on less than Int\$ 10 per day (Figure 1.1), the threshold (or poverty line) associated with a more permanent move out of poverty (Edward and Sumner, 2019). Poverty has not spared individuals in even some of the world's wealthiest countries, with the poverty rate exceeding 10% in many European countries and 15% in the United States (OECD, 2020a). Across the globe, millions more people make a living at levels just above national and international poverty lines and are at significant risk of moving into poverty (Krishna, 2011; World Bank, 2018). Poverty disproportionately affects marginalised groups and has stubbornly persisted across generations (UNDP, 2019).

Poverty has often been perceived using a monetary lens. More refined understandings of poverty emerged in the 1990s that consider the multiple dimensions making up human well-being, including health, safety, food and education, amongst others. These multidimensional definitions of poverty understand it as a state of deprivation or disadvantage that prevents an individual or group from attaining a certain level of well-being and participating fully in society (World Bank, 2001; Smeeding, 2016). Such definitions encompass not only commonly used income or consumption measures of poverty but also a range of non-monetary attributes that directly affect people's capabilities and overall well-being and allow human capabilities to go unrealised (Sen, 1993; 1999; Alkire, 2002; World Bank, 2018).

Given the persistence of poverty in many corners of the world, addressing this problem stands

as one of the most urgent global priorities. Poverty eradication has therefore found a place at the top of the United Nations 2030 Agenda for Sustainable Development: "End poverty in all its forms everywhere" is the first of the 17 Sustainable Development Goals (SDGs) agreed upon by all 193 Member States of the United Nations (UN, 2015). This SDG includes five main targets covering many aspects of poverty, from a focus on extreme poverty measured in monetary terms to nationally-determined definitions of multi-dimensional poverty (Box 1.1). The natural environment is embedded explicitly in two of these targets, which concern the rights to land and natural resources (target 4) and the resilience of the poor in the face of different kinds of shocks and disasters (target 5).

Poor and vulnerable populations often rely heavily on natural resources and *ecosystem services* to support their livelihoods, both for subsistence and income generation (Sunderlin *et al.*, 2005; Angelsen *et al.* 2014). Evidence shows that *forests and tree-based systems* can support rural livelihoods, have a buffer function in maintaining livelihoods, and represent natural insurance (Wunder *et al.*, 2014; Rasmussen *et al.*, 2017). A large body of literature also specifically examines the role that forests (Agrawal *et al.*, 2018; Cheng *et al.*, 2019; Miller and Hajjar, 2020) and trees outside forests (Waldron *et al.*, 2017; Miller *et al.*, 2020) can play in *poverty alleviation*.



An indigenous Kichwa family make a 4-hour journey on the Payamino River to see a doctor in the town of Coca, Ecuadorian Amazon

Photo © Johan Oldekop

¹ Throughout this assessment report, all terms that are defined in the glossary are introduced for the first time in a chapter using italics.

² An 'international dollar' is a hypothetical currency used to enable comparisons across country contexts. This currency could buy in a given country a comparable amount of goods and services that a US dollar would buy in the United States. In other words, it has the same purchasing power parity that the US dollars has at a given point in time.



Note: Poverty thresholds in 'international dollars' (Int\$) at constant 2011 purchasing power parity (PPP) prices to account for inflation and cross-country differences in price levels. Data: World Bank, 2019.

Forests and trees are important assets in virtually all countries of the world, including those where forests cover vast tracts of land like Brazil, Gabon, Papua New Guinea and Russia, and countries where forests are sparse but trees are scattered across portions of the landscape such as Mongolia, Niger and Yemen (Hansen et al., 2013; Zomer et al., 2016). Worldwide, more than a billion people (often referred to as forest-reliant people), many living below the international poverty line, derive direct and indirect benefits from forests (Angelsen et al., 2014; FAO, 2014). These benefits include forest-related employment and income, use of timber and non-timber forest products, and a wide range of ecosystem services. Recent figures suggest that the forest sector contributes at least USD 539 billion directly to the world Gross Domestic Product (GDP), with a total contribution (accounting for direct, indirect and induced effects) of some USD 1.2 trillion (Li et al., 2019). For comparison, this amount is roughly the equivalent of the GDP of Australia, the world's fourteenth largest economy (World Bank, 2020a). These figures are likely significant underestimates given that they do not capture inputs

from the informal use of forests nor many other values of forests (e.g. provision of ecosystem services to support agriculture and nature tourism, and recreation related to forests).

Trees outside forests also provide important benefits to millions of people around the world. In 2010, some 43% of all agricultural land globally had at least 10% tree cover, and this percentage has increased over the last decade (Zomer *et al.*, 2016). Agroforestry – the integration of trees or other woody perennials with crops or livestock in production systems – can offer subsistence and income opportunities (Garrity *et al.*, 2010; Miller *et al.*, 2017), enhance the productivity and resilience of agricultural lands (Reed *et al.*, 2017; Blaser *et al.*, 2018; Amadu *et al.*, 2020; Hughes *et al.*, 2020) and improve food and *nutrition security* (Vira *et al.*, 2015; HLPE, 2017; Rosenstock *et al.*, 2019), among other benefits.

Box 1.1

Targets for Sustainable Development Goal 1: "End poverty in all its forms everywhere"

- **1.1** Eradicate extreme poverty for all people, everywhere.
- **1.2** Reduce at least by half the proportion of all people living in poverty in all its dimensions, by national definitions.
- **1.3** Implement nationally appropriate social protection systems and measures for all with substantial coverage for the poor and vulnerable.
- **1.4** Ensure that all people, particularly the poor and the vulnerable, have equal rights to economic resources, and access to basic services, ownership and control over land and other forms of property, inheritance, natural resources, appropriate new technology and financial services, including microfinance.
- **1.5** Build the resilience of the poor and those in vulnerable situations and reduce their exposure and vulnerability to climate-related extreme events and other economic, social and environmental shocks and disasters.

Source: UN, 2015.

But what do we know about the potential for forests and tree-based systems to contribute to achieving SDG 1? This is the central question that this report seeks to answer. The report synthesises and presents available scientific evidence on the role of forests and tree-based systems to alleviate and, ultimately, eradicate poverty. The scope is global, covering countries classified as low- and middle-income (LMIC) and high income (HIC) by the World Bank (2020b). Although the literature on the relationship between forests and tree-based systems and poverty has grown markedly over the past two decades (Cheng et al., 2019), important knowledge gaps remain. This report summarises current understanding while highlighting areas where further research is needed. In so doing, we seek to describe how forests and trees outside forests have contributed to poverty alleviation - and might do so in the future - and make these connections more visible for decision-makers across the globe.

This global assessment of forests and poverty comes at a critical time. The implementation period for the SDGs is well underway, with less than a decade remaining to reach the goals and targets set (UN, 2015). Even as poverty remains a widespread problem in many parts of the world, significant progress has been made to reduce poverty over the past several decades (World Bank 2018; Figure 1.1). With a growing middle class in many countries, more people are economically prosperous than ever before in human history. However, coronavirus disease 2019 (COVID-19) spread across the globe in 2020, causing not only major health problems and hundreds of thousands of deaths but also a massive economic slowdown that threatens to further impoverish millions of people around the world (OECD, 2020b; Sommer et al., 2020). The UN estimates that at least 71 million people will have been pushed into extreme poverty in 2020 (UN, 2020b). The pandemic has also exacerbated the yawning gap between the richest and the poorest globally and within countries (UN, 2020a; Collins et al., 2020) For these and other reasons, COVID-19 as well as potential other infectious diseases in the future pose a major threat to progress towards the SDGs (Di Marco et al., 2020).

As questions about global prosperity mount, so do pressures on the world's forests and ecosystems that sustain life on earth. For example, 14 of the 18 categories of contributions identified by the Intergovernmental Panel on Biodiversity and Ecosystem Services (IPBES) that nature makes to human well-being – including a range of regulating, material, and non-material contributions - have declined over the past three decades (IPBES, 2019). Many species have gone extinct during this period and a further one million species are estimated to face extinction within decades (IPBES, 2019). In a recent study, not even one out of 150 countries was able to meet the basic needs for its citizens at a globally sustainable level of resource use (O'Neill et al., 2018).



Forested landscape in Malawi Photo © Jennifer Zavaleta Cheek

Although the rate of deforestation has slowed in the decade since 2010, net deforestation has increased across much of the tropics, leading to a net reduction in tree cover of 8% (~9,500,000 ha) in the tropical dry forest and 2% (~8,400,000 ha) in the tropical moist deciduous forest biomes since 1982 (Song et al., 2018). Intact, old-growth forests have seen major declines across the globe (Potapov et al., 2017; Watson et al., 2018), with losses especially acute in biodiversity 'hotspots' in Australia, Brazil, Central America, Madagascar, Southeast Asia and West Africa (Hill et al., 2019). Forests have returned in a number of (mainly temperate) regions, but the world's forest estate is, on average, younger with stands having a faster turnover as old-growth forests, characterised by stable dynamics, decline (McDowell et al., 2020). These trends have meant that the international community will fall well short of the goal of halving tropical deforestation by 2020 and faces a steep challenge in ending it by 2030, as agreed in the 2014 New York Declaration on Forests.

Major drivers of deforestation include the production of commodities (soy, cattle, palm oil), logging, shifting agriculture and fire (Curtis et al., 2018), each of which is affected or exacerbated by climate change. In fact, climate change not only jeopardises gains made in addressing global poverty but also threatens some of the world's poorest and most vulnerable people whose basic necessities and livelihoods often derive directly from forests and tree-based systems (Althor et al., 2016; Hallegatte and Rozenberg, 2017). Human-induced changes to the earth's climate, including increases in frequency and intensity of extreme events, such as floods and fires, have already adversely affected food security and terrestrial ecosystems in many regions (IPCC, 2019). Climate change threatens both the total area covered by forests and forest integrity (Trumbore et al., 2015) which, in turn, imperils the livelihoods of forest-dependent people (Newton et al., 2016). Given that forests are also key to mitigating climate change through avoided deforestation and carbon sequestration, policies that promote sustainable forest management, conservation and restoration while recognising the rights and stewardship of forest-dependent people, promise to help advance multiple sustainable development objectives (Seymour and Busch, 2016; Mansourian, 2018; Katila et al., 2019). However, such policies will have to navigate potential trade-offs between climate mitigation and forest-based livelihoods.

Together, climate change, biodiversity loss and other environmental issues push planetary boundaries (Rockström *et al.*, 2009; Steffen *et al.*, 2015) and threaten the sustainability of progress made to alleviate poverty globally. They also challenge efforts to ensure the social foundations of sustainability are sound, including extending gains made to populations that remain marginalised in countries around the world (Raworth, 2017). In this context, forests take on particular importance for not only expanding human well-being to reduce poverty and bring more widespread *prosperity* but to do so in a way that is sustained over time.



Children in Takamanda, Cameroon, peeling the seed of njansang (Ricinodendron heudelotii), a spice commonly used in local cuisine Photo © Terry Sunderland

1.2 Conceptualising Forests, Trees and Poverty

1.2.1 Forests and tree-based systems

This report examines the relationship of both forests and trees outside of forests to poverty alleviation. Throughout the report we use the term forests and tree-based systems, which encompasses a wide array of different types of land use involving trees along a forest transition curve (Figure 1.2), from intact old growth forests to planted forests, to agroforestry systems, and single species tree crop production. We include different types of forests from tropical dry and wet forests to temperate and boreal forests. Mangroves fall within our definition of forests but oil palm does not as it is considered an agricultural crop (FAO, 2014). We highlight that we do not assume teleological forest change from pristine primary forest to land dominated by agriculture, but seek evidence on the dynamics of diverse forests and tree-based systems and poverty alleviation in different parts of the world.



1.2.2 Poverty

Poverty has multiple dimensions, but it is often measured as material income or consumption and in terms of being below a given income threshold. For example, the World Bank defines poverty at USD 1.90 per day in 2011 purchasing power parity (World Bank, 2018). Income levels associated with absolute poverty are often used to measure poverty because they are quantifiable and allow for comparison across different geographical contexts. This approach has limitations, however. Income levels are indirect measurements of poverty, which assume that if people have a high enough income, they will be able to provide for basic needs. However, differences in consumption patterns and prices based on a given individual's demographic characteristics and location pose a major challenge to identifying a precise, universal poverty line (Alkire and Santos, 2014). Moreover, the experience of poverty is more than having a low level of consumption or income; it is better understood by the inability to satisfy a set of basic needs, rights or functionings.

The *capabilities* approach developed by Sen (1999) and Nussbaum (2000) provides an alternative conception of poverty where wealth is understood as the expansion of citizens' capabilities and ability to pursue what they value. In this conception, wealth is only useful in so far as it translates into a reduction of suffering and increase in substantive freedoms like economic opportunity, political choice, security and health. Since wealth

should not be an end in and of itself, another, more direct way to assess poverty is to measure the indicators associated with meeting minimum standards related to core human functionings. One example of doing so is the Multidimensional Poverty Index (MPI), which attempts to assess the magnitude of poverty in a comparable way across countries by directly measuring three dimensions of poverty: education, health and living standards (Alkire and Santos, 2014). Use of this indicator implies that the share of the world's poor is about 50% higher than when monetary measures for extreme poverty are used, suggesting that poverty is an even more widespread and deeply entrenched global problem than previously thought (World Bank, 2018). Research based on the perspectives of people actually experiencing poverty (Bray et al., 2020) extends the dimensions of poverty (to include, e.g. disempowerment, suffering in body, mind and heart, and institutional maltreatment) and further suggests the expansive and stubborn reach of poverty.

Several other conceptual issues relating to poverty merit brief discussion here to orient the reader to the remainder of this report. First is the distinction between *absolute* and *relative pover*ty. Most countries use an absolute threshold to measure their poverty in which a fixed amount of money needed to meet basic needs such as food, clothing and shelter is specified. Some countries use a relative threshold that defines poverty in relation to the economic status of other members of society using a cutoff point. For example, the European Union typically defines the poor as those whose per capita income falls below 50% of the median (Haughton and Khandker, 2009).

Second, both absolute and relative measures of poverty can be used to describe the incidence of poverty – the number of people who are poor in a given context. However, these 'headcount' approaches are silent on the intensity of poverty (how poor the people below the poverty line are) and on inequalities among the poor and between the poor and the rich (Agrawal and Redford, 2006). The poverty gap index – an estimate of how far on average the poor are from the poverty line – and the Gini coefficient – a measure of income or wealth distribution – are two common approaches used to shed light on poverty intensity and *inequality*.

Third, poverty also has an important temporal dimension. The literature distinguishes between chronic (or persistent) poverty and transient poverty (Haughton and Khandker, 2009). Those who are chronically poor experience deprivation over long periods of time, even their entire lives, and often pass poverty along to their children. The transitory poor are those who are sometimes in poverty but at other times are able to move out of it such that, on average, they are not classified as poor.

Fourth, there are a number of ways to conceptualise a lessening of poverty. Poverty alleviation refers to a lessening of deprivation or disadvantage such that well-being is improved. This lessening may include movement above a certain income or consumption threshold, such as international or country-specific poverty lines (termed poverty reduction or poverty elimination). It may also include a lessening in the degree of poverty experienced or avoiding falling into poverty (termed poverty mitigation) (World Bank, 2001; Sunderlin et al., 2005). The term 'poverty reduction' can also be used to refer to a lessening of the number of people who are considered to be poor based on measures of other dimensions of poverty (World Bank, 2001). This term is frequently used in national poverty strategies and is often understood as a situation in which the poverty rate falls, more or less permanently, based on economic growth (World Bank, 2018). Poverty eradication refers to the complete or near absence of people or households under the international poverty line in a given context. It implies permanent movement out of poverty by addressing the root causes of why people are impoverished (UN, 2020a). 'Poverty alleviation' is the most encompassing of the foregoing terms and, for this reason, it is the main one used in this report when refering to a lessening of poverty. Other terms, notably 'poverty eradication', are used when more specific meaning is required.

Finally, we emphasize that understanding and acting to address poverty requires a relational perspective that recognises and addresses heterogeneity across different social groups (Beck et al., 2020). Particular emphasis must be placed on addressing the needs and aspirations of vulnerable groups, including children, youth, persons with disabilities, people living with HIV, older persons, indigenous peoples, minority populations, refugees, internally displaced persons and migrants (UN, 2016). It is important to note that individuals can be part of several different groups, which can present multiple disadvantages in terms of the assets and capabilities that a person has and the services and opportunities they can access. For example, a poor elderly woman from an indigenous group may face more marginalisation than someone else that has one or fewer of her traits. Additionally, inequality can exist across geospatial scales, including within the urban to rural spectrum as well as across sub-regions within a country and between countries. (Kochendorfer-Lucius and Pleskovic, 2006).

1.3 Policy Context for this Report

The role of forests in sustainable land use approaches that balance poverty alleviation with other management goals is important for the implementation of existing international commitments.

In addition to the 2030 Agenda and the SDGs, the international policy framework related to sustainable development, poverty alleviation and forests is largely based on Agenda 21 and the three Rio Conventions – the UN Framework Convention on Climate Change (UNFCCC), the Convention on Biological Diversity (CBD) and the UN Convention to Combat Desertification (UNCCD) – adopted following the 1992 Earth Summit. The relationship between forests and poverty also intersects with the implementation of the UN Declaration on the Rights of Indigenous Peoples and the ILO Indigenous and Tribal Peoples Convention (1989) No. 169.

The UN Strategic Plan for Forests (ECOSOC Resolution 2017/4) was agreed as part of the follow-up process to Agenda 21 in the UN Forum on Forests (UNFF). The second of the Plan's six goals – "enhance forest-based economic, social and environmental benefits, including by improving the livelihoods of forest-dependent people" – is especially relevant to the subject of this report. Notably, it includes target 2.1 that "extreme poverty for all forest-dependent people is eradicated" (UN, 2017).

The Convention on Biological Diversity represents another international instrument relevant for this assessment. In 2010, the signatories to the CBD adopted a Strategic Plan for Biodiversity 2011-2020, which included the 20 Aichi Biodiversity Targets. Targets 5 and 7 explicitly refer to forests and aim at halting deforestation and sustainably managing forests. Target 2 addresses the need to integrate biodiversity values in poverty reduction strategies (UNEP, 2010). Negotiations for a global Post-2020 Biodiversity Framework are underway, with a 'Zero Draft' version released in early 2020 and a final version expected to be adopted in late 2020 or early 2021. The contribution that forests and biodiversity make to people, as per the IPBES report, is likely to be central to the Post-2020 Biodiversity Framework.

The parties to the UNFCCC responded to the significant role of forests in *climate change mitigation and adaptation* with decisions on land use, land-use change and forestry activities (LULUCF). These included establishing the REDD+ scheme (Reducing Emissions from Deforestation and Degradation and the role of conservation, sustainable management of forests and enhancement of forest carbon stocks in developing countries). Within the framework of the Paris Agreement (2015) the role of forests for sustainable development was further emphasised.

The UNCCD 2018-2030 Strategic Framework aims at restoring the productivity of degraded lands, improving the livelihoods of people, and reducing the impacts of drought on vulnerable populations (UNCCD, 2017). National Action Programmes are the key instruments for realising these goals, and parties to the Convention are requested to include measures in them to conserve natural resources, such as the sustainable management of forests (Wildburger, 2009). The UNCCD emphasises 'land degradation neutrality' as a pathway to sustainable development, within which forests (particularly forest restoration and rehabilitation) play a substantial role.

All of these international commitments are interrelated and form an important basis for the SDGs and their implementation. The SDGs themselves are rooted in the eight Millennium Development Goals (MDGs), which were adopted in the Millennium Declaration in 2000 as the international framework for development policy until 2015. Under this earlier agenda, countries committed to reduce extreme poverty and made major progress toward that goal (Figure 1.1; McArthur and Rasmussen, 2018). The SDG framework has now broadened the scope for addressing poverty to cover most aspects of economic, social and environmental sustainability (Timko et al., 2018; Katila et al., 2019). This framework is also global as opposed to the MDGs, which were mainly relevant for LMICs.

The 17 SDGs and their associated 169 targets cover a wide range of social, economic and environmental issues, and address essential global challenges, including poverty alleviation, food security and nutrition, and the protection, restoration and sustainable use of forests. Given its focus on poverty, SDG 1 is most relevant to this report. Two SDGs explicitly address forests: SDG 15 (Life on Land) aims at the protection, restoration and sustainable use of terrestrial ecosystems and halting the loss of biodiversity, and refers to forests in several targets; SDG 6 (Clean Water and Sanitation) calls for the protection and restoration of forests in one of its targets. However, forests and tree-based systems can be directly or indirectly linked to each of the SDGs (Figure 1.3). A range of scientific assessments have explored the different contributions forests and trees can make to realising the SDGs (Vira et al., 2015; Seymour and Busch, 2016; Creed and van Noordwijk, 2018) while related work has considered how efforts to achieve the SDGs may affect forests and people who rely on them (Katila et al., 2019).



Nepali woman cutting firewood Photo © Conghe Song



Source: Landscapes for People, Food and Nature Initiative, 2015.

Governmental policies to alleviate poverty tend to focus on agriculture, infrastructure and cash transfers while neglecting the roles of forests. This approach is likely to undermine efforts to restore, sustainably manage and conserve forests as well as to sustain gains made in alleviating poverty. At the same time, government actions to protect forests have excluded the poor in many cases (Peluso, 1992; Adams and Hutton, 2007) and people living in or near forests are often politically weak such that they are ignored in the development of relevant policies (Larson and Ribot, 2007).

Effective implementation of the 2030 Agenda for Sustainable Development requires prioritising such marginalised communities and the development of cross-sectoral measures that take into account the role of forests and trees in poverty alleviation (Stafford-Smith *et al.*, 2017; Timko *et al.*, 2018; van Noordwijk et al., 2018). Such a nexus approach holds promise for improved outcomes for people and forests, although in practice it faces political economy challenges, including the historic separation of key government agencies relating to agriculture, forestry and wildlife (Gibson, 1999; Larson and Ribot, 2007; Miller *et al.*, 2017). Growing consensus among policymakers, the business community and the public about the importance and interlinked nature of environment and development issues (Fisher *et al.*, 2020; WEF 2020) may enable the collective action necessary to overcome such challenges.

The outbreak of the COVID-19 global pandemic and the rise of diverse social movements to tackle issues like climate change and racial discrimination are also shaping the international policy arena. Box 1.2 introduces these phenomena, briefly discusses their implications for forests and treebased systems and poverty, and highlights ways in which they are covered in this report.

COVID-19, social movements and forest-poverty dynamics

The coronavirus outbreak (COVID-19) in 2020 has become the first global pandemic in more than a century. At the same time, major social movements seeking urgent action on climate change and to address systemic racism have also gained momentum in countries around the world. These two contemporary phenomena have potentially profound implications for understanding and action relating to forests, trees and poverty.

The COVID-19 pandemic is likely to affect forest-poverty dynamics in multiple ways (FAO, 2020; Wunder, 2020). First, illness and death within households are a major cause of poverty (Krishna, 2011) and this is the most direct way in which COVID-19 is likely to exacerbate poverty worldwide. Many thousands have lingering health effects due to the disease. Second, the COVID-19 pandemic has triggered the most severe global recession in nearly a century, with the global economy expected to contract by more than 6% in 2020 (OECD, 2020b). Current estimates suggest COVID-19 will push up to 71 million people into extreme poverty, and income-based poverty is expected to increase for the first time since 1990 (Sumner et al., 2020; UN, 2020b). Third, unemployment and concerns about contagion in cities is leading to migration to rural areas in many countries, which is beginning to have major environmental impacts as land will be required for agriculture and other uses (Rondeau et al., 2020). Migration, together with more limited enforcement in forest areas, is leading to deforestation in many countries, with Brazil a particularly well-documented case (Londoño et al., 2020). Forest destruction, in turn, increases the likelihood that viruses and other pathogens will transfer from wild animals to humans, in a vicious zoonotic circle (Bloomfield et al., 2020). Illegal activity in forests has also risen with reduced access for enforcement agencies, and governments' attention turned to addressing the pandemic and its immediate impacts on society. Forest exploitation (both legal and illegal) at the expense of local communities dependent on them, is also likely to increase as powerful actors (both public and private) seek to expand their income base. Finally, development aid is likely to be reduced as governments spend substantial amounts propping up their own economies.

Concurrent with the COVID-19 pandemic is increasing social pressure across the globe to address climate change and other environmental issues and to reckon with racism and discrimination. The student-led climate movement is reshaping politics, policies and investments in many countries (Marris, 2019) while a major international social movement has arisen to address institutionalised racism across many sectors. The forestry and environmental field is no exception (e.g., Merchant, 2003; Taylor, 2016; Baker et al., 2019). These movements are spurring reflection on how forest conservation and management might become more inclusive and just, notably, by directly grappling with legacies of colonialism and dispossession that have disproportionately affected indigenous people and people of colour.

The COVID-19 pandemic and current pro-environment, anti-racist social movements are distinct, but they have intertwined to create both enormous uncertainty and considerable opportunity. Each is already having far-reaching impacts that bear on forest-poverty dynamics and possibilities. The possibility of transformational change to more equitable, just and sustainable ways of steering society and the environment is now on the table in a way it has not been before. Detailed discussion of these two issues is beyond the scope of this report, but we do touch on both of them in several chapters. For example, Chapter 3 considers social differences, including class, gender and race³, in the poverty effects of forests and trees. Chapter 6 discusses COVID-19 and human infectious disease, more broadly, as critical trends bearing on forests, trees, and poverty alleviation. Chapter 7 offers concluding reflections on the implications of the current pandemic and movements against systematic racism for the findings and policy recommendations of this report.

³ This term is used here as per the Universal Declaration of Human Rights.

1.4 Assessment Objectives, Scope and Approach

To better understand and recognise the contributions of forests and tree-based systems to the overall 2030 Agenda goals of poverty eradication and sustainable development, government representatives and other policymakers need reliable, policy-relevant scientific information. In turn, this information can help decision-makers identify synergies and navigate potential trade-offs concerning poverty alleviation and forests, trees and land use worldwide.

Recognising this need, IUFRO on behalf of the Collaborative Partnership on Forests (CPF) tasked the Global Forest Expert Panel (GFEP) on Forests and Poverty to carry out a comprehensive global assessment of available scientific evidence about the interactions between forests and poverty. The results are compiled in this report, which seeks to inform relevant international policy processes related to the 2030 Agenda for Sustainable Development, particularly SDG 1.

The overall ambition of this report is to make the contributions of forests and tree-based systems to poverty alleviation more visible while highlighting key equity issues relating to the distribution of the costs and benefits of forest-related interventions. In so doing, we seek to provide a strong and accessible scientific basis for more effective decision-making by policymakers, donors, practitioners and other relevant stakeholders. We recognise that forests and tree-based systems have a wide range of benefits beyond their potential role in poverty alleviation, including biodiversity conservation and reducing greenhouse gas emissions. However, the explicit focus of this report is to understand the extent to which conservation and management of forests and tree-based systems can contribute to poverty alleviation while also maintaining their biophysical benefits. We explore the considerable variation that exists within our broad definition of forests and tree-based systems with regard to this challenge.

The scope of this report is global. We collate and analyse evidence on the relationships between forests and tree-based systems and poverty across a wide range of economic and agro-ecological and economic contexts. This report considers poverty in relation to forests and trees in any country where such resources are prevalent nationally or in particular subnational regions. However, we devote particular attention to LMICs as they are home to the majority of the world's poorest people, many of whom live in forests and tree-based landscapes.

This scientific assessment has been carried out by the members of the GFEP on Forests and Poverty, twenty-one scientists from diverse geographical and cultural backgrounds with established expertise on this topic. More than twenty contributing authors also supported the development of this study. In reviewing the evidence, we have relied primarily on peer-reviewed literature. We have also used relevant published work from a variety of organisations recognised as reliable sources of data, particularly UN institutions. This assessment has revealed important gaps in knowledge on the relationship between forests, trees and poverty, and we identify these, while also pointing out where the evidence base is relatively strong. We also discuss the level of consistency in the literature, and address the degree of certainty regarding conclusions that can be drawn from the evidence.



Villagers who rely on nearby forests take a rest to meet researchers in Rhamechhap District, Nepal Photo © Johan Oldekop

1.5 An Overview of Forest-Poverty Linkages

Approximately 737 million people were estimated to be living in extreme poverty in 2015 (World Bank, 2019). Most of these people live in Africa and Asia (Figure 1.4), with half in just five countries: Bangladesh, Democratic Republic of the Congo,

Figure 1.4



Data: World Bank, 2019

Ethiopia, India and Nigeria (World Bank, 2018). Forests comprise a vital resource in each of these countries. The Democratic Republic of the Congo and India are among the top ten countries in terms of forested area globally, while Bangladesh is home to the largest mangrove forest in the world (FAO and UNEP, 2020). Forest dependence is high and coincides with poverty in different areas of each country. In India, for example, about 27% of the population, or 275 million people, were found to depend on forest resources for subsistence and income generation (Milne, 2006) and, for the poorest in some states, forests provide about 30% of total income, an amount greater than that from agriculture (Damania *et al.*, 2020).

Globally, around 40% of the extreme rural poor – some 250 million people – are estimated to live in forest and savannah areas (FAO and UNEP, 2020). However, further research is needed to confirm and update these numbers. Recent work mapping 'forest proximate people' worldwide (Newton *et al.*, 2020; Figure 1.5) provides information on the possible universe of people who reside in or near forests, many of whom may be poor. Further, research suggests that rural areas with the world's poorest often have high tree cover and high levels of biodiversity (Fisher and Christopher, 2007; Sachs *et al.*, 2009). Country-scale studies also shed important light on patterns of forest-poverty overlap (e.g., Sunderlin *et al.*, 2008), but these have not yet been scaled up to provide a comprehensive global portrait. We know that millions of people rely on trees on farms for their livelihoods around the world (Garrity *et al.*, 2010; Miller *et al.*, 2017) and data are available on tree cover on agricultural land globally (Zomer *et al.*, 2016). However, detailed global estimates of the imbrication of poverty with trees on farms remains elusive.

The relationship between forests, trees and poverty is dynamic, not just spatially but also temporally and contextually. Chapter 2 presents an in-depth analysis of these dynamics and their implications for poverty alleviation, particularly the specific targets articulated under SDG 1. Here, we underscore that different kinds of forests and tree-based systems may have different capacity to address poverty at different times – and that they may have different effects for different social



Source: Newton et al., 2020

groups. This capacity will depend on a series of social, economic, political and biophysical factors (see Chapter 4). A comprehensive assessment of the potential of forests and tree-based systems to alleviate poverty must attend to these dynamics.

For example, given that being lifted out of and descending into poverty may entail different dynamics (Krishna, 2011), forests and trees may play different roles for those seeking a *pathway* out of *poverty* compared to those who may be at risk of becoming poor due to forest degradation or destruction. Without attending to such distinctions, forests risk being seen as inferior to other forms of rural poverty reduction strategies that imply forest conversion to agriculture or other more extractive approaches, which may deliver results in the near term, but have less certain long-term impacts. More holistic analysis of the role of forests and tree-based systems in poverty alleviation over time, across space, and in different biophysical and socio-economic contexts is critical for enhancing synergies between sustainability goals and effectively addressing trade-offs.

1.6 Structure of this Report

This report comprises seven chapters, including this introduction. Figure 1.6 presents a conceptual overview of how the different chapters fit together to form a comprehensive global assessment of current knowledge on the relationships between forests, trees and poverty.

Chapter 2 reviews existing poverty frameworks, discusses the specific targets and indicators under SDG 1, and presents the framing for how we analyse forest-poverty dynamics and human well-being throughout this report. Chapter 3 synthesises current knowledge on the role of forests and treebased systems in affecting poverty alleviation and the well-being of the poor. It considers forest con-



tributions to income, broader human well-being and to reducing risk, and compiles evidence on the socially differentiated effects of conservation, management, and restoration of forests and treebased systems.

Chapter 4 examines a set of key social, economic, political and environmental factors affecting forests and tree-based systems and their ability to alleviate poverty. Chapter 5 identifies and describes a set of 21 potential levers (e.g. policies, programmes, technologies, strategies) used in the forest sector that could plausibly reduce poverty. The chapter then reviews evidence on the effect that each lever has had on alleviating poverty. Chapter 6 identifies six major global trends that are expected to affect poverty in forests and treebased systems. The chapter concludes by discussing how the enabling conditions described in Chapter 4 and policy levers reviewed in Chapter 5 may interact with these global changes to shape the future of forests, trees and poverty. Chapter 7 concludes by summarising the key messages of this report for decision-makers, including those in local and national governments, intergovernmental organisations, the business sector, civil society groups and the research community.

1.7 References

Adams, W. M. and Hutton, J., 2007. People, parks and poverty: political ecology and biodiversity conservation. *Conservation and Society* 5(2): 147-183.

Agrawal, A. and Redford, K.H., 2006. Poverty, development, and biodiversity conservation: Shooting in the dark? New York: Wildlife Conservation Society.

Agrawal, A., Hajjar, R., Liao, C., Rasmussen, L. V. and Watkins, C., 2018. Editorial overview: Forest governance interventions for sustainability through information, incentives, and institutions. *Current Opinion in Environmental Sustainability* 32: A1-A7.

Alkire, S., 2002. Dimensions of Human Development. World Development 30: 181-205.

Alkire, S. and Santos, M. E. 2014. Measuring Acute Poverty in the Developing World: Robustness and Scope of the Multidimensional Poverty Index. World Development 59: 251-274.

Althor, G., Watson, J. E. and Fuller, R. A., 2016. Global mismatch between greenhouse gas emissions and the burden of climate change. *Scientific reports* 6: 20281.

Amadu, F.O., Miller, D.C. and McNamara, P.E., 2020. Agroforestry as a pathway to agricultural yield impacts in climate-smart agriculture investments: Evidence from southern Malawi. *Ecological Economics* 167: 106443.

Angelsen, A., Jagger, P., Babigumira, R., Belcher, B.,
Hogarth, N. J., Bauch, S., Rner, J. B. and Wunder, S.,
2014. Environmental Income and Rural Livelihoods:
A Global-Comparative Analysis. World Development
64: S12-S28.

Baker, K., Eichhorn, M.P. and Griffiths, M. 2019. Decolonizing field ecology. *Biotropica* 51: 288–292.

Beck, V., Hahn, H. and Lepenies, R., 2020. Interdisciplinary Perspectives on Poverty Measurement, Epistemic Injustices, and Social Activism.
In: Dimensions of Poverty, edited by V. Beck, H. Hahn and R. Lepenies. Cham: Springer Nature.

Blaser, W.J., Oppong, J., Hart, S.P., Landolt, J., Yeboah, E. and Six, J., 2018. Climate-smart sustainable agriculture in low-to-intermediate shade agroforests. Nature Sustainability, 1(5), pp.234-239.

Bloomfield, L.S.P., McIntosh, T.L. and Lambin, E.F., 2020. Habitat fragmentation, livelihood behaviors, and contact between people and nonhuman primates in Africa. *Landscape Ecol* 35: 985–1000.

Bray, R., de Laat, M., Godinot, X., Ugarteg, A. and Walker, R., 2020. Realising poverty in all its dimensions: A six-country participatory study. World Development 134: 105025.

Cheng, S.H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Rana, P., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. *Environmental Evidence* 8:3. Collins, C., Ocampo, O. and Paslaski, S., 2020. Billionaire Bonanza 2020: Wealth Windfalls, Tumbling Taxes, and Pandemic Profiteers. Washington, D.C.: Institute for Policy Studies.

Creed, I. F. and Van Noordwijk, M., 2018. Forest and Water on a Changing Planet: Vulnerability, Adaptation and Governance Opportunities: A Global Assessment Report. Vienna: International Union of Forest Research Organizations (IUFRO).

Curtis, P.G., Slay, C.M., Harris, N.L., Tyukavina, A. and Hansen, M.C., 2018. Classifying drivers of global forest loss. 361(6407): 1108-1111.

Damania, R., Joshi, A. and Russ, J., 2020. India's forests– Stepping stone or millstone for the poor? World Development 125: 104451.

Di Marco, M., Baker, M.L., Daszak, P., De Barro, P., Eskew, E.A., Godde, C.M., Harwood, T.D., *et al.*, 2020. Opinion: Sustainable development must account for pandemic risk. PNAS 117(8): 3888-3892.

Edward, P. and Sumner, A., 2019. The End of Poverty: Inequality and Growth in Global Perspective. Cham: Springer Nature.

FAO, 2014. State of the World's Forests 2014: Enchancing the socioeconomic benefits from forests. Rome: FAO.

FAO, 2020. The impacts of COVID-19 on the forest sector: How to respond? Rome: FAO.

FAO and UNEP, 2020. The State of the World's Forests 2020. Forests, biodiversity and people. Rome: FAO.

Fisher, B. and Christopher, T., 2007. Poverty and biodiversity: measuring the overlap of human poverty and the biodiversity hotspots. *Ecological economics* 62(1) : 93-101.

Fisher, J.A., Dhungana, H., Duffy, J., He, J., Inturias, M., Lehmann, I., Martin, A., Mwayafu, D.M., Rodríguez, I. and Schneider, H., 2020. Conservationists' perspectives on poverty: An empirical study. *People* and Nature. 2(2).

Garrity, D. P., Akinnifesi, F. K., Ajayi, O. C.,
Weldesemayat, S. G., Mowo, J. G., Kalinganire,
A., Larwanou, M. and Bayala, J. 2010. Evergreen
Agriculture: a robust approach to sustainable food
security in Africa. Food Security 2: 197-214.

Gibson C.G., 1999. Politicians and Poachers: The Political Economy of Wildlife Policy in Africa. Cambridge, MA: Cambridge University Press.

Hallegatte, S. and Rozenberg, J., 2017. Climate change through a poverty lens. *Nature Climate Change* 7: 250-256.

Hansen, M. C., Potapov, P. V., Moore, R., Hancher, M., Turubanova, S. A., Tyukavina, A., Thau, D., et al., 2013. High-resolution global maps of 21st-century forest cover change. *Science* 342: 850-853.

- Haughton, J., and Khandker, S. R.. 2009. Handbook on poverty and inequality (English). Washington, DC: World Bank. Available at: http://documents. worldbank.org/curated/en/488081468157174849/ Handbook-on-poverty-and-inequality [accessed on 24 September 2020].
- Hill, S. L., Arnell, A., Butchart, S. H., Hilton-Taylor, C., Ciciarelli, C., Davis, C., Dinerstein, E., et al., 2019.
 Measuring forest biodiversity status and changes globally. Frontiers in Forests and Global Change 2: 70.
- HLPE, 2017. Sustainable forestry for food security and nutrition. A report by the High-Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security. Rome: FAO.
- Hughes, K., Morgan, S., Baylis, K., Oduol, J., Smith-Dumont, E., Vågen, T.G. and Kegode, H., 2020.
 Assessing the downstream socioeconomic impacts of agroforestry in Kenya. World Development 128: 104835.
- IPBES, 2019. Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. Bonn: IPBES.
- IPCC, 2019. Climate Change and Land. An IPCC Special Report on Climate Change, Desertification, Land Degradation, Sustainable Land Management, Food Security, and Greenhouse Gas Fluxes in Terrestrial Ecosystems. Summary for Policymakers. Geneva: IPCC.
- Katila, P., Pierce Colfer, C. J., De Jong, W., Galloway, G., Pacheco, P. and Winkel, G., (eds.) 2019. Sustainable Development Goals: Their Impacts on Forests and People, Cambridge, Cambridge University Press.
- Kochendorfer-Lucius, G. (ed.), Pleskovic, B.(ed.), Hofman, M., Ferreira, F., Walton, M., Bourguignon, F., Battaini-Dragoni, G., et al., 2006. Equity and Development (English). Washington, D.C.: World Bank. http://documents.worldbank.org/curated/ en/284631468142167097/Equity-and-Development
- Krishna, A., 2011. One illness away: Why people become poor and how they escape poverty, New York: Oxford University Press.
- Larson A.M., and Ribot J.C., 2007. The poverty of forestry policy: double standards on an uneven playing field. Sustain Sci 2: 189-204.
- Li, Y., Mei, B. and Linhares-Juvenal, T., 2019. The economic contribution of the world's forest sector. *Forest Policy and Economics* 100: 236-253.
- Londoño, E., Andreoni, M. and Casado, L., 2020. Amazon Deforestation Soars as Pandemic Hobbles Enforcement, June 6, 2020. https:// www.nytimes.com/2020/06/06/world/americas/ amazon-deforestation-brazil.html [accessed on 23 September 2020].
- Mansourian, S., 2018. In the eye of the beholder: reconciling interpretations of forest landscape restoration. Land Degradation and Development 29(9): 2888-2898.

- Marris, E., 2019. Why young climate activists have captured the world's attention. *Nature* 573 : 471-472.
- McArthur, J. W. and Rasmussen, K., 2018. Change of Pace: Accelerations and Advances during the Millennium Development Goal era. World Development 105: 132-143.
- McDowell, N. G., Allen, C. D., Anderson-Teixeira, K.,
 Aukema, B. H., Bond-Lamberty, B., Chini, L., Clark,
 J. S., et al., 2020. Pervasive shifts in forest dynamics in a changing world. Science 368(6494): eaaz9463.
- Merchant, C., 2003. Shades of darkness: Race and environmental history. *Environmental History* 8(3): 380-394.
- Miller, D. and Hajjar, R., 2020. Forests as Pathways to Prosperity: Empirical Insights and Conceptual Advances. World Development 125: 104647.
- Miller, D. C., Muñoz-Mora, J. C. and Christiaensen, L., 2017. Prevalence, economic contribution, and determinants of trees on farms across Sub-Saharan Africa. Forest Policy and Economics 84: 47-61.
- Miller, D. C., Ordoñez, P. J., Brown, S. E., Forrest, S., Nava, N. J., Hughes, K. and Baylis, K., 2020. The impacts of agroforestry on agricultural productivity, ecosystem services, and human well-being in low-and middleincome countries: An evidence and gap map. *Campbell Systematic Reviews* 16: e1066.
- Milne, G.R., 2006. India: Unlocking Opportunities for Forestdependent People. Washington D.C.: World Bank.
- Newton, P., Miller, D. C., Byenkya, M. A. A. and Agrawal, A., 2016. Who are forest-dependent people? A taxonomy to aid livelihood and land use decision-making in forested regions. *Land Use Policy* 57: 388-395.
- Newton, P., Kinzer, A., Miller, D.C., Oldekop, J.A. and Agrawal, A., 2020. The number and spatial distribution of forest-proximate people globally. *One Earth* 3(3): 363-370.
- Nussbaum, M., 2000. Women and Human Development. Cambridge and New York: Cambridge University Press.
- O'Neill, D. W., Fanning, A. L., Lamb, W. F. and Steinberger, J. K., 2018. A good life for all within planetary boundaries. *Nature Sustainability* 1: 88-95.
- OECD, 2020a. Poverty rate (indicator). doi: 10.1787/0fe1315d-en (Accessed on 16 August 2020)
- OECD, 2020b. OECD Eonomic Outlook, June 2020: The World Economy on a Tightrope. Paris: OECD.
- Peluso, N.L., 1992. Rich forests, poor people: Resource control and resistance in Java. Berkeley, Los Angeles and London: Univ of California Press.
- Potapov, P., Hansen, M. C., Laestadius, L., Turubanova, S., Yaroshenko, A., Thies, C., Smith, W., et al., 2017. The last frontiers of wilderness: Tracking loss of intact forest landscapes from 2000 to 2013. Science Advances 3: e1600821.

Raworth, K., 2017. Doughnut economics: seven ways to think like a 21st-century economist. White River Junction VT: Chelsea Green Publishing.

Rasmussen, L.V., Watkins, C. and Agrawal, A., 2017. Forest contributions to livelihoods in changing agriculture-forest landscapes. Forest policy and economics 84: 1-8.

Reed, J., Van Vianen, J., Clendinning, J., Petrokovsky, G. and Sunderland, T., 2017. Trees for life: The ecosystem service contribution of trees to food production and livelihoods in the tropics. *Forest Policy and Economics* 84: 62-71.

Rockström, J., Steffen, W., Noone, K., Persson, A.,
Chapin, F. S. III, Lambin, E. F., Lenton, T. et al., 2009.
A Safe operating space for humanity. Nature 461: 472-475.

Rondeau, D., Perry, B. and Grimard, F., 2020. The Consequences of COVID-19 and Other Disasters for Wildlife and Biodiversity. Environmental and Resource Economics 76: 945–961

Rosenstock, T. S., Dawson, I. K., Aynekulu, E., Chomba, S., Degrande, A., Fornace, K., Jamnadass, R., et al., 2019. A Planetary Health Perspective on Agroforestry in Sub-Saharan Africa. *One Earth* 1: 330-344.

Sachs, J.D., Baillie, J.E., Sutherland, W.J., Armsworth, P.R., Ash, N., Beddington, J., Blackburn, T.M., et al., 2009. Biodiversity conservation and the millennium development goals. Science 325(5947): 1502-1503.

Sen, A., 1993. Capability And Well-Being. In: The Quality Of Life, edited by M. Nussbaum and A. Sen. Oxford: Claredon Press.

Sen, A., 1999. Development As Freedom. New York: Anchor Books.

Seymour, F. and Busch, J., 2016. Why Forests? Why Now?: The Science, Economics, And Politics Of Tropical Forests And Climate Change, Washington D.C.: Center For Global Development

Smeeding, T. M., 2016. Poverty Measurement. In: The Oxford Handbook of the Social Science of Poverty, edited by D. Brady and L. M. Burton. New York: Oxford University Press.

Song, X.-P., Hansen, M. C., Stehman, S. V., Potapov, P. V., Tyukavina, A., Vermote, E. F. and Townshend, J. R. 2018. Global land change from 1982 to 2016. *Nature* 560: 639-643.

Stafford-Smith, M., Griggs, D., Gaffney, O., Ullah, F., Reyers, B., Kanie, N., Stigson, B., et al., 2017. Integration: the key to implementing the Sustainable Development Goals. Sustainability Science 12(6): 911-919.

Steffen, W., Richardson, K., Rockström, J., Cornell, S.
E., Fetzer, I., Bennett, E. M., Biggs, R., et al., 2015.
Planetary boundaries: Guiding human development on a changing planet. Science 347(6223): 1259855.

Sumner, A., Hoy, C. and Ortiz-Juarez, E., 2020. Estimates of the Impact of COVID-19 on Global Poverty. UNU-WIDER, April, pp.800-9. Sunderlin, W. D., Angelsen, A., Belcher, B., Burgers, P., Nasi, R., Santoso, L. and Wunder, S. 2005. Livelihoods, forests, and conservation in developing countries: an overview. World Development 33: 1383-1402.

Sunderlin, W.D., Dewi, S., Puntodewo, A., Muller, D., Angelsen, A. and Epprecht M., 2008. Why Forests Are Important for Global Poverty Alleviation: a Spatial Explanation. *Ecology and Society* 1-21.

Taylor, D. E., 2016. The rise of the American conservation movement: Power, privilege, and environmental protection. Durham: Duke University Press.

Timko, J., Le Billon, P., Zerriffi, H., Honey-Rosés, J., De La Roche, I., Gaston, C., Sunderland, T. C. and Kozak, R. A., 2018. A Policy Nexus Approach To Forests And The Sdgs: Tradeoffs And Synergies. Current Opinion In Environmental Sustainability 34: 7-12.

Trumbore, S., Brando, P. and Hartmann, H., 2015. Forest health and global change. *Science* 349: 814-818.

UN, 2015. Sustainable development goals. (https:// sustainabledevelopment.un.org/sdgs [accessed on 23 September 2020]).

UN, 2016. Leaving no one behind: the imperative of inclusive development. A report on the world social situation 2016. Executive summary. https://www.un.org/esa/socdev/rwss/2016/executive-summary.pdf [accessed on 15 August 2020].

- UN, 2017. United Nations strategic plan for forests 2017–2030. New York: United Nations.
- UN, 2020a. World Social Report 2020: Inequality in a rapidly changing world. New York: United Nations.
- UN, 2020b. The Sustainable Development Goals Report 2020. New York: United Nations.

UNCCD, 2017. ICCD/COP(13)/21/Add.1. Decision 7 / COP 13. The future strategic framework of the Convention. New York: UN.

UNDP, 2019. Human Development Report 2019, Beyond income, beyond averages, beyond today: inequalities in human development in the 21st century. New York: UNDP.

UNEP, 2010. UNEP/CBD/COP/DEC/X/2. Decision Adopted By The Conference Of The Parties To The Convention On Biological Diversity At Its Tenth Meeting X/2. The Strategic Plan for Biodiversity 2011-2020 and the Aichi Biodiversity Targets. Nairobi: UNEP.

Van Noordwijk, M., Duguma, L. A., Dewi, S., Leimona, B., Catacutan, D. C., Lusiana, B., Öborn, I., et al., 2018. Sdg Synergy Between Agriculture And Forestry In The Food, Energy, Water And Income Nexus: Reinventing Agroforestry? *Current Opinion In Environmental Sustainability* 34: 33-42.

Vira, B., Wildburger, C. and Mansourian, S. (eds.) 2015. Forests and Food: Addressing Hunger and Nutrition Across Sustainable Landscapes. Cambridge, UK: Open Book Publishers.

- Waldron, A., Garrity, D., Malhi, Y., Girardin, C., Miller, D. C. and Seddon, N. 2017. Agroforestry can enhance food security while meeting other Sustainable Development Goals. *Tropical Conservation Science* 10: 1940082917720667.
- Watson, J. E. M., Evans, T., Venter, O., Williams, B.,
 Tulloch, A., Stewart, C., Thompson, I., et al., 2018.
 The exceptional value of intact forest ecosystems.
 Nature Ecology Evolution 2: 599-610.
- Wildburger, C., 2009. Overview of international policy instruments related to forests and their goals and tools. In: Embracing complexity: Meeting the challenges of international forest governance. A global assessment report, edited by J. Rayner, A. Buck and P. Katila.
 Prepared by the Global Forest Expert Panel on the International Forest Regime. IUFRO World Series Volume 28. Vienna: IUFRO.
- World Bank, 2001. World development report 2000/2001: Attacking poverty. Oxford & New York: Oxford University Press.
- World Bank, 2018. Poverty and Shared Prosperity 2018: Piecing Together the Poverty Puzzle, Washington, D.C.: World Bank.
- World Bank, 2019. PovcalNet: an online analysis tool for global poverty monitoring. (available at: http://iresearch.worldbank.org/PovcalNet/ [accessed on 23 September 2020]).
- World Bank, 2020a. World Development Indicators. (available at : https://datacatalog.worldbank.org/dataset/world-development-indicators. [accessed on 15 August 2020]).
- World Bank, 2020b. World Bank Country and Lending Groups. (available at : https://datahelpdesk.worldbank.org/knowledgebase/articles/906519-worldbank-country-and-lending-groups. [accessed on 15 August 2020]).
- World Economic Forum (WEF). 2020. The Global Risks Report 2020. Geneva: World Economic Forum.
- Wunder, S., 2001. Poverty alleviation and tropical forests – what scope for synergies? World development 29(11): 1817-1833.
- Wunder, S., A. Angelsen and B. Belcher, B., 2014. Forests, Livelihoods, and Conservation: Broadening the Empirical Base. World Development 64: S1-S11.
- Wunder, S., 2020. Coronavirus, markets, economy and forests: what's ahead? Barcelona: European Forest Institute (available at : https://blog.efi.int/coronavirus-markets-economy-and-forests-whats-ahead/ [accessed on 23 September 2020]).
- Zomer, R. J., Neufeldt, H., Xu, J., Ahrends, A., Bossio,
 D., Trabucco, A., Van Noordwijk, M. and Wang, M.
 2016. Global Tree Cover And Biomass Carbon On
 Agricultural Land: The contribution of agroforestry
 to global and national carbon budgets. Scientific
 Reports 6: 29987.



Chapter 2

Key Concepts for Understanding Forest-Poverty Dynamics

Coordinating Lead Authors: Pamela A. Jagger and Jennifer Zavaleta Cheek Lead Authors: Daniel C. Miller, Casey Ryan, Onja H. Razafindratsima, Priya Shyamsundar and Erin O. Sills

TABLE OF CONTENTS

2.1 Introduction	34
2.2 Poverty Dynamics and Human Well-being	34
2.3 Conceptual Framing for Forest-Poverty Dynamics	38
2.4 Towards a Deeper Knowledge Base for Understanding Forest-Poverty Dynamics	43
2.5 Conclusions	46
2.6 References	48

Abstract

In this chapter we review existing poverty and forest-poverty frameworks and present the framing for how we analyse forest-poverty dynamics and human well-being throughout this report. We highlight four ways that forests and tree-based systems can shape poverty outcomes: 1) as a resource that allows people to move out of poverty; 2) supporting livelihoods and thus helping people avoid moving into, or further into, poverty; 3) mitigating risk; and 4) negatively affecting livelihood opportunities. We connect these four roles for forests and tree-based systems with the targets and indicators under SDG1. Existing conceptual frameworks for forest-poverty dynamics primarily focus on the role that forests play in supporting human well-being over relatively short time frames (temporal dynamics). This is a critical dimension that merits further attention as we continue to build the evidence base surrounding forest-poverty dynamics over longer time frames (>2-5 years). Forest-poverty dynamics also have two other important dimensions: spatial and contextual. We highlight the importance of considering temporal, spatial and contextual dimensions of forest-poverty relationships as part of systems frameworks for building our knowledge base.

2.1 Introduction

Poverty⁴ and its converse, well-being, are conceptualised, measured and interpreted in many different ways, which in turn influence our understanding of the role of forests and tree-based systems in poverty alleviation. This chapter provides a framework for reviewing the dynamic relationship between forests and tree-based systems and poverty. Our emphasis is on forest-poverty dynamics, explicitly focusing on the role that forests play in reducing the prevalence and impact of poverty. We focus on how forests contribute to poverty alleviation rather than how poverty alleviation affect forests (see Lawlor et al., 2019 for an extensive review of this relationship). Our review and framework development covers three related, but conceptually distinct dimensions of forest-poverty dynamics temporal, spatial and contextual - which we view as critical for building a case for policy and practice that leverages the full potential of forests and tree-based systems for poverty alleviation.

2.2 Poverty Dynamics and Human Well-being

Global efforts to track progress towards poverty alleviation have brought to the fore questions about how we conceptualise, measure and interpret poverty and poverty dynamics. Multi- and bi-lateral donors and governments historically focused analysis of poverty on monetary measures including dollar a day metrics such as the current USD 1.90/day *poverty line* (World Bank, 2018a). The approach to monitoring progress towards poverty alleviation, which remains an important component of poverty analytics, involves aggregating poverty data at country level, and using static poverty rates (e.g. snapshots at a given point in time using cross-sectional data for households or individuals), across multiple years to describe broad trends observed over time (Yaqab, 2000). The most robust static poverty analysis uses multiple waves of population representative data to establish trends over time. Analysis of poverty dynamics is different from static poverty analysis in that it uses longitudinal or panel data to track the same households or individuals over time.

Measures of dynamic poverty allow us to understand transient poverty (e.g. oscillation above and below the poverty line) and the factors that catalyse movement into or out of poverty for households and individuals. In their path-breaking work to advance the analytical study of poverty dynamics Addison, Hulme and Kanbur (2008) advocate for research that considers poverty dynamics over the life-course of individuals, across generations and between different social groups. They argue that without consideration of changes in poverty status over time we cannot understand transient poverty, or factors that hold people in chronic poverty. They advocate for the use of panel data, carefully selected measures that can provide information about poverty status over time (e.g. objective indicators of nutritional status such as stunting) and use of retrospective data about past circumstances.

Several poverty scholars have proposed frameworks to extend our conception of changes in poverty status beyond monetary poverty and static approaches to data collection and analysis. The overall tenor of these frameworks is movement

away from measures of monetary poverty at single points in time, and towards recognition that a host of factors at the household, community and broader institutional levels (e.g. markets, regulatory frameworks, institutions, etc.) influence movement into and out of poverty, and that change occurs across time, space and context. Chambers and Conway (1992) proposed a livelihoods approach where economic units (e.g. individuals or households) have a portfolio of capabilities, assets and activities required for a means of living. This framing is dynamic in that it captures changes fundamental to sustaining livelihoods over time. Scoones (1998) developed the idea of the five capitals: human, economic or financial, physical, social and natural as endowments that determine human well-being. Each of the five capitals is a potential lever of change affecting the trajectory of movement in and out of poverty. Livelihoods are sustainable when they can cope and recover from stresses and shocks, maintain or enhance capabilities and assets, while not undermining the resource base. An important contribution of Scoones (1998) is to consider scale in the study of livelihood strategies, noting that the net effect of different strategies may accrue differently at the household-level than in aggregate at a coarser scale unit of analysis, for example district or sub-regional-level.

Other scholars take a broader approach to conceptualising poverty dynamics and inequality. Sen (1985) focused normative arguments on missing dimensions of poverty analysis including longevity and education, extending to capabilities inherent in physical and mental characteristics, and social opportunity and influence. The capabilities approach he developed in collaboration with Martha Nussbaum emphasises the opportunities people have (or lack) to fulfil needs including being adequately nourished, receiving appropriate medical care in times of need, and pursuing educational interests (Sen, 1999; Nussbaum 2000). Building on core concepts developed by Sen and Nussbaum, Alkire (2007) further developed the concept of multidimensional poverty, which complements traditional income-based poverty measures focusing on individual deprivations in the areas of employment quality, empowerment, physical safety, the ability to go without shame, and psychological and subjective well-being. Alkire (2007) argues for inclusion of high quality indicators in all of these areas to advance our understanding of poverty and poverty dynamics.

In addition to broader conceptualisations of poverty, the dynamics of inequality have emerged as important to understand and monitor over time. Krishna (2004) in the 'Stages of Progress' method emphasises that the benchmarking of households within communities is critical to understanding poverty trajectories that are often only considered at aggregate scales, and acknowledges that both micro- and macro-level processes are fundamental to understanding poverty trajectories. The Stages of Progress method uses life histories and retrospectives of community events and their impacts gained through participatory approaches. This approach yields rich site-specific contextual information about past and present trajectories as well as expectations about the future. People may be mired in poverty, find ways to move out of poverty, fall into poverty again, or find themselves in poverty for the first time (Krishna, 2006). Krishna and Shariff (2011) note that panel data collection as an approach to understanding poverty dynamics is costly and requires time to establish a period long enough to generate meaningful insights into factors influencing movement into and out of poverty.

Integration of the aggregate effect of changes in poverty status are embodied in the World Bank's (2015) 'twin goals' which frame the majority of the Bank's current lending and policy programming: an end to extreme poverty by 2030; and improvements in metrics of shared *prosperity*, or reduced inequality, in every society. These broad objectives emphasise reducing the number of people living below the global poverty line (as measured by income), supporting countries in setting poverty thresholds that are relevant to particular contexts, and monitoring what share of total income goes to the poorest 40% of people.

Throughout this report we use human well-being as the orienting concept for our discussion of forest-poverty dynamics, with poverty being the converse of well-being (Fisher et al., 2014). Human well-being encompasses many of the contributions of the poverty scholars cited above, the objective material circumstances of people's lives including health, housing, food security and income; social aspects such as community relations and trust; and inclusive of subjective assessments related to how individuals view their own circumstances (McKinnon et al., 2016; OECD, 2017). Following the broad literature on poverty and poverty dynamics, well-being (and thus poverty) are multidimensional, with important dimensions beyond commonly measured economic aspects of poverty (Alkire and Santos, 2013). The concept of multi-dimensional poverty is widely accepted in policy discussion and is consistent with the SDG 1 targets. These targets (Box 1.1; Chapter 1) consider poverty in unidimensional framings focused on cash income (Target 1.1) but also allow scope for

multi-dimensional poverty through consideration of nationally defined indicators (Target 1.2) and access to social protection programmes (Target 1.3), equitable resource access and rights (Target 1.4); and building *resilience* and reducing vulnerability to climate-related and other shocks (Target 1.5).

Figure 2.1 conceptualises the ways that well-being and poverty status, broadly conceived, can change over time. In this stylised framework, people are classified as non-poor, poor, transient poor and extremely poor. The last of these categories is important because the poorest are often most reliant on forest-related products for meeting basic needs and smoothing consumption (Angelsen *et al.*, 2014). A given household may find itself within different groups at different times based on the season, the household 'life cycle', or after experiencing a shock.

Trajectories of well-being over time (Figure 2.1) can be described as:

- Durable improvements in well-being (shifts out of poor or extreme poor status) (Ai, Aii)
- Maintenance of the status quo (any horizontal trajectory)
- Transience around poor or extreme poor status (oscillation within yellow or red bands)
- Durable declines in well-being (into poor or extremely poor status) (Bi, Bii)

2.2.1 Durable shift out of poor or extreme poor status

People are considered to have moved out of poverty when they surpass a certain level of income or consumption (World Bank, 2018b) or move bevond a locally defined state of poverty (Krishna, 2006). Movement up and out of extreme poverty (Ai), or poverty (Aii) typically requires major changes in income earning strategies and/or ability to accumulate assets and capacity to leverage social, human and natural capital in ways that dramatically alter well-being (Brockington, 2019). Given the human capital limitations to dramatically changing income earning strategies, and the pervasiveness of intergenerational transmission of poverty, shifts out of poor and extreme poor status may happen across generations, especially after higher investments in childcare and education (Moran, 2003).

2.2.2 Maintaining the status quo

People could also remain at the same level of poverty or maintain the status quo. Even though the ultimate goal is for households and individuals to move out of poverty permanently, other trajectories include maintaining the status quo and not falling into poverty or from 'poor' to 'extreme poor' status, especially in the face of shocks (see below).



Ai Aii = Moving out of poverty Bi Bii = Moving into poverty
2.2.3 Transience around poor or extreme poor status

Several potential stressors and shocks contribute to transience for the poor and extremely poor groups (e.g. climate change related shocks, morbidity or mortality of a productive household member, etc.) (Dercon and Hoddinott, 2004). Transitory or stochastic poverty can occur due to short-term, adverse labour, price and environmental shocks that can decrease incomes or increase prices of goods (Carter and May, 2001). Studies of household income dynamics find that most households experiencing poverty move in and out of poverty over time (Baulch and Hoddinott, 2000).



Woman collecting fodder from the forest in Nepal Photo © Conghe Song

2.2.4 Durable shift into poor or extreme poor status

Households or individuals can shift from a state of being poor to being extremely poor (Bi) or fall into poverty from non-poor status (Bii). Poverty dynamics of this scale are typically the result of extreme events that have a major impact on livelihood strategies. Typical causes of movement into poverty include health crises of productive aged adults, death in families (Krishna, 2010), loss of access to assets and resources, major disasters or weather events that alter income and consumption streams in the long term (Krishna and Shariff, 2011; Klasen and Waibel, 2015). Political upheaval can also lead to extreme poverty (Goodhand 2001). Given the high frequency of transience among the poor, it is primarily those in the non-poor category and those in *poverty traps* who are likely to maintain their position over time. We address the special case of forest and poverty traps in Box 2.1.

Box 2.1

Forests and poverty traps

A poverty trap is a self-reinforcing mechanism that causes poverty to persist (Azariadis and Stachurski, 2005). Even though most poverty in low-income countries is transitory, meaning that poor households move in and out of poverty over relatively short periods of time, poverty traps exist when poverty persists indefinitely with no promise of movement out. Despite a growing literature on poverty traps, largely emerging from the availability of household-level panel datasets, relatively little attention has been given to conditions under which forests serve as poverty traps (Angelsen et al., 2014). The poverty trap literature considers various measures of well-being including flows of income or expenditures as well as stock measures of assets and human capital, health and nutrition (Barrett et al., 2016). At the macro scale, human populations in remote forest landscapes far from markets may find themselves in a poverty trap by virtue of the lack of income and asset accumulation opportunities. This macro-level relationship highlights a potential correlation between people living at the forest frontier and poverty traps, but does not establish a causal link between forests and poverty. Given the high overlap between the extreme poor and those living in marginal or remote areas often dominated by forests and woodland, forest-dependent people may be at risk of facing poverty traps (Barbier, 2019). For example, depletion of community resources resulting from coordination failures, overexploitation of natural resources, and degradation below recoverable thresholds can lead to a poverty trap (Ostrom, 1990; Baland and Platteau, 1996; Toth, 2015). Overall, there is good evidence that poverty is higher in forested areas, but no evidence that this correlation is caused by the negative impacts of forests on well-being.

2.3 Conceptual Framing for Forest-Poverty Dynamics

The study of forest-poverty dynamics has evolved at the same time as the conceptualisation of poverty by the global community has broadened to include multidimensional aspects of poverty, and as data and analytical tools for monitoring poverty and inequality over time have improved. However, most foundational work on poverty dynamics (c.f. Scoones, 1998) largely fails to give explicit attention to natural resources and environmental sustainability as central to supporting well-being (Miller and Hajjar, 2020). Early conceptual framing of the role of forests and trees in poverty dynamics emerged primarily out of a small number of high quality, mostly static analyses conducted in disparate settings. Seminal efforts include Cavendish (2000) who analysed high frequency household panel survey data with detailed questions on environmental income finding that environmental resources contribute on average to 40% of the income (cash plus subsistence) for poor households in Zimbabwe. Studies by Pattanayak and Sills (2001), McSweeney (2004; 2005) and Shackleton et al. (2007) illustrated that forests play an important role as a safety net, providing not only opportunities for income diversification but also serving as a form of natural insurance against shocks ranging from disruptions in agricultural and livestock production to illness and death related to HIV/AIDS.

These early empirical examples highlighted the 'hidden harvest' of forests and tree-based systems functionally absent from standard analyses of rural livelihoods and poverty dynamics (Campbell and Luckert, 2002). They led Angelsen and Wunder (2003) to conceptualise three ways that forests contribute to rural livelihoods and poverty alleviation: supporting current consumption (e.g. providing fuel, food, fodder, medicine, building materials needed to meet everyday consumption needs); acting as a safety-net or natural insurance against shocks; and serving as a pathway out of poverty. Numerous studies (see Chapter 3 of this volume) review findings related to current consumption and safety-net functions of forests and trees in a range of settings.

Angelsen and Wunder (2003) give considerable attention to forest-poverty dynamics by discussing the hypothesised limitations of relying on forests as a poverty alleviation strategy highlighting three issues. First, the most common ways in which the majority of rural households in low and middle-income countries rely on forests are the production and sale of non-timber forest products (NTFPs) and trade in low value products (e.g. firewood and low value timber). Both of these activities have relatively low return on investment relative to the labour input required. Second, they point out the trade-off inherent in forest based poverty alleviation strategies and forest conservation, suggesting that relying on forest-based strategies other than sustainable forest management for carbon, bio-



Clear-cut in a Chilean forest Photo © Jennifer Zavaleta Cheek

diversity, water or tourism values is not a viable pathway. Third, they flag the need for considerable financial and human capital to engage in most high-value forest sector activities (e.g., extraction of valuable timber species) implying that policies that support forests as a pathway out of poverty might bypass the poorest due to their own lack of capital and elite capture. Sunderlin et al. (2005) add nuance to the discussion by considering the geographic overlap of rural poverty and forests, and suggesting that remoteness plays a fundamental role in forest-poverty dynamics. Forest cover, population density, scarcity of forest products and proximity to markets all play an important role in determining the economic viability of forest based economic strategies. These contributions suggest that understanding the complex relationship between forests and tree-based systems and poverty requires attention to spatial and contextual heterogeneity.

While Angelsen and Wunder (2003) and Sunderlin et al. (2005) provided important new ways of thinking about the role of forests and environmental goods in rural livelihoods, their conceptualisation is limited to the role of subsistence (e.g., the monetised value of goods produced and consumed by households) and cash income, and the contributions of that income to defining poverty status. The first two dimensions of the Angelsen and Wunder (2003) framework were explored across a wide range of settings in the low-income tropics with data from the Poverty Environment Network (PEN) led by CIFOR. Analysis of the PEN data provided confirmation that forests and trees are critical in supporting the current consumption and income needs of households and for reducing income inequality within communities (Angelsen et al., 2014; Lopez-Feldman et al. 2007), but did not confirm the use of forests as safety nets in a wide range of settings (Wunder et al., 2014b). Several of the institutional barriers hypothesised by Angelsen and Wunder (2003) as important mediators in the ability of households and communities to access resources were analysed using the PEN dataset, including tenure rights (Jagger et al., 2014) and gender roles (Sunderland et al., 2014). Notably, the PEN Project was not designed, nor were the data appropriate, for testing theories about forests as a pathway out of poverty.

Two recent frameworks add new dimensions to our conception of forest-poverty dynamics including the indirect effects of forest management on *poverty reduction* (Miller and Hajjar, 2020) and the multidimensional framing of how forests can reduce poverty (Shyamsundar et al., 2020). Following the framing of the twin goals of the World Bank (2015), Miller and Hajjar (2020) describe how forests contribute to prosperity. They emphasise human well-being beyond economic and material dimensions, and suggest a societal movement towards prosperity (e.g., a reduction in inequality), both significant departures from earlier conceptualisations that largely focused on income poverty. The main mechanisms or pathways to prosperity focus on forest-based income diversification through the sale of timber and non-timber forest products, payments for ecosystem services (PES), tourism and the contribution of forest ecosystem services to agricultural productivity. They highlight that the inclusion of indirect pathways through which forest management might influence poverty (e.g. by enhancing agriculture) have not been explored in the literature. The PRIME framework (Shyamsundar et al., 2020) describes five ways through which policy and programmatic interventions can facilitate the contribution of forests to poverty alleviation: improving productivity of forest and labour; strengthening community, household and women's rights; strengthening institutions, infrastructure and public service provision; increasing access to markets; and enhancing ecosystem services. They test the utility of this framework by examining forestry projects within the World Bank's portfolio.

Drawing on this literature, we identify four roles that forests and tree-based systems play in affecting poverty and human well-being: 1) helping people move out of poverty through direct and indirect contributions to income; 2) supporting well-being through subsistence, food security, and cultural and spiritual values; 3) mitigating risks through seasonal gap filling and mitigation of shocks; and 4) decreasing well-being through poverty traps or experiences due to negative externalities related to the presence of forests and trees.

2.3.1 Moving out of poverty

A substantial change in income directly or indirectly derived from forest can catalyse movement out of poverty. Forests and trees outside forests help people move out of poverty by providing a range of goods and services that directly or indirectly enable households to earn income. The former includes goods such as timber, energy and bushmeat; the latter includes the ways in which forests support agriculture income or ecotourism-based livelihoods. Such income and other benefits from forests and trees can lead to asset accumulation for households such that they are able to move out of poverty. Benefits may derive from sustainable or unsustainable exploitation of forests and trees as well as through their conservation, *restoration* and sustainable management.

Households derive income from extracting timber and non-timber products. Timber collection, which is the most common commercially viable and lucrative forest product (FAO, 2014) often requires large economies of scale and high initial investments in technology (Wunder, 2001; Angelsen and Wunder, 2003; Belcher and Kusters, 2004). Economic opportunity for forest extraction lies primarily in tropical countries, with more than two-thirds of the people living in or near forests residing in such countries (Newton et al., 2020). However, comparatively few people live in or near large tracts of tropical rainforest (many live in or near dry forests), so income generation from timber extraction may be limited for the poorest forest-dependent households. Further, elite capture, corruption and unsustainable practices and other political economy challenges frequently hinder the business for commodities with high rents (e.g. Sundström, 2016). Nevertheless, locally-controlled forestry businesses and cooperatives exist in contexts around the world that can contribute to poverty reduction and broader notions of prosperity (Macqueen et al., 2020; Humphries et al., 2020).

Additionally, households earn income from converting forests into agricultural land, but the sustainability of this strategy depends on soil quality, input availability and a host of other factors. Over time, the costs associated with any loss and degradation of ecosystem services may outweigh livelihood benefits (see discussion and evidence in Chapter 3). At the same time, a wide diversity of *agroforestry* systems exist that may provide income and other benefits that enable more sustainable movement out of poverty (Miller *et al.*, 2020a).

Collection of NTFPs typically requires little specialised skill or technology, but it does require proximity and access to the forest and to markets for NTFPs (Neumann and Hirsch, 2000; Shackleton *et al.*, 2011; Angelsen *et al.*, 2014). Often the poorest are the most likely to collect NTFPs (Neumann and Hirsch, 2000; Shackleton *et al.*, 2011; Tincani, 2012; Angelsen *et al.*, 2014), perhaps because they lack better alternatives (Wunder, 2001). Whether NTFPs are a pathway out of poverty is debated (Wunder *et al.*, 2014a; Shyamsundar *et al.*, 2020), given the limited potential to scale-up production and in many cases limited markets for NTFPs. However, extraction of specialised products for urban markets (e.g. rubber, honey, resins, nuts, etc.) has po-

tential to increase incomes (Kusters *et al.*, 2006; IUCN, 2012).

Other ways in which people earn income from forests include PES, ecotourism, and via indirect contributions to agricultural yields. Households can receive payment to keep their forest intact or even reforest through PES, an increasingly widespread policy for supporting forests and livelihoods globally (Salzman et al., 2018). This includes mechanisms such as REDD+ promoted under the UNFCCC (Parrotta et al., 2012). However, land ownership and social capital are often prerequisites for successful engagement in PES programmes (Wunder et al., 2018), acting as a barrier to entry or success in these schemes (Wells et al., 2020). Ecotourism can also increase incomes by providing more job opportunities within the tourism sector, and catalysing the construction of infrastructure like roads, health clinics and schools (den Braber et al., 2018). In Costa Rica, Ferraro and Hanauer (2014) found that revenue from ecotourism was the main mechanism for reducing poverty. Last, forests could additionally contribute to incomes indirectly by providing ecosystem services like regulating water cycles and filtration, weed and pest control, limiting erosion, and pest and disease control that improve yields (MEA, 2005; Gamfeldt et al., 2013).

2.3.2 Support well-being

Forests support well-being because they provide subsistence goods, improve food security, offer cultural and spiritual goods and services, and provide a means for formalising rights and strengthening democratic processes.

Forests provide a wide range of goods in support of subsistence such as food, fuel, fodder and building materials (Sunderlin *et al.*, 2005; Angelsen *et al.*, 2014). Resources collected from forests can support consumption to meet basic needs, which can reduce pressure on stretched incomes and make households less vulnerable to price changes of fuel and other critical consumption goods in the market. Subsistence income is not typically captured in standard measures of poverty, but it does play a critical role in smoothing incomes and mitigating risk, which relates to SDG 1 Target 5, and to multidimensional poverty (Target 1.2).

Forests also support well-being by improving food security and dietary diversity (Vira *et al.*, 2015). Tree cover (Ickowitz *et al.*, 2014), proximity to forests (Golden *et al.*, 2011; Tata *et al.*, 2019), and composition of forests (Rasmussen *et al.*, 2019) are associated with higher quality diets, including higher dietary diversity of fruits, vegetables and meats. In many contexts, the contribution of forests to



Mother and children on the border of W National Park, Benin Photo © Daniel C. Miller

dietary diversity is not easily replaced by higher incomes. Paradoxically, in places where forests were converted to commodity crops leading to increased incomes, there were negative consequences for nutrition and dietary diversity (Ickowitz *et al.*, 2016). For example, after over thirty years of oil palm development in Indonesia, the incidence of child stunting remains very high (Beal *et al.*, 2018; Tiominar, 2011; Santika *et al.*, 2019).

Forests also provide non-material benefits that contribute to culture and spirituality, feelings of empowerment, individual happiness and social relations (Raymond et al., 2009). Forests and trees can be central to the identity of forest-dependent communities (Oteng-Yeboah et al., 2012; Daniel et al., 2016), and forest knowledge is intimately embedded in indigenous knowledge systems (Asselin, 2015). For some indigenous communities, the forest is a sacred place where they perform cultural and religious ceremonies and collect products for traditional ceremonies (Munyi and Mutta, 2007; Rutte, 2011; Ngoufo et al., 2014). Even though non-material benefits are unlikely to influence economic measures of well-being or income and asset poverty, non-material benefits from forests contribute to the broader multi-dimensional understandings of well-being and may contribute to fulfilment of SDG1 targets focused on rights (1.4) and resilience (1.5), in a supporting role in the overall well-being of households.

2.3.3 Mitigate risk

Forests can play a critical role in risk management and coping with shocks by providing a safety net or 'natural insurance' function and through serving as seasonal gap fillers. In the face of shocks, households can turn to forests to increase their income or improve subsistence by collecting forest goods (Pattanayak and Sills, 2001; Sunderlin et al., 2005; Wunder et al., 2018). Households may start harvesting timber and non-timber forest products when they have lost other income earning opportunities, or increase their frequency of visits to collect more timber and NTFPs. Given the barriers to entry for timber production and limited potential to scale up NTFP collection, forests seem most useful in making up shortfalls and providing basic survival for the poor (Neumann and Hirsch, 2000). Often income- and asset-poor (Wunder et al., 2014b) and male-headed households (McSweeny, 2004; Fisher and Shively, 2005) rely more on forests to cope with shocks as compared to more asset-rich households who have other response options. Households with greater access to forest (Fisher and Shively, 2005; Fisher et al., 2010) or that are remote and lack market-based coping options (Godoy et al., 1998) are more likely to use the forest as a safety net. Depending on the availability of other coping strategies, households chose responses other

				Figure 2.2						
	Relationship between forests and tree-based systems and SDG1 targets and indicators									
		Targets	Indicators	Ways forests and tree-based systems may help reach target						
1.1		Eradicate extreme poverty (less than USD 1.90/day)	1.1.1 Proportion of population below the international poverty line	 Cash income from timber and NTFPs Cash income from PES and ecotourism Cash income from increased crop yields due to ecosystem services from forests Enable asset accumulation 						
1.2		Reduce by half all people living in poverty using national definitions	1.2.1 Proportion of people below national poverty line1.2.3 Proportion of all people living in poverty	 Provide subsistence goods Contribute to food security Cultural and spiritual values of forests 						
1.3		Implement sustainable national social protection programmes focusing on the poor and vulnerable	1.3.1 Proportion of people covered by social protection floors/systems with attention to poor and most vulnerable	 Payment for ecosystem services Link social cash transfer programmes with forest related outcomes 						
1.4		Ensure equal access to economic resources	1.4.1 Proportion of population with access to basic services1.4.2 Proportion of adult population with secure tenure rights to land	 Rights to natural resources, especially for indigenous communities Community forestry rights strengthen democratic processes and advance social justice 						
1.5		Build resilience for poor and vulnerable by reducing exposure and vulnerability to climate change, environmental shocks and disasters	 1.5.1 Number of deaths and missing persons and people affected by disaster 1.5.2 Direct economic loss attributed to disasters in relation to global GDP 1.5.3 Number of countries with disaster risk reduction strategies 	 Safety net for shocks Seasonal gap filler that allows people to smooth consumption 						

than forests first to respond to shock (Wunder et *al.*, 2014b). Nevertheless, forests can be particularly important when households face multiple, interacting shocks such as economic collapse and drought (Pritchard *et al.*, 2020).

There is evidence that forests and trees play an important role as seasonal gap fillers allowing households to smooth income and consumption (Pattanayak and Sills, 2001; Sunderlin et al., 2005) over the course of the year. Forests provide a variety of non-seasonal products (e.g. fuelwood, building materials) suitable for collection between agricultural seasons (Byron and Arnold, 1999). However, most NTFPs are seasonally available (Ngane, 2012), suggesting that there may be limitations for NTFPs to support incomes. Logically, more diverse forests provide a wider range of goods at different times throughout the year (Pritchard et al. 2019). When forests act as safety nets, households are not likely to shift out of poverty in a durable way, but they can smooth income and consumption in the short- to medium-term, which relates to SDG 1 target 5 on risk mitigation.

2.3.4 Movement into poverty

Even though the scope of this report is to discuss the potential for forests to alleviate poverty, it is worth noting that forests may be associated with negative consequences for poverty alleviation. Events associated with living in proximity to trees and forests can negatively impact well-being, sometimes in very extreme ways. Examples include forest fires, crop-raiding by wildlife (Naugton-Treves et al., 1998; Hill, 2018; Chen et al., 2019) and prevalence of zoonotic disease (Kilpatrick and Randolph, 2012; Paige et al., 2014). These events are largely a function of living at the forest frontier, or where forest and agricultural systems merge. Forests have also occasionally been referred to as a 'poverty trap', but the causal mechanisms for how forests can keep people in poverty may be more related to their association with remoteness than to the forest itself (Box 2.1).

Policies affecting actions in forest landscapes where human populations reside, particularly those that create incentives to change access to forests and contribute to land use change, have the potential to reduce well-being. Examples include forest policies related to establishment of strict protected area (Brockington and Wilkie, 2015) and other exclusionary land use policies that limit rights of access to forest goods and services like projects that establish large-scale monoculture commercial *plantations* or land grabs that establish long-term access rights for foreign entities (Fairhead *et al.*, 2012). Additionally, policies that convert forests to agricultural lands or for mining practices may have short-term rewards, but may have medium- or long-term implications for forest derived goods and ecosystem services (Rodrigues *et al.*, 2009). We also note that actions or absence of enforcement of policies and laws can foster and sustain elite capture, support rent seeking behaviour, and or support corrupt practices, all of which undermine opportunities for the poor to move up and out of poverty through forest-based pathways.

2.4 Towards a Deeper Knowledge Base for Understanding Forest-Poverty Dynamics

In this section, we consider the extent that existing frameworks for understanding forest-poverty dynamics address temporal, spatial and contextual dimensions of change. Incomplete knowledge about these three dimensions of forest-poverty dynamics limits our understanding of the relationship between forests and poverty, and serves as a barrier to policymakers and other key stakeholders as they weigh the relative effectiveness of relying on forests and trees to support poverty alleviation efforts (Figure 2.2).

2.4.1 Forest-poverty dynamics across time

The conceptual framing of forest-poverty dynamics to date has primarily focused on the role that forests play in supporting human well-being over time (temporal dynamics) (Figure 2.3, x-axis). For example, the concept of the forest transition (Figure 1.2, Chapter 1) provides a framework for understanding the relationship between economic development and forest cover at large spatial and temporal scales (Duchelle et al., 2014; Mather, 1992; Mather and Needle, 1998; Rudel et al. 2005), but provides only very general insights into the temporal dimension of forest-poverty dynamics. Time is a critical dimension of the forest-poverty dynamics framework that merits further attention as we continue to build the evidence base, allowing us to build on knowledge gained from the dominant cross-sectional or static approach of inquiry. A recent review by Miller and Hajjar (2020) found that less than 17% of 150 studies on forests and poverty had an explicit temporal dimension, and that most covered a relatively short time frame (e.g. 1-5 years). More research is needed that involves repeated measures on the same units of analysis over time, acknowledging that there are a range of temporal units with relevance to understanding forest-poverty dynamics. Analysis involving repeated measures of the



same units over short time periods sheds light on within year seasonal variation, which is important to understanding well-being and risk mitigation functions (Wunder *et al.*, 2014a; Wunder *et al.*, 2018), whereas analysis involving data collected over decades is more appropriate for understanding the impact of major changes in land use and land cover and implications for the role of forests in poverty alleviation. In the absence of such efforts to collect this type of data, forests and trees are often seen as inferior to other poverty alleviation strategies, which may deliver results in the near term, but have less certain long-term impacts.

In recent years the emergence of evidence-based policymaking has motivated an increase in experimental and quasi-experimental programme evaluations that use data collected over relatively short time frames (1-5 years), frequently for the same households or other relevant units of analysis. While the conservation and natural resource management sectors lag behind efforts in the health, education and social protection sectors (Ferraro and Pattanayak, 2006; Caplow *et al.*, 2011; Baylis *et al.*, 2016), calls to assess the impact on poverty of protected area management and forest carbon projects using data over time have been addressed (Andam et al., 2010; Sims, 2010; Clements et al., 2015; Sills et al., 2017). Similarly, the use of regularly collected household survey data, including the World Bank Living Standards Measurement Surveys (LSMS) and the Demographic and Health Survey (DHS) data has made it easier to consider analysing changes in forest and tree use and poverty for representative populations over time (e.g. Jagger and Perez-Heydrich, 2016; Miller et al., 2020b). Despite their potential, however, to date, these datasets have had serious limitations with regard to the attention given to forest goods and services (Wunder et al., 2014a; Bakkegaard et al., 2017; Box 3.2 in Chapter 3). Perhaps most challenging, but also most important for understanding the role of forests and trees in movement out of and into poverty, is the use of temporal data on the decadal scale.

2.4.2 Forest-poverty dynamics across space

We consider three dimensions of spatial analysis as highly relevant for further development of our understanding of forest-poverty dynamics (Figure 2.3, left y-axis). The first is to strive to set studies from relatively small or homogeneous geographies in broader context. The majority of studies focused on forests and poverty take a micro-level approach focusing on data collected from a specific community or communities. Even research programmes focused on forests and people that are global in scope, including PEN (Angelsen et al., 2014; Wunder et al., 2014a) and the International Forest Resources and Institutions (IFRI) (Wollenberg et al. 2007; Poteete and Ostrom, 2008) research programmes, rely on study sites opportunistically selected by partners and collaborators. Framing information on the relationship between forests and poverty in the context of the spatial location of the study site along the forest transition curve is one strategy for assisting policymakers with judgements about the relevance of findings. The framework permits researchers to move beyond general 'context matters' statements, to locate sites along a reference measure, the forest transition curve, and test hypotheses related to their location on the curve (Duchelle et al., 2014). Second, outcomes often accrue at different scales. Impacts observed at highly localised scale may not be observable or easily aggregated to broader scale, or conversely, impacts may only be observed and measured at coarser scales. Lastly, whatever limited knowledge of forest-poverty dynamics exists, it is for rural sites. Trends in reliance on forests and trees as urbanisation expands have received limited attention in the literature, apart from studies focused on biomass energy use in growing urban centres (Zulu, 2010) and use of NTFPs (Schlesinger et al., 2014). Increased attention to urban settings is warranted given population projections for sub-Saharan Africa and South Asia, and the poverty dynamics within Africa's growing cities.

2.4.3 Forest-poverty dynamics across contexts

Last, recognising heterogeneity of context (Figure 2.3., right y-axis) is important for acknowledging and understanding the complexity within and between sites. Within social-ecological systems there are a large number of context variables that influence outcomes (Senge, 1990; Ostrom 2007). Context dynamics refers to the unpredictable ways in which this large number of variables interact with one another shaping outcomes (Salafsky *et al.*, 2002). Context includes biophysical characteristics, market access, social, political, economic and demographic factors. It allows us to address critical questions about who benefits and under what circumstances they benefit from access to forest and tree-based goods and services.

Angelsen and Wunder (2003) note that rural populations tend to be heterogeneous exhibiting a range of different interactions with forest resources. Hence, proposals for interventions should take into account multiple scales, providing reference to particular forest and socio-economic and political conditions. Further, relevant demographic and social dynamics need to be unravelled: how do differing gender, class, ethnicity and other dimensions of social differentiation influence forest uses, dependency, and rules and norms over access to and control of forests (Hecht *et al.*, 2015)?



A woman passes an axe to another community member in the Ecuadorian Amazon Photo © Johan Oldekop

Again, the forest transition curve provides a useful analytical tool for situating evidence of the relationship between forests and poverty in broader context. In addition to variation in forest cover, rates of land use change and broad trends in economic development (e.g. the core of the forest transition), variation in infrastructure, market access, property rights, migration patterns, rule enforcement, etc. are critical to building our understanding of the enabling conditions for movement out of or into poverty via a pathway involving forest-based goods and services. How contextual factors vary along and within stages of the forest transition is important to understand. Whether a given household can move out of poverty depends on a coalescence of enabling conditions, which need to be sustained over time. Even though any one of these pathways could lead to poverty alleviation, pronounced and enduring impacts may require engaging with multiple pathways at the same time (Shyamsundar et al., 2020). However, when and where these pathways are best suited to achieve poverty reduction requires empirical evaluation of impacts (Cheng et al., 2019). Situating indepth and site-specific knowledge in broader context should be an explicit objective of future work on forest-poverty dynamics.

In Table 2.1 we connect the four roles for forests and trees and poverty trajectories with our discussion of temporal, spatial and contextual heterogeneity summarising the key considerations for generation of a robust evidence base on the topic of forest-poverty dynamics. Research employing space for time substitution (e.g. using data from different geographic and social-ecological contexts to provide insights into temporal dynamics) is potentially a fruitful tool for both research and policy in the near term (Sunderland *et al.*, 2017). Emphasising use of the forest transition curve as a guiding stylised framework for spatially locating and contextualising sites provides a mechanism through which policymakers can situate findings relative to the geographic distribution of the population across different stages of economic development and land use change.

Our framing here lays out the path for the analysis that follows in Chapters 3, 4 and 5.

Table 2.1

	FOREST AND TREE ROLES	WELL-BEING TRAJECTORY	TEMPORAL	SPATIAL	CONTEXTUAL
1	Moving out of poverty	Up and out of poverty	Annual/decadal	Within and between landscapes	Enabling conditions
2	Support well-being	Status quo	Seasonal/annual	Within landscape	Who benefits under what circumstances
3	Mitigate risk	Transient poverty	Seasonal/annual	Within landscape	Who benefits under what circumstances
4	Increase poverty	(Further) into poverty	Seasonal/annual/ decadal	Within and between landscapes	Enabling conditions

Charting a path operationalising evidence of forest-poverty dynamics

2.5 Conclusion

There is abundant evidence that forests and trees make crucial contributions to poverty alleviation and human well-being. While the role of forests in moving people permanently out of poverty is likely limited, it requires further exploration with more robust data that takes into account temporal, spatial and contextual heterogeneity. Strong support does exist for the role that forests play in supporting current consumption, gap filling and managing risk. A more broadly conceived definition of well-being allows for greater consideration of the role of forests in poverty reduction and allows for greater recognition of the role of forests in meeting SDG 1 targets and indicators. To date, the theoretical work on poverty dynamics does not give much attention to natural resource endowments, including forests and trees, and the relationship of forests to poverty reduction. One of our main observations is that the conceptual framing of forest-poverty dynamics primarily focuses on the role that forests play in supporting human well-being over time (temporal dynamics). This is a critical dimension of the dynamics framework that merits further attention as we continue to build the evidence base surrounding forest-poverty dynamics. However, we note that forest-poverty dynamics have two other important dimensions: spatial and contextual. In this chapter we provided framing for considering temporal, spatial and contextual dimensions of forest-poverty relationships as part of systems frameworks for building our knowledge base.



Handmade cart used for wood transport in Chile Photo © Jennifer Zavaleta Cheek

2.6 References

- Addison, T., Hulme, D. and Kanbur, R., 2008. Poverty dynamics: measurement and understanding from an interdisciplinary perspective. Manchester: Brooks World Poverty Institute.
- Alkire, S. and Santos, M.E., 2013. A multidimensional approach: Poverty measurement & beyond. Social Indicators Research 112 (2): 239-257.
- Alkire, S., 2007. The Missing Dimensions of Poverty Data: Introduction to the Special Issue, Oxford Development Studies 35 (4): 347-359.
- Andam, K. S., Ferraro, P. J., Sims, K. R., Healy, A. and Holland, M. B., 2010. Protected areas reduced poverty in Costa Rica and Thailand. PNAS 107 (22): 9996-10001.
- Angelsen, A., Jagger, P., Babigumira, R., Belcher, B., Hogarth, N.J., Bauch, S., Börner, J., et al., 2014.
 Environmental Income and Rural Livelihoods:
 A Global-Comparative Analysis. World Development 64: S12–S28.
- Angelsen, A. and Wunder, S., 2003. Exploring the forest--poverty link: key concepts, issues and research implications. Bogor: Center for International Forestry Research (CIFOR).
- Asselin, H., 2015. Indigenous forest knowledge.In: Routledge handbook of forest ecology, edited byK. S. H. Peh, R. T. Corlett and Y. Bergeron. Abingdon:Routledge, pp. 602-612.
- Azariadis, C. and Stachurski, J., 2005. Chapter 5 Poverty Traps. In: Handbook of Economic Growth, edited by:
 S. Bowles, S. Durlauf and K. Hoff. Princeton: Princeton University Press, pp. 295–384.
- Bakkegaard, R. K., Hogarth, N. J., Bong, I. W., Bosselmann, A. S. and Wunder, S., 2017. Measuring forest and wild product contributions to household welfare: Testing a scalable household survey instrument in Indonesia. Forest Policy and Economics 84: 20-28.
- Baland, J. M. and Platteau, J. P., 1996. Inequality and collective action in the commons. Namur: CRED, University of Namur.
- Barbier, E. B., 2019. Institutional Constraints and the Forest Transition in Tropical Developing Countries. International Advances in Economic Research 25(1): 1-18.
- Barbier, E. B., Hochard, J.P., 2019. Poverty-Environment Traps. Environ Resource Econ, 74 : 1239–1271.
- Barrett, C. B., Garg, T. and McBride, L., 2016. Well-Being Dynamics and Poverty Traps. Annual Review of Resource Economics 8: 303–327.
- Baulch, B. and Hoddinott, J., 2000. Economic mobility and poverty dynamics in developing countries. *Journal of Development Studies* 3: 1–24.

- Baylis, K., Honey-Rosés, J., Börner, J., Corbera, E.,
 Ezzine-de-Blas, D., Ferraro, P. J., Lapeyre, R., et al.,
 2016. Mainstreaming impact evaluation in nature conservation. Conservation Letters 9(1): 58-64.
- Beal, T., Tumilowicz, A., Sutrisna, A., Izwardy, D. and Neufeld, L. M., 2018. A review of child stunting determinants in Indonesia. *Maternal & Child* Nutrition 14: e12617.
- Belcher, B. and Kusters, K., 2004. Non-timber forest product commercialisation: development and conservation lessons. In: Forest products, livelihoods and conservation: case studies of non-timber forest product systems, edited by S. Shackleton, C. Shackleton and P. Shanley. Berlin; Heidelberg: Springer, Volume 1 – Asia, 1-22.
- Brockington, D., 2019. Persistent peasant poverty and assets. Exploring dynamics of new forms of wealth and poverty in Tanzania 1999–2018. *The Journal of Peasant Studies* 1-20.
- Brockington, D. and Wilkie, D., 2015. Protected areas and poverty. Philosophical Transactions of the Royal Society B 370: 20140271.
- Byron, N. and Arnold, M., 1999. What Futures for the People of the Tropical Forests? World Development 27: 789–805.
- Caplow, S., Jagger, P., Lawlor, K. and Sills, E., 2011. Evaluating land use and livelihood impacts of early forest carbon projects: Lessons for learning about REDD+. Environmental Science & Policy 14 (2): 152-167.
- Campbell, B. M. and Luckert, M. K., 2002. Uncovering the Hidden Harvest: Valuation methods for Woodland and Forest Resources. London: Earthscan Publications Ltd.
- Carter, M. R. and May, J., 2001. One Kind of Freedom: Poverty Dynamics in Post-apartheid SouthAfrica. World Development 29: 1987–2006.
- Cavendish, W., 2000. Empirical regularities in the poverty-environment relationship of rural households: Evidence from Zimbabwe. World Development 28(11): 1979-2003.
- Chambers, R. and Conway, G., 1992. Sustainable rural livelihoods: practical concepts for the 21st century. IDS Discussion Paper No. 296. Sussex, UK: Institute for Development Studies.
- Chen, X., Zhang, Q., Peterson, M. N. and Song, C., 2018. Feedback effect of crop raiding in payments for ecosystem services. *Ambio* 48: 732-740.
- Cheng, S. H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Rana, P., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. *Environmental Evidence* 8: 3.

- Clements, T. and Milner-Gulland, E. J., 2015. Impact of payments for environmental services and protected areas on local livelihoods and forest conservation in northern Cambodia. *Conservation Biology* 29 (1): 78-87.
- Daniel, K.S., Udeagha, A.U. and Jacob, D.E., 2016. Sociocultural importance of sacred forests conservation in south southern Nigeria. *African Journal of Sustainable Development* 6(2): 251-268.
- den Braber, B., Evans, K. L. and Oldekop, J. A., 2018. Impact of protected areas on poverty, extreme poverty, and inequality in Nepal. Conservation Letters 11 (6): 12576.
- Dercon, S. and Hoddinott, J., 2004. Health, shocks, and poverty persistence. In: *Insurance Against Poverty*, edited by S. Dercon. Helsinki: UN Wider.
- Duchelle, A., Zambrano, A. M. A., Wunder, S., Borner, J. and Kainer, K. A., 2014. Smallholder specialization strategies along the forest transition curve in Southwestern Amazonia. World Development 64: S149-S158.
- Fairhead, J., Leach, M. and Scoones, I., 2012. Green grabbing: a new appropriation of nature? *Journal of Peasant Studies* 39 (2): 237-261.
- Ferraro, P. J. and Pattanayak, S. K., 2006. Money for nothing? A call for empirical evaluation of biodiversity conservation investments. PLoS Biology 4(4): 105.
- Ferraro, P. J. and Hanauer, M. M., 2014. Quantifying causal mechanisms to determine how protected areas affect poverty through changes in ecosystem services and infrastructure. PNAS 111: 4332–4337.
- Fisher, J. A., Patenaude, G., Giri, K., Lewis, K., Meir, P., Pinho, P., Rounsevell, M. D. and Williams, M., 2014. Understanding the relationships between ecosystem services and poverty alleviation: a conceptual framework. Ecosystem services 7: 34-45.
- Fisher, M., Chaudhury, M. and McCusker, B., 2010. Do forests help rural households adapt to climate variability? Evidence from Southern Malawi. World Development 38 (9): 1241-1250.
- Fisher, M. and Shively, G., 2005. Can Income Programs Reduce Tropical Forest Pressure? Income Shocks and Forest Use in Malawi. World Development 33: 1115–1128.
- FAO, 2014. State of the World's Forests: Enhancing the Socioeconomic Benefits from Forests. Rome: FAO.
- Gamfeldt, L., Snäll, T., Bagchi, R., Jonsson, M., Gustafsson, L., Kjellander, P., Ruiz-Jaen, M. C., et al., 2013. Higher levels of multiple ecosystem services are found in forests with more tree species. Nature Communications 4: 1340.
- Godoy, R., Jacobson, M. and Wilkie, D., 1998. Strategies of rain-forest dwellers against misfortunes: the Tsimane'Indians of Bolivia. Ethnology 55-69.

- Golden, C. D., Fernald, L. C. H., Brashares, J. S.,
 Rasolofoniaina, B. J. R. and Kremen, C., 2011.
 Benefits of wildlife consumption to child nutrition in a biodiversity hotspot. PNAS 108: 19653–19656.
- Goodhand, J., 2001. Violent Conflict, Poverty and Chronic Poverty. Chronic Poverty Research Centre Working Paper No. 6. London: Overseas Development Institute.
- Hecht, S., Yang, A. L., Basnett, B. S., Padoch, C. and Peluso, N., 2015. People in motion, forests in transition: trends in migration, urbanization, and remittances and their effects on tropical forests. *CIFOR Occasional Paper No.* 142. Bogor: CIFOR.
- Hill, C. M., 2018. Crop foraging, crop losses and crop raiding. Annual Review of Anthropology 47: 377-394.
- Humphries, S., Holmes, T., Andrade, D. F. C., McGrath, D. and Dantas, J. B., 2020. Searching for win-win forest outcomes: Learning-by-doing, financial viability, and income growth for a community-based forest management cooperative in the Brazilian Amazon. World Development 125: 104336.
- Ickowitz, A., Rowland, D., Powell, B. and Salim, M. A., 2016. Forests, trees, and micronutrient-rich food consumption in Indonesia. *PLOS One* 11 (5): 1-15.
- Ickowitz, A., Powell, B., Salim, M. A. and Sunderland, T. C. H., 2014. Dietary quality and tree cover in Africa. Global Environmental Change 24: 287–294.
- IUCN, 2012. Improving Ecosystem Functionality and Livelihood Experiences in Forest Landscape Restoration and Management. Livelihoods and Landscapes Strategy Working Paper No. 2. Gland: IUCN.
- Jagger, P. and Perez-Heydrich, C., 2016. Land use and household energy dynamics in Malawi. Environmental Research Letters 11 (12): 125004.
- Jagger, P., Luckert, M., Duchelle, A. E., Lund, J. F. and Sunderlin, W. D., 2014. Tenure and Forest Income: Observations from a Global Study on Forests and Poverty. World Development 64: S43–S55.
- Kilpatrick, A. M. and Randolph, S. E., 2012. Drivers dynamics, and control of emerging vector-borne zoonotic diseases. The Lancet 380 (9857): 1946-1955.
- Klasen, S., and Waibel, H., 2015. Vulnerability to poverty in South-east Asia: Drivers, measurement, responses and policy issues. World Development 7 (1): 1-3.
- Krishna, A., 2004. Escaping poverty and becoming poor: who gains, who loses, and why? World Development 32 (1): 121-136
- Krishna, A., 2006. Pathways out of and into poverty in 36 villages of Andhra Pradesh, India. World Development 34 (2): 271-288.
- Krishna, A., 2010. One illness away: why people become poor and how they escape poverty. New York and Oxford: Oxford University Press.

Krishna, A. and Shariff, A., 2011. The Irrelevance of National Strategies? Rural Poverty Dynamics in States and Regions of India, 1993–2005. World Development 39: 533–549.

Kusters, K., Achdiawan, R., Belcher, B. and Ruiz Pérez, M., 2006. Balancing development and conservation? An assessment of livelihood and environmental outcomes of non-timber forest product trade in Asia, Africa, and Latin America. Ecology and Society 11 (2): 20.

Lawlor, K., Sill, E. O., Atmadja, S., Lin, L. and Songwathana, K., 2019. SDG 1: No poverty – Impacts of social protection, tenure security and building resilience on forests. In: Sustainable Development Goals: Their Impacts on Forests and People, edited by P. Katila, C. Colfer, W. de Jong, G. Galloway, P. Pacheco and G. Winkel. Cambridge: Cambridge University Press.

López-Feldman, A., Mora, J. and Taylor, J. E., 2007. Does natural resource extraction mitigate poverty and inequality? Evidence from rural Mexico and a Lacandona Rainforest Community. *Environment and Development Economics* 12(2): 251-269.

Macqueen, D., Bolin, A., Greijmans, M., Grouwels, S. and Humphries, S., 2020. Innovations towards prosperity emerging in locally controlled forest business models and prospects for scaling up. World Development 125: 104382.

Mather, A.S., 1992. The forest transition. Area 24 (4): 367-379.

Mather, A.S. and Needle, C. L., 1998. The forest transition: a theoretical basis. *Area* 30 (2): 117-124.

McKinnon, M. C., Cheng, S. H., Dupre, S., Edmond, J., Garside, R., Glew, L., Holland, M. B., et al., 2016.
What are the effects of nature conservation on human well-being? A systematic map of empirical evidence from developing countries. *Environmental Evidence* 5: 8.

McSweeney, K., 2004. Forest product sale as natural insurance: the effects of household characteristics and the nature of shock in eastern Honduras. Society and Natural Resources 17 (1): 39-56.

McSweeney, K., 2005. Natural insurance, forest access, and compounded misfortune: Forest resources in smallholder coping strategies before and after Hurricane Mitch, northeastern Honduras. World Development 33: 1453–1471.

MEA, 2005. Millennium Ecosystem Assessment. Available at: https://www.millenniumassessment.org/en/ index.html. [Accessed on 9 March 2020].

Miller, D. C. and Hajjar, R., 2020. Forests as pathways to prosperity: Empirical insights and conceptual advances. World Development 125: 104647.

Miller, D. C., Ordoñez, P. J., Brown, S. E., Forrest, S., Nava, N. J., Hughes, K. and Baylis, K., 2020a. The impacts of agroforestry on agricultural productivity, ecosystem services, and human well-being in lowand middle-income countries: An evidence and gap map. Campbell Systematic Reviews 16 (1): 1066. Miller, D.C., Muñoz-Mora, J.C., Rasmussen, L.V. and Zezza, A., 2020b. Do Trees on Farms Improve Household Well-Being? Evidence From National Panel Data in Uganda. Frontiers in Forests and Global Change. 3:101.

Moran, R., 2003. Escaping the poverty trap: investing in children in Latin America. Washington, D.C.: Inter-American Development Bank.

Munyi P. and Mutta D., 2007. Protection of community rights over traditional knowledge: implications for customary laws and practices in Kenya. Nairobi: Science Press.

Naughton-Treves, L., Treves, A., Chapman, C. and Wrangham, R., 1998. Temporal patterns of crop raiding by primates: linking food availability in croplands and adjacent forests. *Journal of Applied Ecology* 35: 596-606.

Neumann, R. P. and Hirsch, E., 2000. Commercialization of non-timber forest products: review an analysis of research. Bogor and Rome: CIFOR and FAO.

Newton, P., Kinzer, A., Miller, D.C., Oldekop J.A. and Agrawal, A., 2020. The number and spatial distribution of forest-proximate people globally. *One Earth* 3(3): 363-370.

Ngane B. K., 2012. Seasonality of non-timber forest products in the Kupe mountain region of South Cameroon. Scientific Research and Essays 7 (18): 1786-1797.

Ngoufo, R., Yongyeh, N., Obioha, E., Bobo, K., Jimoh, S. and Waltert, M., 2014. Social norms and cultural services - community belief system and use of wildlife products in the Northern periphery of the Korup National Park, South-West Cameroon. *Change and Adaptation in Socio-Ecological Systems* 1 (1): 26-34.

Nussbaum, M., 2000. Women and Human Development. Cambridge: Cambridge University Press.

OECD, 2017. Bridging the gap: Inclusive growth 2017 update report. Available at: http://www.oecd.org/ inclusive-growth/Bridging_the_Gap.pdf. [Accessed on 7 March 2020].

Ostrom, E., 1990. Governing the commons: the evolution of institutions for collective action, The Political economy of institutions and decisions. Cambridge and New York: Cambridge University Press.

Ostrom, E., 2007. A diagnostic approach for going beyond panaceas. PNAS 104 (39): 15181-15187.

Oteng-Yeboah, A., Mutta, D., Byarugaba, D. and Mala, W. A., 2012. Africa. In: *Traditional Forest-Related Knowledge*, edited by J. A. Parrotta and R. L. Trosper. Dordrecht, Netherlands: Springer, pp. 37-78.

Paige, S. B., Frost, S. D. W., Gibson, M. A., Jones, J. H., Shankar, A., Switzer, W. M., Ting, N. and Goldberg, T. L., 2014. Beyond bushmeat: Animal conflict, injury, and zoonotic disease risk in western Uganda. *EcoHealth* 11: 534-543.

- Parrotta, J., Wildburger, C. and Mansourian, S., 2012. Understanding relationships between biodiversity, carbon, forests and people: The key to achieving REDD+ objectives. A global assessment report prepared by the Global Forest Expert Panel on Biodiversity, Forest Management and REDD+. IUFRO World Series. Vienna: IUFRO.
- Pattanayak, S. K. and Sills, E. O., 2001. Do Tropical Forests Provide Natural Insurance? The Microeconomics of Non-Timber Forest Product Collection in the Brazilian Amazon. Land Economics 77: 595–612.
- Poteete, A. and Ostrom, E., 2008. Fifteen years of empirical research on collective action in natural resource management: Struggling to build large-N databases based on qualitative research. World Development 36 (1): 176-195.
- Pritchard, R., Grundy, I. M., van der Horst, D. and Ryan, C.M., 2019. Environmental incomes sustained as provisioning ecosystem service availability declines along a woodland resource gradient in Zimbabwe. World Development 122: 325-338.
- Pritchard, R., Grundy, I. M., van der Horst, D., Dzobo, N. and Ryan, C. M., 2020. Environmental resources as 'last resort' coping strategies following harvest failures in Zimbabwe. World Development 127: 104741.
- Rasmussen, L., Fagan, M. E., Ickowitz, A., Wood, S. L.
 R., Kennedy, G., Powell, B., Baudron, F., et al., 2019.
 Forest pattern, not just amount, influences dietary quality in five African countries. *Global Food Security* 25: 100331.
- Raymond, C.M., Bryan, B.A., MacDonald, D.H., Cast, A., Strathearn, S., Grandgirard, A. and Kalivas, T., 2009. Mapping community values for natural capital and ecosystem services. *Ecological economics* 68 (5): 1301-1315.
- Rodrigues, R. R., Lima, R. A., Gandolfi, S. and Nave, A.G., 2009. On the restoration of high diversity forests:
 30 years of experience in the Brazilian Atlantic Forest. Biological Conservation 142 (6): 1242-1251.
- Rudel, T. K., Coomes, O. T., Moran, E., Achard, F., Angelsen, A., Xu, J. and Lambin, R., 2005. Forest transitions: towards a global understanding of land use change. *Global Environmental Change* 15 (1): 23-31.
- Rutte, C., 2011. The sacred commons: Conflicts and solutions of resource management in sacred natural sites. *Biological Conservation* 144: 2387–2394.
- Salafsky, N., Margoluis, R., Redford, K. H. and Robinson, J. G., 2002. Improving the practice of conservation: a conceptual framework and research agenda for conservation science. *Conservation Biology* 16 (6): 1469-1479.
- Salzman, J., Bennett, G., Carroll, N., Goldstein, A. and Jenkins, M., 2018. The global status and trends of payments for ecosystem services. *Nature Sustainability* 1 (3): 136-144.

- Santika, T., Wilson, K. A., Meijaard, E., Budiharta,
 S., Law, E. E., Sabri, M., Struebig, M., et al., 2019.
 Changing landscapes, livelihoods and village welfare in the context of oil palm development.
 Land Use Policy 87: 104073.
- Schlesinger, J., Drescher, A. and Shackleton, C. M., 2015. Socio-spatial dynamics in the use of wild natural resources: evidence from six rapidly growing medium-sized cities in Africa. Applied Geography 56: 107–115.
- Scoones, I., 1998. Sustainable Rural Livelihoods. A Framework for Analysis. IDS Working Paper No. 72. Sussex: Institute of Development Studies.
- Sen. A., 1985. Well-being, agency and freedom: The Dewey Lectures 1984. The Journal of Philosophy 82 (4): 169-221.
- Sen, A., 1999. Development as Freedom. New York, NY: Anchor Books.
- Senge, P. M., 1990. The Fifth Discipline: The art and practice of the learning organization. New York, NY: Doubleday.
- Shackleton, C. M., and Shackleton, S. E., 2007. The importance of woodlands and forest in rural livelihoods and poverty alleviation in South Arica. Forest Policy and Economics 9 (5): 558-577.
- Shackleton C., Shackleton S. and Shanley P., 2011. Building a Holistic Picture: An Integrative Analysis of Current and Future Prospects for Non-timber Forest Products in a Changing World. In: Non-Timber Forest Products in the Global Context. Tropical Forestry, vol 7, edited by S. Shackleton, C. Shackleton, and P. Shanley. Berlin: Springer.
- Shyamsundar, P., Ahlroth, S., Kristjanson, P. and Onder, S., 2020. Supporting pathways to prosperity in forest landscapes – A PRIME framework. World Development 125: 104622.
- Sills, E. O., de Sassi, C., Jagger, P., Lawlor, K., Miteva,
 D. A., Pattanayak, S. K. and Sunderlin, W. D., 2017.
 Building the evidence base for REDD+: Study
 design and methods for evaluating the impacts
 of conservation interventions on local wellbeing. Global environmental change 43: 148-160.
- Sims, K. R., 2010. Conservation and development: Evidence from Thai protected areas. Journal of Environmental Economics and Management 60 (2): 94-114.
- Sunderland, T., Abdoulaye, R., Ahammad, R., Asaha, S., Baudron, F., Deakin, E., Duriaux, J. Y., *et al.*, 2017. A methodological approach for assessing cross-site landscape change: Understanding socio-ecological systems. *Forest Policy and Economics* 84: 83-91.
- Sunderland, T., Achdiawan, R., Angelsen, A., Babigumira, R., Ickowitz, A., Paumgarten, F., Reyes-Garcia, V., and Shively, G., 2014. Challenging perceptions about men, women and forest product use: A global comparative study. World Development 64 (1): S56-S66.

- Sunderlin, W. D., Angelsen, A., Belcher, B., Burgers, P., Nasi, R., Santoso, L. and Wunder, S., 2005. Livelihoods, forests, and conservation in developing countries: An Overview. World Development 33: 1383–1402.
- Sundström, A., 2016. Understanding illegality and corruption in forest governance. *Journal of environmental management* 181: 779-790.
- Tata, C. Y., Ickowitz, A., Powell, B. and Colecraft, E.K., 2019. Dietary intake, forest foods, and anemia in Southwest Cameroon. PloS one 14 (4): 0215281.
- Tincani, L. S., 2012. Resilient livelihoods: adaptation, food security and wild foods in rural Burkina Faso. PhD Thesis. London: School of Oriental and African Studies, University of London.
- Tiominar, B., 2011. Plantations and Poverty: Notes from a Village Deep in Oil Palm Territory. Down to Earth. Available at: https://www.downtoearth-indonesia. org/story/plantations-and-poverty-notes-villagedeep-oil-palm-territory. [Accessed on 7 March 2020].
- Toth, R., 2015. Traps and thresholds in pastoralist mobility. *American Journal of Agricultural Economics* 97 (1): 315-332.
- United Nations, 2020. Transforming our World: the 2030 Agenda for Sustainable Development. Available at: https://sustainabledevelopment.un.org/post2015/ transformingourworld/. [Accessed on 7 March 2020].
- Vira, B., Wildburger, C. and Mansourian, S., 2015. Forests, Trees and Landscapes for Food Security and Nutrition: A Global Assessment Report. Vienna: IUFRO.
- Wells, G. J., Corbera, E., Fisher, J. and Ryan, C., 2020. Success at the farm-level in payments for ecosystem services: monetary incentives, participatory processes and institutional adaptation all matter (preprint submitted to EarthArXiv and Elsevier).
- Wollenberg, E., Merino, L., Agrawal, A. and Ostrom, E., 2007. Fourteen years of monitoring communitymanaged forests: learning from IFRI's experience. International Forestry Review 9 (2): 670-684.
- World Bank, 2015. A Measured Approach to Ending Poverty and Boosting Shared Prosperity: Concepts, Data and the Twin Goals. Washington, D.C.: The World Bank.
- World Bank, 2018a. Poverty and Shared Prosperity: Piecing Together the Poverty Puzzle. Washington, D.C.: World Bank.
- World Bank, 2018b. World Bank Annual Report 2018. Washington, D.C.: World Bank.
- Wunder, S., 2001. Poverty Alleviation and Tropical Forests--What Scope for Synergies?. World Development 29 (11): 1817-1833
- Wunder, S., Angelsen, A. and Belcher, B., 2014a. Forests, Livelihoods, and Conservation: Broadening the Empirical Base. World Development 64: S1–S11.

- Wunder, S., Börner, J., Shively, G. and Wyman, M., 2014b. Safety Nets, Gap Filling and Forests: A Global-Comparative Perspective. World Development 64: S29–S42.
- Wunder, S., Noack, F. and Angelsen, A., 2018. Climate, crops, and forests: A pan-tropical analysis of household income generation. Environment and Development Economics 23 (3): 279-297.
- Yaqab, S., 2000. Poverty Dynamics in Developing Countries. Development Bibliography 16. Sussex: Institute for Development Studies.
- Zulu, L. C., 2010. The forbidden fuel: charcoal, urban woodfuel demand and supply dynamics, community forest management and woodfuel policy in Malawi. *Energy Policy* 38: 3717–3730.



Chapter 3

Forest-Poverty Dynamics: Current State of Knowledge

Coordinating Lead Authors: Onja H. Razafindratsima, Judith Kamoto

Lead Authors: Erin Sills, Doris N. Mutta, Conghe Song, Terry Sunderland, Gillian Kabwe and Casey Ryan Contributing Authors: Sarah E. Brown, Patricia M. Kristjanson, Maria Brockhaus, Patrice Bigombe Logo, Blaise-Pascal Ntirumenyerwa Mihigo, Symphorien Ongolo, Jianmin Xiao and Yue Zhao

TABLE OF CONTENTS

3.1 Introduction	56
3.2 Movement Out of Poverty	57
3.3 Role of Forests in Maintaining Human Well-Being	59
3.4 Risk Management	72
3.5 Forest Negative Externalities	73
3.6 Negative impacts of deforestation and forest ecosystem disturbance	76
3.7 Conclusions	77
3.8 References	78

Abstract

This chapter reports on evidence about the role of forests and trees in alleviating poverty and supporting wider human well-being. It considers how, whether, where, when and for whom forests and trees are important in forest-poverty dynamics. We organise the evidence according to four possible relationships between forest products and ecosystem services and poverty: 1) helping households move out of poverty; 2) supporting well-being through subsistence, food security and cultural and spiritual values; 3) mitigating risks; and 4) decreasing well-being by generating negative externalities that could significantly contribute to trapping or moving households into poverty. The evidence shows that these relationships are strongly context-dependent, varying with geography and social, economic and political contexts. However, across contexts, we most commonly observe that forest and tree products and services help the poor to secure and stabilise their livelihoods, rather than either helping them exit poverty or driving them into poverty.

3.1 Introduction

Forests and tree-based systems⁵ are vital resources for sustaining human populations around the world (Sjaastad et al., 2005; Ali, 2018; Cheng et al., 2019). In tropical countries, forests contribute an average of 22% of household income in communities near forests (Angelsen et al., 2014), which generally have high poverty rates (Sunderlin et al., 2008). Agroforestry systems are also commonly used by poor farmers and have the potential to increase their incomes, especially with improved genetics and markets (Leakey et al., 2005). Thus, both sustainable forest management and agroforestry are widely claimed to be important for achieving the first SDG (Griggs et al., 2013; Lawlor et al., 2019). In this chapter, we assess the available evidence for these claims. To set the stage, we first review some of the central narratives and myths about poverty and forests.

One of these central narratives is that rural populations who rely on shifting cultivation and pastoralism, including many forest-proximate populations, need to be 'settled' in fixed communities to develop. This is rooted in the idea that their traditional grazing and small-scale cultivation systems trap them in poverty and are responsible for deforestation, despite evidence to the contrary (Curtis et al., 2018; Thu et al., 2018; Dressler et al., 2020). This framing of smallholders as responsible for deforestation has persisted since colonial rule and has been used to justify claims on forests for large-scale production of global commodities such as timber, at the expense of local forest stewards (Dove, 1983; Doolittle, 2007). This discourse remains common in debates over climate change (Weatherley-Singh and Gupta, 2015; Skutsch and Turnhout, 2020). The potential of shifting cultivation systems to generate joint benefits for *livelihoods* and climate change *mitigation*, e.g. in building below-ground carbon stocks, has been largely ignored (Ickowitz, 2006; van Vliet *et al.*, 2012; Bruun *et al.*, 2017; Dressler *et al.*, 2017; Bruun *et al.*, 2018).

In contrast, agroforestry has been widely accepted as a way to achieve the 2030 development agenda (Garrity, 2004; Waldron *et al.*, 2017; Agroforestry Network, 2018). Agroforestry systems are appealing because they provide a suite of products and services that contribute to *poverty alleviation* and improved *human well-being*. However, agroforestry is just one of a whole spectrum of trees on farms and in landscapes that can make these contributions.

Another theme in the literature has been the potential for harvest and sale of diverse *non-timber forest products* (NTFPs) to both conserve forests and alleviate poverty. As we summarise below, there is substantial evidence that poor households use NTFPs to maintain their socio-economic and cultural status, but less evidence that they can leverage them to move out of poverty. And while forest products have been shown to help smooth income and consumption, it is not clear whether and when they are the preferred insurance mechanism. In many settings, forests may be more important for *ecosystem services* that are inputs to quality of life and agricultural production, rather than as sources of forest products.

This chapter summarises current knowledge of the role of forests and trees in poverty dynamics, considering the full range of products and services that are sold, consumed, or used as production inputs. Other than the formal timber sector (Box 3.1), most contributions of forests are either excluded or not attributed to forests in the official economic statistics that are the basis for national poverty rates (Box 3.2). Thus, we draw on the scientific literature for evidence on how, whether, where, when and for whom forests and trees play the four possible roles in poverty dynamics posited in Chapter 2: (1) helping people move out of poverty; (2) supporting well-being through subsistence, food security and cultural and spiritual values; (3) mitigating risks, and (4) decreasing well-being by generating negative externalities that could significantly contribute to trapping or moving households into poverty. For this fourth role, we focus on evidence regarding whether the forest itself has negative effects on local communities (Lyytimäki, 2015), distinct from the negative effects associated with the process of deforestation (reviewed briefly in Section 3.6) and with the imposition of strict forest protection rules that exclude local people (Byg et al., 2017; Poudyal et al., 2018), which are addressed in the context of protected areas in Chapter 5.

We find ample evidence that forest ecosystem goods and services affect poverty dynamics, with some evidence on how such dynamics vary with geography and socio-demographics. We focus on differences in how the dynamics play out for men and women. There is relatively more evidence on forest products as part of the second and third roles and relatively less evidence both on services and on the first and fourth roles. This is reflected in the varying lengths of the following sections on the four forest-poverty dynamics. In each section, we synthesise the existing literature, with emphasis on regions with the highest poverty rates (sub-Saharan Africa), the highest poverty headcounts (South Asia), and the most dramatic reductions in poverty in recent decades (China).



Firewood is a critical resource for rural households in many countries Photo © Nelson Grima

Box 3.1

Formal timber sector

Focusing on the region with the highest poverty rates, FAO estimates that 79 million m³ of wood was harvested as industrial timber in Africa in 2018 (FAO, 2019), but this is widely recognised as a substantial underestimate due to large-scale illegal felling and trade of logs in many countries. The formal forest sector contributed less than 1% of the total GDP of sub-Saharan Africa in 2011, rising above 10% in only one country (Liberia) (FAO, 2014). These industrial wood harvests are destined both for growing regional markets and for export, largely to China. In 2009, 78% of Africa's timber exports were bound for the Chinese market, having risen from 35% in 2000 (IIED, 2014). In turn, estimates for the extent of illegal logging are high, but difficult to quantify and confirm by its very nature (Kleinschmit et al., 2016).

During the decade from 1990 to 2000, about half a million people were directly employed in the formal, primary wood production and wood industry sector in Africa (Lebedys and Yanshu, 2014). This represented a small (<1%) and declining fraction of the labour force (FAO, 2005; Whiteman and Lebedys, 2006; FAO, 2014). However, the FAO also estimates that at least three times more people are employed in the informal sector, mainly related to fuelwood and charcoal, than in the formal forestry sector (FAO, 2014). In particular, the charcoal trade accounts for a large share of incomes within the informal forest products sector (Mwampamba et al., 2013; FAO, 2014; Jones et al., 2016; Chiteculo et al., 2018) and may provide jobs for millions of people.

3.2 Movement Out of Poverty

Forests and trees in the landscape could help reduce the proportion of people living in poverty, by enabling households to increase their incomes through sales of forest and tree products (Belcher, 2005). It is difficult to evaluate whether, where and for whom this has occurred without longitudinal data. Angelsen *et al.* (2014) and Miller and Hajjar (2020) point to panel survey data as

particularly valuable for understanding this role of forests, trees and agroforestry. Lacking long term panel data, researchers typically examine contributions to household income, expenditures or assets, rather than the role of forests in lifting people out of poverty over time (Miller et al., 2017). Based on this largely cross-sectional and descriptive literature, the consensus view that emerged in the 2000s was that fundamental structural barriers generally prevent poor people from using forest and tree products to exit poverty (Wunder, 2001; Belcher et al., 2005; Pérez, 2005). This is consistent with a recent literature review (Miller and Hajjar, 2020), which found only 12 studies that "described a social group (e.g., household, community or region) as moving out of poverty due at least in part to forests," i.e. through sales of timber or non-timber forest products. In the context of this pessimism about the potential for substantial numbers of people to use forests and trees to exit poverty, we seek to identify conditions that have allowed poor people to leverage tree and forest products to climb out of poverty. In these case studies, we give careful consideration to who has participated, i.e., who has benefitted. Ongoing policy changes discussed in Chapter 5 are also creating new conditions and offering new ways that the poor can leverage forests to raise their incomes above the poverty line, such as payments for ecosystem services (see Box 5.2 on the Conversion of Cropland to Forest Programme in China).



Harvesting bush mango (Irvingia gabonensis) for local consumption and trade, Ekuri, Nigeria Photo © Terry Sunderland

Collecting national information on the socio-economic dimensions of forests and trees

Box 3.2

National statistical offices in most countries conduct household surveys to inform national decision-making on poverty and livelihood issues, but these usually collect little to no information on the use and benefits of forests and trees (FAO *et al.*, 2016; Miller *et al.*, 2017). This absence of information means that the contribution of forests and trees outside forests to households' welfare, and their role in *poverty reduction* often remain hidden (Scoones *et al.*, 1992; Luckert and Campbell, 2012).

Recognition of the need for national information on the socio-economic dimensions of forest and tree resources has spurred the development of multiple new tools in recent years. Building from foundational work on measuring forest livelihoods (Cavendish, 2000; Wollenberg et al., 2007; Angelsen et al., 2012), a team of experts from FAO, CIFOR, IFRI (International Forestry Resources and Institutions) and PROFOR (Program on Forests) and the LSMS-ISA (Living Standards Measurement Study - Integrated Surveys on Agriculture) initiative at the World Bank has now published a set of forestry modules and guidebooks on their use (FAO et al., 2016). The Forestry Modules were piloted in Indonesia, Nepal and Tanzania with national-scale implementation completed in Turkey. These modules are now complemented by a parallel effort to develop a set of modules to capture the socio-economic values of trees on farms, which was piloted in Mali (Miller et al., 2019). They have also informed related tools, such as Forest-SWIFT (Survey of Well-being via Instant and Frequent Tracking) (World Bank, 2019a) and LivWell (FLARE, 2019), capable of more rapidly capturing focused data on forest reliance and poverty.

The main goal of these modules and the guidebooks describing their use is to provide a mechanism for national statistical offices to collect forest- and tree-related information in regular national-level household socio-economic surveys, thereby filling key information gaps on the contribution of forests and other environmental resources to income, welfare and livelihoods. The modules are adaptable to different scales from local communities through to a whole country. Therefore, they are also relevant to a range of other potential users, including researchers, donors, other government agencies (e.g. forestry departments) and non-governmental organisations (NGOs) interested in improved information on the socio-economic dimensions of forests and trees in broader landscapes.

In one of the earliest reviews that concluded forests have a limited role to play in lifting people out of income poverty, Wunder (2001) argued that timber is the product most likely to generate enough profits to reduce poverty, but that those profits are largely captured by capital intensive and politically powerful actors. In Box 3.5, we describe a unique case in which poor communities have profited from timber, partly because they hold secure collective tenure rights. This points to the decentralisation of forest ownership and management rights as one way to enhance the role of forests in poverty alleviation, consistent with systematic reviews that have found decentralisation effectively increased local incomes in the few cases where it has been rigorously evaluated (Samii et al., 2014). While acknowledging that "forest-based development success stories ... are rare, especially among the developing countries," Palo et al. (1999) pointed to Finland where timber exports "played a vital role in economic development." In fact, the timber sector contributed to the economic development of many countries in northern Europe and North America that now have among the highest incomes in the world. These same countries continue to benefit from the trade in timber harvested in the Global South, thus contributing to the structural barriers that limit the ability of the poor in the Global South to leverage timber resources to exit poverty (Box 3.6).

Non-timber forest products are generally more accessible to poor households but offer only low returns to their labour (Wunder, 2001). López-Feldman and Wilen (2008) developed a theoretical model and examined a case study in Mexico showing that the poorest households, with the lowest opportunity costs of time and fewest alternative income-generating opportunities, are the most likely to be engaged in collection and sale of NTFPs. While this means that NTFP income flows disproportionately to the poor, it does not necessarily lift those households out of poverty, especially when forest resources are open access (and therefore susceptible to overexploitation and dissipation of rents) and when access to transportation, markets and other public services are limited (Belcher et al., 2005). As Shackleton et al. (2008) concluded, "while key in enhancing the livelihood security of the poorest households, these products were unlikely to provide a route out of poverty for most, although there were exceptions." These exceptions can be created by more equitable forest policy (Larson and Ribot, 2007) or better market access (Scherr et al., 2003). Market access is in turn shaped by consumer demand, globalisation, demographic trends, and expansion of communications or road infrastructure, which therefore can open windows of opportunity for poor people to harness forest products as a way to exit poverty, as illustrated by shea butter in Box 3.7.

As with forests, the rural poor face structural barriers to harnessing agroforestry to exit poverty. Russell and Franzel (2004) point to the need to expand market opportunities for smallholders, including for high-value products such as vanilla, as described in Box 3.8.

3.3 Role of Forests in Maintaining Human Well-being

This section reviews the evidence on how the poor secure their well-being by drawing on the multiple benefits of forests and trees, including both final and intermediate goods and services, both traded in markets and consumed. Standard measures of income and poverty as reflected in official statistics only credit forests for income from final goods and services traded legally in markets (Box 3.1). However, forests and trees also provide both tangible and intangible inputs to production and to household well-being (e.g. fodder, pollination, food and sacred places), which do not pass through markets and therefore are excluded from national income accounts.

3.3.1 Wood products

The harvest and processing of timber provide employment and income to millions of people worldwide. Exact numbers are difficult to obtain because this mostly occurs through the informal sector. FAO (2014) estimated that there were 54.2 million people employed in forestry, logging, and secondary manufacturing (sawn wood, panels, and paper) worldwide in 2010. In a report for the FAO Farm and Forest Facility, Verdone (2018) estimated that smallholders produced USD 2-4

Box **3.3**

Bushmeat as an example of non-timber forest products

Wild meat or bushmeat harvesting around the world remains an important source of protein and, more importantly, micronutrients, for many rural households and vulnerable populations such as indigenous groups and children (Swamy and Pinedo-Vasquez, 2014). Poor families living in rural areas, isolated from markets, rely heavily on wildlife - including bushmeat, insects and fish - for food, nutrition, as well as an income source (FAO, 2013; Oishi and Hagiwara, 2015; McIntyre et al., 2016; Wilkie et al., 2016; Lo et al., 2019). Bushmeat harvesting is important in helping rural households to achieve healthy nutrition and food security (Golden et al., 2014; Cawthorn and Hoffman, 2015; van Vliet et al., 2015; Reuter et al., 2016; Nielsen et al., 2018). In the Abun region of West Papua, Indonesia, for example, hunting has proved to be an important factor

billion in timber products per year (an order of magnitude less than just the single largest forest products company), but USD 76-309 billion/year worth of charcoal and firewood. Charcoal provides energy for over 80% of urban households in sub-Saharan Africa (Zulu and Richardson, 2013; Agyei *et al.*, 2020), and the informal charcoal industry provides employment to over half a million people in rural areas in Kenya alone (Njenga *et al.*, 2013).

In addition to generating employment and valuable forest products, harvesting timber and fuelwood for household use can reduce household expenditures and offer an additional income source. By growing trees on their own land, in woodlots or agroforestry systems, farmers can reduce the time and labour spent on gathering fuelwood (Njenga et al., 2017), with the additional public benefit of reducing pressure on natural forests (Iiyama et al., 2014). However, the ability of trees on farms to supply enough fuelwood for household consumption is dependent on the size of the landholding (Ndayambaje and Mohren, 2011), and thus many people remain reliant on wood collected off-farm. Even when trees on farms produce sufficient wood, it may be more lucratively sold as timber, poles or specialty products than used for fuelwood (e.g. in the case of Acacia catechu in India). Studies show

in fighting food insecurity, as wild meat accounted for 49% of the diets of respondents (Pattiselanno and Lubis, 2014). While bushmeat is often consumed locally by hunters and their households (Wilkie et al., 2005; Agustino et al., 2011), the surplus is sold to both other community members and traders, with the latter often re-selling in cities (Nasi et al., 2011; Nielsen et al., 2017). In addition, the harvest of bushmeat may have ancillary benefits for agriculture, by reducing predation on crops, livestock and people working in remote fields (Wilkie et al., 2011; Rentsch and Damon, 2013; Lindsey et al., 2015). Harvesting of bushmeat has long been controversial due to concerns over its conservation impacts, when endangered species are targeted, and its long-term sustainability (Agustino et al., 2011; Lindsey et al., 2014).

that harvest and sale of timber (Antinori and Bray, 2005; Sikor and Baggio, 2014) and other wood-related forest products are important ways in which households augment their incomes (Humphries *et al.*, 2020; Macqueen et al., 2020).

3.3.2 Non-timber forest products

Compared to timber, non-timber forest products, including fuelwood (generally defined separately to NTFPs) and a wide range of other products, are less likely to enter the market economy and thus, less likely to be recorded in official economic accounts. This has meant that "the constant and profound reliance on forests by local people was under-observed by both Bureaux of Statistics and Forestry Departments in government" (Shepherd et al., 2020), although there are efforts to remedy this (FAO et al., 2016; Sorrenti, 2017). In contrast, the scientific literature of the past 25 years has provided a much richer understanding of the roles played by NTFPs in forest-poverty dynamics. In reviewing this literature, we include both plant and animal products from both forests and trees outside the forest. In many forest ecosystems across the world, including West and Central Africa, Brazil, Peru, India, and Indonesia, wild animals are important as both high-value market products and critical sources of protein (Box 3.3). In Section 3.2.4, we highlight the ways that households use NTFPs to increase their food security and improve their nutritional status, as perhaps the most important contribution of this category of forest benefits to supporting human well-being.

Numerous studies have found that poor, rural populations are disproportionately dependent on NTFPs to meet their basic needs (Shackleton and Shackleton, 2004; Belcher *et al.*, 2005; Heubach *et al.*, 2011; Vira and Kontoleon, 2012; Wunder *et al.*, 2014; Leßmeister *et al.*, 2018). In many cases, rural people who depend on forest products live in remote areas with limited access to basic infrastructure, such as motorable roads, making it difficult to access markets and other services (Belcher *et al.*, 2005).

NTFPs are generally managed as open access resources and can be harvested using low-cost and/or traditional technologies (Belcher et al., 2005). For rural dwellers with little financial and physical capital, the affordability and low barrier to entry of NTFP collection make them a viable livelihood strategy. As discussed in Section 3.4, their specific role in household livelihoods is often as a safety net and buffer during times of need such as natural disasters, crop failure, or family illnesses and periods of financial struggle (Leßmeister et al., 2018). Closely related to the role of safety net, NTFPs are also used for seasonal gap filling, i.e., they are collected and sold seasonally based on the availability of time and labour that fluctuates with crop harvesting and planting seasons (Arnold et al., 2011; Leßmeister et al., 2018).

NTFPs play multiple roles in rural livelihoods. They can be collected and used directly for food, medicine, home construction and other traditional purposes. Studies have demonstrated that people who live near areas with more forest and tree cover have more diverse and nutritious diets (Powell et al., 2011; Ickowitz et al., 2014; Powell et al., 2015b; Baudron et al., 2017; Hall et al., 2019). The micronutrients provided by forest foods improve health outcomes and prevent stunting and impairments of cognitive development (Johnson et al., 2013; Ruel and Alderman, 2013; Vinceti et al., 2013). NTFPs also serve as inputs to production, e.g. fodder, mulch and poles for constructing fences. Both intermediate and final products may also be sold, helping generate cash to pay school fees, purchase food from markets, and acquire agriculture inputs (Arnold et al., 2011; Kar and Jacobson, 2012; Hall et al., 2019).

One particularly important role of forests is as a 'natural pharmacy' or source of medicinal plants that play important health care roles for people living in remote rural areas where modern medicine is not accessible. A significant proportion of the population in tropical Asia, Africa and Latin America (Colfer et al., 2006) relies on medicinal plants that form an integral part of their primary health care systems because of affordability, access and effectiveness. As the oldest known health care products, they constitute the medicines used by up to 75-80 % (RAFI, 1994; Ten Kate and Laird, 2020) of the population in developing countries, about 3.5 billion people (WHO, 1993). Cunningham (1993) estimates that 70-80% of Africans consult traditional medical practitioners for health care. Colfer et al. (2006) have documented a large body of literature on the wide use of medicinal plants in traditional health care systems thus contributing to the well-being of rural forest-dependent people in most regions of the world. Also, the market for them is large and expanding, meaning that they can also generate cash income. Further, medicinal plants are important for pharmacological research and drug development when they are used as basic materials for the synthesis of drugs or as models for pharmacologically active compounds, thus offering the opportunity for increased income for custodians and knowledge holders of such plants through bioprospecting ventures. However, with increased deforestation and forest degradation, many such plants are threatened with extinction leading to a loss of health care benefits for those who depend on them.

The literature on the contribution of NTFPs to rural livelihoods is dominated by local level case studies (Angelsen et al., 2014). Some studies have found high dependence on NTFPs (Pattanayak and Sills, 2001; McSweeney, 2004; Debela et al., 2012), but results across studies even within the same region can differ drastically (Leßmeister et al., 2018). This is partly because the diversity of NTFPs and the level of dependency vary greatly with the local context. Belcher and Kusters (2004) also attribute this inconsistency to the lack of an agreed-upon definition of NTFPs and variation in focus, scale, approach and methodology. For instance, studies vary in whether they include relatively low-value products collected in high volumes like fodder, mulch and fuelwood. Some studies focus on specific NTFPs and extrapolate claims on NTFP dependence based on those select products (Belcher and Kusters, 2004; Belcher et al., 2005; Ahenkan and Boon, 2011; Leßmeister et al., 2018).

To generate a more global understanding of NTFPs, the Center for International Forestry Research's (CIFOR) Poverty Environment Network (PEN) applied consistent methods to estimate "environmental income" derived from forests and other non-cultivated lands across sites in 24 developing countries (Angelsen *et al.*, 2014). Across their study sites, environmental income accounts for an average of 28% of household income, 77% of which originated from forests (Angelsen *et al.*, 2014). Only 10.4% of households in the sample used environmental resources, predominantly forests, as their primary safety net (Wunder *et al.*, 2014), as discussed further in Section 3.4.

3.3.3. Products from trees on farms

More than 43% of agricultural land globally has at least 10% tree cover on-farm, and thus, trees on farms affect the livelihoods of hundreds of millions of farmers (Zomer et al., 2016). Agroforestry practices directly contribute to increased income through sales of tree products, increased yields, or payments for sustainable land-use practices through payments for ecosystem services (PES) and certification programmes. Agroforestry and trees on farms can produce high-value tree crops, such as rubber, coffee, cacao, leaves, cashews, macadamia and shea nuts. Rubber agroforestry, for instance, is widely practised in South and Southeast Asia as a low-input production system that generates significant income and can enhance tenure security (Gouyon et al., 1993). Rubber can be intercropped with food crops, fruit trees or timber species in diverse agroforestry systems, which can substantially increase net farm incomes as well as provide resiliency to rubber price fluctuations and promote environmental benefits (Viswanathan, 2008; Somboonsuke et al., 2011; Jessy et al., 2017; Kenney-Lazar et al., 2018). Smallholder farmers may earn two to six times more from rubber monoculture systems with a high-value intercrop, such as pineapples, corn, custard apple, salacca or rice (Somboonsuke et al., 2011). Barriers limiting the adoption of agroforestry systems including their technical complexity and increased labour and input requirements. The economic viability of these systems is also debated (Kenney-Lazar et al., 2018).

Agroforestry can increase income and food security by providing various foods for household consumption and sale, particularly from the use of multi-strata agroforestry systems surrounding houses, known as home gardens (Soemarwoto, 1987). In Vietnam, for example, home gardens were found to contribute between 13-54% of total household income (Trinh *et al.*, 2003), and in Indonesia, to 7-56% of total income (Soemarwoto, 1987). In Brazil, small home gardens had the highest net

income per hectare and highest income-to-cost ratio, followed by medium-sized home gardens, as compared to commercial agroforestry enterprises, commercial agroforestry by smallholder farmers, enriched fallow, pasture with babassu and swidden cultivation (Cardozo et al., 2015). Brazilian small home gardens generated the equivalent of 7.47 minimum wages per hectare, and medium-sized home gardens generated 6.77 minimum wages per hectare, indicating high productivity of these systems, while also maintaining high levels of biodiversity (Cardozo et al., 2015). For comparison, pastures with babassu only generated 0.77 minimum wages per hectare and shifting cultivation systems generated 1.85 minimum wages per hectare during the cultivation phase (not including fallow phase) (Cardozo et al., 2015). In sub-Saharan Africa, fruit trees provide a significant source of income for many families, in some cases acting as a safety net and provide supplemental income to cover everyday household expenditures and education costs (Schreckenberg et al., 2006). Recent evidence from Uganda using data from a national survey of nearly 1,400 households over a 10-year period shows that households that increased the area they allocated to trees on farms - particularly fruit trees saw a significant increase in their total consumption (Miller et al., 2020).

Many tree-crop-based systems are transitioning towards more intensive plantation systems, e.g. shifting along a gradient from high shade to unshaded coffee production systems. This change towards intensification can increase yields and incomes but at the cost of biodiversity and system resilience (Steffan-Dewenter et al., 2007). Agroforestry systems managed for a greater diversity of products, as opposed to only one commodity crop like coffee, can increase resilience to market shocks and fluctuations. Crop diversity has also been shown to significantly reduce the probability of household poverty in some contexts (Michler and Josephson, 2017). There are trade-offs between productivity and environmental sustainability, but optimal configurations across both objectives may be found, such as with low-shade agroforestry systems (<30% shade) that maintain productivity while creating benefits for climate adaptation, climate mitigation and biodiversity (Blaser et al., 2018). Coffee agroforestry can diversify and decrease expenditures or increase household incomes, through the consumption or sale of fuelwood, fruit and lumber beyond the sale of coffee (Rice, 2008). Under sustainable management, these types of agroforestry practices yield high-value products along with the commodity crops, while maintaining tree cover that delivers ecosystem

services. Exploiting these resources from agroforestry can help relieve pressures on *primary forests* to supply these products (Rice, 2008).

3.3.4 Forests and trees for food security and nutrition

Forests and tree-based systems contribute to food security and *nutrition*, particularly for vulnerable groups such as children and pregnant and nursing women. In this section, we focus on the direct provisioning of wild and cultivated foods such as edible plants, fruit, nuts and seeds from forests and trees, as one of a suite of ways that these resources support food security and nutrition (Vira *et al.*, 2015; HLPE, 2017; Rasmussen *et al.*, 2017; in addition, see Box 3.3 on wild meat harvesting). The indirect contributions of forests and treebased systems are discussed in the next section.

Food harvested from the wild contributes to food security because of its nutritional value

(Boedecker *et al.*, 2014). While wild foods may not necessarily contribute to the caloric intake of rural households, studies have indicated that their role is particularly important in providing essential vitamins and minerals (Powell *et al.*, 2015a; Vira *et al.*, 2015; Asprilla-Perea and Díaz-Puente, 2019). The collection of such wild foods is also a means of mitigating the risks and shocks that poor people face due to, for example, droughts, illness, conflict, a poor harvest (Pouliot and Treue, 2013; Clements *et al.*, 2014) or forced displacement for the creation or enforcement of strictly protected areas (Sunderland and Vasquez, 2020).

Evidence shows that in many countries rural populations living in or around forested areas rely, to diverse extents, on the harvesting of wild foods for their dietary needs (Sunderland, 2011; Sunderland *et al.*, 2013; Boedecker *et al.*, 2014; Rowland *et al.*, 2017). Hickey *et al.* (2016) carried out a comparative analysis at a global scale, which concluded that 77% of rural households surveyed

Box 3.4

Sacred groves in Africa

Around the world, sacred groves represent a traditional form of community-based conservation, known to preserve areas that hold strong cultural and religious importance to local people (Oviedo and Jeanrenaud, 2007; Ormsby and Bhagwat, 2010; Bulkan, 2017). These sites can be individual trees, forest remnants, rivers, waterfalls, meadows, wildlife, sacred caves, lakes, hills and other sites (Bhagwat and Rutte, 2006; Ormsby, 2012; Liljeblad and Verschuuren, 2018) managed and sustained by a system of enduring religious beliefs and socio-cultural practices (Mgumia and Oba, 2003; Aniah and Yelfaanibe, 2016). The practice of establishing sacred groves is widespread across many countries in Africa, including Tanzania (Mgumia and Oba, 2003; Sheridan, 2009), Cameroon (Fru, 2014; Kemeuze et al., 2016), Nigeria (Onyekwelu and Olusola, 2014; Oyelowo et al., 2014; Daniel et al., 2016), Ghana (Ormsby, 2012; Aniah and Yelfaanibe, 2016) and Ethiopia (Aerts et al., 2016; Orlowska and Klepeis, 2018). Sacred groves exist throughout tropical Africa, and are usually designated as places for rituals of initiation and sacrifice. Generally, they consist of patches of forest in agrarian landscapes and are commonly found in the long arc of forest-savannah transition zone (Sheridan, 2009). The ecological status of African sacred groves is associated not only with their spiritual significance but also with political, economic and legal processes (Sheridan, 2009).

Ghana is considered as having the highest number and concentration of sacred natural sites in Africa (Ormsby, 2012). It was estimated that over 1,900 sacred groves of varying sizes ranging from very small patches (less than 1 ha) to larger expanses of several thousand hectares are spread across the country (Ntiamoa-Baidu, 1995). In the central and northern Ethiopian highlands, 'church forests' are a strong and longstanding tradition where small fragments of forests, mostly the remaining native forests, are managed by the Ethiopian Orthodox Tewahido Churches and monasteries as sacred groves (Aerts et al., 2016; Woods et al., 2016; Orlowska and Klepeis, 2018). In Nigeria, sacred groves are a symbol of identity for most of the Yoruba People in the south-west region of the country, and were historically established outside their settlements (National Commission for Museums and Monuments, 2005).



Harvesting acai (Euterpe oleracea) in the state of Amapá, Brazil Photo © Reem Hajjar

collected food from the wild, highlighting the extent to which such harvesting is an integral part of many livelihood strategies, particularly in developing countries. Both in Malawi (Johnson et al., 2013; Hall et al., 2019) and Indonesia (Ickowitz et al., 2016) a positive correlation has been found between tree cover and dietary diversity. A number of multi-country meta-analyses have also served to confirm this positive relationship between forests and diets. Ickowitz et al. (2014), for example, in their meta-analysis of 21 African countries, found a statistically significant correlation between tree cover and dietary diversity among the diets of children in all 21 countries. Similarly, Rasolofoson et al. (2018) also identified a positive relationship in their analysis of 27 African countries. In contrast, Galway et al. (2018) found that deforestation and the loss of forest around villages and agricultural fields resulted in poorer dietary outcomes for children in sub-Saharan Africa.

Farmed produce are often unable to fulfil all the dietary requirements of a rural family, and wild food collection often serves to complement their nutritional needs (Fischer *et al.*, 2017; Nakamura and Hanazaki, 2017). In fact, reliance on agricultural production may lead to lower quality diets lacking in vitamins and micronutrients such as iron, zinc and vitamin B12 (Sunderland *et* al., 2013; Cawthorn and Hoffman, 2015; Powell et al., 2015a). This lack of quality in the diet has been termed the 'hidden hunger' (Ickowitz et al., 2014; Fa et al., 2015; Powell et al., 2015a) and leads to malnutrition, which can have severe impacts on the development of young children, leading for example to childhood stunting with life-long consequences (Golden et al., 2011; Temsah et al., 2018). A study by Blaney et al. (2009) exploring the contribution of natural resources to the nutrition of the local population in a protected area in Gabon, found that the consumption by children aged 5 to 9 of products stemming directly from their environment, was the best predictor for nutritional status. Similarly, although overall natural foods were found to contribute only 12% of the energy requirements of villagers in the Gamba Complex of Gabon, they contributed an estimated 82% of protein, 36% of vitamin A and 20% of iron requirements (Blaney et al., 2009). In this context, the role of wild foods collected from the forest is all the more important to help to combat micronutrient deficiencies.

Fruit trees, both wild and cultivated, are an important source of dietary diversity for many rural households. They have the advantage that they are often easy for households to domesticate and manage on their land (Willett *et al.*, 2019). Fruits like baobab, mango, papaya and orange are par-

ticularly vitamin-rich sources of nutrients (Vira *et al.*, 2015). A nationally-representative panel data study from nearly 1,400 households in Uganda, for example, showed that those that increased the share of trees on their farms, especially fruit trees, saw improved child health and nutrition outcomes (e.g. less child wasting and stunting) (Miller *et. al.*, 2020).

Box **3.5**

An example of how poor communities have profited from timber harvesting

In Mexico, as a result of a national agrarian reform that occurred in several waves in the 1900s (Bray et al., 2006), many communities have control over and log forests with commercial timber potential. In the Yucatan region, forestry authorities, civil society organisations and the state government of Quintana Roo cooperated under the Forestry Pilot Plan, or Plan Piloto Forestal (PPF), to provide technical assistance to communities for commercial forest management. Bray et al. (2007) found a suggestive correlation of lower poverty rates with direct control over more of the value chain, i.e., via the establishment of sawmills rather than simply selling stumpage or logs. The distribution of benefits from logging also varies across communities, with some choosing to share profits within workgroups that manage particular stands and others investing in local social services or public goods, such as schools, drinking water systems and health posts, that benefit all families living in the community (Huelsz and Negreros-Castillo, 2014). This is particularly relevant for women, who typically do not participate directly in commercial forestry but do benefit from local social services.

3.3.5 Forest and tree inputs to production in other sectors

In addition to the sale and consumption of goods from forests and trees, many households derive benefits from the contributions of forests and trees to production in other sectors, most notably agriculture and fisheries. These indirect benefits are particularly important to poor households who cannot afford to purchase substitutes for the free inputs provided by forests and trees (Chavarría et al., 2018). For example, hundreds of thousands of smallholders, half of them women, plant fodder trees in East Africa, where they reduce the cost of producing milk (Franzel et al., 2014). The inputs include services generated as externalities of forests, such as reduced sedimentation downstream, and as a result of deliberate management of trees, such as shade and nitrogen fixation. They also include inputs gathered from forests, such as fodder and poles, and produced in tree-based agricultural systems. These systems are diverse, ranging from trees retained on farms following forest clearance, to simple agroforestry systems such as improved rotational fallow, alley cropping, intercropping and hedgerow systems, to complex agroforestry systems that mimic natural forest ecosystems (McNeely and Schroth, 2006). While there have been numerous initiatives to promote these tree-based systems because of their benefits to farmers and society in general, farmers must balance these expected benefits against potential costs of competition for resources and negative effects on the microclimate, such as increasing relative humidity and lowering air temperature in sub-humid zones (Kuyah et al., 2016).

Forests and trees can increase crop and livestock productivity (Baudron et al., 2019). Tree-based systems can increase agricultural yield and nutritional quality through various provisioning and regulating ecosystem services (Reed et al., 2017; Barrios et al., 2018). The resulting product diversification and regulating services that maintain productivity can increase resiliency to climate change and other shocks (Thorlakson and Neufeldt, 2012; Kenney-Lazar et al., 2018; Quandt et al., 2019). A review of 438 studies spanning over 20 countries across sub-Saharan Africa shows that crop yields increased under tree-based systems such as fallowing, tree-crop intercrop and alley cropping compared to treeless systems in 68% of the studies due to improved microclimate, nutrient cycling and soil fertility (Kuyah et al., 2016). However, 18% of these studies also reported a decline in crop yields mainly due to trees competing with crops for nutrients, water and light (Kuyah et al., 2016).

Likewise, at a landscape level, crop yields can be maintained or enhanced at a level comparable to intensive monoculture when forests and trees are incorporated effectively in an agricultural landscape (Reed *et al.*, 2017; Baudron *et al.*, 2019). The presence of trees and forests in agricultural landscapes showed an overall positive or neutral effect on crop yields in 52% of the case studies in a pan-tropical review of 74 studies (Reed *et al.*, 2017). In two studies in Ethiopia, livestock productivity

Box **3.6**

The forestry sector in the Congo Basin as an inequality machine? A brief overview of the issues

In many tropical forest-rich countries, among them, Cameroon and the Democratic Republic of the Congo (DRC), the exploitation of forests and forestland is justified by the promise of development and increased societal welfare.

When investigating the specific cases of Cameroon and DRC, we find that this development discourse is contradicted by the long stagnating national incomes (Alvaredo et al., 2018) in both countries, and documented violation of rights and exclusion of indigenous people from access to forest resources and related benefits (Logo, 2010; Assembe-Mvondo et al., 2013). Paradoxically, since the colonial period, the marginalisation of indigenous people and forest communities' rights at domestic level co-exists with an increasing proliferation of policies driven by international forest-related agreements that promote an improvement of their participation, land tenure security and benefit-sharing (Maggio, 1997; Adams and Hulme, 2001; Larson et al., 2010; Schroeder, 2010; Sikor et al., 2010). Over the longer term, the forest sector appears to have contributed more to the economic prosperity of European countries that have historically dominated forest exploitation and forestland conversion in the Congo Basin as colonial powers, namely Belgium, France and Germany (Hardin and Bahuchet, 2011; Coquery-Vidrovitch, 2017). More recently, China emerged as a new power in the region, engaging with a similar dual agenda of linking promises of development for societal welfare in exchange for the (over) exploitation of forests that are combined with forestlands and large-scale land acquisitions (Sautman and Yan, 2008; Germain et al., 2018). Not least, state bureaucracies and national elites are also entangled in rent-seeking behaviour (Ross, 2015), with powerful groups having gained access and using their influence and power to capture and/or enlarge the forest rent. This is reflected throughout a set of dominant strategies in the politics of land acquisitions, forest concessions, trade and investment patterns (Ribot, 1999; Ekoko, 2000; Karsenty and Ongolo, 2012).

Social inequality within and among societies in different parts of the world is current-

ly part of many public debates (Alvaredo et al., 2018; UNDP, 2019) and often narrowly expressed in (economic) opportunities and outcomes, for example in access to education or participation in decision-making over the use of natural forests (Sen, 1997; Obeng-Odoom, 2020). Underlying those inequalities as starting points (opportunities) and finishing lines (outcomes) are multidimensional, socio-political processes that often feed into a machinery of increased 'production' of social inequalities (Afonso et al., 2015). Inequality resulting from uneven distribution of, and access to, the many materials and immaterial benefits from forests in the tropics has been discussed to some extent in recent literature, for example with regard to global North-South dynamics driving and justifying access to forests and large scale conversion of forestlands in the name of development (Ribot and Peluso, 2003). Here, social inequality is manifested in institutional path-dependencies and power relations, defining who has the right to access and benefit. Other scholars refer to benefits and burdens in the context of climate change and specific policies and programmes (Ribot, 1999; Phelps et al., 2010; Luttrell et al., 2013; Pham et al., 2014). Detailed accounts of the role of colonial exploitation in generating inequality, such as Peluso (1991) for the case of Java, are limited for the forest sector, in the Congo Basin and elsewhere. Much of the literature focuses on particular inequalities in benefit sharing within a particular tropical forest country, while often missing the link between major financial actors in the Global North investing in industries driving deforestation in the Global South (Galaz et al., 2018). Scholars also highlight the underlying long-term dynamics of power and politics in the global forest and land-use sector, and the political economy establishing incentive structures and discursive practices which drive and justify unequal outcomes from tropical deforestation (Angelsen and Kaimowitz, 1999; Rudel, 2007; Dauvergne and Neville, 2010; Burgess et al., 2012). Meanwhile, today's decision-making over forests and forestlands in the tropics seems still to be shaped by persistent myths that create

barriers to transformation towards global forest sustainability (Delabre *et al.*, 2020). With emerging datasets on forest change, global trade, investments and related inequalities among countries, as well as increasing access to digital information in colonial archives, it is now crucial more than ever to examine the as

and nutrient balances, and the nutritional value of crops both improved with proximity to a forest (Chavarría *et al.*, 2018; Wood et al., 2018).

The key ecosystem services from forests and trees that support crop production include nutrient cycling (Power, 2010), pollination (Garibaldi et al., 2011), seed dispersal (Thrupp, 2000), soil formation (Hurni et al., 2015), reduced erosion and leaching (Mbow et al., 2014), natural pest and disease control (Karp et al., 2013) and climate and water regulation (Daily and Matson, 2008). In particular, 'fertiliser trees' (that are grown in agricultural fields or pastures to increase nitrogen availability) can offer an alternative or supplement to fertiliser application, which can reduce expenditure on fertiliser and increase income through higher yields. Nitrogen-fixing trees maintain and enhance soil fertility by cycling atmospheric nitrogen, thereby increasing yields (Akinnifesi et al., 2010; Ajayi et al., 2011). A review of 90 studies suggests that maize yields increased, and crop production stabilised during drought after the integration of nitrogen-fixing trees on farms in Eastern and Southern Africa (Sileshi et al., 2007). Similarly, incorporating trees in wheat fields increased nitrogen availability in soil, water use efficiency, reduced heat stress and increased yield significantly compared to wheat fields without trees (Sida et al., 2018), resulting in higher net income (Place et al., 2005; Kuntashula and Mungatana, 2013; Coulibaly et al., 2017; Amadu et al., 2020).

Forests and trees support pollinators and natural predators of crop pests. Although many major crops are self- or wind-pollinated, wild pollinators such as bees, butterflies, birds and bats directly affect the productivity of 75% of globally important crops (Potts *et al.*, 2016). For instance, yields in coffee crops in Costa Rica and watermelons in California increased in sites near forest fragments due to more frequent visits by pollinators (Scherr and McNeely, 2008). Similarly, a global study found that pollinator richness increased crop yield across 89 crop systems (Dainese *et al.*, 2019). The stability of pollination services declines in crop fields with increasing distance from forests and trees (Garibaldi *et al.*, 2011). yet largely unanswered question (McDermott, 2017) of 'who, and whose societies, benefit from tropical forest exploitation and deforestation?' And to what extent do current global forest governance arrangements reinforce or break with existing patterns of inequality?

Another key service is pest control. Incorporating forests and trees within agricultural landscapes creates heterogeneity in the habitat and supports diverse natural predators of crop pests (Maas et al., 2016; Chaplin-Kramer et al., 2019; Kebede et al., 2019), especially in perennial crops (Pumariño et al., 2015). For instance, forest cover in farmland improved pest control by increasing natural predators such as bats and birds in Costa Rica and Western Kenya (Karp et al., 2013; Guenat et al., 2019). Similarly, effective management of ants and shade trees increased crop yields in cocoa agroforestry in Indonesia (Gras et al., 2016). Conversely, forest cover loss reduced agricultural production by 45% due to the loss of biological pest control in Indonesia (Yamamoto et al., 2019).

Forests and trees can also increase crop productivity and resilience by improving microclimate conditions in agricultural landscapes (Pramova *et al.*, 2012). For instance, trees can buffer extreme climatic fluctuations such as temperature spikes that have negative impacts on crop growth (Hatfield, 2016). Shade trees have been found to enhance production by regulating temperature and humidity fluctuations in coffee agroforestry systems in Latin America (Lin *et al.*, 2008) and India (Nesper *et al.*, 2017). Trees in agricultural landscapes can also enhance understory growth by reducing air and soil temperature and by regulating water retention and gas exchange (Lott *et al.*, 2009).

In livestock systems, trees provide both the key service of shade and the key input of fodder. These systems include grazing livestock on pastures with trees and allowing livestock to graze on the trees or shrubs, as well as supplying tree cuttings as fodder for livestock. Fodder trees, when used as a protein supplement, improve milk and meat production, livestock growth, and livestock health and reproduction (Franzel et al., 2014). This increase in productivity leads to improved incomes and food security. In East Africa, for example, fodder trees and shrubs contributed about USD 3.8 million annually to farmers' incomes by 2006 (Franzel et al., 2008). At the household level, this translated to an increase in net returns of between USD 13- 334 per year in Zimbabwe, USD 30-114 per year in Kenya

Box **3.7**

Shea butter market as an example of NTFPs as a way out of poverty

The nuts of the shea tree (Vitellaria paradoxa), found in the dry savannah and grassy woodlands of Africa, are both consumed at home and sold in the market for end uses including food, oils and cosmetics. Shea nut prices increased five-fold from the 1990s to 2013 (Rousseau et al., 2015). This strong market means that a single shea nut tree has an estimated net present value of USD 211 (IUCN Uganda, 2016). In particular, shea collection, processing and subsequent sale of shea-based products generate income and offer employment to rural women and children (Aboyella, 2002; Abdul-Mumeen et al., 2013; Mohammed et al., 2013; Sarkodie et al., 2016). Laube (2015) finds it "unlikely that shea nut pickers will be able to substantially increase their production with labour shortages and

and Uganda, and USD 68-503 per year in the Philippines due to increased production and income from cattle (Franzel *et al.*, 2014).

While not as well recognised as contributions to crop and livestock productivity, forests and trees also support fisheries. Fish and other products (e.g. freshwater prawn, crayfish and crabs) have long been recognised as an important source of protein for the poor, who consume less but are more dependent on (and have fewer substitutes for) these products in their diets, as compared to wealthier populations (Kent, 1997; Jones et al., 2006; HLPE, 2014). A growing body of literature highlights the contribution that 'blue forests', notably mangroves, make in supporting local community well-being, livelihoods and food security (Himes-Cornell et al., 2018). McIntyre et al. (2016) report that hundreds of millions of people globally benefit from low-cost protein and commerce that freshwater fisheries provide, particularly where alternative sources of protein and employment are scarce.

Deforestation, overexploitation or contamination of water by agriculture, mining or other land use changes can have drastic effects on aquatic foods with subsequent impacts on downstream people reliant on these systems for income, nutrition and food security (Carignan and Steedman, 2011). Forests play a role in the maintenance and regulation of aquatic food webs by regulating flow, dwindling access to shea trees". However, women involved in the shea business are more likely to effectively increase their family income when they have access to microcredit, e.g. through the Community Life Improvement Programme (CLIP) in northern Ghana (Robinson, 2001; Bawa et al., 2017). At the country level, FAO estimated that Ghana exported 42,424 metric tonnes (MT) of shea worth USD 14.8 million in 2008 (FAOSTAT, 2008). This same quantity of shea nuts could have yielded 21,212 MT of shea butter at a total value of USD 21.2 million (Omane, 2014). The implication is that value addition through the processing of shea nuts into butter presents an opportunity for increasing income, improving livelihood outcomes and alleviating poverty.

controlling sedimentation rates, regulating instream temperature and contributing energy flows through terrestrial resource subsidies in the form of terrestrial fauna entering the aquatic food web. Losing this regulating function of forests impacts on the health of the people whose food security and nutritional needs rely on them. For example, in the Amazon Basin, fish is in many cases the most important source of protein consumed by traditional rural peoples. In freshwater streams and rivers, fish are dependent on fruits and seeds from riparian vegetation for their survival (Goulding, 1981).

Mangroves play an important role in the productivity of marine fisheries, providing habitat, spawning grounds and nutrients for a variety of fish and shellfish, including many commercial species. Falling leaves and woody matter from mangroves are essential to the marine food chain that supports fisheries (FAO, 2007; Hutchison et al., 2014). Juvenile fishery species can hide among the roots of mangrove trees and grow to a size where they are less prone to predation, leading to higher survival rates. Fish and shellfish from mangroves support a large number of fishing and rural communities around the world providing them with income and food security. Mangroves contribute to the employment of an estimated 38.4 million people globally, of whom 90% are artisanal fishers (Hutchison et al., 2014). As mangrove forests

Box **3.8**

Vanilla production as an example of agroforestry as a way out of poverty

Approximately 80% of the world's vanilla is produced in Madagascar, largely in the northeastern SAVA region, constituting up to 26% of Madagascar's crop export revenue and up to 6.8% of the country's national revenue (Organisation Internationale du Travail, 2016; World Bank, 2019b). Vanilla orchids (Vanilla planifolia) are grown on other vegetation for support, including in native forest that has not been significantly altered (Hending *et al.*, 2019). In the SAVA region, these agroforestry plantations have become the main source of income for many farmers. Hänke *et al.* (2018) report that the cultivation of vanilla has improved the socio-economic status of smallholders.

are destroyed, local fish catches drop, leading to a direct loss in livelihoods. More broadly, the loss of mangroves may have significant negative impacts on the fisheries sector. Mangroves also contribute to aquaculture, both open-water estuarine mariculture (e.g. oysters and mussels) and pond culture (mainly for shrimps) (FAO, 2007). Shrimp farming, for example, has a high economic rate of return and has been promoted in several countries to boost the national economy and alleviate poverty.

3.3.6 Non-material contributions

The first SDG recognises that poverty is multi-dimensional, calling for a reduction "at least by half [in] the proportion of men, women, and children of all ages living in poverty in all its dimensions according to national definitions." Culture, religion and spiritual values are clearly important to human well-being and thus their loss is a form of impoverishment.

Forests and trees are significant to spiritual and cultural traditions central to the identity of forest-proximate communities, especially indigenous peoples (Oteng-Yeboah *et al.*, 2011; Asselin, 2015; Daniel *et al.*, 2016). Information on the cultural significance of forest resources can be gleaned from anthropological, ethnobotanical and ethnoecological studies (Toledo, 2002; Alexiades, 2003; Cocks, 2006). These cultural values manifest in ways ranging from forests being objects of animist-based beliefs to traditional forest products marketed globally based on their joint natural and as indicated by income, education, access to electricity and ownership of assets. These benefits generally arise from contracts with vanilla exporters or collectors and thus are concentrated among smallholders able to obtain those contracts. Female-headed households are much less likely to get contracts because of their significant social disadvantages (e.g. lower labour availability and smaller fields). Additionally, tight integration with the export market results in both unstable prices (Zhu, 2018) and perceived exploitation due to the wide spread between the prices offered to smallholders and the export value.

cultural attributes. Box 3.4 provides an example of the spiritual role of forests. Specifically, for people who traditionally lived near and with forests, one dimension of poverty alleviation is restoring the cultural, spiritual and religious values of forests.

Cultural identity and integrity

Forests are culturally important to the self-identification of indigenous peoples in the role they play in their well-being, and are an important factor in non-material aspects of quality of life of many indigenous peoples (IPBES, 2019). They symbolise cultural cohesion in a rapidly changing environment and, hence, cultural integrity. Intimately linked with ancestry and cultural heritage, forest symbols strengthen social and cultural identity. For example, most sacred forests in southeastern Nigeria and coastal Kenya are important sites for the coronation of paramount rulers, exclusive meetings for spiritual leaders, traditional rites and celebrations (Kibet and Nyamweru, 2008; Umazi et al., 2013; Daniel et al., 2016).

This value is not limited to forests or indigenous populations. For example, studies of immigrants from lower-income countries living in Europe highlight the spiritual and cultural importance of agroforestry as a connection to their culture and traditions (Mazumdar and Mazumdar, 2012). In the western Brazilian Amazon, social movements and a state government have emphasised cultural connections to the forest among descendants of migrants who came to the region to

tap rubber from the native Hevea brasiliensis trees (in the late 1800s and during WWII). Rubber tappers lived in the forest and walked forest trails daily to tap trees for rubber. Their livelihoods thus required conservation of the forest, and this brought them into conflict with farmer and cattle ranchers who migrated to the region in the 1970s and 1980s. Thus, as recounted by Gomes et al. (2012), "the term 'rubber tapper' acquired a new emphasis that highlighted sustainable resource use and traditional claims to forested lands, set against an unsustainable development model involving land speculation and deforestation." In 1998, the state of Acre elected a new government that emphasised 'florestania' or forest citizenship, thus recognising the centrality of forests to the culture and identity of rubber tappers and claiming those cultural traditions and that identity as a source of pride for the entire state. To support both the ecosystem services and the cultural values associated with rubber tapping, the government offered a subsidy for rubber as a mechanism to simultaneously increase rubber-tapper incomes, incentivise protection of the forests with rubber trees and recognise the cultural value of rubber-tapping (Sills and Saha, 2010; Jaramillo-Giraldo et al., 2017). As Gomes et al. (2012) argue, "the cultural content of rubber tapper identity is rooted in historical material conditions" and thus, its continuation depends on the competitiveness of the forest economy (Hoelle, 2015).

Importance of forests as sacred spaces

Traditionally managed by indigenous communities, in many regions forests are considered sacred and are governed by a set of traditional norms and rules (Munyi and Mutta, 2007; Rutte, 2011; Ngoufo *et al.*, 2014). Preserved forest patches are usually close to human settlement, thus, forming an integral part of traditional closely-knit rural communities (Ray *et al.*, 2014). They provide the venue for social, cultural and religious ceremonies and a range of products for traditional ceremonies from food and beverages to costumes and musical instruments.

Most sacred forests in south-eastern Nigeria are used for the coronation of paramount rulers and are deemed sacred to non-initiates, as an exclusive meeting place for the members of the Ekpe occult society (Umazi *et al.*, 2013; Daniel *et al.*, 2016). In Kenya, the Kayas (sacred forests) of the Mijikenda tribal group fulfil many roles: they are burial sites of an ancestral or founding figure, or of revered community elders, are former battlegrounds or the sites at which a community leader first established title to the location. They are also sites of seclusion for initiates, meeting places for secret societies and areas where community rituals and celebrations are held (Kibet and Nyamweru, 2008).

Many religions that originated in Central and South Asia, China and Japan (including Buddhism, Daoism and Hinduism) integrate nature as a critical component of their belief systems (Dudley et al., 2009). Sacred forests provide essential spiritual services to Tibetan Buddhists, who believe in both the Buddha and local deities. For them, sacred forests are naturally forested Holy Hills where village gods that protect a person for their entire life (Liu, 2006) and spirits are believed to reside (Taylor and Kaplan, 2005). Improper actions or disrespect of these forests are punished by misfortunes. Rituals are practised each year to consecrate the sacred forests and honour the gods and spirits that live there. The traditional annual rituals provide an essential mechanism to integrate widely scattered households into a close-knit community (Liu, 2006). Sacred forests also provide similar spiritual services to the Dai people who live in Southwest Yunnan Province of China, Northwest Vietnam, Northern Thailand and upper Laos (Taylor and Kaplan, 2005). To the extent that cultural practices contribute to a shared sense of belief among communities, they are essential complements to economic approaches to livelihoods through the collective community capabilities.

Importance of forests in customary and religious rituals

In a study of the uses of fallow tree species in Ho (Ghana), Asamoah (1985) found that half of the identified species were valued in customary rites. Most musical instruments are made from forest products. For example, the seed shells of *Chrysophyllum albidum* and *Mammea africana* are worn by dancers as rattles and the wooden strips of *Ricinodendron heudelottii* are used to make xylophones in Igboland, Nigeria (Okigbo, 1980). The long history of the sites and the related rituals, and the reference to the ancestors give these forests their high value (Darr et al., 2009).

In Nepal and India, all Hindu families have to perform pujas (religious rituals) on certain occasions that require plants and their products. Traditional Hindu books such as *Ramayana*, *Mahabharata* and *Veds*, all call for preservation of the forest as a part of the cultural heritage. In Hindu theology, some plant species (such as *Ficus religiosa*) are considered as "incarnations or symbols of deities and other supernatural forces" and therefore must be



Wildlife is a main attractive for ecotourism businesses (Giraffe at Massai Mara, Kenya) Photo © Daniel C. Miller

worshipped (Ingles, 1997). Consequently, harvesting of such sacred species is "thought to be against the god, a belief that is still common". Forests and trees are also often linked to some cultural events. For instance, "plates made from sal leaves are essential for all ritual functions and are regarded as *chokho* (uncontaminated)." Another example of forest or tree contribution to the cultural well-being of Nepalese people is the practice that "a dead body must be carried in a green bamboo casket to the place of cremation where it has to be burned with firewood from the plant *Ficus benjamina*" (Acharya, 2003).

There are also ecological implications of these cultural practices which directly or indirectly contribute to people's well-being. It has been reported that the maintenance of religious forests, especially in hilly regions of Nepal, has had positive impacts on soil conservation and microclimate preservation. There are 40 religious forests in the Kathmandu Valley alone (NBAP, 2001). In Borneo, Meijaard et al. (2013) found that forest use and cultural values are highest among people who live close to the remaining forest, and especially among older Christian residents. In their study, perceived values of forests were generally high, with 48% of respondents considering the importance of forests for cultural and spiritual purposes to be very significant and 26% considering them quite significant. A study by Melnykovych and Soloviy (2014) showed that economic, environmental, social, cultural and aesthetic functions of forests contribute considerably to the well-being of forest-dependent communities in the Ukrainian Carpathians.

3.3.7 Gender considerations

Taking gender into consideration in relation to forest landscapes matters because how, why and where men and women access, use and manage forests and trees differ (Mai et al., 2011; Mwangi et al., 2011; Kristjanson et al., 2019). Further, the feminisation of agriculture is a global trend, making gender a particularly important variable for understanding the role of trees on farms. A review of the literature on forests and gender identified persistent gender gaps across regions in access to services, access to markets and value-addition activities, land and tree tenure voice and agency, and hiring labour (Colfer et al., 2016). In addition to these, gender differences in the capacity for addressing climate change have been recognised as an issue that affects not only productivity but widen existing gender gaps in many places (Pérez et al., 2014). And in some areas, men's migration from rural areas has left women to assume the spectrum of agricultural and forest management roles, often without the resources or agency to do so successfully (Giri and Darnhofer, 2010; Jaquet et al., 2015).

CIFOR's pan-tropical PEN study found evidence of distinct male and female roles in relation to the collection of forest products that vary across regions (Sunderland *et al.*, 2014). In Africa, they found that women are the main collectors of subsistence-oriented forest products, while in Latin America, they found that men dominated firewood collection. Men were also more involved in fuelwood collection in Africa than often assumed. In all regions, men were more involved in hunting, wood harvesting and mineral extraction than women. They found that in Latin America, men earn seven times more income than women from unprocessed forest products, while in Asia earnings are similar for men and women, and in Africa the share of income from forests is greater for women. With respect to income from processed forest products (e.g. furniture), the share of overall income is higher for men (61%) than women (25%) across the three regions (Sunderland et al., 2014). Women were also found to collect more forest products than men from common property resources in Latin America and Asia, but not in Africa (Jagger et al., 2014; Sunderland et al., 2014). This is, however, not always the case; for example, relatively few differences between men and women were found regarding the role of NTFPs in household coping strategies in South Africa (Paumgarten and Shackleton, 2011).

Harvesting from forests is often dangerous and exhausting, and collecting wood on-farm reduces the distance women have to travel and affords more time for leisure (Njenga et al., 2017). Women are often responsible for managing livestock, so available shrubs also reduce the time required to gather fodder, allowing women more time for leisure activities and to prepare nutritious food for their families (Kiptot et al., 2014). Women also directly earn money through the sale of milk, and the income they have control over often goes directly towards their children's education and providing nutritious food to their families (Kiptot et al., 2014). Additionally, in some communities, women might benefit from the sale of fruit and fruit products (Kiptot et al., 2014). Many agroforestry studies that consider nutrition outcomes highlight the importance of women and women's empowerment in decision-making as key factors determining household nutrition and dietary diversity along with agroforestry practices.

On the other hand, there is considerable variability in how the incorporation of fertiliser trees affects different population sub-groups, such as smallholder farmers, women and poorer or more marginalised households (Place *et al.*, 2005; Kuntashula and Mungatana, 2013; Coulibaly *et al.*, 2017). In many of these cases, women are often restricted in their ability to participate in agroforestry programmes due to social norms or programme design, and they often experience fewer benefits from the participation in agroforestry programmes compared to their male counterparts (Place *et al.*, 2005; Hegde and Bull, 2011). For high-value tree crop systems, such as rubber and coffee, men often control these tree crops with high commercial value, and women are often excluded from these high-value enterprises (Kiptot and Franzel, 2012).

3.4 Risk Management

Forests and trees help the poor manage risk by reducing exposure, and by providing a means to smooth income and consumption across seasons and years. In this way, they help to prevent transitory poverty and enable investments that are high risk but high return, by effectively offering insurance in the form of forest products (Shackleton and Shackleton, 2004; Paumgarten, 2005). This is especially relevant to the rural poor because they often do not have access to other forms of insurance and they often rely on activities that are subject to covariate shocks, such as variable weather, that affect entire communities (e.g. see Noack et al., 2019 for the role of forests in stabilising incomes during droughts). Climate change is expected to exacerbate this situation in Africa, the region with the highest poverty rates.

Forests can help to reduce the vulnerability of households to climate change. For example, in coastal regions, mangroves buffer human settlements from tropical cyclones and storm surge (Sierra-Correa and Kintz, 2015) and on steep slopes, forests help prevent landslides in response to extreme precipitation events (de Jesús Arce-Mojica *et al.*, 2019). Unlike annual crops, many trees are able to tap into deeper water sources through their roots and produce leaves, fruits and other products during periods of water shortage or high temperatures, which also contributes to households' capacity to cope with weather and climate-related shocks (Shackleton and Shackleton, 2004; Fisher *et al.*, 2010; Place et al., 2016).

Most frameworks for risk management include both ex-ante and ex-post actions, where ex-ante may include diversification through tree-based systems (Krishna, 2011; Kristjanson *et al.*, 2019), and ex-post may include capturing income from sources that are otherwise too labour intensive or too long-term to be competitive, such as harvest of NTFPs. Investing in harvest systems for natural assets that are either slow-growing or that produce low but very consistent yields is also a form of ex-ante adaptation. Because these activities are not the first choice of households, they tend not to be recognised as an important part of local economies. However, preventing shortfalls in con-
sumption is both a worthy end in and of itself and can contribute to long-term poverty alleviation, by helping to maintain human capital.

A large case study literature demonstrates that people use forests as safety nets, increasing their collection of NTFPs to smooth shortfalls in other income sources, especially in response to covariate shocks that limit options for the sale of assets or borrowing from neighbours. This does not appear to be the most common or the preferred strategy employed by rural people in general (Wunder et al., 2014), and there is little evidence available on the relative quality or efficiency of forest-based versus other risk management strategies, such as crop insurance (for an exception, see Mbiba et al., 2019). However, forests can be particularly important for remote rural populations that are poor and have few alternatives. For example, for people without access to financial services, forests and trees may act as stores of wealth in terms of both food and income sources during droughts and other events that would otherwise increase debt loads (Thorlakson and Neufeldt, 2012; Angelsen et al., 2014; Wunder et al., 2014).

Public demand and initiatives to conserve forest are providing new ways for poor households to diversify their incomes directly from forest landscapes, including income from wildlife conservancies and ecotourism (e.g. Andam *et al.*, 2010; Bedelian and Ogutu, 2017). Payments for environmental services, such as carbon sequestration and watershed restoration, are increasingly important income sources for local and indigenous communities in many regions (Pagiola *et al.*, 2005). Conservation policy can also affect sensitivity to climate shocks by determining access and management rights to forests that buffer income shortfalls (Lawlor *et al.*, 2019).

3.5 Forest Negative Externalities

Standing forests and trees also generate negative externalities for forest-proximate populations. That is, in addition to benefits, there are costs originating from the existence of forests, leading to harmful, unpleasant or unwanted consequences for people (Lyytimäki, 2015). The role of forest as habitat for wild animal populations leads to negative outcomes including crop-raiding, livestock predation and transfer of diseases from wildlife to livestock and humans. The effects of invasive tree species also can be considered a negative externality of forests (McGarry *et al.*, 2005; Sun *et al.*, 2006; von Dohren and Haase, 2015). Over the last 8,000 years, about half of the forests on the planet were cleared by human activities (Foley *et al.*, 2005). This extensive loss of forests has meant an equally dramatic loss of wildlife habitat which increases the potential for human-wildlife conflict. Interaction with wildlife can also pass dangerous pathogens to livestock or human beings, such as bovine tuberculosis and rabies (Megaze *et al.*, 2017; Matseketsa *et al.*, 2019). Both the SARS-CoV, the virus that caused the SARS epidemic in China in 2003, and SARS-CoV-2, the virus that caused the 2020 COVID-19 pandemic, are believed to have originated from wildlife living in the forest (Li *et al.*, 2005; Hu *et al.*, 2017; Zhang *et al.*, 2020).

Forests generally have much higher leaf area per unit ground area than other vegetation types (Gray and Song, 2012) and some evergreen tree species have long growing seasons. In areas with limited precipitation, the water demand by trees reduces water availability for agricultural and domestic use (Sun *et al.*, 2006; Li *et al.*, 2016). Similarly, trees on farms compete with crops for water and light. However, the net effects of trees on farms depends on the balance between this increased competition and improvements in the microclimate and soil fertility (Kuyah *et al.*, 2016).

3.5.1 Crop raiding and livestock depredation

Crop raiding and livestock depredation happen wherever people live close to forests, but especially near protected areas with high wildlife density (Naughton-Treves, 1998; Karanth and Ranganathan, 2018). Crops and livestock in the buffer zones around those protected areas are convenient sources of food for some wildlife. The elephant, which is the largest crop-raiding mammal, not only destroys crops but may also cause human injuries and even death. Human-elephant conflicts happen primarily near protected areas in both Africa and South Asia. For example, Neupane et al. (2017) found that elephants are responsible for more than 40% of crop-raiding, causing the loss of 25% of crop production in the Terai region of Nepal. Harich et al. (2013) found that 84% of farmers experience crop damage from elephants around the Bia Conservation Area in Ghana.

Primates also cause serious crop damage, and it is difficult to guard against their opportunistic crop-raiding behaviour. Baboons, chimpanzees and numerous monkey species inflict damage on crops (Naughton-Treves, 1998; Tweheyo *et al.*, 2005; Mwakatobe *et al.*, 2014; Mackenzie *et al.*, 2015; Mohammed *et al.*, 2017). For example, Tweheyo *et* *al.* (2005) found that 73% of people living around the Budongo Forest Reserve in Uganda reported crop damage by primates, and 79% of the residents consider baboons to be the most destructive.

Countless other species cause crop damage. Flying fox, squirrels, birds, field rats, rabbits, porcupines, bears, wild boars and peccaries, can all cause serious crop damage. For example, wild boars are reported to cause the most damage around Tianma National Nature Reserve in Anhui, China (Zhang et al., 2018) and around Kerinci Seblat National Park in Sumatra, Indonesia (Linkie et al., 2007), Ramnagar Forest Division, Uttarakhand, India (Kumar et al., 2017), and the Kibale National Park, Uganda (Naughton-Treves, 1998). De Carvalho et al. (2019) reported that nearly every household interviewed suffered from crop-raiding in a landscape with cropland intermixed with small forest patches in southeastern Brazil. The primary cause of damage was the white-eyed parakeet (Psittacara leucophthalmus), which attacks maize and fruit crops.

Livestock predation by wildlife is another major problem for forest-proximate people, leading to losses as high as two-thirds of household cash income (Wang and Macdonald, 2006; Holmern et al., 2007). The species most commonly responsible for predation are large felids (e.g., lions, tigers, pumas, cheetahs, leopards, snow leopards and jaguars), whose predation on livestock is most widely reported (Inskip and Zimmermann, 2009). For example, in the buffer zone of the Chitwan National Park (Nepal), Dhungana et al. (2019) found that more than 87% of livestock lost during 2007-2016 were goats taken by leopards. Moreover, people in disadvantaged social groups suffered disproportionately more attacks in the buffer zone of the Chitwan National Park according to data records, notably because they live closer to the forest and do not have suitable protection facilities (Lamichhane et al., 2018). Around the Serengeti National Park in Tanzania, Holmern et al. (2007) found 27.4% of livestock owners experienced loss to wild predators with an average of 4.5% or 5.3 heads of stock in a year, 97.7% of which were taken by spotted hyenas, followed by leopard (1.6%), baboons (0.5%) and lions (0.1%). Wang and Macdonald (2006) reported that leopards, tigers, Himalayan black bear and dhole are the primary animals that attack livestock in the Jigme Singye Wangchuck National Park, central Bhutan, causing an average 17% loss in cash income by affected households. At higher elevations, snow leopards are the primary predator of livestock (e.g. Chetri et al., 2019 for the Central Himalayas of Nepal). Demonstrating that this is a long-standing problem Mishra, (1997) reported that in the Indian trans-Himalaya (Kibber Wildlife Sanctuary), snow leopard and wolf are the primary livestock predators, causing a loss of 18% of livestock holdings over a period of 18 months. Although some governments or conservation management agencies provide financial compensation to livestock owners for the loss of livestock to wildlife depredation, the compensation is often far below the actual cost, e.g. only accounting for 3% of the perceived annual loss in the Kibber Wildlife Sanctuary (Mishra, 1997).

While the most prominent and well-publicised cases of human-wildlife conflict are often those around protected areas, conflicts can happen anywhere in forested landscapes. Michalski et al. (2006) reported that jaguars and pumas were the main animals attacking cattle in a fragmented forest landscape in the southern Brazilian Amazonia, with damages of up to USD 885 per year per farm. Bista and Song (under review) found local residents suffered significant economic loss from wildlife in the Middle Hills of Nepal where forest conditions had improved significantly from community forestry. On the other hand, around the Nilgiris Biosphere Research and Bhadra Tiger Reserve in the Western Ghats of India, Puyravaud et al. (2019) found that deforestation increased the frequency of crop-raiding by elephants.

3.5.2 Negative effects of trees

Forests, and especially high productivity forest plantations, are major water users and thus compete with other downstream uses of water (Calder, 2007). This may be especially true of the fast-growing and non-native species planted in industrial plantations, such as ponderosa pine (Pinus ponderosa) plantations in the forest-steppe ecotone in western Patagonia (Licata et al., 2008). Sun et al. (2006) estimated that extensive forest plantations in China could reduce the water yield by 50 mm/year or 50% in the semi-arid region of the Loess Plateau, and as much as 300 mm/year or 30% in the tropical south. Yu et al. (2010) estimated that a 10% increase in forest cover would lead to 25.6 mm/year or 13% of water yield reduction in a watershed in the Qilian Mountains of northwest China. Zhang et al. (2018) identified a lack of water as a major factor contributing to cropland abandonment in a rural community in China.

This competition for water must be seen in the context of the overall relationship between water availability and forests and trees, as reviewed by the 2018 Global Forest Expert Panel of the Collaborative Partnership on Forests (Creed and van Noordwijk, 2018). On the one hand, forests – especially natural forests – contribute to the resilience of water supply through conservation of soil and water resources, providing freshwater during dry seasons and mitigating floods during wet seasons in many parts of the world. On the other hand, forests – especially fast-growing plantations – use water themselves, reducing freshwater availability (Kim *et al.*, 2014). A systems approach that integrates hydrological processes at all scales is needed to understand the role of forests in water availability and the subsequent impact on people's livelihood under a changing climate. If properly managed, forests can help enhance the resilience and quality of water supplies.

Some tree species that are planted can also become invasive, and in turn, compete for water and growing space. This is particularly true of species that have deep roots, high transpiration rate and a long growing season. In South Africa, the invasive species black wattle, Eucalyptus and pines can blanket the landscape with non-native forests (McGarry et al., 2005). Le Maitre et al. (2019) estimated that forests of these invasive alien species use as much as 970 m³/ha/year of water, having a significant negative effect on the Western Cape water supply system. Guava (Psidium quajava) and its sister species strawberry guava (Psidium cattlenianum), native species to tropical America, have become invasive in many parts of the world, including Australia, southern Africa, southeast US, Hawaii, Galapagos Islands and Madagascar, following human introduction. Many are dispersed by seeds and regenerate by root suckering and are extremely difficult to remove and almost impossible to eradicate (Walsh et al., 2008; DeSisto et al., 2020). These tree species drastically change the character of the ecosystems that they invade, including cropland and pastures.

This is related to the phenomena of woody encroachment and expansion (also known as "bush encroachment" or "woody thickening"). This process, whereby trees become more numerous, larger or expand into open ecosystems such as grasslands, has been widely reported in the dry tropics of Africa and to a lesser extent in Australia and Latin America (Liu et al., 2015; Skowno et al., 2017; Stevens et al., 2017). The causes of encroachment and its heterogeneous manifestation remain controversial, but CO₂ fertilisation, climate change and land use management have all been suggested as possibilities (Bond and Midgley, 2012; Abreu et al., 2017; Venter et al., 2018), and some models predict it will increase rapidly over the coming century (Higgins and Scheiter, 2012).

Woody encroachment is often associated with negative impacts on biodiversity (Parr et al., 2012;

Ratajczak et al., 2012; Parr et al., 2014) as many species require the open habitat that is lost. There are also large potential impacts on social well-being and rural economic activities. These include the loss or reduction in productivity of grazing land and the extra costs incurred by pastoralists or ranchers for 'debushing' or thicket clearance (Stafford *et al.*, 2017). These impacts can have severe consequences for livelihoods during drought years (Angassa and Oba, 2008). There is also concern that if open savannahs become more woody, then they will be less attractive to tourists interested in viewing large mammals (Gray and Bond, 2013).

3.5.3. Winners and losers

Most people lose in human-wildlife conflicts, but some more than others. Poor people bear the brunt of the impacts and stand to lose the most with respect to their total income. Losses of crops and livestock and human injuries or casualties are the direct costs of human-wildlife conflicts. There are also numerous indirect costs as reviewed by Barua et al. (2013). Crop damage contributes to food insecurity among the rural poor. The potential of wildlife-caused human injury or mortality stokes fears in residents, damaging psychosocial well-being. For example, Jadhav and Barua (2012) argue that the fear of an elephant attack exacerbates the mental illness of marginalised people, imposing greater health damage than the physical threat. To mitigate wildlife impacts, farmers engage in extensive guarding, sometimes day and night, and divert limited financial resources to purchase materials for fences and stalls. Schoolage boys may be deployed to guard crops during peak crop-raiding time, compromising their performance at school (Mackenzie et al., 2015).

There are also some hidden ecosystem service benefits of crop-raiding. In South African macadamia orchards, bats and birds directly reduce yields by 26%, but at the same time, they serve as a biocontrol for insects. Exclusion of bats and birds resulted in losses of up to 60% of yield to insects (Linden et al., 2019). Therefore, the presence of bats and birds provides a net gain in macadamia orchards. Byg et al. (2017) argued for a 'disaggregated accounting' of both forest services and disservices, and their distribution across people and places. For example, some of the species involved in human-wildlife conflicts are highly attractive to tourists and may, therefore, generate higher revenues from ecotourism than losses from crop-raiding and livestock predation.

Likewise, the impacts of invasive alien tree species on local people's livelihoods are complicated,



Smallholder logging activities in the state of Amapá, Brazil Photo © Reem Hajjar

as reviewed by McGarry *et al.* (2005). Depending on the traits of an invasive species and its invasion history, its impacts on people's livelihoods can be extremely disruptive, neutral, or positive. For the same invasive alien species, it may have a large range of influences for different people depending on their culture and adaptability. For example, some invasive alien tree species may take up precious space, disrupting subsistence agriculture. But other people can take advantage of the species by harvesting valuable components, such as fruit, nuts or wood to burn as firewood or convert to charcoal for sale. This further depends on people's ability to manage these species to neutralise their negative impacts (McGarry *et al.*, 2005).

3.6 Negative Impacts of Deforestation and Forest Ecosystem Disturbance

The negative impacts of deforestation lie on the other side of the coin from the ecosystem ser-

vices provided by standing forests. Such negative impacts include both the direct effects of landuse changes and the effects of other simultaneous changes. The direct effects of deforestation include the physical effects of road construction, reduction in tree cover, and habitat *fragmentation* (Laurance, 1999; Evans, 2016). These may, for example, change the hydroclimate, with potentially more variable rainfall (De Sales *et al.*, 2020). As this report is written in the context of the COVID-19 pandemic that is believed to have originated in wildlife markets (Mackenzie and Smith, 2020), we focus here on the public health implications of deforestation.

A causal chain links land use change and the spread of zoonotic diseases. For example, changes in land-use associated with mining, agriculture and plantations bring about new risk factors that affect the transmission of diseases (Bauch *et al.*, 2015; Whitmee *et al.*, 2015). As habitats are altered, so is the predator-prey relationship leading

to a change in the ecological regulation of parasitic diseases. This change also leads to a reduction in the diversity of organisms, and facilitates the flow of pathogens. In-migration and human population growth accompany deforestation, resulting in greater exposure to disease. In turn, diseases that may not have been viable in small populations may thrive and become endemic. In addition, and in similar ways, deforestation may also increase the likelihood of emergent infectious diseases by increasing contact between forest animals, domestic animals and humans (Patz et al., 2005). For example, encroachment into forest lands is thought to have been a factor in the emergence of several viral diseases including Ebola, Marburg, Nipiah and Ross River Viruses (Chua et al., 2002). Similarly, the loss of forests and the construction of roads leading to an increase in bushmeat hunting and trade are thought to have contributed to the original zoonosis of HIV (Wolfe et al., 2004).

Conversion of forests and the alteration of physical characteristics may create new breeding sites for populations of disease-carrying organisms, alter micro-climatic conditions and eventually lead to the emergence of zoonotic diseases (Dobson et al., 2020; Gibb et al., 2020). Road building, mining pits and logging can all create new breeding grounds for insect vectors such as mosquitoes. For example, in the Peruvian Amazon, the biting rate of the malaria carrying mosquito Anopheles darlingi was found to be proportional to the area of land modification and inversely proportional to the area of remaining forest (Vittor et al., 2006). In some instances changes in habitat have had positive effects on the prevalence of infectious diseases. For example, in many parts of sub-Saharan Africa, a reduction in the prevalence of malaria has been traced back to the draining of wetlands (Keiser et al., 2005). Equally, in Thailand, deforestation is thought to have reduced the overall burden of malaria (Yasuoka and Levins, 2007). Overall, negative effects of deforestation in terms of disease prevalence, however, outweigh positive effects (Melrose, 2011). In their study in the Amazon, MacDonald and Mordecai (2019) found that deforestation increases the risk of malaria transmission. Garg (2019) found similar results in Indonesia, concluding that primary forest loss increased the incidence of malaria based on robust counterfactual estimation methods. In contrast, in their study across sub-Saharan Africa, Bauhoff and Busch (2020) found no relationship between deforestation and malaria, suggesting that the socioeconomics of deforestation may be different in that region compared to Latin American and Asian contexts.

3.7 Conclusions

Evidence from around the world shows that the goods and ecosystem services of forests and treebased systems can play important roles in poverty alleviation, especially by consistently contributing to income or consumption, thereby helping poor households secure their socio-economic and cultural status. In particular contexts, poor households also use forests and trees on farms to exit poverty and to mitigate risks, thereby avoiding transient poverty. On the other hand, those forests and trees may also generate negative externalities that contribute to trapping or moving households into poverty. All four forest-poverty relationships are strongly context-dependent: which relationship manifests in a particular location depends on the natural resource endowment and cultural, religious, economic, political and institutional setting. However, regardless of context, there are relatively few documented cases of forests or trees either being the primary pathway out of poverty or generating significant negative externalities, and thus it is not possible to draw any general conclusions about those dynamics.

The contrast between the widespread dependence of the poor on forests and trees for their livelihoods and well-being, but their limited ability to use those same resources to exit poverty, begs an explanation. Possibilities that deserve more consideration include the influence of international investment and trade on the allocation of benefits from forests and the influence of the rules governing access to forests on the global distribution of prosperity. More research on the dynamics between forests and *inequality* across countries is merited.

While forests and trees also do not offer a 'silver bullet' for securing or stabilising well-being, there is more evidence that the poor have been able to harness them to meet these objectives. The role of forests and trees is relatively more important in locations that are remote and thus offer limited access to markets and public services. Further, this role may become more important with climate change, as a way to maintain livelihoods and manage risks in a future that presents ever more challenges for the rural poor.

3.8 References

- Abdul-Mumeen, I., Zakpaa, H. D. and Mills-Robertson, F. C., 2013. Biochemical and microbiological analysis of Shea nut cake: A waste product from Shea butter processing. Journal of Agricultural Biotechnology and Sustainable Development 5(4): 61-68.
- Aboyella, C., 2002. An economic analysis of Sheanuts and cashew production in the Bawku East District. A BSc. Dissertation submitted to the Department of Economics and Agribusiness, University of Ghana.
- Abreu, R. C. R., Hoffmann, W. A., Vasconcelos, H. L., Pilon, N. A., Rossatto, D. R. and Durigan, G., 2017. The biodiversity cost of carbon sequestration in tropical savanna. Science Advances 3(8): e1701284.
- Acharya, K. P., 2003. Religious and spiritual values of forest plants in Nepal. A paper submitted to the XII Forest Congress, Quebec City Canada.
- Adams, W. M. and Hulme, D., 2001. Conservation & Community: Changing Narratives, Policies
 & Practices in African Conservation. In: African
 Wildlife & Livelihoods: The Promise and Performance of Community Conservation, Oxford: James Currey, edited
 by D. Hulme and M. Murphree. Cambridge MA: Harvard University Press, pp. 73-98.
- Aerts, R., Van Overtveld, K., November, E., Wassie, A., Abiyu, A., Demissew, S., Daye, D.D., et al., 2016. Conservation of the Ethiopian church forests: Threats, opportunities and implications for their management. Science of the Total Environment 551: 404-414.
- Afonso, H., LaFleur, M. and Alarcón, D., 2015. Development Issues No. 1: Concepts of Inequality. UN/ DESA. (available online at: https://www.un.org/ development/desa/dpad/document_gem/wessreport/. [accessed on: 1 September 2020]).
- Agroforestry Network, 2018. Achieving the Global Goals through Agroforestry (available at: https: // agroforestrynetwork.org/database_post/achievingthe-global-goals-through-agroforestry/ [accessed on: 1 September 2020]).
- Agustino, S., Mataya, B., Senelwa, K. and Achigan-Dako, G. E., 2011. Non-wood forest products and services for socio-economic development. A Compendium for Technical and Professional Forestry Education. Nairobi: The African Forest Forum.
- Agyei, F. K., Hansen, C. P. and Acheampong, E., 2020. Access along Ghana's charcoal commodity chain. Society & Natural Resources 33(2): 224-243.
- Ahenkan, A. and Boon, E., 2011. Non-timber forest products (NTFPs): Clearing the confusion in semantics. *Journal of Human Ecology* 33(1): 1-9.
- Ajayi, O. C., Place, F., Akinnifesi, F. K. and Sileshi, G. W., 2011. Agricultural success from Africa: the case of fertilizer tree systems in southern Africa (Malawi, Tanzania, Mozambique, Zambia and Zimbabwe). International Journal of Agricultural Sustainability 9(1): 129-136.

- Akinnifesi, F. K., Ajayi, O. C., Sileshi, G., Chirwa, P. W. and Chianu, J., 2010. Fertiliser trees for sustainable food security in the maize-based production systems of East and Southern Africa. A review. Agronomy for Sustainable Development 30(3): 615-629.
- Alexiades, M. N., 2003. Ethnobotany in the third millennium: expectations and unresolved issues. *Delpinoa* 45: 15-28.
- Ali, A., 2018. Forest-based livelihoods, income, and poverty: Empirical evidence from the Himalayan region of rural Pakistan. *Journal of Rural Studies* 57: 44-54.
- Alvaredo, F., Chancel, L., Piketty, T., Saez, E., and Zucman, G., 2018. World Inequality Report 2018. Cambridge, MA: Harvard University Press.
- Amadu, F.O., Miller, D.C. and McNamara, P.E., 2020. Agroforestry as a pathway to agricultural yield impacts in climate-smart agriculture investments: Evidence from southern Malawi. Ecological Economics 167: 106443.
- Andam, K. S., Ferraro, P. J., Sims, K. R. E., Healy, A. and Holland, M. B., 2010. Protected areas reduced poverty in Costa Rica and Thailand. PNAS 107: 9996-10001.
- Angassa, A. and Oba, G., 2008. Herder Perceptions on Impacts of Range Enclosures, Crop Farming, Fire Ban and Bush Encroachment on the Rangelands of Borana, Southern Ethiopia. *Human Ecology* 36(2): 201-215.
- Angelsen, A. and Kaimowitz, D., 1999. Rethinking the causes of deforestation: Lessons from economic models. The World Bank Research Observer 14(1): 73-98.
- Angelsen, A., Jagger, P., Babigumira, R., Belcher, B., Hogarth, N. J., Bauch, S., Börner, J., Smith-Hall, C. and Wunder, S., 2014. Environmental income and rural livelihoods: A global-comparative analysis. World Development 64(Supplement 1): S12-S28.
- Angelsen, A., Larsen, H. O., and Olsen, C. S., 2012. Measuring livelihoods and environmental dependence: Methods for research and fieldwork. Washington, D.C.: Earthscan.
- Aniah, P. and Yelfaanibe, A., 2016. Learning from the past: the role of sacred groves and shrines in environmental management in the Bongo District of Ghana. Environmental Earth Sciences 75(10): 916.
- Antinori, C. and Bray, D. B., 2005. Community forest enterprises as entrepreneurial firms: Economic and institutional perspectives from Mexico. World Development 33: 1529-1543.
- Arnold, M., Powell, B., Shanley, P. and Sunderland, T. C. H., 2011. Forests, biodiversity and food security. The International Forestry Review 2011(13): 3.
- Asamoah, R. K. F., 1985. Uses of fallow trees and farm practices in Ho forest districts (Ghana). Unpublished PhD thesis. Kumasi (Ghana): Institute of Renewable Natural Resources, University of Science and Technology.

Asprilla-Perea, J. and Díaz-Puente, J. M., 2019. Importance of wild foods to household food security in tropical forest areas. *Food Security* 11(1): 15-22.

Asselin, H., 2015. Indigenous Forest Knowledge. In: Routledge handbook of forest ecology, edited by K. Peh, R. Corlett and Y. Bergeron. New York: Earthscan.

Assembe-Mvondo, S., Brockhaus, M. and Lescuyer, G., 2013. Assessment of the effectiveness, efficiency and equity of benefit-sharing schemes under large-scale agriculture: Lessons from land fees in Cameroon. The European Journal of Development Research 25(4): 641-656.

Barrios, E., Valencia, V., Jonsson, M., Brauman, A.,
Hairiah, K., Mortimer, P.E. and Okubo, S., 2018.
Contribution of trees to the conservation of
biodiversity and ecosystem services in agricultural
landscapes. International Journal of Biodiversity Science,
Ecosystem Services & Management 14(1): 1-16.

Barua, M., Bhagwat, S.A. and Jadhav, S., 2013. The hidden dimensions of human-wildlife conflict: Health impacts, opportunity and transaction costs. *Biological Conservation* 157: 309-316.

Bauch, S. C., Birkenbach, A. M., Pattanayak, S. K., and Sills, E. O., 2015. Public health impacts of ecosystem change in the Brazilian Amazon. PNAS 112(24): 7414-7419.

Baudron, F., Chavarría, J.Y.D., Remans, R., Yang, K. and Sunderland, T., 2017. Indirect contributions of forests to dietary diversity in Southern Ethiopia. Ecology and Society 22(2): 28.

Baudron, F., Schultner, J., Duriaux, J.Y., Gergel, S.E. and Sunderland, T., 2019. Agriculturally productive yet biodiverse: human benefits and conservation values along a forest-agriculture gradient in Southern Ethiopia. Landscape ecology 34(2): 341-356.

Bauhoff, S. and Busch, J., 2020. Does deforestation increase malaria prevalence? Evidence from satellite data and health surveys. World Development 127: 104734.

Bawa, A., Atengdem, P. B. and Abukari, S. C., 2017. Assessing the Impact of Community Life Improvement Program (CLIP) on the Livelihood Assets of Women in the Sheabutter Processing Business in Karaga District of Northern Ghana. Journal of Social Sciences 13(2): 108-117.

Bedelian, C. and Ogutu, J. O., 2017. Trade-offs for climate-resilient pastoral livelihoods in wildlife conservancies in the Mara ecosystem, Kenya. *Pastoralism* 7(1): 10.

Belcher, B. and Kusters, K., 2004. Non-timber forest product commercialisation: Development and conservation lessons. In: Forest products, livelihoods and conservation: Case studies of non-timber forest product systems, edited by K. Kusters and B. Blecher. Bogor: CIFOR, pp. 1-22.

Belcher, B. M., 2005. Forest product markets, forests and poverty reduction. International Forestry Review 7(2): 82-89. Belcher, B., Ruíz-Pérez, M. and Achdiawan, R., 2005. Global patterns and trends in the use and management of commercial NTFPs: Implications for livelihoods and conservation. World Development 33(9): 1435-1452.

Bhagwat, S. A. and Rutte, C., 2006. Sacred groves: potential for biodiversity management. Frontiers in Ecology and the Environment 4(10): 519-524.

Bista, R. and Song, C., (under review). Crop raiding and livestock depredation: Understanding the feedback effects of community forestry in the Middle Mountains of Nepal. *Environment, Development and Sustainability*, under review.

Blaney, S., Beaudry, M. and Latham, M., 2009. Contribution of natural resources to nutritional status in a protected area of Gabon. Food and nutrition bulletin 30(1): 49-62.

Blaser, W.J., Oppong, J., Hart, S.P., Landolt, J., Yeboah, E. and Six, J., 2018. Climate-smart sustainable agriculture in low-to-intermediate shade agroforests. Nature Sustainability 1(5): 234-239.

Boedecker, J., Termote, C., Assogbadjo, A.E., Van Damme, P. and Lachat, C., 2014. Dietary contribution of wild edible plants to women's diets in the buffer zone around the Lama forest, Benin – an underutilized potential. Food. Sec 6: 833-849.

Bond, W. J. and Midgley, G. F., 2012. Carbon dioxide and the uneasy interactions of trees and savannah grasses. Philosophical Transactions of the Royal Society B: Biological Sciences 367(1588): 601-612.

Bray, D.B., Antinori, C. and Torres-Rojo, J.M., 2006. The Mexican model of community forest management: The role of agrarian policy, forest policy and entrepreneurial organization. *Forest Policy and Economics* 8(4): 470-484.

Bray, D.B., Durán Medina, E., Merino Pérez, L., Torres Rojo, J.M. and Velázquez Montes, A., 2007. Nueva evidencia: los bosques comunitarios de México protegen el ambiente, disminuyen la pobreza y promueven la paz social. Mexico City: Agrupación Sierra Madre/Consejo Civil Mexicano para la Silvicultura Sostenible.

Bruun, T.B., Berry, N., De Neergaard, A., Xaphokahme, P., McNicol, I. and Ryan, C.M., 2018. Long rotation swidden systems maintain higher carbon stocks than rubber plantations. Agriculture, Ecosystems & Environment 256: 239-249.

Bruun, T.B., de Neergaard, A., Burup, M.L., Hepp, C.M., Larsen, M.N., Abel, C., Aumtong, S., Magid, J. and Mertz, O., 2017. Intensification of upland agriculture in Thailand: development or degradation? *Land Degradation & Development* 28(1): 83-94.

Bulkan, J. 2017. Social, cultural and spiritual needs and values. In: Sustainable forest management: from concept to practice, edited by J. Innes and A. Tikina. London: Earthscan, pp. 241-256.

Burgess, R., Hansen, M., Olken, B.A., Potapov, P. and Sieber, S., 2012. The political economy of deforestation in the tropics. *The Quarterly Journal of Economics* 127(4): 1707-1754. Byg, A., Novo, P., Dinato, M., Moges, A., Tefera, T., Balana, B., Woldeamanuel, T. and Black, H., 2017. Trees, soils, and warthogs: Distribution of services and disservices from reforestation areas in southern Ethiopia. Forest Policy and Economics 84: 112-119.

Calder, I.R., 2007. Forests and water-ensuring forest benefits outweigh water costs. Forest ecology and management 251(1-2): 110-120.

Cardozo, E.G., Muchavisoy, H.M., Silva, H.R., Zelarayán, M.L.C., Leite, M.F.A., Rousseau, G.X. and Gehring, C., 2015. Species richness increases income in agroforestry systems of eastern Amazonia. Agroforestry Systems 89(5): 901-916.

Carignan, R. and Steedman, R., 2011. Impacts of major watershed perturbations on aquatic ecosystems. *Canadian Journal of Fisheries and Aquatic Sciences* 57: 1-4.

de Carvalho, A.L.C., Araújo, A.R., Machado, T.M.M., Ribon, R. and Lopes, L.E., 2019. Wildlife and damage to agriculture: An ethnobiological approach with rural producers in southeastern Brazil. Ornithology Research 27(1): 17-26.

Cavendish, W., 2000. Empirical regularities in the poverty-environment relationship of rural households: Evidence from Zimbabwe. World Development 28: 1979-2003.

Cawthorn, D.-M. and Hoffman, L. C., 2015. The bushmeat and food security nexus: A global account of the contributions, conundrums and ethical collisions. *Food Research International* 76: 906-925.

Chaplin-Kramer, R., O'Rourke, M., Zhang, W., Robinson, B., Schellhorn, N., Gratton, C., Rosenheim, J.A., et al., 2019. Measuring what matters: actionable information for conservation biocontrol in multifunctional landscapes. Frontiers in Sustainable Food Systems 3: 60.

Chavarría, J.Y.D., Baudron, F. and Sunderland, T., 2018. Retaining forests within agricultural landscapes as a pathway to sustainable intensification: Evidence from Southern Ethiopia. Agriculture, Ecosystems & Environment 263: 41-52.

Cheng, S.H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Rana, P., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. *Environmental Evidence* 8(1).

Chetri, M., Odden, M., Devineau, O. and Wegge, P., 2019. Patterns of livestock depredation by snow leopards and other large carnivores in the Central Himalayas, Nepal. Global Ecology and Conservation 17.

Chiteculo, V., Lojka, B., Surový, P., Verner, V., Panagiotidis, D. and Woitsch, J., 2018. Value Chain of Charcoal Production and Implications for Forest Degradation: Case Study of Bié Province, Angola. *Environments* 5(11): 113.

Chua, K.B., Chua, B.H. and Wang, C.W., 2002. Anthropogenic deforestation, El Nino and the emergence of Nipah virus in Malaysia. *Malaysian Journal of Pathology* 24(1): 15-21. Clements, T., Suon, S., Wilkie, D.S. and Milner-Gulland, E.J., 2014. Impacts of protected areas on local livelihoods in Cambodia. *World Development* 64: S125-S134.

Cocks, M. L., 2006. Wild resources and practices in rural and urban households in South Africa: implications for bio-cultural diversity conservation. PhD Thesis, Wageningen: Wageningen University.

Colfer, C. J. P., Basnett, B. S. and Elias, M., 2016. Gender and forests: climate change, tenure, value chains and emerging issues. Abingdon and New York: Routledge.

Colfer, C.J.P., Sheil, D. and Kishi, M., 2006. Forests and human health: assessing the evidence. Bogor: CIFOR.

Coquery-Vidrovitch, C., 2017. Le Congo au temps des grandes compagnies concessionnaires 1898-1930. Walter de Gruyter GmbH & Co KG.

Coulibaly, J.Y., Chiputwa, B., Nakelse, T. and Kundhlande, G., 2017. Adoption of agroforestry and the impact on household food security among farmers in Malawi. Agricultural Systems 155: 52-69.

Creed, I. F. and van Noordwijk, M., 2018. Forest and Water on a Changing Planet: Vulnerability, Adaptation and Governance Opportunities: A Global Assessment Report. Vienna: IUFRO.

Cunningham, A. B., 1993. African medicinal plants. Paris: UNESCO.

Curtis, P.G., Slay, C.M., Harris, N.L., Tyukavina, A. and Hansen, M.C., 2018. Classifying drivers of global forest loss. *Science* 361(6407): 1108-1111.

Daily, G. C. and Matson, P. A., 2008. Ecosystem services: from theory to implementation. PNAS 105(28): 9455-9456.

Dainese, M., Martin, E.A., Aizen, M.A., Albrecht, M., Bartomeus, I., Bommarco, R., Carvalheiro, L.G., et al., 2019. A global synthesis reveals biodiversitymediated benefits for crop production. Science advances 5(10): eaax0121.

Daniel, K.S., Udeagha, A.U. and Jacob, D.E., 2016. Sociocultural importance of sacred forests conservation in south southern Nigeria. African Journal of Sustainable Development 6(2): 251-268.

Darr, B., Pretzsch, J., Depzinsky, T., Darr, B., Pretzsch, J. and Depzinsky, T., 2009. Traditional forest perception and its relevance to forest conservation among the Tiriki in Kenya. In: Traditional forest-related knowledge and sustainable forest management in Africa. IUFRO world series 23, edited by J.A. Parrotta, A. Oteng-Yeboah and J. Cobbinah. Accra: IUFRO.

Dauvergne, P. and Neville, K. J., 2010. Forests, food, and fuel in the tropics: the uneven social and ecological consequences of the emerging political economy of biofuels. *The Journal of peasant studies* 37(4): 631-660.

Debela, B., Shively, G., Angelsen, A. and Wik, M., 2012. Economic shocks, diversification, and forest use in Uganda. *Land Economics* 88(1): 139-154. Delabre, I., Boyd, E., Brockhaus, M., Carton, W., Krause, T., Newell, P., Wong, G.Y. and Zelli, F., 2020. Unearthing the myths of global sustainable forest governance. *Global Sustainability* 3: 1-10.

DeSisto, C.M., Park, D.S., Davis, C.C., Ramananjato, V., Tonos, J.L. and Razafindratsima, O.H., 2020. An invasive species spread by threatened diurnal lemurs impacts rainforest structure in Madagascar. *Biological Invasions* (in press).

Dhungana, R., Lamichhane, B.R., Savini, T., Dhakal, M., Poudel, B.S. and Karki, J.B., 2019. Livestock depredation by leopards around Chitwan National Park, Nepal. *Mammalian Biology* 96(1): 7-13.

Dobson, A.P., Pimm, S.L., Hannah, L., Kaufman, L., Ahumada, J.A., Ando, A.W., Bernstein, A., et al., 2020. Ecology and economics for pandemic prevention. Science 369(6502): 379-381.

von Dohren, P. and Haase, D., 2015. Ecosystem disservices research: A review of the state of the art with a focus on cities. *Ecological indicators* 52: 490-497.

Doolittle, A. A., 2007. Native land tenure, conservation, and development in a pseudo-democracy: Sabah, Malaysia. *The Journal of Peasant Studies* 34(3-4): 474-497.

Dove, M. R., 1983. Theories of swidden agriculture, and the political economy of ignorance. *Agroforestry Systems* 1(2): 85-99.

Dressler, W.H., Smith, W., Kull, C.A., Carmenta, R. and Pulhin, J.M., 2020. Recalibrating burdens of blame: Anti-swidden politics and green governance in the Philippine Uplands. *Geoforum*.

Dressler, W.H., Wilson, D., Clendenning, J., Cramb, R., Keenan, R., Mahanty, S., Bruun, T.B., Mertz, O. and Lasco, R.D., 2017. The impact of swidden decline on livelihoods and ecosystem services in Southeast Asia: A review of the evidence from 1990 to 2015. Ambio 46(3): 291-310.

Dudley, N., Higgins-Zogib, L. and Mansourian, S., 2009. The links between protected areas, faiths, and sacred natural sites. *Conservation Biology* 23(3): 568-577.

Ekoko, F., 2000. Balancing politics, economics and conservation: The case of the Cameroon forestry law reform. Development and Change 31(1): 131-154.

Evans, M.C., 2016. Deforestation in Australia: drivers, trends and policy responses. Pacific Conservation Biology 22(2): 130-150.

Fa, J.E., Olivero, J., Real, R., Farfán, M.A., Márquez, A.L., Vargas, J.M., Ziegler, S., et al., 2015. Disentangling the relative effects of bushmeat availability on human nutrition in central Africa. Scientific Reports 5: 8168.

FAO, 2005. FAOSTAT, the FAO 's on-line statistical database for agriculture, fisheries and forestry. (available at: http: // faostat.external.fao.org. [accessed on: 1 September 2020]).

FAO, 2007. The world's mangroves 1980-2005. FAO Forestry Paper 153. Rome: FAO.

- FAO, 2013. Edible insects: future prospects for food and feed security. FAO Forestry Paper. Rome: FAO.
- FAO, 2014. State of the World's Forests. Enhancing the socioeconomic benefits from forests. Rome: FAO.
- FAO, 2019. Global forest products and figures 2018. Rome: FAO.

FAO, CIFOR, IFRI and World Bank, 2016. National socioeconomic surveys in forestry: guidance and survey modules for measuring the multiple roles of forests in household welfare and livelihoods, edited by R.K. Bakkegaard, A. Agrawal, I. Animon, N. Hogarth, D. Miller, L. Persha, E. Rametsteiner, S. Wunder and A. Zezza. FAO Forestry Paper No. 179. Rome: FAO.

FAOSTAT, 2008. The Food and Agricultural Organisation of the United Nation Production Databases [Online]. (available at: http://www.fao.org. [accessed on 1 September 2020]).

Fischer, J., Abson, D.J., Bergsten, A., Collier, N.F.,
Dorresteijn, I., Hanspach, J., Hylander, K., et al., 2017.
Reframing the food-biodiversity challenge. *Trends in Ecology & Evolution* 32(5): 335-345.

Fisher, M., Chaudhury, M. and McCusker, B., 2010. Do forests help rural households adapt to climate variability? Evidence from southern Malawi. World Development 38(2010): 1241-1250.

FLARE, 2019. Livelihoods and Well-being tool (LivWell). (available at: http://www.forestlivelihoods.org/ resources/livwell/. [accessed on 12 December 2019]).

Foley, J.A., DeFries, R., Asner, G.P., Barford, C., Bonan, G., Carpenter, S.R., Chapin, F.S., et al., 2005. Global consequences of land use. Science 309(5734): 570-574.

Franzel, S., Carsan, S., Lukuyu, B., Sinja, J. and Wambugu, C., 2014. Fodder trees for improving livestock productivity and smallholder livelihoods in Africa. *Current Opinion in Environmental Sustainability* 6: 98-103.

Franzel, S., Wambugu, C., Arimi, H. and Stewart, J., 2008. Fodder shrubs for improving livestock productivity and sustainable land management in East Africa. World Bank, Sustainable Land Management Sourcebook. Washington D.C.: World Bank.

Fru, B. S., 2014. Contribution of traditional institutions on the sustainable management of sacred forests: Case Study of Mankon Sacred Forests Northwest Region, Cameroon. University of Dschang, Cameroon.

Galaz, V., Crona, B., Dauriach, A., Scholtens, B. and Steffen, W., 2018. Finance and the Earth system-Exploring the links between financial actors and non-linear changes in the climate system. *Global Environmental Change* 53: 296-302.

Galway, L.P., Acharya, Y. and Jones, A.D., 2018. Deforestation and child diet diversity: A geospatial analysis of 15 Sub-Saharan African countries. *Health and Place* 51: 78-88.

Garg, T., 2019. Ecosystems and human health: the local benefits of forest cover in Indonesia. *Journal of* Environmental Economics and Management 98: 102271. Garibaldi, L.A., Steffan-Dewenter, I., Kremen, C., Morales, J.M., Bommarco, R., Cunningham, S.A., Carvalheiro, L.G., et al., 2011. Stability of pollination services decreases with isolation from natural areas despite honey bee visits. *Ecology Letters* 14(10): 1062-1072.

Garrity, D.P., 2004. Agroforestry and the achievement of the Millennium Development Goals. Agroforestry systems 61(1-3): 5-17.

Germain, R., Ghosh, C. and Jayasuriya, M., 2018. Community forestry in the State of Uttarakhand, India: Not meeting the needs of the villagers. Smallscale Forestry 17(2): 225-242.

Gibb, R., Redding, D. W., Chin, K. Q., Donnelly, C. A., Blackburn, T. M., Newbold, T. and Jones, K. E., 2020. Zoonotic host diversity increases in humandominated ecosystems. *Nature* 584: 398-402.

Giri, K. and Darnhofer, I., 2010. Outmigrating men: A window of opportunity for women's participation in community forestry?. Scandinavian journal of forest research 25(sup9): 55-61.

Golden, C.D., Fernald, L.C., Brashares, J.S., Rasolofoniaina, B.R. and Kremen, C., 2011. Benefits of wildlife consumption to child nutrition in a biodiversity hotspot. Proceedings of the National Academy of Sciences 108(49): 19653-19656.

Golden, C.D., Bonds, M.H., Brashares, J.S., Rodolph Rasolofoniaina, B.J. and Kremen, C., 2014. Economic valuation of subsistence harvest of wildlife in Madagascar. *Conservation Biology* 28: 234-243.

Gomes, C.V.A., Vadjunec, J.M. and Perz, S.G., 2012. Rubber tapper identities: Political-economic dynamics, livelihood shifts, and environmental implications in a changing Amazon. *Geoforum* 43(2): 260-271.

Goulding, M., 1981. Natural history of the flood fishes.In: Man and Fisheries on an Amazon Frontier, edited byM. Goulding. Dordrecht: Springer, pp. 69-115.

Gouyon, A., De Foresta, H. and Levang, P., 1993. Does 'jungle rubber' deserve its name? An analysis of rubber agroforestry systems in southeast Sumatra. Agroforestry systems 22: pp.181-206.

Gras, P., Tscharntke, T., Maas, B., Tjoa, A., Hafsah, A. and Clough, Y., 2016. How ants, birds and bats affect crop yield along shade gradients in tropical cacao agroforestry. *Journal of Applied Ecology* 53(3): 953-963.

Gray, E. F. and Bond, W. J., 2013. Will woody plant encroachment impact the visitor experience and economy of conservation areas? *Koedoe* 55(1): 1106.

Gray, J. M. and Song, C., 2012. Mapping leaf area index using spatial, spectral and temporal information from multiple sensors. *Remote Sensing of Environment* 119: 173-183.

Griggs, D., Stafford-Smith, M., Gaffney, O., Rockström, J., Öhman, M.C., Shyamsundar, P., Steffen, W., et al., 2013. Sustainable development goals for people and planet. Nature 495(7441): 305-307. Guenat, S., Kaartinen, R. and Jonsson, M., 2019. Shade trees decrease pest abundances on brassica crops in Kenya. Agroforestry Systems 93(2): 641-652.

Hall, C., Macdiarmid, J.I., Matthews, R.B., Smith, P., Hubbard, S.F. and Dawson, T.P., 2019. The relationship between forest cover and diet quality: a case study of rural southern Malawi. *Food Security* 11(3): 635-650.

Hänke, H., Barkmann, J., Blum, L., Franke, Y., Martin, D.A., Niens, J., Osen, K., et al., 2018. Socio-economic, land use and value chain perspectives on vanilla farming in the SAVA Region (north-eastern Madagascar): The Diversity Turn Baseline Study (DTBS). Diskussionsbeitrag.

Hardin, R. and Bahuchet, S., 2011. Concessionary politics: Property, patronage, and political rivalry in Central African forest management. *Current Anthropology* 52(S3): S113-S125.

Harich, F.K., Treydte, A.C., Sauerborn, J. and Owusu, E.H., 2013. People and wildlife: Conflicts arising around the Bia Conservation Area in Ghana. *Journal for Nature Conservation* 21: 342-349.

Hatfield, J. L., 2016. Increased Temperatures Have Dramatic Effects on Growth and Grain Yield of Three Maize Hybrids. Agricultural & Environmental Letters 1(1): 150006.

Hegde, R. and Bull, G.Q., 2011. Performance of an agro-forestry based Payments-for-Environmental-Services project in Mozambique: A household level analysis. *Ecological Economics* 71: 122-130.

Hending, D., Andrianiaina, A., Maxfield, P., Rakotomalala, Z. and Cotton, S., 2019. Floral species richness, structural diversity and conservation value of vanilla agroecosystems in Madagascar. African Journal of Ecology 58(1): 100-111.

Heubach, K., Wittig, R., Nuppenau, E.A. and Hahn, K., 2011. The economic importance of non-timber forest products (NTFPs) for livelihood maintenance of rural west African communities: A case study from northern Benin. *Ecological Economics* 70(11): 1991-2001.

Hickey, G.M., Pouliot, M., Smith-Hall, C., Wunder, S. and Nielsen, M.R., 2016. Quantifying the economic contribution of wild food harvests to rural livelihoods: A global-comparative analysis. *Food Policy* 62: 122-132.

Higgins, S. I. and Scheiter, S., 2012. Atmospheric CO₂ forces abrupt vegetation shifts locally, but not globally. Nature 488(7410): 209-212.

Himes-Cornell, A., Pendleton, L. and Atiyah, P., 2018. Valuing ecosystem services from blue forests: a systematic review of the valuation of salt marshes, sea grass beds and mangrove forests. *Ecosystem Services* 30: 36-48.

HLPE, 2014. Sustainable fisheries and aquaculture for food security and nutrition. A report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security. Rome: Committee on World Food Security. HLPE, 2017. Food systems. A report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security. Rome: Committee on World Food Security.

Hoelle, J. 2015. Rainforest cowboys: the rise of ranching and cattle culture in Western Amazonia. Austin: University of Texas Press.

Holmern, T., Nyahongo, J. and E. Roskafa, E., 2007. Livestock loss caused by predators outside the Serengeti National Park, Tanzania. *Biological Conservation* 135: 518-526.

Hu, B., Zeng, L.P., Yang, X.L., Ge, X.Y., Zhang, W., Li, B., Xie, J.Z., et al., 2017. Discovery of a rich gene pool of bat SARS-related coronaviruses provides new insights into the origin of SARS coronavirus. PLOS Pathogens 13(11): e1006698.

Huelsz, J. S. and Negreros-Castillo P., 2014. Communitybased forest management in Quintana Roo, Mexico. *IUFRO World Series* 32: 131-151.

Humphries, S., Holmes, T., de Andrade, D.F.C., McGrath, D. and Dantas, J.B., 2020. Searching for win-win forest outcomes: Learning-by-doing, financial viability, and income growth for a community-based forest management cooperative in the Brazilian Amazon. World Development 125: 104336.

Hurni, H., Giger, M., Liniger, H., Studer, R.M., Messerli, P., Portner, B., Schwilch, G., et al., 2015. Soils, agriculture and food security: the interplay between ecosystem functioning and human well-being. *Current Opinion in Environmental Sustainability* 15: 25-34.

Hutchison, J., Spalding, M. and zu Ermgassen, P.,
 2014. The role of mangroves in fisheries enhancement.
 Washington D.C. and Wageningen: The Nature
 Conservancy and Wetlands International.

Ickowitz, A., 2006. Shifting cultivation and deforestation in tropical Africa: critical reflections. Development and Change 37(3): 599-626.

Ickowitz, A., Powell, B., Salim, M.A. and Sunderland, T.C., 2014. Dietary quality and tree cover in Africa. *Global Environmental Change* 24: 287-294.

IIED, 2014. Building bridges for sustainable development. International Institute for Environment and Development Annual Report 2013/14. London: IIED.

Ingles, A. W., 1997. The influence of religious beliefs and rituals on forest conservation in Nepal. In: Nature Is Culture: Indigenous Knowledge and Socio-cultural Aspects of Trees and Forests in Non-European Cultures, edited by K. Seeland. Exeter, UK: Intermediate Technology Publications, pp 57–66.

Inskip, C. and Zimmermann, A., 2009. Human-felid conflict: A review of patterns and priorities worldwide. Oryx 43(01): 18.

IPBES, 2019. Global assessment report on biodiversity and ecosystem services, edited by E. S. Brondizio, J. Settele, S. Díaz, and H. T. Ngo. Bonn: IPBES secretariat. IUCN Uganda, 2016. Implementation of the Community Environment Conservation Fund (CECF) to enhance Forest Landscape Restoration in Uganda. Emerging lessons and recommendations for scaling up. Kampala: IUCN.

Ickowitz, A., Rowland, D., Powell, B., Salim, M.A. and Sunderland, T., 2016. Forests, trees, and micronutrient-rich food consumption in Indonesia. PloS one 11(5): p.e0154139.

Iiyama, M., Neufeldt, H., Dobie, P., Njenga, M., Ndegwa, G. and Jamnadass, R., 2014. The potential of agroforestry in the provision of sustainable woodfuel in sub-Saharan Africa. Current Opinion in Environmental Sustainability 6: 138-147.

Jaquet, S., Schwilch, G., Hartung-Hofmann, F., Adhikari, A., Sudmeier-Rieux, K., Shrestha, G., Liniger, H.P. and Kohler, T., 2015. Does outmigration lead to land degradation? Labour shortage and land management in a western Nepal watershed. Applied geography 62: 157-170.

Jadhav, S. and Barua, M., 2012. The elephant vanishes: Impact of human-elephant conflict on people's wellbeing. *Health & Place* 18(6): 1356-1365.

Jagger, P., Luckert, M.M.K., Duchelle, A.E., Lund, J.F. and Sunderlin, W.D., 2014. Tenure and forest income: Observations from a global study on forests and poverty. World Development 64: S43-S55.

Jaramillo-Giraldo, C., Soares Filho, B., Ribeiro, S.M.C. and Gonçalves, R.C., 2017. Is it possible to make rubber extraction ecologically and economically viable in the Amazon? *The Southern Acre and Chico Mendes Reserve case study* 134: 186-197.

Jessy, M. D., Joseph, P. and George, S., 2017. Possibilities of diverse rubber based agroforestry systems for smallholdings in India. *Agroforestry Systems* 91(3): 515-526.

de Jesús Arce-Mojica, T., Nehren, U., Sudmeier-Rieux, K., Miranda, P.J. and Anhuf, D., 2019. Nature-based solutions (NbS) for reducing the risk of shallow landslides: Where do we stand? *International Journal* of Disaster Risk Reduction 41: 101293.

Johnson, K. B., Jacob, A. and Brown, M. E., 2013. Forest cover associated with improved child health and nutrition: Evidence from the Malawi Demographic and Health Survey and satellite data. *Global Health: Science and Practice* 1(2): 237-248.

Jones, D., Ryan, C. M. and Fisher, J., 2016. Charcoal as a diversification strategy: The flexible role of charcoal production in the livelihoods of smallholders in central Mozambique. *Energy for Sustainable Development* 32: 14-21.

Jones, J.P. andriahajaina, F.B., Ranambinintsoa, E.H., Hockley, N.J. and Ravoahangimalala, O., 2006. The economic importance of freshwater crayfish harvesting in Madagascar and the potential of community-based conservation to improve management. Oryx 40(2): 168-175. Karanth, K.K. and Ranganathan, P., 2018. Assessing human-wildlife interactions in a forest settlement in Sathyamangalam and Mudumalai Tiger Reserves. Tropical Conservation Science 11: 1-14.

Kar, S. P. and Jacobson, M. G., 2012. NTFP income contribution to household economy and related socio-economic factors: Lessons from Bangladesh. Forest Policy and Economics 14(1): 136-142.

Karp, D.S., Mendenhall, C.D., Sandí, R.F., Chaumont, N., Ehrlich, P.R., Hadly, E.A. and Daily, G.C., 2013. Forest bolsters bird abundance, pest control and coffee yield. Ecology letters 16(11): 1339-1347.

Karsenty, A. and Ongolo, S., 2012. Can "fragile states" decide to reduce their deforestation?: The inappropriate use of the theory of incentives with respect to the REDD mechanism. Forest Policy and Economics 18: 38-45.

Kebede, Y., Bianchi, F.J., Baudron, F. and Tittonell, P., 2019. Landscape composition overrides field level management effects on maize stemborer control in Ethiopia. Agriculture, ecosystems & environment 279: 65-73.

Keiser, J., Singer, B. H., and Utzinger, J., 2005. Reducing the burden of malaria in different eco-epidemiological settings with environmental management: a systematic review. *The Lancet Infectious Diseases* 5(11): 695-708.

Kemeuze, V.A., Sonwa, D.J., Nkongmeneck, B.A. and Mapongmetsem, P.M., 2016. Sacred groves and biodiversity conservation in semi-arid area of Cameroon: case study of Diamare plain. In: Botanists of the twenty-first century: roles, challenges and opportunities, edited by N. R. Rakotoarisoa, S. Blackmore and B. Riera. Paris: UNESCO, pp. 171-183.

Kenney-Lazar, M., Wong, G., Baral, H. and Russell, A.J., 2018. Greening rubber? Political ecologies of plantation sustainability in Laos and Myanmar. *Geoforum* 92: 96-105.

Kent, G., 1997. Fisheries, food security, and the poor. Food Policy 22(5): 393-404.

Kibet, S. and Nyamweru, C., 2008. Cultural and biological heritage at risk; the case of the Rabai Kaya forests in Coastal Kenya. *Journal of Human Ecology* 24(4): 287-295.

Kim, Y., Band, L. E. and Song, C., 2014. The influence of forest re-growth on the stream discharge in the North Carolina Piedmont watersheds. *Journal of the American Water Resources Association* 50: 57-73.

Kiptot, E., Franzel, S. and Degrande, A., 2014. Gender, agroforestry and food security in Africa. *Current Opinion in Environmental Sustainability* 6: 104-109.

Kiptot, E. and Franzel, S., 2012. Gender and agroforestry in Africa: a review of women's participation. Agroforestry systems 84(1): 35-58.

Kleinschmit, D., Mansourian, S., Wildburger, C. and Purret, A. eds., 2016. Illegal Logging and Related Timber Trade: Dimensions, Drivers, Impacts and Responses: A Global Scientific Rapid Response Assessment Report. Vienna: IUFRO. Krishna, A., 2011. One illness away: Why people become poor and how they escape poverty. New York: Oxford University Press.

Kristjanson, P., Bah, T., Kuriakose, A., Shakirova, M., Segura, G., Siegmann, K. and Granat, M., 2019. Taking Action on Gender Gaps in Forest Landscapes. Program on Forests (PROFOR). Washington, D.C.: World Bank.

Kumar, A., Bargali, H.S., David, A. and Edgaonkar, A., 2017. Patterns of crop raiding by wild ungulates and elephants in Ramnagar Forest Division, Uttarakhand. Human-Wildlife Interactions 11(1): 41-49.

Kuntashula, E. and Mungatana, E., 2013. Estimating the causal effect of improved fallows on farmer welfare using robust identification strategies in Chongwe, Zambia. Agroforestry Systems 87(6): 1229-1246.

Kuyah, S., Öborn, I., Jonsson, M., Dahlin, A.S., Barrios,
E., Muthuri, C., Malmer, A., et al., 2016. Trees
in agricultural landscapes enhance provision
of ecosystem services in Sub-Saharan Africa.
International Journal of Biodiversity Science, Ecosystem
Services & Management 12(4): 255-273.

Lamichhane, B.R., Persoon, G.A., Leirs, H., Poudel, S., Subedi, N., Pokheral, C.P., Bhattarai, S., Thapaliya, B.P. and de Iongh, H.H., 2018. Spatio-temporal patterns of attacks on human and economic losses from wildlife in Chitwan National Park, Nepal. *PLOS ONE* 13(4): e0195373.

Larson, A.M., Barry, D., Dahal, G.R. and Colfer, C.J.P., 2010. Forests for people: *Community rights and forest tenure reform*. Abingdon and New York: Earthscan.

Larson, A. M. and Ribot, J. C., 2007. The poverty of forestry policy: double standards on an uneven playing field. Sustainability Science 2(2): 189-204.

Laube, W., 2015. Global shea nut commodity chains and poverty eradication in Northern Ghana: Myth or reality? UDS International Journal of Development 2(1): 128-147.

Laurance, W.F., 1999. Reflections on the tropical deforestation crisis. *Biological conservation* 91(2-3): 109-117.

Lawlor, K., Sills, E., Atmadja, S., Lin, L. and Songwathana, K., 2019. SDG 1: No Poverty – Impacts of social protection, tenure security and building resilience on forests. In: Sustainable Development Goals: Their Impacts on Forests and People, edited by C. J. Pierce Colfer, G. Winkel, G. Galloway, P. Pacheco, P. Katila, and W. de Jong. Cambridge: Cambridge University Press, pp. 17-47

Leakey, R.R., Tchoundjeu, Z., Schreckenberg, K., Shackleton, S.E. and Shackleton, C.M., 2005. Agroforestry tree products (AFTPs): targeting poverty reduction and enhanced livelihoods. International Journal of Agricultural Sustainability 3(1): 1-23.

Lebedys, A. and Yanshu, L., 2014. Contribution of the forestry sector to national economies, 1990-2011. Forest Finance Working Paper (FAO) eng no. 09. Rome: FAO.

- Leßmeister, A., Heubach, K., Lykke, A.M., Thiombiano, A., Wittig, R. and Hahn, K., 2018. The contribution of non-timber forest products (NTFPs) to rural household revenues in two villages in south-eastern Burkina Faso. Agroforestry systems 92(1): 139-155.
- Li, S., Liang, W., Fu, B., Lü, Y., Fu, S., Wang, S. and Su, H., 2016. Vegetation changes in recent large-scale ecological restoration projects and subsequent impact on water resources in China's Loess Plateau. *Science of Total Environment* 569-570: 1032-1039.
- Li, W., Shi, Z., Yu, M., Ren, W., Smith, C., Epstein, J.H., Wang, H., Crameri, G., Hu, Z., Zhang, H. and Zhang, J., 2005. Bats are natural reservoirs of SARS-like coronaviruses. *Science* 310(5748): 676-679.
- Licata, J.A., Gyenge, J.E., Fernández, M.E., Schlichter, T.M. and Bond, B.J., 2008. Increased water use by ponderosa pine plantations in northwestern Patagonia, Argentina compared with native forest vegetation. Forest Ecology and Management 255: 753-764.
- Liljeblad, J. and Verschuuren, B., 2018. Indigenous perspectives on sacred natural sites: culture, governance and conservation. Abingdon and New York: Routledge.
- Lin, B. B., Perfecto, I. and Vandermeer, J., 2008. Synergies between Agricultural Intensification and Climate Change Could Create Surprising Vulnerabilities for Crops. BioScience 58(9): 847-854.
- Linden, V.M., Grass, I., Joubert, E., Tscharntke, T., Weier, S.M. and Taylor, P.J., 2019. Ecosystem services and disservices by birds, bats and monkeys change with macadamia landscape heterogeneity. *Journal of Applied Ecology* 56(8): 2069-2078.
- Lindsey, P., Balme, G., Becker, M., Begg, C., Bento, C., Bocchino, C., Dickman, et al., 2015. Illegal hunting and the bush-meat trade in savanna Africa: Drivers, impacts and solutions to address the problem. Rome: FAO.
- Lindsey, P.A., Nyirenda, V.R., Barnes, J.I., Becker, M.S., McRobb, R., Tambling, C.J., Taylor, et al., 2014.
 Underperformance of African protected area networks and the case for new conservation models: Insights from Zambia. *PLoS one* 9(5): e94109.
- Linkie, M., Dinata, Y., Nofrianto, A. and Leader-Williams, N., 2007. Patterns and perceptions of wildlife crop raiding in and around Kerinci Seblat National Park, Sumatra. Animal Conservation 10: 127-135.
- Liu, Y.Y., Van Dijk, A.I., De Jeu, R.A., Canadell, J.G., McCabe, M.F., Evans, J.P. and Wang, G., 2015. Recent reversal in loss of global terrestrial biomass. *Nature Climate Change* 5(5): 470-474.
- Liu, Z. 2006. Tibetan rural cultural traditions and its reconstruction. *Mingzu Press* 978: 105-07960.
- Lo, M., Narulita, S. and Ickowitz, A., 2019. The relationship between forests and freshwater fish consumption in rural Nigeria. *PloS One* 14(6): e0218038.

- Logo, P. B., 2010. Governance of decentralized forest revenue in Central Africa: For better or for worse?
 In: Governing Africa's forests in a globalized world, edited by L. A. German, A. Karsenty, and A. M. Tiani. London: Earthscan, pp. 174-186
- López-Feldman, A. and Wilen, J. E., 2008. Poverty and spatial dimensions of non-timber forest extraction. Environment and Development Economics 13(5): 621-642.
- Lott, J. E., Ong, C. K. and Black, C. R., 2009. Understorey microclimate and crop performance in a Grevillea robusta-based agroforestry system in semi-arid Kenya. Agricultural and Forest Meteorology 149(6): 1140-1151.
- Luckert, M. K. and Campbell, B. M., 2012. Uncovering the hidden harvest: valuation methods for woodland and forest resources. Abingdon and New York: Routledge.
- Luttrell, C., Loft, L., Gebara, M.F., Kweka, D., Brockhaus, M., Angelsen, A. and Sunderlin, W.D., 2013. Who should benefit from REDD+? Rationales and realities. *Ecology and Society* 18(4): 52-70.
- Lyytimäki, J., 2015. Ecosystem disservices: Embrace the catchword. Ecosystem Services 12: 136.
- Maas, B., Karp, D.S., Bumrungsri, S., Darras, K., Gonthier, D., Huang, J.C.C., Lindell, C.A., *et al.*, 2016. Bird and bat predation services in tropical forests and agroforestry landscapes. *Biological Reviews* 91(4): 1081-1101.
- MacDonald, A. J. and Mordecai, E. A., 2019. Amazon deforestation drives malaria transmission, and malaria burden reduces forest clearing. PNAS 116(44): 22212-22218.
- Mackenzie, C. A., Sengupta, R. R. and Kaoser R., 2015. Chasing baboons or attending class: Protected areas and childhood education in Uganda. *Environmental Conservation* 42(4): 373-383.
- Mackenzie, J.S. and Smith, D.W., 2020. COVID-19: a novel zoonotic disease caused by a coronavirus from China: what we know and what we don't. *Microbiology Australia* 41(1): 45-50.
- Macqueen, D., Bolin, A., Greijmans, M., Grouwels, S. and Humphries, S., 2020. Innovations towards prosperity emerging in locally controlled forest business models and prospects for scaling up. *World Development* 125: 104382.
- Maggio, G. F., 1997. Recognizing the vital role of local communities in international legal instruments for conserving biodiversity. UCLA Journal of Environmental Laws and Policy 16: 179.
- Mai, Y. H., Mwangi, E. and Wan, M., 2011. Gender analysis in forestry research: Looking back and thinking ahead. International Forestry Review 13(2): 245-258.
- le Maitre, D., Görgens, A., Howard, G. and Walker, N., 2019. Impacts of alien plant invasions on water resources and yields from the Western Cape Water Supply System (WCWSS). Water SA 45(4): 568-579.

Matseketsa, G., Muboko, N., Gandiwa, E., Kombora, D.M. and Chibememe, G., 2019. An assessment of humanwildlife conflicts in local communities bordering the western part of Save Valley Conservancy, Zimbabwe. *Global Ecology and Conservation* 20: e00737.

Mazumdar, S. and Mazumdar, S., 2012. Immigrant home gardens: Places of religion, culture, ecology, and family. Landscape and Urban Planning 105(3): 258-265.

Mbiba, M., Collinson, M., Hunter, L. and Twine, W., 2019. Social capital is subordinate to natural capital in buffering rural livelihoods from negative shocks: Insights from rural South Africa. *Journal of Rural* Studies 65: 12-21.

Mbow, C., Van Noordwijk, M., Luedeling, E., Neufeldt, H., Minang, P.A. and Kowero, G., 2014. Agroforestry solutions to address food security and climate change challenges in Africa. *Current Opinion in Environmental Sustainability* 6: 61-67.

McDermott, C. L., 2017. Whose forests, whose gain? Nature Climate Change 7(6): 386-387.

McGarry, D., Shackleton, C.M., Fourie, S., Gambiza, J., Shackleton, S.E. and Fabricius, C.F., 2005. A rapid assessment of the effects of invasive species on human livelihoods, especially of the rural poor. Grahamstown: Department of Environmental Science, Rhodes University.

McIntyre, P.B., Liermann, C.A.R. and Revenga, C., 2016. Linking freshwater fishery management to global food security and biodiversity conservation. Proceedings of the National Academy of Science 113: 12880-12885.

McNeely, J. A. and Schroth, G., 2006. Agroforestry and biodiversity conservation-traditional practices, present dynamics, and lessons for the future. Biodiversity & Conservation 15(2): 549-554.

McSweeney, K., 2004. Forest product sale as natural insurance: the effects of household characteristics and the nature of shock in eastern Honduras. Society and Natural Resources 1(17): 39-56.

Megaze, A., Balakrishnan, M. and Belay, G., 2017. Human-wildlife conflict and attitude of local people towards conservation of wildlife in Chebera Churchura National Park, Ethiopia. *African Zoology* 52(1): 1-8.

Meijaard, E., Abram, N.K., Wells, J.A., Pellier, A.S., Ancrenaz, M., Gaveau, D.L., Runting, R.K. and Mengersen, K., 2013. People's perceptions about the importance of forests on Borneo. *PloS One* 8(9): e73008.

Melnykovych, M. and Soloviy, I., 2014. Contribution of forestry to wellbeing of mountain forest dependent communities' in the Ukrainian Carpathians. *Наукові* праці Лісівничої академії наук України (12): 233-241.

Melrose, W., 2011. Deforestation in Papua New Guinea: Potential impact on health care. Rural and Remote Environmental Health 1: 79-88. Mgumia, F. H. and Oba, G., 2003. Potential role of sacred groves in biodiversity conservation in Tanzania. *Environmental Conservation* 30(3): 259-265.

Michalski, F., Boulhosa, R.L.P., Faria, A. and Peres, C.A., 2006. Human-wildlife conflicts in a fragmented Amazonian forest landscape: Determinants of large felid depredation on livestock. *Animal Conservation* 9: 179-188.

Michler, J. D. and Josephson, A. L., 2017. To Specialize or Diversify: Agricultural Diversity and Poverty Dynamics in Ethiopia. World Development 89: 214-226.

Miller, D. C. and Hajjar, R., 2020. Forests as pathways to prosperity: Empirical insights and conceptual advances. World Development 125: 104647.

Miller, D.C., Muñoz-Mora, J.C. and Christiaensen, L., 2017. Prevalence, economic contribution, and determinants of trees on farms across Sub-Saharan Africa. Forest Policy and Economics 84: 47-61.

Miller, D.C., Muñoz-Mora, J.C., Zezza, A. and Durazo. J., 2019. Trees on Farms: Measuring their Contribution to Household Welfare. A *Guidebook for Designing Household Surveys*. Washington D.C.: World Bank.

Miller, D.C., Muñoz-Mora, J.C., Rasmussen, L.V. and Zezza, A., 2020. Do Trees on Farms Improve Household Well-Being? Evidence From National Panel Data in Uganda. Frontiers in Forests and Global Change. 3:101.

Mishra, C., 1997. Livestock depredation by large carnivores in the Indian trans-Himalaya: conflict perceptions and conservation prospects. Environmental Conservation 24(4): 338-343.

Mohammed, F., Boateng, S. and Al-hassan, S., 2013. Effects of adoption of improved shea butter processing technology on women's livelihoods and their microenterprise growth. *American Journal of Humanities and Social Sciences* 1(4): 244-250.

 Mohammed, N., Goudar, K.S., Getachew, G. and Ibrahim, H., 2017. Human-wildlife conflict: intensity of domestic damage caused by wild animals around Yegof National Forest Priority Area, South Wollo, Amhara region, Ethiopia. International Journal of Conservation Science 8(3): 519-528.

Munyi, P. and Mutta, D., 2007. Protection of community rights over traditional knowledge: Implications for customary laws and practices. *ICIPE Science Press* 92: 9064.

Mwakatobe, A., Nyahongo, J., Ntalwila, J. and Roskaft, E., 2014. The impact of crop raiding by wild animals in communities surrounding the Serengeti National Park, Tanzania. International Journal of Biodiversity and Conservation 6(9): 637-646.

Mwampamba, T.H., Ghilardi, A., Sander, K. and Chaix, K.J., 2013. Dispelling common misconceptions to improve attitudes and policy outlook on charcoal in developing countries. *Energy for Sustainable Development* 17(2): 75-85. Mwangi, E., Meinzen-Dick, R. and Sun, Y., 2011. Gender and sustainable forest management in East Africa and Latin America. *Ecology and Society* 16(1): 17-32.

Nakamura, E. M. and Hanazaki, N., 2017. Protected area establishment and its implications for local food security. *Human Ecology Review* 23(1): 101-122.

Nasi, R., Taber, A. and van Vliet, N., 2011. Empty forests, empty stomachs? Bushmeat and livelihoods in the Congo and Amazon Basins. *International Forestry Review* 13(3): 355-368.

National Commission for Museums and Monuments (NCMM), 2005. Nomination to the World Heritage List of Osun-Osogbo Sacred Grove, Osogbo, Osun State, Nigeria. Abuja: National Commission for Museums and Monuments.

Naughton-Treves, L., 1998. Predicting patterns of crop damage by wildlife around Kibale National Park, Uganda. *Conservation Biology* 12(1): 156-168.

NBAP, 2001. Nepal Biodiversity Action Plan. Kathmandu: Ministry of Forest and Soil Conservation.

Ndayambaje, J.D. and Mohren, G.M.J., 2011. Fuelwood demand and supply in Rwanda and the role of agroforestry. *Agroforestry systems* 83(3): 303-320.

Nesper, M., Kueffer, C., Krishnan, S., Kushalappa, C.G. and Ghazoul, J., 2017. Shade tree diversity enhances coffee production and quality in agroforestry systems in the Western Ghats. *Agriculture, Ecosystems* & Environment 247: 172-181.

Neupane, D., Johnson, R.L. and Risch, T.S., 2017. How do land-use practices affect human-elephant conflict in Nepal? Wildlife Biology 4: wlb00313.

Ngoufo, R., Yongyeh, N.K., Obioha, E.E., Bobo, K.S., Jimoh, S.O. and Waltert, M., 2014. Social norms and cultural services-community belief system and use of wildlife products in the Northern periphery of the Korup National Park, South-West Cameroon. Change and Adaptation in Socio-Ecological Systems 1(1): 26-34.

Nielsen, M.R., Meilby, H., Smith-Hall, C., Pouliot, M. and Treue, T., 2018. The importance of wild meat in the global south. *Ecological Economics* 146: 696-705.

Nielsen, M.R., Pouliot, M., Meilby, H., Smith-Hall, C. and Angelsen, A., 2017. Global patterns and determinants of the economic importance of bushmeat. *Biological Conservation* 215: 277-287.

Njenga, M., Karanja, N., Munster, C., Iiyama, M., Neufeldt, H., Kithinji, J. and Jamnadass, R., 2013. Charcoal production and strategies to enhance its sustainability in Kenya. *Development in Practice* 23(3): 359-371.

Njenga, M., Mendum, R., Gitau, J., Iiyama, M., Jamnadass, R. and Watson, C., 2017. Trees on farms could satisfy household firewood needs. *Tree Farmers Magazine for Africa* 33: 20-23.

Noack, F., Riekhof, M.C. and Di Falco, S., 2019. Droughts, biodiversity, and rural incomes in the tropics. Journal of the Association of Environmental and Resource Economists 6(4): 823-852. Ntiamoa-Baidu, Y., 1995. Indigenous vs. introduced biodiversity conservation strategies: The case of protected area systems in Ghana. Washington D.C.: Biodiversity Support Program.

Obeng-Odoom, F., 2020. COVID-19, inequality, and social stratification in Africa. African Review of Economics and Finance 12(1): 3-37.

Oishi, T. and Hagiwara, M., 2015. A preliminary report of the distribution of freshwater fish of the Congo River based on the observation of local markets in Brazzaville, Republic of Congo. *African Study Monographs* 51: 93-105.

Okigbo, B., 1980. Plants and food in Igbo culture. Ahiajoku Lecture series. Owerri: Igbo Language Culture Centre.

Omane, B. M., 2014. The effect of the adoption of Improved Sheanut Processing Technology on women's income in the Savelugu-Nanton Municipality. PhD Thesis. Legon: University of Ghana, Department of Agricultural Economics and Agribusiness, College of Agriculture and Consumer Science.

Onyekwelu, J. C. and Olusola, J. A., 2014. Role of sacred grove in in-situ biodiversity conservation in rainforest zone of south-west Nigeria. *Journal of Tropical Forest Science* 26(1): 5-15.

Organisation Internationale du Travail, 2016. Etat des lieux du travail des enfants dans la filière vanille dans la région de la SAVA. Antananarivo: OIT (available at: http://www.ilo.org/global/docs/WCMS_520913/langfr/index.htm, [accessed on: 1 September 2020]).

Orlowska, I. and Klepeis, P., 2018. Ethiopian church forests: a socio-religious conservation model under change. Journal of Eastern African Studies 12: 674-695.

Ormsby, A., 2012. Cultural and conservation values of sacred forests in Ghana. Sacred species and sites: advances in biocultural conservation. Cambridge: Cambridge University Press.

Ormsby, A. A. and Bhagwat, S. A., 2010. Sacred forests of India: a strong tradition of community-based natural resource management. *Environmental Conservation* 37(3): 320-326.

Oteng-Yeboah, A., Mutta, D., Byarugaba, D. and Mala, W. A., 2011. The African view of TK and SFM in traditional forest related knowledge and forest science interface. In: Traditional Forest-Related Knowledge: Sustaining Communities, Ecosystems and Biocultural Diversity, edited by J. A. Parrotta and R. L. Trosper. New York: Springer, pp. 37-78.

Oviedo, G. and Jeanrenaud, S., 2007. Protecting sacred natural sites of indigenous and traditional peoples. In: Protected Areas and Spirituality, Proceedings of the First Workshop of the Delos Initiative, edited by J. M. Mallarach and T. Papayannis. Gland: IUCN and Publicacions de l'Abadia de Montserrat, pp. 77-100.

Oyelowo, O.J., Aduradola, A.M., Onadeko, S.A. and Agboola, D.A., 2014. Socio-economic contribution of selected sacred groves in South-West Nigeria. International Journal of Novel Research in Civil Structural and Earth Sciences 1(1): 36-45.

- Pagiola, S., Arcenas, A. and Platais, G., 2005. Can payments for environmental services help reduce poverty? An exploration of the issues and the evidence to date from Latin America. World Development 33(2): 237-253.
- Palo, M., Uusivuori, J. and Lehto, E., 1999. Forest-based Development in Finland: a Unique Success? In: World forests, society and environment, edited by M.
 Palo and J. Uusivuoiri. Dordrecht: Springer, pp. 300-318.
- Parr, C.L., Gray, E.F. and Bond, W.J., 2012. Cascading biodiversity and functional consequences of a global change-induced biome switch. *Diversity and Distributions* 18(5): 493-503.
- Parr, C.L., Lehmann, C.E., Bond, W.J., Hoffmann, W.A. and Andersen, A.N., 2014. Tropical grassy biomes: misunderstood, neglected, and under threat. *Trends* in Ecology & Evolution 29(4): 205-213.
- Pattanayak, S. K. and Sills, E. O., 2001. Do tropical forests provide natural insurance? The microeconomics of non-timber forest product collection in the Brazilian Amazon. *Land Economics* 77(4): 595-612.
- Pattiselanno, F. and Lubis, M. I., 2014. Hunting at the Abun regional marine protected areas: a link between wildmeat and food security. HAYATI Journal of Biosciences 21(4): 180-186.
- Patz, J.A., Confalonieri, U.E., Amerasinghe, F.P., Chua,
 K.B., Daszak, P., Hyatt, A.D., Molyneux, D., et al., 2005.
 Human health: ecosystem regulation of infectious diseases. In: Ecosystems and Human Well-Being:
 Current State and Trends: Findings of the Condition and Trends Working Group of the Millennium Ecosystem
 Assessment, edited by R. Hassan, R. Scholes and N.
 Ash. Washington, Covelo and London: Island Press, pp. 391-415.
- Paumgarten, F., 2005. The role of non-timber forest products as safety-nets: A review of evidence with a focus on South Africa. *GeoJournal* 64(3): 189-197.
- Paumgarten, F. and Shackleton, C. M., 2011. The role of non-timber forest products in household coping strategies in South Africa: The influence of household wealth and gender. *Population and Environment* 33(1): 108.
- Peluso, N. L., 1991. The history of state forest management in colonial Java. Forest & Conservation History 35(2): 65-75.
- Pérez, C., Jones, E., Kristjanson, P.M., Cramer, L., Thornton, P.K., Förch, W. and Barahona, C., 2014. How resilient are farming households, communities, men and women to a changing climate in Africa? Copenhagen: CGIAR Research Program on Climate Change, Agriculture and Food Security (CCAFS).
- Pérez, M. R., 2005. Poverty alleviation and forest conservation: The role of non-timber forest products. In: Non-Timber Forest Products between Poverty Alleviation and Market Forces, edited by J.-L. Pfund and P. Robinson. Zurich: Intercooperation, pp. 8-13.

- Pham, T.T., Moeliono, M., Brockhaus, M., Le, D.N., Wong, G.Y. and Le, T.M., 2014. Local preferences and strategies for effective, efficient, and equitable distribution of PES revenues in Vietnam: Lessons for REDD+. Human Ecology 42(6): 885-899.
- Phelps, J., Webb, E.L. and Agrawal, A., 2010. Does REDD+ threaten to recentralize forest governance? *Science* 328(5976): 312-313.
- Place, F., Adato, M., Hebinck, P. and Omosa, M.,
 2005. Impact of agroforestry-based soil fertility
 replenishment practices on the poor in western Kenya (Vol. 142). Washington D.C.: International Food Policy
 Research Institute
- Place, F., Garrity, D. and Agostini, P., 2016. Tree-based systems: multiple pathways to boosting resilience.
 In: Confronting Drought in Africa's Drylands: Opportunities for Enhancing Resilience, edited by R. Cervigni and M. Morris. Washington D.C.: World Bank.
- Potts, S.G., Imperatriz-Fonseca, V., Ngo, H.T., Aizen, M.A., Biesmeijer, J.C., Breeze, T.D., Dicks, L.V., Garibaldi, L.A., Hill, R., Settele, J. and Vanbergen, A.J., 2016. Safeguarding pollinators and their values to human well-being. *Nature* 540(7632): 220-229.
- Poudyal, M., Jones, J.P., Rakotonarivo, O.S., Hockley, N., Gibbons, J.M., Mandimbiniaina, R., Rasoamanana, A., et al., 2018. Who bears the cost of forest conservation? *PeerJ* 6: e5106.
- Pouliot, M. and Treue, T., 2013. Rural people's reliance on forests and the non-forest environment in West Africa: Evidence from Ghana and Burkina Faso. World Development 43: 180-193.
- Powell, B., Hall, J. and Johns, T., 2011. Forest cover, use and dietary intake in the East Usambara Mountains, Tanzania. International Forestry Review (13) 3: 305-317.
- Powell, B., Thilsted, S.H., Ickowitz, A., Termote, C., Sunderland, T. and Herforth, A., 2015a. Strategies to improve diets with wild and cultivated biodiversity from across the landscape. *Food Security* 7(3): 535-554.
- Powell, B., Thilsted, S.H., Ickowitz, A., Termote, C., Sunderland, T. and Herforth, A., 2015b. Improving diets with wild and cultivated biodiversity from across the landscape. *Food Security* 7(3): 535-554.
- Power, A.G., 2010. Ecosystem services and agriculture: tradeoffs and synergies. Philosophical Transactions of the Royal Society B: *Biological Sciences* 365(1554): 2959-2971.
- Pramova, E., Locatelli, B., Djoudi, H. and Somorin, O.A., 2012. Forests and trees for social adaptation to climate variability and change. Wiley Interdisciplinary Reviews: Climate Change 3(6): 581-596.
- Pumariño, L., Sileshi, G.W., Gripenberg, S., Kaartinen, R., Barrios, E., Muchane, M.N., Midega, C. and Jonsson, M., 2015. Effects of agroforestry on pest, disease and weed control: a meta-analysis. *Basic and Applied Ecology* 16(7): 573-582.

Puyravaud, J.P., Gubbi, S., Poornesha, H.C. and Davidar, P., 2019. Deforestation increases frequency of incidents with elephants (Elephas maximus). Tropical Conservation Science 12: 1-11.

Quandt, A., Neufeldt, H. and McCabe, J.T., 2019. Building livelihood resilience: what role does agroforestry play? Climate and Development 11(6): 485-500.

RAFI, 1994. The Rural Advancement Foundation International. Conserving indigenous knowledge: Integrating two systems of innovation. An independent study commissioned by the United National Development Programme. New York: UNDP.

Rasmussen, L.V., Watkins, C. and Agrawal, A., 2017. Forest contributions to livelihoods in changing agriculture-forest landscapes. Forest policy and economics 84: 1-8.

Rasolofoson, R.A., Hanauer, M.M., Pappinen, A., Fisher, B. and Ricketts, T.H., 2018. Impacts of forests on children's diet in rural areas across 27 developing countries. Science Advances 4(8): eaat2853.

Ratajczak, Z., Nippert, J.B. and Collins, S.L., 2012. Woody encroachment decreases diversity across North American grasslands and savannas. *Ecology* 93(4): 697-703.

Ray, R., Chandran, M.S. and Ramachandra, T.V., 2014. Socio-cultural protection of endemic trees in humanised landscape. *Biodiversity and Conservation* 23(8): 1977-1994.

Reed, J., van Vianen, J., Foli, S., Clendenning, J., Yang, K., MacDonald, M., Petrokofsky, G., Padoch, C. and Sunderland, T., 2017. Trees for life: The ecosystem service contribution of trees to food production and livelihoods in the tropics. Forest Policy and Economics 84: 62-71.

Rentsch, D. and Damon, A., 2013. Prices, poaching, and protein alternatives: An analysis of bushmeat consumption around Serengeti National Park, Tanzania. Ecological Economics 91: 1-9.

Reuter, K.E., Randell, H., Wills, A.R. and Sewall, B.J., 2016. The consumption of wild meat in Madagascar: Drivers, popularity and food security. *Environmental Conservation* 43: 273-283.

Ribot, J. C., 1999. Decentralisation, participation and accountability in Sahelian forestry: legal instruments of political-administrative control. *Africa* 69(1): 23-65.

Ribot, J. C. and Peluso, N. L., 2003. A theory of access. Rural Sociology 68(2): 153-181.

Rice, R.A., 2008. Agricultural intensification within agroforestry: the case of coffee and wood products. Agriculture, ecosystems & environment 128(4): 212-218.

Robinson, M., 2001. The Micro-finance Revolution Sustainable Finance for the Poor. Lessons from Indonesia. The Emerging Industry. Washington, D.C.: The World Bank. Ross, M. L., 2015. What have we learned about the resource curse? *Annual Review of Political Science* 18: 239-259.

Rousseau, K., Gautier, D. and Wardell, D.A., 2015. Coping with the upheavals of globalization in the shea value chain: The maintenance and relevance of upstream shea nut supply chain organization in western Burkina Faso. World Development 66: 413-427.

Rowland, D., Ickowitz, A.M.Y., Powell, B., Nasi, R. and Sunderland, T., 2017. Forest foods and healthy diets: Quantifying the contributions. *Environmental Conservation* 44: 102-114.

Rudel, T. K., 2007. Changing agents of deforestation: from state-initiated to enterprise driven processes, 1970-2000. Land use policy 24(1): 35-41.

Ruel, M. T. and Alderman, H., 2013. Nutrition-sensitive interventions and programmes: how can they help to accelerate progress in improving maternal and child nutrition? *The Lancet*: 382: 536-51.

Ruel, M.T., Garrett, J., Yosef, S. and Olivier, M., 2017. Urbanization, Food Security and Nutrition. In: Nutrition and Health in a Developing World, edited by S. de Pee, D. Taren, and M. W. Bloem. New York: Springer International Publishing, pp. 705-735.

Russell, D. and Franzel, S., 2004. Trees of prosperity: Agroforestry, markets and the African smallholder. *Agroforestry Systems* 61(1): 345-355.

Rutte, C., 2011. The scared commons: conflicts and solutions of resource management in sacred natural sites. *Biological Conservation* 144: 2387-2394.

de Sales, F., Santiago, T., Biggs, T., Mullan, K. and Sills, E.O., 2020. Impacts of protected area deforestation on dry-season regional climate in the Brazilian Amazon. *Geophysical Research: Atmospheres* (in press).

Samii, C., Lisiecki, M., Kulkarni, P., Paler, L. and Chavis, L., 2014. Effects of decentralized forest management (DFM) on deforestation and poverty in low-and middle-income countries: A systematic review. *Campbell Systematic Reviews* 10(1): 1-88.

Sarkodie, P. A., Agyapong, D., Mumuni, S., and Amponsah, F. Y., 2016. Assessing the impact of indigenous shea butter processing activities in Northern Ghana. International Research Journal of Environment Sciences 5(3): 18-26.

Sautman, B. and Yan, H., 2008. The forest for the trees: Trade, investment, and the China-in-Africa discourse. *Pacific Affairs* 81(1): 9-29.

Scherr, S. J. and McNeely, J. A., 2008. Biodiversity conservation and agricultural sustainability: towards a new paradigm of 'ecoagriculture' landscapes. Philosophical Transactions of the Royal Society B: Biological Sciences 363(1491): 477-494.

Scherr, S.J., White, A. and Kaimowitz, D., 2003. Making markets work for forest communities. *The International Forestry Review* 5(1): 67-73. Schroeder, H., 2010. Agency in international climate negotiations: the case of indigenous peoples and avoided deforestation. International Environmental Agreements: Politics, Law and Economics 10(4): 317-332.

Scoones, I., Melnyk, M. and Pretty, J.N., 1992. The hidden harvest: wild foods and agricultural systems. A literature review and annotated bibliography. London: IIED.

Sen, A., 1997. Human capital and human capability. World Development 25(12): 1959-1961.

Shackleton, C. and Shackleton, S., 2004. The importance of non-timber forest products in rural livelihood security and as safety nets: A review of evidence from South Africa. South African Journal of Science 100: 658-664.

Shackleton, S., Campbell, B., Lotz-Sisitka, H. and Shackleton, C., 2008. Links between the local trade in natural products, livelihoods and poverty alleviation in a semi-arid region of South Africa. World Development 36(3): 505-526.

Shepherd, G., Warner, K. and Hogarth, N., 2020. Forests and poverty: how has our understanding of the relationship been changed by experience? International Forestry Review 22(1): 29-43.

Sheridan, M. J., 2009. The Environmental and Social History of African Sacred Groves: A Tanzanian Case Study. African Studies Review 52: 01.

Sida, T.S., Baudron, F., Kim, H. and Giller, K.E., 2018. Climate-smart agroforestry: Faidherbia albida trees buffer wheat against climatic extremes in the Central Rift Valley of Ethiopia. Agricultural and forest meteorology 248: 339-347.

Sierra-Correa, P.C. and Kintz, J.R.C., 2015. Ecosystembased adaptation for improving coastal planning for sea-level rise: A systematic review for mangrove coasts. *Marine Policy* 51: 385-393.

Sikor, T. and Baggio, J. A., 2014. Can smallholders engage in tree plantations? An entitlements analysis from Vietnam. World Development 64: S101-S112.

Sikor, T., Stahl, J., Enters, T., Ribot, J.C., Singh, N., Sunderlin, W.D. and Wollenberg, L., 2010. REDDplus, forest people's rights and nested climate governance. Global Environmental Change 20(3): 423-425.

Sileshi, G., Akinnifesi, F.K., Ajayi, O.C., Chakeredza, S., Kaonga, M. and Matakala, P.W., 2007. Contributions of agroforestry to ecosystem services in the Miombo eco-region of eastern and southern Africa. African journal of environmental science and technology 1(4): 68-80.

Sills, E.O. and Saha, S., 2010. Subsidies for rubber: conserving rainforests while sustaining livelihoods in the Amazon? *Journal of Sustainable Forestry* 29(2-4): 152-173.

Sjaastad, E., Angelsen, A., Vedeld, P., and Bojö J., 2005. What is environmental income? *Ecological Economics* 55(1): 37-46. Skowno, A.L., Thompson, M.W., Hiestermann, J., Ripley, B., West, A.G. and Bond, W.J., 2017. Woodland expansion in South African grassy biomes based on satellite observations (1990-2013): general patterns and potential drivers. *Global Change Biology* 23(6): 2358-2369.

Skutsch, M. and Turnhout, E., 2020. REDD+: If communities are the solution, what is the problem? World Development 130: 104942.

Soemarwoto, O., 1987. Homegardens: a traditional agroforestry system with a promising future. In: *Agroforestry: A decade of development*, edited by H.A. Steppler and P.K.R. Nair. Nairobi: ICRAF, pp. 57-170.

Somboonsuke, B., Wetayaprasit, P., Chernchom, P. and Pacheerat, K., 2011. Diversification of smallholding rubber agroforestry system (SRAS) Thailand. *Kasetsart Journal* (Soc. Sci) 32: 327-339.

Sorrenti, S., 2017. Non-wood forest products in international statistical systems. Non-Wood Forest Products (FAO) no. 22. Rome: FAO. (available at: http://www.fao.org/3/a-i6731e.pdf [accessed on 23 September 2020]).

Stafford, W., Birch, C., Etter, H., Blanchard, R., Mudavanhu, S., Angelstam, P., Blignaut, J., Ferreira, L. and Marais, C., 2017. The economics of landscape restoration: Benefits of controlling bush encroachment and invasive plant species in South Africa and Namibia. Ecosystem Services 27: 193-202.

Steffan-Dewenter, I., Kessler, M., Barkmann, J., Bos, M.M., Buchori, D., Erasmi, S., Faust, H., et al., 2007. Tradeoffs between income, biodiversity, and ecosystem functioning during tropical rainforest conversion and agroforestry intensification. PNAS 104(2): 4973-4978.

Stevens, N., Lehmann, C.E., Murphy, B.P. and Durigan, G., 2017. Savanna woody encroachment is widespread across three continents. *Global Change Biology* 23(1): 235-244.

Sun, G., Zhou, G., Zhang, Z., Wei, X., McNulty, S.G. and Vose, J.M., 2006. Potential water yield reduction due to forestation across China. *Journal of Hydrology* 328: 548-558.

Sunderland, T., Achdiawan, R., Angelsen, A., Babigumira, R., Ickowitz, A., Paumgarten, F., Reyes-García, V. and Shively, G., 2014. Challenging perceptions about men, women, and forest product use: a global comparative study. World Development 64: S56-S66.

Sunderland, T. C. H., 2011. Food security: why is biodiversity important? *International Forestry Review* 13(3): 265-274.

Sunderland, T.C. and Vasquez, W., 2020. Forest Conservation, Rights, and Diets: Untangling the Issues. Frontiers in Forests and Global Change 3: 29.

Sunderland, T., Powell, B., Ickowitz, A., Foli, S., Pinedo-Vasquez, M., Nasi, R. and Padoch, C., 2013. Food security and nutrition: The role of forests. Bogor: CIFOR. Sunderlin, W.D., Dewi, S., Puntodewo, A., Müller, D., Angelsen, A. and Epprecht, M., 2008. Why forests are important for global poverty alleviation: a spatial explanation. Ecology and society 13(2): 24.

Sutherst, R. W., 2004. Global change and human vulnerability to vector-borne diseases. *Clinical Microbiology Reviews* 17(1): 136-173.

Swamy, V. and Pinedo-Vasquez, M., 2014. Bushmeat harvest in tropical forests: Knowledge base, gaps and research priorities. Bogor: CIFOR.

Taylor, B. and Kaplan, J., 2005. The Encyclopedia of Religion and Nature. *Continuum Books* 978: 3-4.

Temsah, G., Johnson, K., Evans, T. and Adams, D.K., 2018. Benefits of biodiverse marine resources to child nutrition in differing developmental contexts in Hispaniola. PLoS One 13: 1-24.

Ten Kate, K. and Laird, S. A., 2020. The commercial use of biodiversity: access to genetic resources and benefitsharing. London: Routledge.

Thorlakson, T. and Neufeldt, H., 2012. Reducing subsistence farmers' vulnerability to climate change: evaluating the potential contributions of agroforestry in western Kenya. *Agriculture & Food Security* 1(1): 15.

Thrupp, L. A., 2000. Linking agricultural biodiversity and food security: the valuable role of agrobiodiversity for sustainable agriculture. *International affairs* 76(2): 265-281.

Thu, T.P., Moeliono, M., Wong, G.Y. and Brockhaus,M., 2018. The politics of swidden: A case study fromNghe An and Son La in Vietnam. Land Use Policy.

Toledo, V. M., 2002. Ethnoecology: a conceptual framework for the study of indigenous knowledge of nature. In: Ethnobiology and biocultural diversity, edited by F. S. Wyndham and R. K. Zarger. International Society of Ethnobiology. Athens, GA: University of George Press, pp. 511-522.

Tweheyo, M., Hill, C.M. and Obua, J., 2005. Patterns of crop raiding by primates around the Budongo Forest Reserve, Uganda. Wildlife Biology 11(3): 237-247.

Umazi, A., Iwa, S. and Etim, D., 2013. Cultural and socioeconomic perceptive of the conservation of Asanting Ibiono Sacred Forests in Akwa Ibom State, Nigeria. International Journal of Biodiversity and Conservation 5(11): 696-703.

UNDP, 2019. Human Development Report 2019. Beyond income, beyond averages, beyond today: Inequalities in human development in the 21st century. New York: UNDP.

Venter, Z.S., Cramer, M.D. and Hawkins, H.J., 2018. Drivers of woody plant encroachment over Africa. Nature Communications 9(1): 2272.

Verdone, M., 2018. The world's largest private sector? Recognizing the Cumulative Economic Value of Small-scale Forest and Farm Producers. Gland: IUCN, FAO, IIED and AgriCord. Vinceti, B., Termote, C., Ickowitz, A., Powell, B., Kehlenbeck, K. and Hunter, D., 2013. The contribution of forests and trees to sustainable diets. *Sustainability* 5(11): 4797-4824.

Vira, B. and Kontoleon, A., 2012. Dependence of the poor on biodiversity: which poor, what biodiversity.
In: Conservation and Poverty Alleviation: Exploring the Evidence for a Link edited by D. Roe, J. Elliott,
C. Sandbrook and M. Walpole. Chichester: Wiley Blackwell, pp. 52-84.

Vira, B., Wildburger, C. and Mansourian, S. (eds.) 2015. Forests and Food: Addressing Hunger and Nutrition Across Sustainable Landscapes. Cambridge: Open Book Publishers.

Viswanathan, P. K., 2008. Emerging smallholder rubber farming systems in India and Thailand: A comparative economic analysis. Asian Journal of Agriculture and Development 5(2): 1-19.

Vittor, A.Y., Gilman, R.H., Tielsch, J., Glass, G., Shields, T.I.M., Lozano, W.S., Pinedo-Cancino, V. and Patz, J.A., 2006. The effect of deforestation on the humanbiting rate of Anopheles darlingi, the primary vector of falciparum malaria in the Peruvian amazon. The American Journal of Tropical Medicine and Hygiene 74(1): 3-11.

van Vliet, N., Mertz, O., Heinimann, A., Langanke, T., Pascual, U., Schmook, B., Adams, C., et al., 2012.
Trends, drivers and impacts of changes in swidden cultivation in tropical forest-agriculture frontiers: a global assessment. Global Environmental Change 22(2): 418-429.

van Vliet, N., Nebesse, C. and Nasi, R., 2015. Bushmeat consumption among rural and urban children from Province Orientale, Democratic Republic of Congo. Oryx 49: 165-174.

Waldron, A., Garrity, D., Malhi, Y., Girardin, C., Miller, D.C. and Seddon, N., 2017. Agroforestry can enhance food security while meeting other sustainable development goals. *Tropical Conservation Science* 10: 1-6.

Walsh, S.J., McCleary, A.L., Mena, C.F., Shao, Y., Tuttle, J.P., González, A. and Atkinson, R., 2008. QuickBird and Hyperion data analysis of an invasive plant species in the Galapagos Islands of Ecuador: Implications for control and land use management. Remote Sensing of Environment 112(5): 1927-1941.

Wang, S. W. and Macdonald, D. W., 2006. Livestock predation by carnivores in Jigme Singye Wangchuck National Park, Bhutan. *Biological Conservation* 129: 558-565.

Weatherley-Singh, J. and Gupta, A., 2015. Drivers of deforestation and REDD+ benefit-sharing: A metaanalysis of the (missing) link. Environmental Science & Policy 54: 97-105.

West, P., Igoe, J. and Brockington, D., 2006. Parks and peoples: the social impact of protected areas. Annual Review of Anthropology 35: 251-277. Whiteman, A. and Lebedys, A., 2006. The contribution of the forestry sector to African economies. *International Forestry Review* 8(1): 31-43.

Whitmee, S., Haines, A., Beyrer, C., Boltz, F., Capon, A.G., de Souza Dias, B.F., Ezeh, A., et al., 2015. Safeguarding human health in the Anthropocene epoch: report of The Rockefeller Foundation–Lancet Commission on planetary health. *The Lancet* 386(10007): 1973-2028.

WHO, 1993. Guidelines for the evaluation of herbal medicine. Manila: WHO regional office.

Wilkie, D.S., Bennett, E.L., Peres, C.A. and Cunningham, A.A., 2011. The empty forest revisited. Annals of the New York Academy of Sciences 1223(1): 120-128.

Wilkie, D.S., Starkey, M., Abernethy, K., Effa, E.N., Telfer, P. and Godoy, R., 2005. Role of prices and wealth in consumer demand for bushmeat in Gabon, Central Africa. Conservation Biology 19: 1-7.

Wilkie, D.S., Wieland, M., Boulet, H., Le Bel, S., van Vliet,
N., Cornelis, D., BriacWarnon, V., Nasi, R. and Fa,
J.E., 2016. Eating and conserving bushmeat in Africa.
African Journal of Ecology 54(4): 402-414.

Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, S., Garnett, T., et al., 2019. Food in the Anthropocene: the Eat–Lancet commission on healthy diets from sustainable food systems. *Lancet* 393: 447–492.

Wolfe, N.D., Switzer, W.M., Carr, J.K., Bhullar, V.B., Shanmugam, V., Tamoufe, U., Prosser, A.T., et al., 2004. Naturally acquired simian retrovirus infections in central African hunters. *The Lancet* 363(9413): 932-937.

Wollenberg, E., Merino, L., Agrawal, A. and Ostrom, E., 2007. Fourteen years of monitoring communitymanaged forests: learning from IFRI's experience. International Forestry Review 9(2): 670-684.

Wood, S.A., Tirfessa, D. and Baudron, F., 2018. Soil organic matter underlies crop nutritional quality and productivity in smallholder agriculture. Agriculture, Ecosystems & Environment 266: 100-108.

Woods, C.L., Cardelús, C.L., Scull, P., Wassie, A., Baez, M. and Klepeis, P., 2016. Stone walls and sacred forest conservation in Ethiopia. *Biodiversity and Conservation* 26: 209-221.

World Bank, 2019a. Forest-SWIFT (Survey of Wellbeing via Instant and Frequent Tracking) (available at: https: //www.profor.info/knowledge/forest-swiftmethodology-high-frequency-forest-poverty-datacollection. [accessed on 23 September 2020]).

World Bank. 2019b. Madagascar economic update: A new start. (available at : https://www.worldbank.org/ en/country/madagascar/publication/madagascareconomic-update-new-start-long-standingchallenges [accessed on 23 September 2020]).

Wunder, S., 2001. Poverty alleviation and tropical forests – What scope for synergies? World Development 29(11): 1817-1833. Wunder, S., Börner, J., Shively, G. and Wyman, M., 2014. Safety nets, gap filling and forests: a globalcomparative perspective. World Development 64: S29-S42.

Yamamoto, Y., Shigetomi, Y., Ishimura, Y. and Hattori, M., 2019. Forest change and agricultural productivity: Evidence from Indonesia. World Development 114: 196-207.

Yasuoka, J. and Levins, R., 2007. Impact of deforestation and agricultural development on anopheline ecology and malaria epidemiology. *The American Journal of Tropical Medicine and Hygiene* 76(3): 450-460.

Yu, P., Wang, Y., Wu, X., Dong, X., Xiong, W., Bu, G., Wang, S., et al., 2010. Water yield reduction due to forestation in arid mountainous regions, northwest China. International Journal of Sediment Research 25: 423-430.

Zhang, T., Wu, Q. and Zhang, Z., 2020. Probable pangolin origin of SARS-CoV-2 associated with the COVID-19 outbreak. *Current Biology* 30(7): 1346-1351.

Zhang, Q., Song, C., and Chen, X., 2018. Effects of China's payment for ecosystem services programs on cropland abandonment: a case study in Tiantangzhai township, Anhui, China. Land Use Policy 73: 239-248.

Zhu, A., 2018. Hot money, cold beer: Navigating the vanilla and rosewood export economies in northeastern Madagascar. American Ethnologist 45(2): 253-267.

Zomer, R.J., Neufeldt, H., Xu, J., Ahrends, A., Bossio, D., Trabucco, A., Van Noordwijk, M. and Wang, M., 2016. Global tree cover and biomass carbon on agricultural land: The contribution of agroforestry to global and national carbon budgets. *Scientific Reports* 6(1): 29987.

Zulu, L. C. and Richardson, R. B., 2013. Charcoal, livelihoods, and poverty reduction: Evidence from sub-Saharan Africa. *Energy for Sustainable Development* 17(2): 127-137.



Chapter 4

Contextual Factors Shaping Forest-Poverty Dynamics

Coordinating Lead Authors: Mónica Gabay and Johan A. Oldekop Lead Authors: David Humphreys, Judith Kamoto, Doris N. Mutta, Nitin D. Rai, Conghe Song, Joleen Timko, Laura Vang Rasmussen and Jennifer Zavaleta Cheek Contributing Authors: Dikshya Devkota and Dietmar Stoian

TABLE OF CONTENTS

4.1 Introduction	96
4.2 Frameworks to Contextualise Forest-Poverty Dynamics	96
4.3 Social Context	97
4.4 Economic Context	102
4.5 Political and Policy Context	104
4.6 Environmental Context	108
4.7 Implications for Forest-Poverty Dynamics, Conclusions and Gaps	109
4.8 References	114

Abstract

Forests and tree-based systems represent complex social-ecological systems. Gaining a better understanding of how contextual factors influence forest-poverty dynamics is essential for the design, targeting and implementation of policy instruments and interventions to alleviate poverty. In this chapter we explore key social, economic, political and environmental factors affecting forest-poverty dynamics, and use a series of illustrative examples to demonstrate how factors can take multiple roles in causal chains of processes of social and environmental change in forest and tree-based systems. We conclude the chapter by highlighting how future research can provide a better understanding of the processes and contexts shaping forest-poverty dynamics, including elucidating the relative effects of different drivers of change on multiple social and environmental outcomes.

4.1 Introduction

Major advances have been made over the past three decades to identify and characterise the socio-economic, political and biophysical processes and conditions that influence forest-poverty dynamics (Geist and Lambin, 2002; Phelps et al., 2009; Meyfroidt et al., 2018; Miller and Hajjar, 2020). In addition to shaping forest-poverty dynamics directly, these processes and conditions create diverse contexts within which policy operates to affect social, economic, and environmental change. Advances in understanding forest-poverty dynamics have been catalysed by the development of theoretical frameworks that enable us to categorise social, economic and environmental systems and their components, and accompanying empirical work to understand how these factors influence the livelihoods6 of forest-reliant households and communities. A better understanding of contexts and how they shape change in forests and tree-based systems is essential for the design and implementation of forest-based policies and interventions aiming to address poverty.



Formerly forested hills in Indonesia converted into agricultural land Photo © Reem Hajjar In this chapter, we bring together elements from research strands focusing on land use change, political science, economics and political ecology to identify social, economic, political, and environmental factors (Figure 4.1) operating in forests and tree-based systems that constrain or enable *poverty alleviation*. We also demonstrate how interrelationships among contextual factors can be conceptualised and analysed to better understand how contexts shape forest-poverty dynamics using a series of illustrative examples.

4.2 Frameworks to Contextualise Forest-Poverty Dynamics

Over the past thirty years, scholars have produced various analytical frameworks to understand connections among the multiple components of social, economic, and environmental systems. Notable examples include the Ecosystem Services and Human Well-being framework developed as part of the Millennium Ecosystem Assessment (MEA, 2005), Elinor Ostrom's Social-Ecological System's framework (Ostrom, 2009) and the conceptual framework for the recent Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (IPBES 2013). These frameworks highlight linkages between the various dimensions of the natural environment and social, economic, and political systems. Linkages include direct relationships (e.g. human well-being benefits derived from provisioning ecosystem services), feedback loops linking multiple mechanisms (e.g. relationships among governance systems, resource users and ecosystem outcomes) and dynamics across temporal and spatial scales. Other frameworks have aimed to describe causal pathways and use overlapping terminologies to disentangle their individual components and mechanisms (e.g. Shaw, 1989; Lüdeke et al., 2004).

Geist and Lambin's (2002) characterisation of deforestation drivers, is perhaps the most commonly used classification of socio-economic, political and biophysical factors influencing forest cover change, and distinguishes between underlying and proximate causes. Geist and Lambin (2002) define proximate and underlying causes in terms of their position within a causal pathway, with proximate drivers being directly linked to outcomes (e.g. logging) and underlying drivers preceding them (e.g. road construction, which facilitates logging). These frameworks are useful analytical tools to conceptualise and empirically analyse relationships among disparate components of socio-economic, political, and environmental systems. However, they provide insufficient insight to identify potential levers of change (see Chapter 5), and the factors that constrain/ enable the efficacy of such levers.

Frameworks derived from causal inference analyses provide additional insight into the relationship among different factors by classifying them as drivers, mediators and moderators to understand and describe causal pathways within a system (Ferraro and Hanauer, 2014a). In this context, 'mediators' are defined as the mechanisms or intermediate steps through which drivers (and potential levers of change) exert their effect (Figure 4.1). In contrast, 'moderators' act as contextual processes and conditions that influence the magnitude of the effect that a driver can exert (Figure 4.1).

Social, economic, political, and environmental factors are often missing from analyses of forest-poverty dynamics (see Chapter 2), but are important because they provide some of the spatial, temporal and contextual elements shaping both forest-poverty dynamics as well as potential levers of change. Furthermore, social, economic, political, and environmental factors can act as any of drivers, mediators, moderators or outcomes (see Figures 4.2. and 4.3.). These factors may simultaneously operate as drivers, mediators, moderators and outcomes: their respective position within causal chains is specific to the analysis being conducted (see Box 4.1 for examples).

The factors that we present in this chapter were jointly identified through literature reviews by the chapter authors and iteratively discussed over multiple meetings. They are classified as relating to the social, economic, political and policy, and environmental contexts which all influence the forest-poverty relationship. Our aim in this chapter is to present a wide range of key social, economic, political, and environmental factors that have influenced, and are likely to continue to influence, forest-poverty dynamics over short and long time horizons. In so doing, we also demonstrate how some of these factors can simultaneously operate as drivers, mediators and moderators in forests and tree-based systems using a handful of illustrative examples in Box 4.1. Given the changing and contested nature of forest-poverty dynamics, our list of factors is not likely exhaustive. However, the factors presented here are broadly discussed in the literature and are key to understanding potential variation in forests-poverty dynamics in diverse settings around the world.

The categorisation and ordering of these factors into social, economic, political, and environmental contexts is but one way of organising them. We recognise that factors often straddle multiple contexts. Furthermore, social, economic, political, and environmental contexts (and the multiple factors within them) are not independent from each other: they often co-occur in space and time, and interact in multiple complex ways (Figure 4.1). Our empirical understanding of how these contexts and factors work together to shape forest-poverty dynamics still remains limited due to analytical limitations (see Chapter 2 and Section 4.7). By using the illustrative examples in Box 4.1 we aim to provide a better analytical perspective of how social, economic, political, and biophysical contexts and factors can shape forest-poverty dynamics, and influence levers of change.

4.3 Social Context

Large-scale social factors are key determinants of local forest-poverty dynamics. These factors can originate outside forests and tree-based systems and may operate at national or international scales (e.g. migration), exert influence at more localised scales (e.g. population dynamics), and be intrinsic to individuals, communities or societies (e.g. identity and culture). Feedback loops signify that these factors both shape and are shaped by the social context within which they take place; they may shape local forest-poverty dynamics through national-level policies that are often implemented at sub-national levels. More intrinsic characteristics and elements, such as identity, culture, social capital and associated local norms, have profound implications for livelihoods and forest-use.

4.3.1 Population dynamics and consumption

Population growth and density have often been considered key factors influencing deforestation because of increased local pressure to convert land to agricultural production, and because of



Figure 4.2

Causal chain linking drivers, mediators, moderators and outcomes



the increased demand for forest products. Agricultural expansion and associated deforestation patterns between the 1960s and 1980s were often linked to state-sponsored agricultural frontier colonisation programmes and land reforms in Latin America and Southeast Asia (Rudel *et al.*, 2009). This pattern is well exemplified by the demographic dynamics in the Brazilian Amazon that followed the National Integration Plan and the associated Transamazon Highway Colonisation Scheme (Moran, 1990).

The historical link between population growth, population density and deforestation has weakened. Rural populations globally have substantially declined over the past decades (World Bank, 2020). In their synthesis of 152 sub-national case studies Geist and Lambin (2002) found that increases in population due to fertility rates in forest areas was not a primary driver of deforestation (Geist and Lambin, 2002). While smallholder agriculture continues to be a driver of deforestation (Godar et al., 2014), most tropical deforestation and associated biodiversity loss in recent decades have been driven by well-resourced farmers, ranchers and loggers as well as international agricultural corporations meeting consumer culture and market demands for timber, soy, beef and palm oil, much of it in high- and middle-income countries (Henders et al., 2015; Green et al., 2019; see Chapter 6).

Population shifts from rural areas to urban population centres, and the rise of the middle classes in lower- and middle-income countries have created new urban demands (African Development Bank, 2019). These demands have been linked to deforestation (DeFries et al., 2010) and have been predominantly met by large industrial agricultural projects (Davis et al., 2020). Critically, large-scale agricultural investments and land transactions, often termed 'land grabs', have been linked to multiple negative outcomes for rural and *forest-dependent* communities, including forced displacement and resettlements, and loss of access to land, resources and livelihoods (Agrawal et al., 2019; Hajjar et al., 2020).

4.3.2 Rural outmigration

Outmigration from rural areas represents one of the most important changes in population dynamics globally in recent decades (Hecht *et al.*, 2015; World Bank, 2020). Rural outmigration has been shown to have multiple and complex effects on forests and forest livelihoods. For example, outmigration in Nepal rose from near zero in 1980 to 29% in 2010 and has been a key driver of *reforestation* and poverty alleviation (Fox, 2018; Oldekop et al., 2018), with remittances accounting for 26% of Nepal's gross domestic product (GDP) in 2019 (World Bank, 2020). We explore how rural outmigration can act as a moderator of *policy instruments*, as well as a driver of change in forests and treebased systems in Box 4.1.

The effects of outmigration vary significantly at the household level (Sunam, 2017), and longterm development outcomes of outmigration in places of origin remain unclear (de Haas, 2010; Oldekop et al., 2018). The notion that outmigration of male community members increases women's participation in forest management decisions is also debated (Giri and Darnhofer, 2010; Lama et al., 2017; Prateek et al., 2019). While remittances may reduce household dependence on agriculture and forests resources and help to diversify income sources (Hunter et al., 2015), outmigration has also been linked to labour shortages and the weakening of community management institutions and the capacity for sound decision-making through a reduction in social capital (Robson and Nayak, 2010; Poudel, 2019). Migration effects also depend on whether migration is seasonal, national or international, with seasonal and national migrants often retaining close links to areas of origin, and international migrants sending larger remittances (Davis et al., 2010). Remittance effects also appear to be context dependent, with evidence suggesting that remittances may be invested to switch to less labour intensive and land extensive agricultural systems, such as cattle ranching in some South American countries (Garcia-Barrios et al., 2009). These investments and changes in production systems can deepen rural inequalities and increase deforestation as wealthier landowners consolidate landholdings, and clear additional tracts of forest as part of their agricultural expansion activities (VanWey et al., 2012; Alix-Garcia et al., 2013; Davis and López-Carr, 2014; Taylor et al., 2016).

The 2020 COVID-19 pandemic has heavily impacted migration flows and remittances, and in many countries has driven the return migration of thousands of people to rural areas. The effects that these new migration dynamics will have on incomes, livelihoods and forests remain to be seen.

4.3.3 Armed conflict

Armed conflict and forest-poverty dynamics are interlinked in many contexts across the globe (Brainard and Chollet, 2007). Geography and poverty are both significant predictors of conflict intensity (Do and Iyer, 2010). For example, 80% of armed conflicts in recent history have occurred in biodiversity hotspots, many of them tropical forest regions (Ordway, 2015), leading some to speculate that violent conflicts are endemic to forests (Chomitz *et al.*, 2007). Conflict-related deaths are significantly higher in poorer districts and in geographical locations favoured by insurgent groups, such as mountains and forests, as they enable insurgents to hide easily from government forces (Do and Iyer, 2010).

Poverty is both a cause of insecurity and a consequence of it (Brainard and Chollet, 2007). The lack of economic opportunities is significantly and robustly correlated with a higher intensity of violence (Do and Iyer, 2010). Communities often fall into vicious cycles of conflict, environmental degradation and poverty, leading to what Brookings scholar Susan Rice calls a "doom spiral" that may be nearly impossible to overcome without outside aid (Brainard and Chollet, 2007; Bob and Bronkhorst, 2010).

Although literature explicitly untangling the connections between forests, poverty and conflict is scarce, three key linkages have been identified. First, conflict and warfare can drive deforestation (Ordway, 2015). Insurgent groups in forests frequently destroy forests during violent clashes, cleared for military purposes, and felled for valuable resources such as "conflict timber" to fund and sustain conflict (Harwell, 2011; Kleinschmit et al., 2016). Forests are vital sources of refuge and emergency subsistence for both local and displaced peoples (Harwell, 2011), yet forest-dwellers frequently experience violence, displacement and livelihood insecurity as a result of conflict, leading to increased levels of poverty, malnutrition, illness and death (Buvinic et al., 2013). Massive migrations of internally displaced peoples can also result in a redistribution of pressure on forests in settlement regions (Ordway, 2015). Second, conflict may lead to the active exclusion of activities from certain geographic regions, favouring forest recovery in what has been referred to as war zone refugia or 'gunpoint conservation' (Ordway, 2015). As agricultural fields are abandoned and wood extraction is hindered, forests are able to recover (Kaimowitz, 2006; Sánchez-Cuervo and Aide, 2013). Third, conflicts may lead to the collapse of institutions (Ordway, 2015). Extreme poverty exhausts governing institutions and depletes resources, fuelling a volatile mix of desperation and instability (Brainard and Chollet, 2007). Weakened institutions, plagued by ineffective governance, are unable to meet people's basic needs or to allow them to control their territory, leaving lawless areas and natural resources vulnerable to hijacking by predatory actors (Brainard and Chollet, 2007).

Post-conflict processes see governments prioritising socio-economic recovery, peacekeeping and *poverty reduction* (Pérez and Garzón, 2015) over natural resource management and environmental sustainability. Recognition that natural resources can be an opportunity and a challenge for peace has led international peacekeepers to intervene in post-conflict countries to establish governance reforms that promote sustainable peace and development (Beevers, 2015).

4.3.4 Identity and culture

Identity politics have strong links to forest conservation, rural livelihoods, and poverty. Anthias and Radcliffe (2015) use the term 'ethno-environmental fix', Adam (2010) and Hall *et al.* (2011) use the terms 'ethnoterritorialization' and 'ethnoterritorial claims' to show how identity is used to exclude others from forest titles and lay territorial claim based on ethnicity. Some governments use these ethnicity-based communal titling to retain territorial control while giving local people the illusion of self-government (Tubbeh and Zimmerer, 2019).

The history of forest use, governance, and dependence has been intrinsically linked with the making of a 'forest subject' (Li, 2010). Forest dwelling people have often been cast as non-market subjects who are dependent on the forest for subsistence and, therefore, needing to be protected from the market and micro-economic processes (Li, 2014). States have often used this identity of the non-market, subsistence-based forest dwellers to deny them ownership of land and assets (Li, 2014). Forest residents who are denied these rights may find their labour used in resource extraction efforts, such as timber operations (Münster, 2014). The lack of access to the market and land has 'arborealised' people, making them further dependent on the forests for such products as fruits, resins, tubers and fuelwood and, in some cases, swidden agriculture to meet their food needs (Walker, 2004). This has produced an identity of rural people as being 'ecosystem people' or 'living in harmony with the forest' (Gadgil and Guha, 1992). Such an identity has thus been 'fixed' in current forest governance narratives despite mounting evidence of their historical connections with markets and agrarian society. Although most governance regimes tend to simplify the community for ease of interventions, the 'community' is not homogeneous (Agrawal and Gibson, 1999). At the same time, the creation of an indigenous identity for forest dwellers might divide them and prevent them from making stronger claims for a redistributive regime (Shah, 2010).



Forest-dwellers obtain most of their daily resources from their immediate surroundings. Here, an ethnic Chakma woman stands in front of her house, Chittagong, Bangladesh Photo © Terry Sunderland

4.3.5 Gender norms

Participation in forest usage and decision-making is largely determined by societal rules, norms, and perceptions, including those pertaining to gender. Predicated on gender norms and rules –especially the division of labour and economic endowment – women and men differ in the nature and extent of their reliance on, and use of, forests (Agarwal, 2009). Such norms shape the role of men and women, what they are expected to do or not to do, their preferences in terms of forest products, their knowledge of forests, and ultimately, the ability to use and protect forests.

Since agriculture and forests provide an essential resource to livelihoods in many parts of the world, including sub-Saharan Africa, South Asia, and Southeast Asia, gender patterns of work also play a crucial role in poverty reduction and food security (FAO, 2010; Mudege *et al.*, 2017). *Pathways out of poverty* differ for men and women as they require different types of interventions depending on their reliance on forests and resource usage (Cheng *et al.*, 2019).

Gendered patterns of participatory exclusion in resource access and decision-making persist in many contexts around the world due to the lower bargaining power women have based on pre-existing socio-economic inequalities and relations of power (Agarwal, 2001; Badola and Hussain, 2003; Deda and Rubian, 2004). Often, societal norms define women's role as centred around domestic work and childcare, and societal perceptions discount women's abilities and opinions (Agarwal, 2001). However, opportunities for effective participation are important as a measure of citizenship and a means of empowerment, as well as for equity, efficiency, and sustainability of resource usage (Agarwal, 2001). For example, land is a key determinant of production and central to agricultural and economic development in developing countries, but women are less likely to own land compared to their male counterparts (Kiptot et al., 2014). Even where the law might grant equal land ownership and inheritance rights, customary laws often bar women from land ownership. Research in African contexts has also shown that women have less access than men to productive resources and opportunities such as labour, education, extension, financial services, and technology (Kiptot et al., 2014). Likewise, women in Nepal and India are often excluded from decision-making consultations and training related to community forest management, resulting in the exclusion of



The lack of resources creates harsh living conditions in the mountains of Nepal Photo © Nelson Grima

their existing knowledge in forestry programmes, and women having less chance of acquiring new knowledge (Agarwal, 2001). Such unequal distribution of access and benefits of resources further worsen women's plight, particularly in already poor households.

4.4 Economic Context

Economic factors shaping social and environmental change in forests and tree-based systems reflect multiple political, financial and market-related processes. Typically, economic and political forces originate outside forests and tree-based systems and represent contexts shaped primarily by external actors and institutions, including international markets and international agreements. Important economic factors include overseas development assistance and global financial flows, global and local market dynamics, and investments in infrastructure that facilitate the movement of goods and people.

4.4.1 Overseas development assistance and global financial flows

Global financial flows are key drivers of forest cover and poverty dynamics (Meyfroidt *et al.*, 2013; Folke *et al.*, 2019). Key flows include overseas development assistance (ODA), export credits, and tax avoidance and evasion.

An estimated USD 70 to 160 billion annually is needed to sustainably manage the world's forests (World Bank, 2014). Financial estimates for the effective conservation of biodiversity, much of which resides in forests in low- and middle-income countries (LMICs), are of a similar magnitude, ranging between USD 76 to 440 billion per year (McCarthy et al., 2012; Drutschinin and Ockenden, 2015). However, since 2014 only USD 7 billion in international aid has been allocated to forest-related projects. To assess recent trends in forest aid, we collated a dataset on funding to all forest and tree-related projects from the databases of major aid organisations (e.g. GEF, OECD, World Bank) using keywords (such as 'forest', 'agroforestry', 'deforestation', 'tree'). Furthermore, many ODA disbursements often support actors that favour the extraction of natural resources, which may conflict with forest resource conservation aims.

Export credits to businesses are an important source of loans for LMICs, and are managed by government agencies, private sector bodies or a mixture of the two. Loans are often tied to business contracts with favoured companies in lending countries (Clapp and Dauvergne, 2011). Evidence suggests that such forest developments have often disregarded the traditional lands and forests claimed by indigenous peoples and communities and, in the process, undermined local livelihoods. For example, Asia Pulp and Paper, the holding company for the interests in pulp and paper of Sinar Mas, a major Indonesian business, has been implicated in various deforestation scandals, despite formal commitments to sustainability (Jacobson, 2018).

Tax avoidance (through the use of legal loopholes) and tax evasion (through the use of illegal loopholes) deprive states of funds that could be applied for a range of social and environmental measures (Global Financial Integrity, 2019). The financial secrecy surrounding tax havens makes it difficult to track the money that flows through them and to hold financiers accountable for the environmental and social consequences of their investments (Galaz *et al.*, 2018). However, although estimates are difficult to come by, money channelled through offshore tax havens has been used to finance deforestation that impoverishes communities (Galaz *et al.*, 2018).

4.4.2 Global market dynamics

Global production and trade of principal woodbased products recorded their highest ever values in 2018 (FAO, 2019). This provides forest-dependent people with increased opportunities for the marketing of forest products. However, many marginal communities require technical assistance for post-harvest processing, and stronger support to facilitate market integration, strengthen bargaining power, and access to technology and credit (Belcher, 2005).

Unlike agricultural markets where the advance of voluntary sustainability standards, such as organic, Fairtrade or Rainforest Alliance/UTZ certification, has allowed smallholder farmers to strengthen their position in global value chains (Potts, 2018), international markets for forest products tend to offer less opportunities for resource-poor people to differentiate their offer of timber or non-timber forest products (NTFPs). Timber markets, in particular, provide for little differentiation of logs or sawn wood originating from forest-based communities. Demands for Forest Stewardship Council (FSC) certification are often too costly for forest-based communities, and local communities continue to be at a market disadvantage (Macqueen et al., 2006; Wiersum et al., 2013).

NTFPs and agroforestry products linked to ethical trade schemes in global value chains tend to be more accessible options for gaining added value to commodities and improving revenues (Nelson *et al.*, 2002; Duchelle *et al.* 2014). In some cases, the orientation toward global markets for certified wood products has also allowed community forest enterprises (CFEs) to thrive, with important benefits for resource-poor members. In the Maya Biosphere Reserve in the Petén, Guatemala, for example, CFEs managing forest concessions rich in mahogany and other woods valued in international markets have generated employment and income that have enabled community members to move out of extreme poverty, if not poverty altogether (Stoian *et al.*, 2018).

With limited opportunities for differentiating community-based forest products in the market as such, a viable pathway out of poverty for forest-based communities is adding value to timber and NTFPs through processing and orientation toward emerging markets. Prominent examples for CFEs involved in processing logs into dried sawn wood, plywood, furniture and other value-added products include Mexico (Antinori, 2000) and Brazil (Humphries et al., 2012). Certain NTFPs, in turn, may be directed toward emerging markets for 'superfoods', such as acai, camu camu, sancha inchi, aguaje and other fruits from the Amazon. Adding value to timber and NTFPs requires CFEs to have a sawmill and other diversified processing infrastructure, such as driers, furniture and mouldings factories, or chip mills (Bray and Merino-Pérez, 2002). Such infrastructure is based on investments that often imply donor or government funding, or a combination thereof.

Global markets for forest and tree crop products are entering a new phase of commitments and sustainability standards, such as 'zero net deforestation' (promoted under the UNCCD). In light of the emerging sector of impact investments which often relate to such standards, there are increasing opportunities for CFEs managing forests sustainably to attract this new type of investment to produce more semi-finished and finished products and, based on the value added, to enable their members to move out of poverty.

4.4.3 Local market dynamics

The exact proportion of forest products traded in local markets vis-à-vis those traded in global markets (see Section 4.4.2) is not readily available, nor is the exact value of trade in NTFPs which are critical for many forest-dependent communities for both income generation and household consumption (see Chapter 3). Regional market trends are available for some of the principal forest product markets in the Global North (Europe, Commonwealth of Independent States, North America), with demand for several wood-based products on the rise (UNECE/FAO, 2019), but less so for regional and local markets in the Global South. The latter provide viable outlets for diverse forest products for individual forest users, collective businesses such as CFEs, and other types of small and medium forest enterprises (SMFEs). Compared with international markets, local markets for timber and NTFPs tend to be less demanding in terms of quality and volume requirements, certification or other sustainability standards. Barriers to enter such markets are generally lower, as are potential rewards in the form of quality or other price premiums. Even in domestic timber markets, formal and informal barriers relating to forest and tree tenure, high transaction costs due to cumbersome regulations, and bribes to speed up bureaucratic procedures, may restrict growth and obstruct opportunities for CFEs (Southgate et al., 2000; Gritten et al., 2015; Pulhin and Ramirez, 2016).

In view of growing populations in most of the Global South, local demand is growing for agricultural products, timber, wood-based products and NTFPs such as traditional medicines, fruits, fibres, dyes, seeds, oils, resins and gums (Kusters et al., 2006). Recent studies of NTFP markets in the Brazilian Amazon, for example, have shown economic potential, along with a need for investments in infrastructure for production, training and organisation of extractive communities, and marketing support (Angelo et al., 2018). Better positioning of CFEs and other SMFEs in national and international markets requires upgrading their technical, business and financial capacities to add value to timber and NTFPs, reduce production and administration costs, engage in new business relationships, and negotiate more favourable terms of trade (Donovan et al., 2006). Timber and NTFP marketing may also be combined with recreational ecosystem services in pursuit of multifunctional livelihoods and enhanced social values of communities engaged in forest product extraction (Carvalho Ribeiro et al., 2018). More generally speaking, there is potential for using a forest-based bioeconomy frame for NTFPs to contribute to human nutrition, renewable materials, cultural and experiential services, job creation and income opportunities in rural areas (Weiss et al., 2020).

4.4.4 Infrastructure development

The development of infrastructure has profound implications for forests and tree-based systems (Laurance *et al.*, 2015). Yet, improved access to commodity and labour markets through the construction of roads, clean water and energy can lead to substantial reductions in poverty (Collier, 2007). These benefits are often driven by lower

transportation costs to and from markets, the diversification of livelihoods as rural households complement agricultural incomes with other income sources, and better health outcomes through improvements in sanitation and access to health facilities. However, market integration is often associated with increases in deforestation (Geist and Lambin, 2002) driven both by the expansion of agricultural production of already resident households (Oldekop et al., 2013), as well as through access to forest frontiers that facilitate in-migration for access to new lands, and illicit activities such as illegal timber extraction (Barber et al., 2014). In-migration to forest frontiers has also been linked to land disputes, especially when land tenure and rights are unclear or contested (Messina et al., 2005). We explore how roads can act as both moderators of policy instruments, as well as drivers of change in forests and treebased systems in Box 4.1.

The development of other types of infrastructure, including the construction of hydroelectric dams and expansion of extractive industries such as mining, is often associated with pollution and rural livelihood losses due to conflicts over land and access to natural resources, although beneficial effects may include electrification and increases in incomes through the creation of low-skilled labour markets. Dams and mining activities are also often associated with increases in deforestation (Bebbington *et al.*, 2019), and the down-grading, down-sizing and de-gazettement of protected areas (Golden Kroner *et al.*, 2019).

The development of mega-infrastructure projects such as the Chinese-led Belt and Road Initiative, the Lamu Port and Lamu – Southern Sudan – Ethiopia Transport Corridor (LAPSSET Kenya) and India's 'Make in India' initiative, which aim to increase national and international connectivity and secure access to energy and natural resources, will be transformational for forest landscapes (Ascenção *et al.*, 2018), and likely generate both positive and negative effects for forests and rural communities.

4.5 Political and Policy Context

The political context frames the ways in which stakeholders will interact within the forest-poverty space. Many of the actions and resources required to maximise the potential of forests to contribute to economic growth and poverty alleviation are linked to the effective design and implementation of institutions. Commonly understood as "humanly devised constraints that shape human interaction or the rules of society" (North, 1990), institutions determine the structure of formal and informal power arrangements and include social norms, customs, and informal rules that are locally devised and locally enforced, as well as more formalised laws, agreements and policies implemented by governments, non-governmental and international organisations at local, national and international scales. Effective institutions are a necessary condition for well-functioning social and economic relations, and economic development (North, 1990), and are thus an important part in the analysis of natural resource use and poverty (Box 4.2). Numerous factors operate within the political and policy context. Here we highlight tenure and property rights and decentralisation, elite capture, illegal activities and corruption, and a range of government support programmes such as subsidies, credits, social protection mechanisms and agricultural technologies.

4.5.1 Tenure and property rights and decentralisation

Contemporary policy issues in the forest sector related to tenure and property rights include: the devolution of tenure from centralised governments to communities and private entities, and the decentralisation of forest management to local governments and movements to formalise property rights throughout the developing world (Agrawal and Ostrom, 2001; Jagger *et al.*, 2014, Galik and Jagger, 2015; Sikor *et al.*, 2017; Miller *et al.*, 2019). Tenure and property rights thus act as levers of change (Chapter 5) and also provide a policy environment linked to other types of interventions and agendas (e.g. *forest restoration* initiatives) (Erbaugh *et al.*, 2020).

Indigenous peoples and local communities have legally recognised rights to an estimated 15.3% of the world's forests (RRI, 2017), although a much larger share is contested. Theoretical and empirical work has shown that where tenure and property rights are unambiguous, justly enforced and secure, rightsholders are more likely to invest in forest-enhancing behaviours because they are more likely to capture the benefits of their investments (Ostrom, 1990; Somanathan et al., 2009; Mogoi et al., 2012). This improved efficiency, accountability, equity and sustainability in the production and provision of forest goods and services presents opportunities for poverty alleviation (Adam and Eltayeb, 2016). Forest benefits and household income can come from the sustainable use of forest products, such as timber, construction materials and firewood. Forest revenue streams can account for as much as half of a household's income (Hill, 1999). Access and withdrawal rights to productive resources for forest-dependent people, indigenous people and women are therefore considered a crucial factor in poverty alleviation (Schlager and Ostrom, 1992).

Decentralisation is the process by which a central government cedes powers to actors and institutions at lower levels of government (Ribot 2002). Many natural resource management decentralisation reforms involve changes in ownership or changes in property rights structures (Larson and Soto, 2008; Jagger et al., 2014). Although there is evidence that decentralisation and tenure reforms can lead to reductions in deforestation and poverty (Blackman et al., 2017; Oldekop et al., 2019; Miller et al., 2019), the extent to which decentralisation programmes in developing countries should incorporate goals of poverty alleviation continues to be debated (Samii et al., 2015). Tenure reforms continue to face challenges in lower- and middle-income countries, including: states retaining control of high-value forest (Barrow et al., 2016) and decentralising low-value degraded forestland in need of restoration (de Royer et al., 2018); persistent marginalisation of women's rights to resources (Namubiru-Mwaura, 2014; Elias et al., 2017); and differential livelihood impacts on ethnic minorities, and other marginalised groups (McElwee, 2009; Jagger et al., 2014).

4.5.2 Elite capture

Elite capture refers to the process of corruption through which individuals with high-level political status derived from their wealth, education, ethnicity or other social characteristics reap a disproportionately large share of benefits from resources (Bardhan, 2002; Persha and Andersson, 2014). Elite capture is thus a leading driver of *inequality* in access to resources.

In forests and tree-based systems, elite capture manifests itself through systematic forest policy biases that benefit an influential group (elites), and has often been linked to decentralisation. tenure reforms and community-driven initiatives (Platteau, 2004; Dasgupta and Beard, 2007; Lund and Saito-Jensen, 2013; Persha and Andersson, 2014). For example, despite evidence that community forestry can reduce poverty (Rahut et al., 2015; Bijaya et al., 2016; Oldekop et al., 2019), and in some instances also reduce income inequalities (López-Feldman et al., 2007), many studies have also demonstrated that community forestry can make life harder for the poor and marginalised through various forms of unequal benefit distribution. In Kenya, Chomba et al. (2015) and Mutune et al. (2017) found that community forestry has led to increases in inequality by restricting forest access for the poorest community members. Similar inequities have also been observed in numerous other countries with community forestry programmes, including Ghana (Baruah, 2017), India (Mukherjee et al., 2017), Indonesia (Bong et al., 2019), Madagascar (Brimont et al., 2015), Mexico (Garcia-Lopez, 2019) and Nepal (Bijaya et al., 2016). Failure to account for heterogeneities in political power, socio-economic status, knowledge among forest resources users, and vulnerable groups in the decision-making processes allows wealthier elite members to exercise power over poor and disadvantaged households, and capture the majority share of benefits from community forestry (Adhikari et al., 2014; Persha and Andersson, 2014; Sunam and McCarthy, 2016; Essougong et al., 2019). Critically, elite capture can constrain the implementation of forest conservation and development policy instruments that leverage land titling and community institutions (To et al., 2012; Chomba et al., 2015). For example, payment for ecosystem services schemes require secure titles. In Kenya and Vietnam colonial land tenure legacies have disproportionately disadvantaged households with few or no land entitlements (To et al., 2012; Chomba et al., 2015). Similarly, communities with weaker internal governance structures in Ecuador were less likely to perceive benefits from payments for ecosystem services to be equitable (Hayes and Murtinho, 2018).

4.5.3 Forest crime and corruption

Forests can conceal a number of illicit activities, including cultivation of illegal drugs and illegal mineral extraction as well as harbouring insurgent groups. Illegal logging is perhaps the most widespread 'forest crime', and forms part of a broader problem of malpractice and crime associated with the timber trade (Kleinschmit et al., 2016; Tacconi et al., 2016). Illegal logging has significant negative environmental and social consequences. It results in biodiversity depletion, soil erosion and increased carbon dioxide emissions (Putz et al., 2012; Edwards et al., 2014). Economically, it deprives governments of tax revenue and increases the global supply of timber, depressing prices and placing businesses that trade in legally-sourced timber at a comparative disadvantage (Pacheco et al., 2016). Socially, the problem erodes the lifestyles of traditional forest communities and often accompanies criminal activities that lead to poverty in forests, such as violence against communities that resist illegal logging, illicit drug

cultivation, armed insurgency and the forced exploitation of labour (Reboredo, 2013; Pacheco *et al.*, 2016).

Globally, forest crimes were estimated to total USD 30-100 billion per year, or 10 to 30% of the global timber trade (Nellemann et al., 2014). Pellegrini (2011) reported that illegal logging made up 70-80% of the total timber volumes extracted from forests in Bolivia, Honduras and Nicaragua. The countries most afflicted are the tropical regions of Latin America (Brazil, Colombia, Peru), Africa (Cameroon, Democratic Republic of the Congo, Republic of Congo), Asia and the Pacific (Indonesia, Laos, Malaysia, Myanmar, Papua New Guinea), and Russia (Humphreys et al., 2006; Smirnov et al., 2013; Gan et al., 2016). Corruption and weak governance of the forest sector remain widespread (Sündstrom, 2016). In Indonesia alone, a report from Human Rights Watch (2013) found that forest sector corruption cost the government USD 7 billion per annum between 2007 and 2011, equivalent to the country's entire spending on health care, with half of all timber logged illegally.

Well-equipped and armed illegal loggers are often the most powerful organised force in many remote forest regions, with a greater visibility and presence than the state. Forests also conceal other criminal activities such as drug cultivation, guerrilla armies, illegal mining and enslavement. For example, habilitación in Peru is a form of debt servitude. A lender, or habilitador, will lend money to an intermediary (the patrón) who will lend it to loggers at a high rate of interest. From the loan the logger must buy equipment and tools, hire labour, such as chain saw operators and cooks, and procure fuel and food. The logger must then hand over to the *patrón* an agreed volume and type of timber in order to pay off the debt. Often repayment of the full debt is impossible, especially when the patrón deliberately undervalues the timber harvested in order to perpetuate the debt. The logger must often ask for a further loan to repay the original debt, so the vicious cycle of poverty and crime continues (Urrunaga et al., 2012).

4.5.4 Government support programmes for rural areas

Subsidies, credits, and social protection programmes (e.g. conditional cash transfers or public work programmes) can help reduce vulnerabilities of rural households and have important implications for poverty and livelihoods in forests and tree-based systems. While the adoption of new technologies also occurs independently from the implementation of public policies, often



Simple huts made with locally available materials are a common sight on the hills of Vietnam Photo © Terry Sunderland

subsidies, credits, social protection programmes and agricultural technologies are intertwined. For example, rural credits and cash transfers are common components of social protection programmes (Dyngeland *et al.*, 2020), and are often invested in agricultural technologies. Similarly, public work programmes can specifically target the development of irrigation technologies to improve agricultural production and household *resilience* to environmental shocks.

Understanding these factors is critical, because the adoption of agricultural technologies (e.g. mechanisation, agricultural inputs and irrigation) can help improve agricultural production, increase income through improved yields, and reduce household reliance on forests. Critically, the uptake and use of inputs has, in turn, been catalysed by advances and adoption of mobile phone technologies. In Kenya, the M-PESA mobile phonebased money transfer service (now operating in several other African countries), has significantly eased and lowered financial transaction costs in rural areas (Mbiti and Weil, 2016) and significantly increased the use of agricultural inputs by rural households (Kirui *et al.*, 2013). Information and communication technologies more broadly, are having transformational impacts in forest communities allowing them to map and monitor forest resources (Oldekop *et al.*, under review; see Chapter 6).

Similarly, subsidies (e.g. in the form of agricultural inputs and technologies) and the availability of rural credits can provide financial incentives to support investments in agricultural production by rural households (Druilhe and Barreiro-Hurlé, 2012). However, the ability of subsidies and credits to reduce rural poverty and deforestation - at least in the absence of additional land-use conditionalities - remains disputed (Hemming et al., 2018; Vang Rasmussen et al., 2018). Critically, subsidies and other types of policy incentives can also generate perverse outcomes, if they are not well designed. For example, the Grain-for-Green programme in China was developed to combat soil erosion on sloping lands in the wake of the Yangtze River flood (Liu et al., 2008). Despite evidence of some positive socio-economic impacts and enhanced reforestation efforts (Peng, 2007; Liu et *al.*, 2008), much of the reforestation efforts of the Grain-for-Green programme has been monoculture *plantations*, which have little ecological or biodiversity value (Hua *et al.*, 2016).

Social protection policies have improved education outcomes of forest-dependent communities in Mexico (De Janvry et al., 2006), and nutrition in the Brazilian Amazon (Piperata et al., 2011). In addition to formal state-led social protection programmes, forest producer organisations can provide informal social protection by offering financial support in savings and credits as seen in China, India and Uganda (Tirivayi et al., 2018). However, the effects of social protection programmes on deforestation remain poorly understood (Alpízar and Ferraro, 2020). Mexico's Oportunidades programme was found to increase deforestation (Alix-Garcia, 2013). Brazil's Fome Zero programme has been shown to have both positive and negative effcts on forest cover (Dyngeland et al., 2020), while recent evidence from Indonesia suggests that social protection has helped to reduce deforestation (Ferraro and Simorangkir, 2020).

4.6 Environmental Context

Environmental factors are predominantly time invariant and, in the case of climate change, effects are often of slow onset and characterised by long time lags. These factors and processes characterise boundaries, constraints, and limits to forest dynamics and livelihood activities, and are thus extremely difficult to change through interventions and policy instruments. For the purposes of this analysis, we highlight both geographical factors and climate change.

4.6.1 Geographical factors: location, topography and climatic conditions

Geographical factors – including location, topography and climatic conditions – have a strong influence on the relationship between forests and people (Sunderlin *et al.*, 2005). This link is perhaps most starkly evident in Geist and Lambin's (2002) seminal study of the drivers of tropical deforestation across 152 cases. They found striking regional differences, with only the development of market economies and the expansion of permanently cropped land for food to have geographically invariant effects on deforestation. All other factors (e.g. institutional, technological, and demographic) were found to have distinct regional patterns in the ways that they influence forest loss and, by extension, livelihoods.

Topography affects forest structure, function, dynamics - including post-disturbance recovery rates - in its regulation of temperature, precipitation and moisture, as well as energy and nutrients along elevation gradients (Hadley, 1994; Muscarella et al., 2020) and thus indirectly influences the forest resources available to local people. For example, in a mountainous community in a subtropical climate zone in China studied by Song et al. (2018), ample heat and water resources led to high forest coverage. The vast majority of households studied relied on fuelwood as the primary source of energy despite tremendous economic growth, and a significant number of households cultivating cash crops used fresh logs inside forests as a major source of income (Zhang et al., 2019). In contrast, in a semi-arid community in the Loess Plateau, China, the major source of income is from apples and walnuts (Song et al., 2014).

In addition to influencing forest-poverty dynamics through integration into market economies and access to amenities, geographical location, in particular distance to urban centres and transportation hubs (e.g. seaports), has also been shown to influence political agency. Remote communities are often at a disadvantage, because political relations are often formed and maintained in urban settings (Sunderlin et al., 2005). Despite broad acknowledgement that biophysical factors are key determinants of forest-poverty dynamics, our understanding of factors like location, topography and climatic conditions remains limited due to biases of where studies are conducted (Hajjar et al., 2016; Cheng et al. 2019; Miller and Hajjar, 2020) and the variables that are included in analyses (Hajjar et al., 2016).

4.6.2 Climate change and related policies

Contemporary climate change policies and actions place both forests and the communities that depend on them in the global spotlight – both in terms of the impact of climate change on forests and the role of forests in reducing or exacerbating climate change (Parrotta *et al.*, 2012; Griscom *et al.*, 2017; Watson *et al.*, 2018).

First, climate change – manifested through increasing temperatures, variations in rainfall, and more frequent and intense natural disasters – threatens both forest cover and the integrity of forests' biological functions (Trumbore *et al.*, 2015). For example, excessive precipitation (Hubbart *et al.*, 2016) and exceptional droughts (Millar and Stephenson, 2015) increase forests' susceptibility to floods, wildfires and diseases. When climatic stressors affect biological processes, this in turn
threatens ecosystem services provided by forests, like nutrient cycling, soil and water conservation, and preservation of biodiversity (Ellison et al., 2017; IPBES, 2019; Piao et al., 2019). Moreover, climatic stressors might cause defoliation and tree mortality leading to declining forest productivity over large spatial scales (Brienen et al., 2015). Severe droughts have caused widespread forest degradation in Amazonia (Xu et al., 2011) and the Congo Basin (Zhou et al., 2014), and drought-induced large-scale tree mortality in the western United States (van Mantgem et al., 2009). Forest fires are also expected to increase with subsequent consequences for forest-dependent livelihoods and human health (Barbero et al., 2015; Alencar et al. 2015; Tan-Soo and Pattanayak, 2019). A separation of cause and effect may be observed, with the consequences of climatic change on forests and large-scale forest clearance often experienced thousands of kilometres from where the deforestation occurs (Lawrence and Vandecar, 2015). The cumulative effects of climate change on biodiversity threaten more than one million species (IPBES, 2019). Finally, negative impacts on natural resource dependent livelihood strategies might be an immediate consequence of extreme climatic events, although households may also be able to recover over the longer term (Bauer et al., 2018).

Second, forests are a key mechanism for mitigating climate change through forest restoration, protection and sustainable management because forests (including peatlands) are the most important biome that removes CO₂ from the atmosphere for long-term storage (Dixon et al., 1994; Pan et al., 2011; Bastin et al., 2019). The Bonn Challenge on forest landscape restoration, launched in 2011 and extended in 2014 by the New York Declaration on Forests, has a target of restoring 350 million hectares across the globe by 2030 - corresponding to 3% of the global ice-free land area. These forest-based climate change mitigation and adaptation efforts (i.e., wide-scale reforestation and restoration initiatives) have the potential to sequester large amounts of carbon and will provide new opportunities and challenges for forest-dependent communities - notably concerning alignment with other sustainability agendas, including poverty-alleviation, land rights and food security (Mansourian et al., 2020).

4.7 Implications for Forest-Poverty Dynamics, Conclusions and Gaps

The factors that we highlight in this chapter represent a wide range of social, economic, political, and environmental contexts that shape forest-poverty dynamics. Although they influence forest-poverty dynamics at very local scales, they represent processes and forces that are often external to local forest communities (e.g. labour markets driving rural outmigration), exert influence over large geographical scales (e.g. climate change) and are subject to political and economic forces operating at subnational, national and international scales (e.g. large-scale infrastructure investments). Critically, many of these contexts and factors co-occur in time and space, and our illustrations in Box 4.1 demonstrate how different factors can simultaneously act as drivers, mediators, and moderators within causal chains. The multi-scalar nature of contexts and their complex interactions have implications for the design, implementation and evaluation of levers of change (Chapter 5). This includes the need for policy instruments that can operate across scales (e.g. international climate agreements that influence the implementation of different interventions at local scales).

However, despite substantial theoretical advances, our empirical understanding of complex social-ecological systems, and how different social, economic, political and environmental factors intersect to shape forest-poverty dynamics remains limited. Equally, there is a dearth of evidence on the effectiveness of policy instruments in the forest sector to reduce poverty. This is due to four key features of scholarly work on social-ecological systems study so far.

First, studies so far have been largely mono-deterministic (Alix-Garcia *et al.*, 2015; Schleicher *et al.*, 2017; Oldekop *et al.*, 2019). Research has predominantly focused on trying to estimate the effects of single drivers or processes of change (e.g. how do remittances affect poverty in rural households? or, what is the role of community forest institutions in reducing poverty?). While these approaches have highlighted some key drivers of change in forests and tree-based systems, they have been unable to shed light on the relative effects of different drivers of change or how they interact to jointly shape socio-environmental outcomes.

Second, studies have so far largely been monoconsequential (Agrawal and Chhatre, 2011). Research has focused on estimating single effects (e.g. how does market integration affect rural incomes?). While these studies have highlighted key relationships between drivers and outcomes, they have been unable to account for multiple effects, including positive and negative feedbacks, or unintended outcomes that might signal synergies, positive joint outcomes or substantial trade-offs (Dyngeland *et al.*, 2020).

Third, studies have largely been unable to account for local socio-economic contexts. This is due to: i) a lack of comparative case studies (either within or between countries) that explicitly focus on elucidating the role of local socio-economic, political, and biophysical contexts in shaping both forest-poverty dynamics and how forest-based interventions and policy instruments are implemented (Angelsen et al., 2014); ii) insufficient integration of socio-economic, political, and cultural variables into quantitative analyses (Hajjar et al., 2016); and iii) insufficient use of analytical tools to assess how local contexts shape the outcomes of forest-based interventions and policy instruments. Finally, most socio-economic and policy contexts are fluid and dynamic. Despite this, most studies of forest-poverty dynamics have been conducted at single points in time (e.g. Persha et al., 2011), thus failing to capture important shifts in the relationship between forests and poverty that play out over long time horizons. While the number of studies assessing changes over two time-points has increased in the past decade (e.g. Andam et al., 2011; Alix-Garcia et al., 2018; Oldekop et al., 2019), there is a critical dearth of studies tracking forest-poverty dynamics over multiple time points (Jung et al., 2019; see also Chapter 2).

These gaps in our knowledge hamper our abil-

ity to understand important relationships among the numerous components of social-ecological processes, and how these change over time, and the importance of social and environmental feedback loops. Many studies conclude that forest-poverty dynamics are the outcome of irreducible complexity because they have neglected to examine long-term change. We therefore recommend that future research should focus explicitly on comparisons over time, and not simply comparisons between different forest spaces. Failing to capture the temporal dimension of forest-poverty dynamics in future studies will continue to hamper our ability to identify the levers for positive change that maximise joint and lasting outcomes for forests and people.

Advances in analytical tools, and the increasing availability of social and environmental data from multiple sources (see Chapter 6) provide a potential way of bridging existing knowledge gaps. To better understand the role of socio-economic, political, and biophysical contexts, future research should place greater emphasis on longitudinal approaches, causal chains and comparative analyses. In addition to leveraging novel data sources, gaining a deeper understanding also requires closer alignment between qualitative and quantitative approaches.



Protected areas attract tourism, becoming a source of income for local communities. Photo of a landscape in the Tian Shan Mountains, Kyrgyzstan Photo © Nelson Grima

Empirical examples of drivers, mediators and moderators

We illustrate how factors discussed in the social, economic, policy, and biophysical contexts relate to each other and can take on multiple roles as drivers, mediators, and moderators of social and environmental change in forests and tree-based systems using two examples. Our examples are centred on the effects of two policy levers discussed in Chapter 5, protected areas and community forestry, and how they intersect with multiple factors.

Protected areas

Protected areas are a cornerstone of forest conservation (Watson et al., 2014). Several large-scale data-driven studies that specifically sought to control for numerous socio-economic factors, have demonstrated that protected areas can be key drivers of poverty reduction and improved well-being (Andam et al., 2010; den Braber et al., 2018; Naidoo et al., 2018, Figure 4.2a). One of the mechanisms or mediators through which protected areas reduce poverty is tourism-related infrastructure (Ferraro and Hanauer, 2014; den Braber et al., 2018), which generates opportunities for income generation. The poverty alleviation effect of protected areas is influenced or moderated by roads, which facilitate access to protected areas (Ferraro et al., 2011): the poverty alleviation effect of protected areas that are more remote and more difficult to access is lower than the poverty alleviation effect of protected areas that are more accessible. In addition to being a moderator of protected area effects, roads can also act as a driver of deforestation (Geist and Lambin, 2002; Laurance et al., 2015). This effect is mediated by various factors, including market integration which stimulates agricultural production and expansion (Oldekop et al., 2014), and facilitated access to forest frontiers (Laurance et al., 2015). In this example of protected areas, roads act both as moderators and drivers of social and environmental change.

Community forestry

Both community forestry and rural outmigration have been shown to act as drivers of poverty alleviation and reductions in deforestation (e.g. Oldekop et al., 2018; Oldekop et al., 2019; Figure 4.2b). One of the mechanisms or mediators through which outmigration affects poverty alleviation in rural households and reductions in deforestation is through remittances, which provide direct cash revenue for rural households and reduce dependence on agricultural production and forests. In the case of community forestry, one of the mediators through which poverty alleviation and reductions in deforestation are affected is through greater access to forest resources, such as timber and non-timber forest products (e.g. composting materials) that can be sold or used as inputs for agricultural production, and thus providing an incentive for long-term management. However, both drivers co-occur in time and space, and thus have the potential to interact to jointly shape poverty alleviation and deforestation outcomes. Evidence from Mexico suggests that rural outmigration also acts as mediator of community forestry. Outmigration has weakened community social capital and negatively influenced community forestry management institutions (Robson and Berkes 2011a), with potential negative environmental effects (Robson and Berkes, 2011b). Our understanding of the interactions between outmigration and community forests, or indeed how other drivers of change interact to affect social and environmental outcomes in forest landscapes, and how these interactions are influenced by broader socio-economic and biophysical contexts remains a key research frontier. A better understanding of these relationships has implications for the design of forest-based interventions that are more attuned to local, social, and environmental dynamics.

Box **4.1**

Figure 4.3



Relationship between drivers, mediators and moderators of protected areas (a), and community forestry effects (b) on poverty and forest outcomes

Forest management institutions: Examples from Africa

Analyses of the forestry sector in diverse countries in Africa show that the sector has changed considerably in recent times. New actors have emerged to fulfil unique roles in forestry. The private sector, local communities and NGOs are increasingly taking up forest management roles (FAO, 2003; Cheboiwo et al., 2018) thus expanding the democratic space for public and private sector interests. While NGOs and the private sector play a role in ensuring that forestry activities improve economic outcomes for forest dependent people, governments' facilitating role has become crucial to ensuring that policies create conditions to promote economic growth, increase incomes and minimise inequalities.

Most African countries have official forest agencies whose mandates mostly relate to protection and management of government forests, law enforcement, and advisory services. In the past few decades, reforms have focused on decentralisation and the devolution of powers, and placed increased emphasis on community participation and benefits. In Mali, Mozambique, and Uganda, national poverty eradication strategies have included public investment plans that empower local communities (Greely and Jenkins, 2000), although tangible benefits for many forests and rural communities are yet to materialise (e.g. Banana et al., 2014). In contrast, decentralisation efforts in most other African countries have not been accompanied by such efforts to build human, financial and technical capacities (Lundgren, 2015). Serious staff shortages at field operation levels (Lundgren, 2015) have severely constrained support to communities' forestry activities, and local organisations created to improve incomes feature weakly in national plans and budgets. The relationship with governments remains weak and community activities are considered only informally.

Close collaboration between the state and non-state actors championing the interests of

the poor, weak and marginalised remains an important enabling factor for realising poverty reduction in many parts of Africa. Comprising "a sphere of public life beyond the control of the state" (Colas, 2002), NGOs have an economic role to play in providing the collective goods that would otherwise be undersupplied by the private market. Their increasing role in influencing forest resources management and governance through advocacy for, and support of, community participation is prominent. Their capacity building efforts on livelihoods development facilitate the participation of poor communities in economic activities. Many NGOs have supported communities in establishing community forestry associations as locally based platforms to engage in agroforestry as a pathway out of poverty. Ghana, Somalia, and Uganda have reported positive contributions by NGOs towards improved representation of poor and marginalised communities in governance processes (Adjei et al., 2012; Dahie, 2019).

Private sector participation in forestry in Africa has increased substantially over the past few decades with substantial investments in concessions for timber and agricultural production (Brandt et al., 2016). Although large-scale land acquisitions (often termed 'land grabs') have become integral to national development strategies in Africa (and beyond), there is evidence that concessions - including those under sustainable management - are active contributors to deforestation (Brandt et al., 2016). Further, evidence on the livelihood impacts of concessions is mixed. Land acquisitions have been linked to losses in agricultural lands and livelihood displacements that particularly disadvantage women (Hajjar et al., 2020). However, other studies suggest that land acquisitions can contribute to indirect, non-subsistence job creation through increased demand for goods and services (Jung et al., 2019).

4.8 References

- Adam, J., 2010. Post-Conflict Ambon: Forced Migration and the Ethno-Territorial Effects of Customary Tenure. Development and Change 41 (3): 401-19.
- Adam, Y. O. and Eltayeb, A. M., 2016. Forestry decentralization and poverty alleviation: A review. *Forest Policy and Economics* 73: 300-307.
- Adhikari, S., Kingi, T. and Ganesh, S., 2014. Incentives for community participation in the governance and management of common property resources: the case of community forest management in Nepal. *Forest Policy and Economics* 44:1-9.
- Adjei, P. O. W., Agyemang, S. and Afriyie, K. 2012. Non-Governmental Organizations and Rural Poverty Reduction in Northern Ghana: Perspectives of Beneficiaries on Strategies, Impact and Challenges. Journal of Poverty Alleviation and International Development 3 (2): 47-73.
- Andam, K. S., Ferraro, P. J., Sims, K. R. E., Healy, A. and Holland, M., 2010. Protected areas reduced poverty in Costa Rica and Thailand. PNAS 107 (22): 9996-10001.
- African Development Bank, 2019. Towards Large-Scale Commercial Investment in African Forestry. A Study for the Climate Investment Funds Evaluation & Learning Initiative. Abidjan: African Development Bank.
- Agarwal, B., 2001. Participatory Exclusions, Community Forestry, and Gender: An Analysis for South Asia and a Conceptual Framework. *World Development* 29 (10): 1623-1648.
- Agarwal, B., 2009. Gender and forest conservation: The impact of women's participation in community forest governance. *Ecological Economics* 68(11): 2785-2799.
- Agrawal, A. and Gibson, C.C. 1999. Enchantment and Disenchantment: The Role of Community in Natural Resource Conservation. World Development 27 (4):629-649.
- Agrawal, A., Chhatre, A. and Hardin, R., 2008. Changing governance of the world's forests. *Science* 320 (5882): 1460-1462.
- Agrawal, A. and Chhatre, A., 2011. Against monoconsequentialism: Multiple outcomes and their drivers in social-ecological systems. *Global Environmental Change* 21 (3): 1-3.
- Agrawal, A., Brown, D. G. and Sullivan, J. A., 2019. Are global land grabs ticking socio-environmental bombs or just inefficient investments. *One Earth* 1 (2): 159-162.
- Alencar, A. A., Brando, P. M., Asner, G. P. and Putz, F. E., 2015. Landscape fragmentation, severe drought, and the new Amazon forest fire regime. *Ecological Applications* 25 (6): 1493-1505.

- Agrawal, A. and Ostrom, E., 2001. Collective action, property rights, and decentralization in resource management in India and Nepal. *Politics and Society* 29 (1): 485-514.
- Alix-García, J. M., McIntosh, C. and Sims, K. R. E., 2013. The ecological footprint of poverty alleviation: evidence from Mexico's Oportunidades program. *Review of Economic Statistics* 95: 417-435.
- Alix-García, J. M., Sims, K. R. E. and Yañez-Pagans, P., 2015. Only One Tree from Each Seed? Environmental Effectiveness and Poverty Alleviation in Mexico's Payments for Ecosystem Services Program. American Economic Journal: Economic Policy 7 (4): 1-40.
- Alix-García, J. M., Simes, K. R. E., Orozco-Olvera, V. H., Costica, L. E., Fernández Medina, J. D. and Romo Monroy, S., 2018. Payment for environmental services supported social capital while increasing land management. PNAS 115 (27): 7016-7021.
- Alpízar, F. and Ferraro, P. J., 2020. The environmental effects of poverty programs and the poverty effects of environmental programs: The missing RCTs. World Development 127: 104783.
- Angelo, H., de A Calderon, R., de Almeida, A.N., de Paula, M.F., Meira, M., Miguel, E.P. and Vasconcelos, P.G., 2018. Analysis of the non-timber forest products market in the Brazilian Amazon. Australian Journal of Crop Science 12 (10): 1640.
- Angelsen, A., Jagger, P., Babigumira, R., Belcher, B., Hogarth, N.J., Bauch, N., Börner, J., Smith-Hall, C. and Wunder, S., 2014. Environmental Income and Rural Livelihoods: A Global-Compartive Analysis. World Development 64: S12-S28.
- Anthias, P. and Radcliffe, S. A., 2015. The Ethno-Environmental Fix and Its Limits: Indigenous Land Titling and the Production of Not-Quite-Neoliberal Natures in Bolivia. *Geoforum* 64: 257-69.
- Antinori, C. M., 2000. Vertical Integration in Mexican Common Property Forests. PhD dissertation, University of California.
- Ascencão, F., Fahrig, L., Clevenger, A. P., Corlett, R. T., Jaeger, J. A. G., Laurance, W. F. and Pereira, H. M., 2018. Environmental challenges for the Belt and Road Initative. Nature Sustainability 1: 206-209.
- Badola, R. and Hussain, S. A., 2003. Conflict in paradise - Women and protected areas in the Indian Himalayas. Mountain Research and Development 23 (3): 234-237.
- Banana, A. Y., Namwaala J. J., Byakagaba, P. Waiswa, D., Buyinza, M. and Gombya-Ssembajwe, W., 2014. The impacts of decentralisation reforms on sustainable forest management in Central Uganda. In: Forests under pressure – Local responses to local issues, edited by P. Katila, G. Galloway, W. de Jong, P. Pacheco and G. May, IUFRO World Series Volume 32.

- Barber, C. P., Cochrane, M. A., Souza Jr., C. M., Laurance, W. M. 2014. Roads, deforestation, and the mitigating effect of protected areas in the Amazon. *Biological Conservation* 177: 203-209.
- Bardhan, P., 2002. Decentralization of governance and development. *Journal of economic perspectives* 16 (4): 185-205.
- Barrow, E., Kamugisha-Ruhombe, J., Nhantumbo, I., Oyono, R. and Savadogo, M., 2016. Who owns Africa's forests? Exploring the impacts of forest tenure reform on forest ecosystems and livelihoods. Forests, Trees and Livelihoods 25 (2): 132-156.
- Baruah, M., 2017. Facipulation and elite formation: community resource management in Southwestern Ghana. *Conservation and Society* 15 (4): 371-383.
- Bastin, J. F., Finegold, Y., Garcia, C., Mollicone, D., Rezende, M., Routh, D., Zohner, C. M., et al., 2019. The global tree restoration potential. *Science* 365 (6448): 76-79.
- Bauer, T., Ingram, V., De Jong, W. and Arts, B., 2018. The socio-economic impact of extreme precipitation and flooding on forest livelihoods: evidence from the Bolivian Amazon. International Forestry Review 20 (3): 314-330.
- Barbero, R., Abatzoglou, J. T., Larkin, N. K., Kolden, C. A. and Stocks, B., 2015. Climate change presents increased potential for very large fires in the contiguous United States. *International Journal of* Wildland Fire 24 (7): 892-899.
- Bebbington A, J., Humphreys Bebbington, D., Sauls, L. A., Rogan, J., Agrawal, S., Gamboa, C., Imhof, A., et al., 2018. Resource extraction and infrastructure threaten forest cover and community rights. PNAS 115 (52): 13164-13173.
- Beevers, M. D., 2015. Peace Resources? Governing Liberia's Forests in the Aftermath of Conflict. International Peacekeeping 22 (1): 26-42.
- Belcher, B. M., 2005. Forest product markets, forests and poverty reduction. International Forestry Review 7 (2): 82-89.
- Bijaya, G.D., Cheng, S., Xu, Z., Bhandari, J., Wang, L. and Liu, X., 2016. Community forestry and livelihood in Nepal: A review. *The Journal of Animal and Plant Sciences* 26 (1): 1-12.
- Blackman, A., Corral, L., Santos Lima, E. and Asner, G., 2017. Titling indigenous communities protects forests in the Peruvian Amazon. PNAS 114 (16): 4123-4128.
- Bob, U. and Bronkhorst, S., 2010. Environmental conflicts: Key issues and management implications. *African Journal on Conflict Resolution* 10 (2): 9-30.

- Bong, I., W., Moeliono, M., Wong, G. Y. and Brockhaus, M., 2019. What is success? Gaps and trade-offs in assessing the performance of traditional social forestry systems in Indonesia. *Forest and Society* 3 (1): 1-21.
- Brainard, L. and Chollet, D., 2007. Too poor for peace? Global poverty, conflict, and security in the 21st century. Washington D.C.: The Brookings Institution.
- Brandt, J. Nolte, C. and Agrawal, A., 2016. Deforestation and timber production in Congo after implantation of sustainable forest management policy. *Land Use Policy* 52: 15-22.
- Bray, D. B. and Merino-Pérez, L., 2002. The rise of community forestry in Mexico: History, concepts, and lessons learned from twenty-five years of community forestry timber production. Mexico City: The Ford Foundation.
- Brienen, R. J. W., Phillips, O. L., Feldpausch, T. R., Gloor, E., Baker, T. R., Lloyd, J., Lopez-Gonzalez, G., et al., 2015. Long-term decline of the Amazon carbon sink. Nature 519: 334-348.
- Brimont, L., Ezzine-de-Blas, D., Karsenty, A. and Toulon, A., 2015. Achieving conservation and equity amidst extreme poverty and climate risk: the Makira REDD+ project in Madagascar. Forests 6 (3): 748-768.
- Buvinic, M., Das Gupta, M. and Shemyakina, O. N., 2013. Armed conflict, gender, and schooling. The World Bank Economic Review 28 (2): 311-319.
- Beevers, M. D., 2015. Peace Resources? Governing Liberia's Forests in the Aftermath of Conflict. International Peacekeeping 22 (1): 26-42.
- Carvalho Ribeiro, S.M., Soares Filho, B., Costa, W.L., Bachi, L., de Oliveira, A.R., Bilotta, P., Saadi, A., et al., 2018. Can multifunctional livelihoods including recreational ecosystem services (RES) and non timber forest products (NTFP) maintain biodiverse forests in the Brazilian Amazon? Ecosystem Services 31: 517-526.
- Cheboiwo, J.K., Nasroun, T.H., Mwamakimbullah, R., Kyeyune, R.K. and Mutaganda, A., 2018. Public Private Partnership (PPP) in forest sector in Eastern Africa. Synthesis of primary and secondary production actors and trade. *Journal of Economics and Sustainable Development* 9: 43-58.
- Cheng, S.H., Ahlroth, S., Onder, S., Shyamsundar, P., Garside, R., Kristjanson, P., et al., 2017. What is the evidence for the contribution of forests to poverty alleviation? A systematic map protocol. Environmental Evidence 6: 10.
- Cheng, S.H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Garside, R., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. Environmental Evidence 8: 3.

Chomba, S. W., Nathan, I., Minang, P. A. and Sinclair, F., 2015. Illusions of empowerment? Questioning policy and practice of community forestry in Kenya. *Ecology and* Society 20 (3): 2.

Chomitz, K. M., Piet, B., Giacomo, D. L., Timothy, S. T. and Sheila, W. K., 2007. At Loggerheads? Agricultural expansion, poverty reduction and environment in the tropical forests. Washington D.C.: The World Bank.

Clapp, J. and Dauvergne, P., 2011, Paths to a Green World: The Political Economy of the Global Environment (2nd edition), Cambridge, MA: MIT Press.

Collier, P., 2007. The bottom billion. Oxford: Oxford University Press.

Colas, A., 2002. International Civil Society. Cambridge: Polity Press.

Dahie A. M., 2019. Role of nongovernmental organizations on poverty reduction: empirical study from local NGOs in Mogadishu-Somalia. *European Journal of Research in Social Sciences* 7 (1): 43-55.

Dasgupta, A. and Beard, V. A., 2007. Community driven development, collective action and elite capture in Indonesia. *Development and Change* 35 (2): 229-249.

Davis, K. F., Koo, H. I., Dell'Angelo, J., D'Odorico, P.,
 Estes, L, Kehoe, L. J., Karratzadeh, M., et al., 2020.
 Tropical forest loss enhanced by large-scale land
 acquisitions. Nature Geoscience 13: 482-488.

Davis, B., Carletto, G. and Winters, P. C., 2010. Migration, transfers, and economic decision making among agricultural households: an introduction. *The Journal* of Development Studies 46 (1): 1-13.

Davis, J. and López-Carr, D., 2014. Migration, remittances and smallholder decision-making: implications for land use and livelihood change in Central America. *Land Use Policy* 36: 319-329.

Deda, P. and Rubian, R., 2004. Women and biodiversity: The long journey from users to policy-makers. Natural Resources Forum 28 (3): 201-204.

DeFries, R. S., Rudel, T., Uriarte, M. and Hansen, M., 2010. Deforestation driven by urban population growth and agricultural trade in the twenty-first century. *Nature Geoscience* 3: 178-181.

den Braber, B., Evans K. L. and Oldekop, J. A., 2018. Impact of protected areas on poverty, extreme poverty, and inequality in Nepal. *Conservation Letters* 11: e12576.

de Haas, H., 2010. Migration and Development: A theoretical perspective. International Migration Review 44 (1): 227-264.

De Janvry, A., Finan, F., Sadoulet, E. and Vakis, R., 2006. Can conditional cash transfer programmes serve as safety nets in keeping children at school and from working when exposed to shocks? *Journal of Development Economics* 79 (2): 349–373.

de Royer, S., Van Noordwijk, M. and Roshetko, J. M., 2018. Does community-based forest management in Indonesia devolve social justice or social costs? International Forestry Review 20 (2): 167-180. Dixon, R. K., Solomon, A. M., Brown, S., Houghton, R. A., Trexier, M.C. and Wisniewski, J., 1994. Carbon pools and flux of global forest ecosystems. *Science* 263 (5144): 185-190.

Do, Q. T. and Iyer, L., 2010. Geography, poverty and conflict in Nepal. *Journal of Peace Research* 47 (6): 735-748.

Donovan, J., Stoian, D., Grouwels, S., Macqueen, D., van Leeuwen, A., Boetekees, G. and Nicholson, K., 2006. Towards an enabling environment for small and medium forest enterprise development. CATIE/FAO/IIED/SNV/ ICCO Policy Brief. Rome: FAO.

Duchelle, A.E., Kainer, K.A. and Wadt, L. H., 2014. Is certification associated with better forest management and socioeconomic benefits? A comparative analysis of three certification schemes applied to Brazil nuts in Western Amazonia. Society & Natural Resources 27 (2): 121-139.

Druilhe, Z. and Barreiro-Hurlé, J., 2012. Fertilizer subsidies in sub-Saharan Africa. ESA Working paper No. 12-04. Rome: FAO.

Drutschinin, A. and Ockenden, S., 2015. Financing for development in support of biodiversity and ecosystem services. OECD Development Co-operation Working Papers, No. 23, Paris: OECD Publishing.

Dyngeland, C., Oldekop, J. A. and Evans, K. L., 2020. Assessing multidimensional sustainability: Lesson's from Brazil's social protection programs. PNAS (in press).

Edwards, D. P., Sloan, S., Weng, L., Sayer, J., Dirks, P. and Laurance, W.F., 2014. Mining and the African environment. *Conservation Letters* 7: 302-311.

Elias, M., Jalonen, R., Fernandez, M. and Grosse, A., 2017. Gender-responsive participatory research for social learning and sustainable forest management. Forests, Trees and Livelihoods 26(1): 1-12.

Ellison, D., Morris, C.E., Locatelli, B., Sheil, D., Cohen, J., Murdiyarso, D., Gutierrez, V., et al., 2017. Trees, forests and water: Cool insights for a hot world. Global Environmental Change 43:51-61.

Erbaugh, J. T., Pradhan, N., Adams, J., Oldekop, J. A., Agrawal, A., Brockington, D., Pritchard, R. and Chhatre, A., 2020. Global forest restoration and the importance of prioritizing local communities. *Nature Ecology and Evolution* (in press).

Essougong, U. P. K., Foundjem-Tita, D. and Minang, P. A., 2019. Addressing equity in community forestry lessons from 20 years of implementation in Cameroon. Ecology and Society 24 (1): 9.

FAO, 2003. Forestry Outlook study for Africa. Regional report opportunities and challenges towards 2020. FAO Forestry Paper 141. Rome: African Development Bank, European Commission, FAO.

FAO, 2010. Gender Dimensions of Agricultural and Rural Employment: Differentiated Pathways out of Poverty. Status, Trends and Gaps. Rome: FAO. FAO, 2019. Global forest products facts and figures 2018. Rome: FAO.

Ferraro, P. J., Hanauer, M. M. and Sims, K. R. E., 2011. Conditions associated with protected area success in conservation and poverty reduction. PNAS 108 (34): 13913-13918.

Ferraro, P. J. and Hanauer, M. M., 2014a. Advances in measuring the environmental and social outcomes of environmental programs. *Annual Reviews of Environment and Resources* 39: 495-517.

Ferraro, P. J. and Hanauer, M. M., 2014b. Quantifying causal mechanisms to determine how protected areas affect poverty through changes in ecosystem services and infrastructure. PNAS 111 (11): 4332-4337.

Ferraro, P. J. and Simorangkir, R., 2020. Conditional cash transfers to alleviate poverty also reduced deforestation in Indonesia. *Science Advances* 6 (24): eaaz1298

Folke, C., Österblom, H., Jouffray, J.B., Lambin, E.F., Adger, W.N., Scheffer, M., Crona, B.I., et al., 2019. Transnational corporations and the challenge of biosphere stewardship. Nature Ecology & Evolution 3 (10): 1396-1403.

Fox, J., 2018. Community forestry, labor migration and agrarian change in a Nepali village: 1980-2010. The Journal of Peasant Studies 45(3): 610-629.

Gadgil. M. and Guha, R., 1992. This Fissured Land: An Ecological History of India. Berkeley: University of California Press.

Galaz, V., Crona, B., Dauriach, A., Jouffray, J. B., Österblom, H. and Fichtner, J., 2018. Tax havens and global environmental degradation. *Nature Ecology and Evolution* 2: 1352-1357.

Galik, C.S. and Jagger, P., 2015. Bundles, duties, and rights: A revised framework for analysis of natural resource property rights regimes. *Land Economics* 91(1): 76-90.

Gan, J., Cerutti, P. O., Masiero, M., Pettenella, D.,
Andrighetto, N. and Dawson, T., 2016. Quantifying illegal logging and related timber trade. In: Illegal Logging and Related Timber Trade: Dimensions, Drivers, Impacts and Responses: a Global Scientific Rapid Response Assessment Report, edited by D. Kleinschmit, S. Mansourian, C. Wildburger and A. Purret.
Vienna: International Union of Forest Research Organizations (IUFRO).

Garcia-Barrios, L., Galvan-Yiyoshi, Y. M., Valdivieso-Perez, I. A., Masera, O. R., Bacco, G. and Vandermeer, J. 2009. Neotropical forest conservation, agricultural intensification, and rural out-migration: The Mexico experience. BioScience 59(10): 863-873.

Garcia-Lopez, G. A., 2019. Rethinking elite persistence in neoliberalism: Foresters and techno-bureaucratic logics in Mexico's community forestry. World Development 120: 169-181. Geist, H. D. and E. F. Lambin. 2002. Proximate causes and underlying driving forces of tropical deforestation. *BioScience* 52(2): 143-150.

Godar, J., Gardner, T. A., Tizado, E. J. and Pacheco, P., 2014. Actor-specific contributions to the deforestation slowdown in the Brazilian Amazon. PNAS 111 (43): 15591-15596.

Golden Kroner, R. E., Qin, S., Cook, C. N., Krithivasan, R., Pack, S. M., Bonilla, O. D. and Cort-Kansinally, K. A., 2019. The uncertain future of protected lands and waters. Science 364 (6443): 881-886.

Giri, K. and Darnhofer, I., 2010. Outmigrating men: A window of opportunity for women's participation in community forestry? *Scandinavian Journal of Forest Research* 25 (9): 55-61.

Global Financial Integrity, 2019. Illicit Financial Flows to and from 148 Developing Countries: 2006-2015. Washington DC: Global Financial Integrity (available at: https://gfintegrity.org/report/2019-iff-update/ [accessed on: 26 November 2019]).

Greely, M. and Jenkins, R., 2000. Mainstreaming the poverty-reduction agenda. An analysis of institutional mechanisms to support pro-poor policy making and implementation in six African countries. Brighton: Institute of Development Studies, University of Sussex.

Green, J. M. H., Croft, S. A., Durán, P. D., Balmford, A. P., Burgess, N. D., Fick, S., Gardner, T. A., et al., 2019. Linking global drivers of agricultural trade to onthe-ground impacts on biodiversity. PNAS 116 (46): 23202-23208.

Griscom, B. W., Adams, J., Ellis, P. W., Houghton, R. A., Lomax, G., Miteva, D. A., Schlesinger, W. H., et al., 2017. Natural climate solutions. PNAS 114 (44): 11645-11650.

Gritten, D., Greijmans, M., Lewis, S.R., Sokchea, T.,
Atkinson, J., Quang, T.N., Poudyal, B., et al., 2015.
An uneven playing field: regulatory barriers to
communities making a living from the timber from
their forests–examples from Cambodia, Nepal and
Vietnam. Forests 6 (10): 3433-3451.

Hadley, K. S., 1994. The role of disturbance, topography, and forest structure in the development of a montane forest landscape. Bulletin of the Torrey Botanical Club 121 (1):47-61.

Hajjar R., Ayana, A. N., Rutt, R., Hinde, O. Liao, C., Keene, S., Bandiaky-Badji, S. and Agrawal, A., 2020. Capital, labor, and gender: the consequences of large-scale land transactions on household labor allocation. *Journal of Peasant Studies* 47 (3): 566-588.

Hajjar, R., Oldekop, J. A., Cronkleton, P., Etue, E., Newton, P., Russel, A. J. M., Tjajadi, J. S., et al., 2016. The data not collected on community forests. *Conservation Biology* 30 (6): 1357-1362.

Harwell, E. 2011. Forests in fragile and conflict-affected states. Washington D.C.: World Bank.

- Hayes, T. and Murtinho, F., 2018. Communal governance, equity and payment for ecosystem services. *Land Use Policy* 79: 123-136.
- Hecht, S., Yang, A. L., Basnett, B. S., Padoch, C. and Peluso, N. L., 2015. People in motion, forests in transition: Trends in migration, urbanization, and remittances, and their effect on tropical forests. Occasional Paper 142. Bogor: CIFOR
- Hemming, D. J., Chirwa, E. W., Dorward, A., Ruffhead,
 H. J., Hill, R., Osborn, J., Langer, L., et al., 2018.
 Agricultural input subsidies for improving productivity, farm income, consumer welfare and wider growth in low- and lower-middle-income countries: a systematic review. Campbell Systematic Reviews 14 (1):1-153.
- Hill, I., 1999. Forest Management in Nepal. Washington D.C.: World Bank.
- Hubbart, J. A., Guyette, R. and Muzika, R. M., 2016. More than Drought: Precipitation Variance, Excessive Wetness, Pathogens and the Future of the Western Edge of the Eastern Deciduous Forest. *Science of the Total Environment* 566: 463-467.
- Human Rights Watch, 2013. The dark side of green growth: Human rights impacts of weak governance in Indonesia's forestry sector (Available at: https://www.hrw.org/ report/2013/07/15/dark-side-green-growth/humanrights-impacts-weak-governance-indonesiasforestry#3900 [accessed on 28.10.2020]).
- Humphreys, D., 2006. Logjam: Deforestation and the Crisis of Global Governance. London: Earthscan.
- Humphries, S., Holmes, T. P., Kainer, K., Koury, C. G. G., Cruz, E. and de Miranda Rocha, R., 2012. Are community-based forest enterprises in the tropics financially viable? Case studies from the Brazilian Amazon. Ecological Economics 77: 62-73.
- Hunter, L. M., Luna, J. K. and Norton, R. M., 2015. Environmental dimensions of migration. *Annual Reviews of Sociology* 41: 377-397.
- Henders, S., Persson, U. M. and Kastner, T., 2015. Trading forests: land-use change and carbon emissions embodied in production and exports of forest-risk commodities. *Environmental Research Letters* 10: 125012.
- Hua, F., Wang, X., Zheng, X., Fischer, B., Wang, L., Zhu, J., Tang, Y., et al., 2016. Opportunities for biodiversity gains under the world's largest reforestation programme. Nature Communications 7: 12717.
- IPBES, 2013. Decision IPBES-2/4: Conceptual Framework for the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (Available at: https://www.ipbes.net/ system/tdf/downloads/Decision%20IPBES_2_4. pdf?file=1&type=node&id=14649 [accessed on 12 December 2020]).

- IPBES, 2019. Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. S. Díaz, J. Settele, E. S. Brondizio E.S., H. T. Ngo, M. Guèze, J. Agard, A. Arneth, P. Balvanera, K. A. Brauman, S. H. M. Butchart, K. M. A. Chan, L. A. Garibaldi, K. Ichii, J. Liu, S. M. Subramanian, G. F. Midgley, P. Miloslavich, Z. Molnár, D. Obura, A. Pfaff, S. Polasky, A. Purvis, J. Razzaque, B. Reyers, R. Roy Chowdhury, Y. J. Shin, I. J. Visseren-Hamakers, K. J. Willis and C. N. Zayas (eds.). Bonn: IPBES secretariat.
- Jagger, P., Luckert, M. M. K., Duchelle, A. E., Lund, J. F. and Sunderlin, W. D., 2014. Tenure and forest income: Observations from a global study on forests and poverty. World Development 64: S43-S55.
- Jacobson, P., 2018. Revealed: Paper giant's ex-staff say it used their names for secret company in Borneo. Mongabay. (Available at: https://news.mongabay.com/2018/07/ revealed-paper-giants-ex-staff-say-it-used-theirnames-for-secret-company-in-borneo/ [accessed on 28 August 2020]).
- Jung, S., Liao, C., Agrawal, A. and Brown, D.G., 2019. Evidence on Wealth-Improving Effects of Forest Concessions in Liberia. *Journal of Association of* Environmental and Resource Economists 6 (5): 961-1000.
- Kaimowitz, D., 2006. Forests and armed conflict. ETFRN News 43/44.
- Kiptot, E., Franzel, S. and Degrande, A., 2014. Gender, agroforestry and food security in Africa. *Current Opinion in Environmental Sustainability* 6: 104-109.
- Kirui, O. K., Okello, J. J., Nyikal, R. A. and Njiraina, G. W., 2013. Impact of mobile phone-based money transfer services in agriculture: Evidence from Kenya. Quarterly Journal of International Agriculture 52: 141-162.
- Kleinschmit, D., Mansourian, S., Wildburger, C. and Purret, A., 2016. Illegal Logging and Related Timber Trade: Dimensions, Drivers, Impacts and Responses: a Global Scientific Rapid Response Assessment Report. Vienna: International Union of Forest Research Organizations (IUFRO).
- Kusters, K., Achdiawan, R., Belcher, B. and Ruiz Pérez, M., 2006. Balancing development and conservation? An assessment of livelihood and environmental outcomes of non-timber forest product trade in Asia, Africa, and Latin America. Ecology and Society 11(2): 20.
- Lama, A. S., Kharel, S. and Ghale, T., 2017. When the mean are away: migration and women's participation in Nepal's community forestry. *Mountain Research and Development* 37 (3):263-270.
- Larson, A. M. and Soto, F., 2008. Decentralization of natural resource governance regimes. Annual Review of Environment and Resources 33: 213-239.

- Laurance, W.F., Peletier-Jellema, A., Geenen, B., Koster, H., Verweij, P., Van Dijck, P., Lovejoy, T. E., et al., 2015. Reducing the global environmental impacts of rapid infrastructure expansion. *Current Biology* 25 (7): R259-R262.
- Lawrence, D. and Vandecar, K., 2015. Effects of tropical deforestation on climate and agriculture. *Nature Climate Change* 5: 27-36.
- Li, T. M., 2010. Indigeneity, Capitalism, and the management of dispossession. *Current Anthropology* 51(3): 385-414.
- Li, T. M., 2014. Fixing Non-market Subjects: Governing Land and Population in the Global South. *Foucault Studies* 18: 34-48.
- Liu, J., Li, S., Ouyang, Z., Tam, C. and Chen, X., 2008. Ecological and socioeconomic effects of China's policies for ecosystem services. PNAS 105 (28): 9477-9482.
- López-Feldman, A., Mora, J. and Taylor, J. E., 2007. Does natural resource extraction mitigate poverty and inequality? Evidence from rural Mexico and a Lacandona Rainforest Community. Environment and Development Economics 12(2): 251-269.
- Lüdeke, M. K. B., Petschel-Held G. and Schnellnhuber, H. J., 2004. Syndromes of global change: The first panoramic view. *Gaia* 13 (1): 42-39.
- Lund, J. F. and Saito-Jensen, M., 2013. Revisiting the issue of elite capture of participatory initiatives. *World Development* 46: 104-112.
- Lundgren, B, 2015. Forests and trees: their roles and opportunities in Africa's development, food security and environmental health. Nairobi: African Forest Forum.
- Macqueen, D.J., Dufey, A. and Patel, B. 2006. Exploring fair trade timber: A review of issues in current practice, institutional structures and ways forward. IIED Small and Medium Forestry Enterprise Series 19. Edinburgh: IIED.
- Mansourian, S., Parrotta, J., Balaji, P., Bellwood-Howard, I., Bhasme, S., Bixler, R.P., Boedhihartono, A.K., et al., 2020. Putting the pieces together: integration for forest landscape restoration implementation. Land Degradation & Development 31(4): 419-429.
- Mbiti, I. and Weil, D. N., 2016. Mobile banking: The case of M-Pesa in Kenya. In: African Successes, Volume III: Modernization and Development, edited by S. Edwards, S. Johnson and D. N. Weil. Chicago: University of Chicago Press. 247-293.
- McCarthy, D.P., Donald, P.F., Scharlemann, J.P., Buchanan, G.M., Balmford, A., Green, J.M., Bennun, L.A., et al., 2012. Financial costs of meeting global biodiversity conservation targets: current spending and unmet needs. Science 338 (6109): 946-949.
- McElwee, P., 2009. Reforesting "bare hills" in Vietnam: Social and environmental consequences of the 5 million hectare reforestation program. *Ambio* 38 (6): 325-333.

- MEA, 2005. Millennium Ecosystem Assessment, Ecosystems and Human Wellbeing: A Framework for Assessment. Washington D.C.: Island Press.
- Messina, J. P., Walsh, S. P., Mena, C. F. and Delamater, P. L., 2006. Land tenure and deforestation patterns in the Ecuadorian Amazon: Conflicts in land conservation in frontier settings. *Applied Geography* 26 (2): 113-128.
- Meyfroidt, P., Lambin, E.F., Erb, K.H. and Hertel, T.W., 2013. Globalization of land use: distant drivers of land change and geographic displacement of land use. *Current Opinion in Environmental Sustainability* 5 (5): 438-444.
- Meyfroidt, P., Chowdhury, R.R., de Bremond, A., Ellis, E.C., Erb, K.H., Filatova, T., Garrett, R.D., et al., 2018. Middle-range theories of land system change. Global Environmental Change 53: 52-67.
- Millar, C. I. and Stephenson, N. L., 2015. Temperate forest health in an era of emerging megadisturbance. *Science* 349 (6250): 823-826.
- Miller, D. C. and Hajjar, R. 2020. Forests as pathways to prosperity: Empirical insights and conceptual advances. World Development 125: 104647.
- Miller, D. C., Rana, P., Nakamura, K., Forrest, S., Cheng, S. H., Ahlroth, S. E. and Perge, E. B., 2019. A Systematic Review of the Impact of Forest Property Rights Interventions on Poverty. Washington DC: World Bank.
- Mogoi, J., Obonyo, E., Ongugo, P., Oeba, V. and Mwangi, E., 2012. Communities, property rights and forest decentralisation in Kenya: Early lessons from participatory forestry management. *Conservation and* Society 10 (2): 182.
- Moran, E. F., 1990. Public and private colonization schemes in Amazonia. In: *The Future of Amazonia*, edited by D. Goodman and A. Hall. London: Palgrave McMillan. 70-89.
- Mudege, N. N., Mdege, N., Abidin, P. E. and Bhatasara, S., 2017. The role of gender norms in access to agricultural training in Chikwawa and Phalombe, Malawi. Gender, Place & Culture 24 (12): 1689-1710.
- Mukherjee, P., Ray, B. and Bhattacharya, R., 2017. Status differences in collective action and forest benefits: evidence from joint forest management in India. Environment, Development and Sustainability 19: 1831-1854.
- Münster, U., 2014. Invisible Labor: Adivasi Workers in the History of South Indian Forest Conservation.
 In: Asian Environments: Connections across Borders, Landscapes, and Times, edited by U. Münster, S.
 Satsuka and G. Cederlöf, RCC Perspectives, no. 3, pp. 53-58.
- Muscarella, R., Kolyaie, S., Morton, D.C., Zimmerman, J.K. and Uriarte, M., 2020. Effects of topography on tropical forest structure depend on climate context. *Journal of Ecology* 108 (1): 145-159.

Mutune, J. M., Hansen, C. P., Wahome, R. G. and Mungai, D. N., 2017. What rights and benefits? The implementation of participatory forest management in Kenya: The case of Eastern Mau Forest Reserve. *Journal of Sustainable Forestry* 36 (3): 230-249.

Naidoo, R., Gerkey, D., Hole, D., Pfaff, A., Ellis, A. M., Golden, C. D. and Herrera, D., 2019. Evaluating impacts of protected areas on human well-being across the developing world. *Science Advances* 5 (4): eaav3006.

Namubiru-Mwaura, E., 2014. Land Tenure and Gender: Approaches and Challenges for Strengthening Rural Women's Land Rights. Washington DC: World Bank.

Nellemann, C., Henriksen, R., Raxter, P., Ash, N. and Mrema, N., 2014. The Environmental Crime Crisis: Threats to Sustainable Development from Illegal Exploitation and Trade in Wildlife and Forest Resources. Nairobi: UNEP GRID-Arendal.

Nelson, V., Tallontire, A. and Collinson, C., 2002. Assessing the benefits of ethical trade schemes for forest dependent people: comparative experience from Peru and Ecuador. *International Forestry Review* 4 (2): 99-109.

North D. C., 1990. Institutions, institutional change and economic performance. London, New York and Melbourne: Cambridge University Press.

Oldekop, J. A., Bebbington, A. J., Hennermann, K., McMorrow, J., Springate, D. A., Torres, B. and Preziosi, R. F., 2013. Evaluating the effects of common-pool resource institutions and market forces on species richness and forest cover in Ecuadorian indigenous Kichwa communities. *Conservation Letters* 6 (2): 107-115.

Oldekop, J. A., Sims, K. R. E., Whittingham, M. J. and Agrawal, A., 2018. An upside to Globalization: International migration drives reforestation in Nepal. Global Environmental Change 52: 66-74.

Oldekop, J. A., Sims, K. R., Karna, B. K., Whittingham, M. J. and Agrawal, A., 2019. Reductions in deforestation and poverty from decentralized forest management in Nepal. Nature Sustainability 2 (5): 421.

Oldekop, J. A., Vang Rasmussen, L., Agrawal, A., Bebbington, A. J., Meyfroidt, P., Bengston, D., Blackman, A., *et al.*, 2020. Forest-linked livelihoods in a globalized world (*in review*).

Ordway, E. M., 2015. Political shifts and changing forests: Effects of armed conflict on forest conservation in Rwanda. *Global Ecology and Conservation* 3: 448-460.

Ostrom, E. 2009. A general framework for analyzing sustainability of social-ecological systems. *Science* 325 (5939): 419-422.

Pacheco, P., 2012. Smallholders and communities in timber markets: Conditions shaping diverse forms of engagement in tropical Latin America. *Conservation* and Society 10 (2): 114-123. Pacheco, P., Cerutti, P. O., Edwards, D. P., Lescuyer, G., Mejia, E., Navarro, G., Obidzinski, K., et al., 2016.
Multiple and intertwined impacts of illegal forest activities. In: Illegal Logging and Related Timber Trade: Dimensions, Drivers, Impacts and Responses: a Global Scientific Rapid Response Assessment Report, edited by D. Kleinschmit, S. Mansourian, C. Wildburger and A. Purret. Vienna: International Union of Forest Research Organizations (IUFRO).

Pan, Y., Birdsey, R. A., Fang, J. Houghton, R., Kauppi, P. E., Kurz, W. A., Phillips, O. L., et al., 2011. A large and persistent carbon sink in the world's forests. *Science* 333(6045): 988-993.

Pellegrini, L., 2011. Forest management and poverty in Bolivia, Honduras and Nicaragua: Reform failures? European Journal of Development Research, 23(2): 266-283.

Peng, H., Cheng, G., Xu, Z., Yin, Y. and Xu, W., 2007. Social, economic, and ecological impacts of the "Grain for Green" project in China: a preliminary case in Zhangye, Northwest China. Journal of Environmental Management 85 (3): 774-784.

Pérez, J. J. N. and Garzón, C. A. D., 2015. Inversión en el Posconflicto: Fortalecimiento institucional y reconstrucción del capital social. Revista de Relaciones Internacionales, Estrategia y Seguridad 10 (1): 203-224.

Persha, L. and Andersson, K., 2014. Elite capture risk and mitigation in decentralized forest governance regimes. *Global Environmental Change* 24: 265-276.

Persha, L., Agrawal, A. and Chhatre, A., 2011. Social and ecological synergy: local rule-making, forest livelihoods, and biodiversity conservation. *Science* 331 (6024): 1606-1608.

Phelps, J., Webb, E. L. and Agrawal, A,. 2010. Does REDD+ threaten to recentralize forest governance. *Science* 328 (5976): 312-313.

Piao, S., Fang, J., Zhou, L., Ciais, P. and Zhu, B., 2006. Variations in satellite-derived phenology in China's temperate vegetation. *Global Change Biology* 12 (4):672-685.

Piperata, B. A., Spence, J. E., Da-Gloria, P., Hubbe, M., 2011. The nutrition transition in Amazonia: rapid economic change and its impacts on growth and development in Ribeirinhos. *American Journal of Physical Anthropology* 146 (1): 1-13.

Platteau, J. P., 2004. Monitoring elite capture in community-driven development. *Development and Change* 35 (2): 223-246.

Potts, J., 2018. State of Sustainability Initiatives Review: Standards and the Extractive Economy. London: International Institute for Sustainable Development (IISD).

Poudel, D. P., 2019. Migration, forest management and traditional institutions: Acceptance of and resistance to community forestry models in Nepal. *Geoforum* 106: 275-286.

- Prateek, G., Knopf, R. C. and Aggarwal, R. M., 2019. Impact of out-migration of men on women's participation in community-based forestry: insights from Van Panchayats of Uttarakhand, India. International Forestry Review 21 (2): 212-224.
- Pulhin, J.M. and Ramirez, M.A.M., 2016. Timber regulation and value chain in community-based timber enterprise and smallholder forestry in the Philippines. Forests 7 (8): 152.
- Putz, F. E., Zuidema, P.A., Synnott, T., Peña-Claros, M., Pinard, M.A., Sheil, D., Vanclay, J.K., et al., 2012.
 Sustaining conservation values in selectively logged tropical forests: the attained and the attainable. Conservation Letters 5 (4): 296-303.
- Rahut, D. B., Ali, A. and Behera, B., 2015. Household participation and effects of community forest management on income and poverty levels: Empirical evidence from Bhutan. Forest Policy and Economics 61: 20-29.
- Reboredo, F., 2013. Socio-economic, environmental and governance impacts of illegal logging. Environment Systems and Decisions 33: 2950-394.
- Ribot, J. C., 2002. Democratic decentralization of natural resources: institutionalizing popular participation. Washington DC: World Resources Institute.
- Robson, J. P. and Berkes, F., 2011a. How does outmigration affect community institutions? A study of two indigenous municipalities in Oaxaca, Mexico. *Human Ecology* 39 (2): 179-190.
- Robson, J. P., Berkes, F., 2011b. Exploring some myths of land use change: Can rural to urban migration drive declines in biodiversity? *Global Environmental Change* 21 (3): 844-854.
- Robson, J. P. and Nayak, P. K., 2010. Rural out-migration and resource-dependent communities in Mexico and India. *Population and Environment* 32: 263-284.
- RRI. 2017. Tenure Data Tool. Washington DC: Rights and Resources Initiative.
- Rudel, T. K., DeFries, R. Asner, G. P., Laurance, W. F. 2009. Changing drivers of deforestation, and new opportunities for conservation. *Conservation Biology* 23 (6): 1396-1405.
- Samii, C., Lisiecki, M., Kulkarni, P., Paler, L. and Chavis, L., 2015. Decentralised forest management for reducing deforestation and poverty in low-and middle-income countries. Systematic Review 16. London: International Initiative for Impact Evaluation.
- Sánchez-Cuervo, A. M. and Aide, T. M., 2013. Consequences of the armed conflict, forced human displacement, and land abandonment on forest cover change in Colombia: A multi-scaled analysis. *Ecosystems* 16: 1052-1070.
- Schlager, E. and Ostrom, E., 1992. Property-rights regimes and natural resources: a conceptual analysis. *Land economics* 68 (3): 249-262.

- Schleicher, J., Peres, C. A., Amano, T., Llactayo, W., Leader-Williams, N., 2017. Conservation performance of different conservation governance regimes in the Peruvian Amazon. Scientific Reports 7 (1): 1-10.
- Shah, Alpa 2010. In the Shadows of the State indigenous politics, environmentalism, and insurgency in Jharkhand, India. Durham and London: Duke University Press.
- Shaw, R. P. 1989. Rapid Population Growth and Environmental Degradation: Ultimate versus Proximate Factors. Environmental Conservation 16 (3): 199-208.
- Sikor, T., He, J. and Lestrelin, G., 2017. Property rights regimes and natural resources: a conceptual analysis revisited. World Development 93: 337-349.
- Smirnov, D.Y., Kabanets, A.G., Milakovsky, B.J., Lepeshkin, E.A. and Sychikov, D.V., 2013. Illegal Logging in the Russian FarEast: Global Demand and Taiga Destruction. Moscow: WWF Russia
- Somanathan, E., Prabhakar, R. and Mehta, B. S. 2009. Decentralization for cost-effective conservation. PNAS 106 (11): 4143-4147.
- Song, C., Bilsborrow, R., Jagger, P., Zhang, Q., Chen, X. and Huang, Q., 2018. Rural household energy use and its determinants in China: how important are influences of payment for ecosystem services vs. other factors? *Ecological Economics* 145: 148-159
- Song, C., Zhang, Y., Mei, Y., Liu, H., Zhang, Z., Zhang, Q., Zha, T., et al., 2014. Forests Sustainability and China's Sloping Land Conversion Program (SLCP): A Comparison among Three Sites in Anhui, Hubei and Shanxi. Forest Policy and Economics 38: 161-167.
- Stoian, D., Rodas, A., Butler, M., Monterroso, I. and Hodgdon, B., 2018. Forest concessions in Petén, Guatemala: A systematic analysis of the socioeconomic performance of community enterprises in the Maya Biosphere Reserve. Bogor: CIFOR.
- Southgate, D., Salazar-Canelos, P., Camacho-Saa, C. and Stewart, R., 2000. Markets, institutions, and forestry: the consequences of timber trade liberalization in Ecuador. World Development 28 (11): 2005-2012.
- Sunam, R. K. and McCarthy, J. F., 2016. Reconsidering the links between poverty, international labour migration, and agrarian change: critical insights from Nepal. *The Journal of Peasant Studies* 43 (1): 39-63.
- Sunam, R. K., 2017. In search of pathways out of poverty: Mapping the role of international labour migration, agricultural and rural labour. *Journal of Agrarian Change* 17 (1): 67-80.
- Sunderlin, W.D., Angelsen, A., Belcher, B., Burgers, P., Nasi, R., Santoso, L. and Wunder, S., 2005. Livelihoods, forests, and conservation in developing countries: An overview. World Development 33 (9): 1383-1402.
- Sündstrom, A., 2016. Understanding illegality and corruption in forest governance. *Journal of* Environmental Managament 181: 779-790.

- Taylor, M. J., Aguilar-Støen, M., Castellanos, E., Moran-Taylor, M. J. and Gerkin, K., 2016. International migration, land use change and the environment in Ixcán, Guatemala Land Use Policy 54: 290-301.
- Tacconi, L., Cerutti, P.O., Leipold, S., Rodrigues, R.J., Savaresi, A., Phuc, T. and Weng, X., 2016. Defining Illegal Forest Activities and Illegal Logging. In: Illegal Logging and Related Timber Trade: Dimensions, Drivers, Impacts and Responses: A Global Scientific Rapid Response Assessment Report, edited by D. Kleinschmit, S., Mansourian, C., Wildburger and A. Purret. Vienna: International Union of Forest Research Organizations (IUFRO).
- Tan-Soo, J. S. and Pattanayak, S. K., 2019. Seeking natural capital projects: Forest fires, haze, and early life exposure in Indonesia. PNAS 116 (12): 5239-5245.
- Tiriviyi, N., 2017. Social protection for building resilience of forest-dependent people: Evidence, linkages, practices, and potential applications. Social Protection and Forestry Working Paper 1. Rome: FAO.
- To, P. X., Dressler, W., Mahanty, S., Thuy, P. T. and Zingerli, C., 2012. The prospect for Payment for Ecosystem Services (PES) in Vietnam: A look at three payment schemes. Human Ecology 40 (2): 237-249.
- Trumbore, S., Brando, P. and Hartmann, H., 2015. Forest health and global change. *Science* 349 (5250): 814-818.
- Tubbeh, R. M. and Zimmerer, K. S., 2019. Unraveling the Ethnoterritorial Fix in the Peruvian Amazon: Indigenous Livelihoods and Resource Management after Communal Land Titling (1980s-2016). Journal of Latin American Geography 18 (2): 33-59.
- UNECE/FAO, 2019. Forest Products Annual Market Review, 2018-2019. Geneva and Rome: United Nations Economic Commission for Europe (UNECE) / Food and Agricultural Organization of the United Nations (FAO).
- Urrunaga, J. M., Johnson, A., Dhaynee Orbegozo, I. and Mulligan, F., 2012. The Laundering Machine: How Fraud and Corruption in Peru's Concession System are Destroying the Future of Its Forests. Washington D.C. and London: Environmental Investigation Agency.
- van Mantgem, P. J., Stephenson, N. L., Byrne, J. C., Daniels, L. D., Franklin, J. F., Fulé, P. Z., Harmon, M. E., et al., 2009. Widespread increase tree mortality rates in the western United States. Science 323 (5913): 521-524.
- Vang Rasmussen, L., Coolsaet, B., Martin, A., Mertz, O., Pascual, U., Corbera, E., Dawson, N., Fisher, J. A., et al., 2018. Social-ecological outcomes of agricultural intensification. Nature Sustainability 1: 275-282.
- VanWey, L. K., Guedes, G. R. and D'Antona. Á. O., 2012. Out-migration and land-use change in agricultural frontiers: insights from Altamira settlement project. *Population and Environment* 34: 44-68.
- Walker, A., 2004. Seeing farmers for the trees: Community forestry and the arborealisation of agriculture in northern Thailand. Asia Pacific Viewpoint 45 (3): 311-324

- Watson, J. E. M., Dudley, N., Segan, D. B. and Hockings, M., 2014. The performance and potential of protected areas. *Nature* 515: 67-73.
- Watson, J. E. M., Evans, T., Venter, O., Williams, B., Tulloch, A., Stewart, C., Thompson, I., et al., 2018. The exceptional value of intact forest ecosystems. Nature ecology & evolution 2(4): 599-610.
- Weiss, G., Emery, M.R., Corradini, G. and Živojinović, I., 2020. New Values of Non-Wood Forest Products. Forests 11 (2): 165.
- Wiersum, K.F., Humphries, S. and Van Bommel, S., 2013. Certification of community forestry enterprises: experiences with incorporating community forestry in a global system for forest governance. *Small-scale Forestry* 12 (1): 15-31.
- World Bank, 2020. Indicators Open Data. Available at: https://data.worldbank.org/indicator [accessed on: 28 August 2020].
- Xu, L., Samanta, A., Costa, M. H., Ganguly, S. Nemani, R. R. and Myneni, R. B., 2011. Widespread decline in greenness of Amazonian vegetation due to the 2010 drought. *Geophysical Research Letters* 38 (7): L07402.
- Zhang, Q., Bilsborrow, R. E., Song, C., Tao, S. and Huang, Q., 2019. Rural household income distribution and inequality in China: Effects of payments for ecosystem services policies and other factors. *Ecological Economics* 160: 114-127
- Zhou, L., Tian, Y., Myneni, R. B., Ciais, P., Saatchi, S., Liu, Y. Y., Piao., S., et al., 2014. Widespread decline of Congo rainforest greenness in the past decade. *Nature* 509: 86-90.



Chapter 5

Levers for Alleviating Poverty in Forests and Tree-Based Systems

Coordinating Lead Authors: Reem Hajjar and Peter Newton Lead Authors: Markus Ihalainen, Arun Agrawal and Mónica Gabay Contributing Authors: Jennifer Alix-Garcia, Sarah E. Brown, James T. Erbaugh, Karl Hughes, Samuel Mawutor, Pablo Pacheco, George Schoneveld, Conghe Song and Joleen Timko

TABLE OF CONTENTS

5.1	Introduction	.126
5.2	Overview of the Levers	. 127
5.3	Rights-based Levers	.133
5.4	Regulatory Levers	. 138
5.5	Market and Supply Chain Levers	.141
5.6	Forest and Tree Management Levers	. 153
5.7	Summary of Key Findings	. 156
5.8	Knowledge Gaps and Future Research	. 158
5.9	Conclusions	. 158
5.10	References	. 160

Abstract

An extensive set of policies, programmes, technologies and strategies have been implemented in the forest sector. Collectively, these 'levers' cover a diverse range of approaches, at a variety of scales and are governed by many different stakeholders. It is important for decision-makers to understand which levers might be most useful in achieving poverty alleviation. This chapter seeks to answer the question: which forest management policies, programmes, technologies and strategies have been effective at alleviating poverty? We studied 21 different rights-based, regulatory, market and supply chain, and forest and tree management levers for which we could identify a plausible theory of change of how implementation of that lever might alleviate poverty. For every lever we: define and describe the lever; describe the logic or theory of change by which the lever might plausibly be expected to alleviate poverty; summarise the available evidence showing how the lever has alleviated poverty; and discuss the variables that explain heterogeneity in outcomes. Overall, we found substantial, varied and context-dependent evidence of these levers being associated with mitigating poverty (i.e., by improving well-being). We found limited evidence of these levers being associated with reducing poverty (i.e., moving people out of poverty). Some of the strongest evidence for poverty reduction came from ecotourism, community forest management, agroforestry and, to a lesser extent, payments for ecosystem services (PES). A multitude of cases showing positive outcomes for poverty mitigation came from community forest management, forest producer organisations, small and medium forest enterprises, PES, and tree crop contract production. A combination of more rigorous and long-term research designs, along with examinations of the cost-effectiveness of different levers, would go a long way to contributing to the design of effective interventions for poverty alleviation.

5.1 Introduction

Governments, donors, international organisations, companies and communities have used a diverse array of policies, programmes, technologies and strategies in the forest⁷ sector to reduce poverty (or at least, improve human well-being) as the primary or secondary goal. These 'levers' include regulatory and voluntary strategies, are implemented at a variety of scales, and affect and are governed by many different stakeholders. Many levers affect poverty and well-being in conjunction with other aims: to reduce deforestation, conserve biodiversity or reduce greenhouse gas emissions. The purpose of this chapter is to assess how and to what extent different forest sector policies, programmes, technologies and strategies (hereafter 'levers') have been effective in alleviating poverty. This chapter identifies forest-sector levers that could plausibly alleviate poverty, and evaluates the strength of available evidence for the effect that each lever has had on reducing poverty (moving people above a certain threshold of income or consumption) and mitigating poverty (lessening deprivation or disadvantage such that well-being is improved). That is, we focus on two roles that forests and treebased systems play in poverty alleviation as identified in Chapter 2: 1. movement out of poverty and 2. supporting well-being. To a certain extent, we also consider where a lever may have led to an increase in poverty. We identified 21 levers, and reviewed them individually.

To identify the key levers, firstly Coordinating Lead Authors and Lead Authors for the chapter brainstormed the full range of possible levers. All other members of the Global Forest Expert Panel on Forests and Poverty subsequently reviewed the list to suggest any additional levers that were missed during the first step. No new levers emerged. While this provides some confidence in the robustness of the initial list, we cannot be certain that some relevant levers escaped our initial search and scan.

We selected levers for assessment and analysis if they met two criteria. First, the lever had to be clearly related to forests and/or trees within a wider landscape as per the remit of this report: that is, they had to specifically address the management, use, conservation or *restoration* of forests or trees. Levers that were principally related to the agricultural sector or to other landscapes were not considered, even if they in principle could affect the poverty of people living in or around forests. For example, certification programmes that target agricultural products may affect people living in and around forests, but were not included. Second, the lever had to have some plausible expectation of alleviating poverty, even when alleviating poverty was not its primary purpose. This was interpreted broadly, and included any lever with an identifiable theory of change supporting the provision of one or more socio-economic benefits from forest products and services.

5.2 Overview of the Levers

The chapter considers four main categories of levers: 1) rights-based levers; 2) other regulatory levers; 3) market and supply chain levers; and 4) forest and tree management levers (Table 5.1), acknowledging that alternative taxonomies of levers relevant to this review have been developed by others (e.g. Newton et al., 2013; Agrawal et al., 2018). We also acknowledge that some levers could fall into multiple categories (e.g. community forest management as an intervention often combines aspects of rights-based and regulatory reforms while engaging in markets and introducing new forest management practices). The taxonomic division of levers into different categories would only become pertinent if one were trying to understand whether, for example, regulatory levers were more or less effective than market and supply chain levers as an aggregate category, or if rights-based levers as a whole might be more appropriate than regulatory levers for particular country contexts. We discourage use of this review to try to extract such high-level conclusions. For every lever we: define and describe the lever; describe the logic or theory of change by which the lever might plausibly be expected to alleviate poverty; summarise the available evidence showing how the lever has alleviated poverty (e.g. by increasing income, assets or well-being) and, where available, the magnitude of those changes; and discuss the variables that explain heterogeneity in outcomes. Some of this information is summarised in Table 5.1.

5.2.1 Caveats

There are some important caveats to note. First, drawing lines between different interventions was sometimes partially arbitrary. There is considerable overlap between some of the levers. For example, third-party certification can be considered as a form of supply chain initiative, REDD+ can be conceived as a particular type of *payment for ecosystem services* (PES) programme, community forest management (CFM) can emerge through tenure reform, and small forest management enterprises (SMFEs) can include ecotourism. As such, separating literature, and consequently the effects of these levers on poverty, into discrete categories is somewhat interpretative. It seems at least conceivable that two or more of these levers in tandem could have greater impacts on poverty than any one of them alone. We did not explore such multiplicative interactions, except to the degree that any of the literature did so by virtue of the cases or sites that they studied.

Second, many of the reviewed levers were implemented based on multiple objectives and a win-win logic: improving both conservation and well-being outcomes. In this review, we have not taken into consideration poverty outcomes in relation to other potential programmatic or policy objectives. Thus, while the levers presented may not have been the most impactful or cost-effective from a *poverty reduction* perspective, they should not be discounted as they may have had multiple positive outcomes in other realms.

Finally, it bears repeating that we did not conduct a systematic review of all available literature on each of these levers. As such, some relevant evidence may have been missed.



Wood processing plant in the state of Oaxaca, Mexico Photo © Reem Hajjar

Table 5.1 Forest-sector levers that may alleviate poverty	ORY OF CHANGE SUMMARY OF THE EVIDENCE: SUMMARY OF THE EVIDENCE: CONCLUSIONS		re access to land and forest resources is often seen as a Mostly case studies and some quasi-experimental studies. A systematic review of forest property reforms found setep for forest-reliant poor to be able to reliably benefit, and non-monetarily, from forests. Very limited assessments of tree tenure reform on poverty rights. Very limited assessments of tree tenure reform on poverty is enhanced by the presence of a number of enabling conditional intervention levers discussed in this chapter. Social differentiation in the tenure reform in the reform in th	cognising the rights of local user groups to common tresources, it is expected that the users will benefit that indirectly from forest products and services for the undirectly from forest products and services for the nost countries. Rigorous national-level analyses have shown that CFM has reduced poverty or provided analytical gaps prevent generalisable conclusions about observed socio-economic and environmental outcomes of community forest management.	ral governments or forest departments provideNational-level statistics on concessions' contributions to panies and communities with forest resource (typically er) extraction rights in commercially valuable forests in ange for a stream of revenues. Besides stumpage or taxes to governments, concession agreements often include sions for local public goods such as employment, number of case studies.National-level statistics on concessions' contributions to concessions are available, but contributions of Cameroon and Liberia. In Gabon, NTFPs from forest concessions provide limited evidence of their contributions to povertry reduction even as they generate substantial benefits and profits for large logging companies.ral governments of sions for local sions for local bing and healthcare.National incomes are available, but contributions of Cameroon and Liberia. In Gabon, NTFPs from forest concessions minimally affected livelihoods. Case studies in general only provide limited evidence of their contributions to povertry reduction even as they generate substantial benefits and profits for large logging companies.	can support livelihoods by securing rights of people to Several national-level studies, using quasi-experimental Several studies show that PAs can reduce poverty, tands, supplying ecosystem services, generating income quantitative methods. Few multi-national, quantitative particularly where ecotourism opportunities exist (e.g. in tourism opportunities and improving rural infrastructure. studies.
	THEORY OF CHANGE	ed levers	Secure access to land and forest first step for forest-reliant poor to monetarily and non-monetarily, f	In recognising the rights of local forest resources, it is expected the directly and indirectly from forest subsistence and commercial pur	Central governments or forest de companies and communities wit timber) extraction rights in comn exchange for a stream of revenu paid to governments, concession provisions for local public goods provisions and healthcare.	as PAs can support livelihoods by so forest lands, supplying ecosysten from tourism opportunities and i
	LEVER	Rights-base	Tenure reform	Community for management interventions	Forest concessions	Protected area (PAs)

Regulatory leve	ſS		
Decriminalisation and formalisation of informal workers	Formalisation can: allow the poor to convert their possessions and labour into capital, which can in turn be used to generate added value (e.g. through accessing credit); enhance protection of rights; encourage productive investments; fetch higher prices for products; and minimise risks from forest law enforcement.	A few studies (mostly case studies) in the forest sector in the tropics have focused explicitly on the relationship between formalisation and poverty alleviation.	Mixed results, as formalisation alone does not guarantee success of enterprises. Some formalisation efforts have further marginalised poor small-scale workers; others have improved access to credit and markets, and have supported social projects.
Modifying/simpli- fying regulatory frameworks, including management plans	Overly burdensome regulations keep the forest-reliant poor from engaging in formal forestry sector. Simplified management plans can make it easier for them to engage and benefit from formal activities.	A few case studies in Latin America and Africa have examined the effects of simplified forest management plans on poverty, but none have attempted to empirically disentangle the effects of simplified management plans on poverty from the effects of other factors (such as tenure reform, market access and other barriers to SMFEs).	Mixed results. One study found that simplified manage- ment plans brought financial benefits to some communi- ties, but did not compare the effects of simplified plans relative to non-simplified plans. Many studies continue to point to the difficulties associated with overly bureaucratic and technical processes to participate in the formal sector.
Log export bans (LEBs)	Log export bans are put in place to enhance domestic forest industries and thus domestic employment.	Empirical studies and economic models have examined the effects of LEBs on domestic processing and employment. One study specifically modelled effects on households in poverty.	Empirical and economic models have found no evidence that LEBs target the poor, or increase overall employment in the country. One model indicated that a LEB in Indonesia would result in decreased incomes across agricultural and rural households.
Procurement policies	Sourcing legal timber in international trade has resulted in bilateral trade agreements that have pushed for domestic governance reform – an opportunity for pro-poor policy reforms. Domestic procurement policies can also favour small scale or community-owned forest businesses.	No studies have effectively traced the effects of bilateral or international agreements on poverty reduction. We found one case study of a domestic procurement policy enhancing community forestry.	Authors have pointed to negative effects of international procurement policies affecting small-scale producers. There are cases showing improved small-scale production with domestic procurement policies that purchase from community forests, but specific links to poverty were not examined.

iary of the evidence: Lusions		ositive impacts on household incomes or assets. dy found a small but significant decrease in in a PES programme in Mexico. Evidence of contributions to food security. Where there is income variation, timing payments to the moment icomes are lowest may generate important impacts stty.	nprehensive reviews showed small or insignificant contributions to income across cases. Case studies ed, with many showing small increases in incomes t in the short term), and others showing increasing ties in communities following REDD+. One operimental study found negative effects on ng. Local tenure security was enhanced in many	udies have focused on measures of development rerty-related impacts in terms of generation of local d incomes. A number of studies have examined ects of ecotourism and point to positive outcomes on to livelihoods, socio-economic development, rerty reduction. Many studies suggest that those is better off will be more likely to benefit, exacerbat- l income inequalities.	to isolate evidence of impacts of the presence of as a number of other levers are relevant to creating opriate enabling environment for SMFEs to thrive.	nd context-dependent evidence that this lever poverty, due to the number of additional factors at producers' ability to make use of enhanced market
SUMIN		Small po One stu poverty positive annual i when in on pove	Two cor REDD+ are mixe (at least inequali quasi-ex well-bei cases.	Case stu and pov jobs and local eff in relation who are ing local	Difficult SMFEs, an appr	Mixed a reduces play in p access.
SUMMARY OF THE EVIDENCE: QUANTITY AND TYPE OF STUDIES		A number of large-scale, rigorously designed studies.	No comprehensive and rigorous assessments of the effects of REDD+ on poverty but several localised case studies, comparative analyses and reviews.	Estimates of its economic contributions to national economies and some local communities are available, but not specific to poorer groups. Evidence tracked through number of visitors and their effects on local and national economies. Thousands of case studies at the local level.	Many case studies showing their positive contributions to local prosperity but few impact assessments linking SMFEs directly to poverty reduction.	Primarily case studies.
THEORY OF CHANGE	ply-chain levers	Payments for ecosystem services are expected to either have no impact on poverty (if they exactly compensate for lost profits from forgone environmental activities) or increase incomes.	REDD+ initiatives provide monetary compensation in exchange for reductions in terrestrial emissions through fostering conservation, sustainable management of forests and enhancement of forest carbon stocks. REDD+ initiatives may influence poverty via two pathways: 1) through REDD+ payments to governments and local bodies and benefit sharing of such payments, and 2) as a result of changes in forest benefits to local users and governments as they limit use of forests to conform to REDD+ objectives.	Ecotourism can contribute to poverty reduction in four different ways: 1) improvements in employment and wages of those who find employment, 2) visitor fees for forested locations in protected areas, 3) revenues from visitor purchases of local goods and services, and 4) infrastructure development with spillover effects in areas with high numbers of travellers and visitors.	SMFEs generate local employment opportunities in rural areas and spread wealth locally.	Enhanced market participation can lead to positive impacts on household income and poverty alleviation.
LEVER	Market and supp	Payments for Ecosystem Services (PES) programmes	Reducing Emissions from Deforestation and forest Degrada- tion (REDD+)	Ecotourism	Small & medium forest enterprises (SMFEs)	Market access

Several studies show that producers who were members of a larger organisation or cooperative had higher incomes than non-members.	Mixed results. Some partnerships resulted in increased incomes and employment and improvement to social infrastructure. Other, poorly negotiated contracts, resulted in greater inequities, dependency and other negative effects.	Considerable evidence of positive effects with agricultural tree crops. Less evidence with respect to timber or NTFPs. Some evidence that contract production can exacerbate social differentiation.	There is no robust evidence that certification has reduced poverty, particularly because of the difficulties that small-scale producers have in acquiring and maintaining certification. Some evidence of improved income and well-being from cacao certification.	We found no evidence that supply-chain commitments have reduced poverty or improved human well-being.	Some suggestion that boycotts led to wider adoption of FSC certification. Impacts on poverty depend on whether FSC certification in turn has led to poverty reduction.
A large number of case studies, both econometric and qualitative.	Despite many examples of CCPs, only a few case studies have carefully examined the explicit impacts of company-community contracts on poverty alleviation.	Mostly case studies of particular contracting relations. Some quasi-experimental studies on tree crops. Most studies of timber were largely qualitative, published as grey literature, relied on descriptive statistics, and/or failed to consider counterfactuals.	Mostly case studies. Many studies are grey literature, with unclear methods and analytical rigour. Few studies on certified community forest management conform with standards for impact assessment.	No evidence.	No evidence.
Producer organisations can help forest producers overcome a number of challenges they face in deriving economic benefits from forests (including market access, technical services and information, and collective bargaining).	Partnerships provide leverage for local forest communities to enter capital-intensive timber production or better access markets, potentially improving incomes and net returns from land and labour.	Contracts between producers and processing or marketing companies help poor producers overcome many market and technical barriers, potentially translating into higher incomes and more resilient livelihoods.	Certified products are expected to either fetch a higher price or help producers to reach dedicated markets. Adoption of practices prescribed by certification standards may improve productivity and reduce production risks.	Zero deforestation commitments frequently include guarantees to improve a company's conduct towards various groups of people, including indigenous and other forest-de- pendent people who live in and around forests used for commodity production; labourers employed by commodi- ty-producing or processing companies; and smallholders who produce commodities and sell them into larger supply chains. Therefore, if companies that adopt zero deforestation commitments honour their pledges, then poverty may be reduced in one or more ways.	Coordinated consumer action can hurt the profitability of a company, nudging it to adopt more sustainable production practices for timber, including social standards with potential poverty reduction ramifications (e.g. adoption of FSC standards).
Forest producer organisations	Company-com- munity partner- ships	Contract production	Certification	Zero deforestation commitments	Boycotts

LEVER	THEORY OF CHANGE	SUMMARY OF THE EVIDENCE: QUANTITY AND TYPE OF STUDIES	SUMMARY OF THE EVIDENCE: CONCLUSIONS
Forest and tree I	management levers		
Agroforestry	Agroforestry and tree planting can deliver additional income directly through sale of tree products or indirectly through increasing crop and livestock productivity, PES, and value-add-ing certification systems.	Many studies available on contributions to incomes, food security: some impact assessments with high risk of bias; very limited randomised control trials; and several studies using non-randomised regression analysis.	Several studies show that with extension and training, agroforestry adoption can lead to increased yields, household income, food security and dietary diversity, and tree planting can lead to diversifying incomes and improving livelihoods. A few studies have found that agroforestry programmes are associated with significant poverty reduction.
Forest restoration, reforestation afforestation	Restoration, reforestation or afforestation can reduce poverty through transfer of payments for tree planting activities, PES, improvements in forests goods and services, securing tenure rights and trainings.	Multiple case studies. Few studies with robust counterfac- tual analysis.	Evidence shows that forest restoration, reforestation or afforestation can result in short-term livelihood benefits from direct involvement in tree planting (e.g. via payments and increased asset ownership). There is little evidence that livelihood benefits from services provided by restored forests meaningfully benefit proximate households to alleviate poverty.

5.3 Rights-based Levers

Rights-based levers tend to be developed and implemented by local, sub-national or national governments, with their implementation often supported by civil society actors. We review different types of tenure reform, community forest management, concessions, and protected areas, and examine the evidence that these levers have demonstrably affected poverty.

5.3.1 Tenure and property rights reform

Tenure can be defined as any arrangement that allocates rights to those who hold land and may also establish conditions for access and use of natural resources (FAO, 2011). Tenure may also be viewed as including a bundle of different property rights (Schlager and Ostrom, 1992; Galik and Jagger, 2015; Sikor et al., 2017) with a combination of rights and responsibilities assigned to individuals or groups, permanently or temporarily, exclusively or not, to land or resources on that land (Bromley and Cernea, 1989; Bromley, 1992; Pacheco et al., 2009; FAO, 2011; Cronkleton et al., 2012). The bundle of rights comprises access, withdrawal, management, alteration, exclusion and alienation rights (Schlager and Ostrom, 1992; Galik and Jagger, 2015). Tenure reforms and forest property rights interventions can include a devolution of one or more of these rights to households, communities or other actors, as well as activities such as individual or collective land titling, tenure administration (such as cadastral or other recordkeeping), dispute resolution and enforcement relating to property rights and their security or redistributive land reforms (Miller et al., 2020). Also of importance to forest-reliant people is tree tenure reform, and the devolution of the accompanying bundle of rights related to trees and tree products, which in some countries is considered separately from land tenure (where a person or community has ownership of land but not the naturally occurring trees on that land – Fortman, 1985).

In the context of poverty alleviation, the formal recognition and allocation of these rights to local communities and other *forest-dependent* people is expected to improve their livelihoods and well-being through secured access to resources, enable investments as a result of increased tenure security and, as a consequence, reduce poverty and *inequality* (Deininger, 2003; Meinzen-Dick, 2009; Lawry et *al.*, 2017; Miller *et al.*, 2019). Hence, SDG 1, target 1.4 aims to "ensure that all men and women, in particular the poor and the vulnerable, have equal rights to [...] ownership and control over land and

other forms of property, inheritance, natural resources [...]."

The proportion of forests with secure tenure rights for local communities and other forest dependent people is currently considered a thematic metric of forests' role in ensuring equal rights to economic resources for all (FAO, 2018). Tenure reforms here range from partial devolution of forest management rights to local communities resulting in co-management systems (e.g. Senegal) to strong management and exclusion rights empowering local community forest organisations (e.g. Mexico), to formal titling of Indigenous territories (e.g. Peru). Currently, approximately 15.3% of forests worldwide, and approximately 28% of forests in low- and middle-income countries (LMICs), are formally owned or managed by Indigenous peoples and local communities (RRI, 2018). Yet it is estimated that Indigenous peoples and local communities hold as much as 65% of the world's land area under customary systems, only a fraction of which are formally recognised by governments (RRI, 2015).

Much of the literature on securing land rights has focused on environmental degradation, productivity and incomes in agricultural settings. A recent systematic review of this literature on the impacts of interventions to recognise individual/ private land tenure on agricultural productivity showed substantial productivity and income gains, although these differed by region (Lawry et al., 2017). In the context of forests, a systematic review of the impact of forest property rights interventions on poverty reported generally positive or mixed impacts on income consumption and capital, although quasi-experimental assessments in the review reported positive and negative impacts in equal proportions (Miller et al., 2019). The review noted that most of the studies focused on the devolution of access and withdrawal rights to communities, and that interventions providing more limited rights are less likely to alleviate poverty than those devolving more extensive rights.

Impacts of reform of tree tenure outside forests are much less studied. In Ghana, where much of the studies on tree tenure focus and where naturally occurring trees remain under the purview of the government regardless of land ownership, several studies call for tree tenure reform to address loss of forests and trees on farms and to enhance related livelihoods (Acheampong and Marfo, 2011; Marfo *et al.*, 2012; Hajjar, 2015). There has been some progress in implementation of tree registration programmes and other rights documentation of planted trees, aiming to provide landowners and farmers with the security that they will be able to benefit from the tree when it matures (Fisher *et al.*, 2012; Sullivan *et al.*, 2018), but poverty impacts of such programmes have yet to be adequately measured.

Overall, the available evidence shows that tenure reform can play a role in poverty reduction, but that it seems to work best when combined with other policy instruments (Carter, 2003; Werner and Kruger, 2007; Meinzen-Dick, 2009; Shyamsundar et al., 2020). Hence, the effectiveness of tenure reform is enhanced by interventions on access to justice and the rule of law, enforcement of property rights, technical support, and access to finance and basic infrastructure, e.g. water, electricity, roads, communications, schools, and healthcare (Werner and Kruger, 2007; Prosterman et al., 2009; Meinzen-Dick, 2009; Akinola and Wissink, 2019; Gabay and Rekola 2019). Indeed, tenure reform, including devolution of forest rights and enhancing tenure security, is often a necessary but not sufficient enabling factor for the successful implementation of several levers discussed in this chapter.

The available evidence of the impacts of land tenure reform on vulnerable groups is variable and less promising. Tenure reforms have increased conflicts and tension between communal and individual rights, commodification of land subjecting it to market forces, and elite capture of benefits, as well as having limited benefits for women and Indigenous peoples (Benjaminsen et al., 2008; Prosterman et al., 2009; Okuro, 2011; Akinola and Wissink, 2019). Smallholders lacking requisite social and financial capital have great difficulty in registering trees to claim tenure over them (Johnson Gaither et al., 2019). Even in studies that reported positive outcomes of tenure, many noted that wealthier households, better educated individuals, or men benefited more from tenure reforms (Miller et al., 2020). In 30 of the most forested LMICs, over 50% have laws protecting women's property rights, but for community tenure regimes, only 29% protect women's membership, 10% protect inheritance rights and 3% their right to vote (Bose, 2011; Monterroso et al., 2019). More work is necessary to protect and enforce women's tenure and access rights as women are still left behind (Prosterman et al., 2009; Bose, 2011; Monterroso et al., 2019; Gabay and Rekola, 2019).

5.3.2 Community forest management interventions

Community forests, or forest commons, are "forests used in common by a large number of heterogeneous users... [who] have a stake in good governance... [that] central governments formally or informally recognize" (Chhatre and Agrawal, 2008). Such forests constitute more than 18% of the global forest area and make important contributions to carbon sequestration, biodiversity conservation and livelihoods (Agrawal *et al.* 2008). Their management historically predates more centralised forms of forest *governance*. But the postcolonial version of community forest management (CFM) has become a key instrument of forest policies since the late 1970s. Community forestry is now one among several strategies of forest management on which governments rely throughout the world for achieving dual objectives of forest conservation and livelihood improvement.

In recognising the rights of local user groups to manage common forest resources, it is expected that users will benefit directly and indirectly from forest products and services for subsistence and commercial purposes. However, the ecological diversity of forest commons, the institutional diversity of their forms of management, the variable national and local market and policy contexts in which they exist, and the wide range of benefits and ecosystem services that multi-functional community forests provide, have meant that there are few reliable national level assessments of the contributions of community forests to poverty alleviation. Such assessments are especially difficult because many of the benefits community forests provide are not sold in markets but harvested directly by users. It is therefore difficult both to quantify them and to price them.

At the same time, the role of community forests in providing subsistence, livelihood and commercial benefits to users has been a core focus of research for more than two decades (Antinori and Bray, 2005; Anderson *et al.*, 2006; Bray *et al.*, 2006; Ali and Behera, 2015). There is thus a wealth of both case literature and reviews of research on community forestry. This literature enables some generalisations, despite patchiness in country and regional coverage, about whether, how and to what extent CFM improves well-being.

Much case study evidence points to clear material benefits from community forest management for the poor (Thoms, 2008; Beauchamp and Ingram, 2011). McDougall *et al.* (2013) show that community forests in Nepal helped improve incomes, financial and forest assets, and employment. Reporting on findings from a multi-country study of forests and poverty, Jagger *et al.* (2014) suggested community forests do contribute in important ways to household incomes but less than do state-owned forests. In reviewing 40 years of community-based forestry, Gilmour (2016) states that there is much potential for CFM to produce benefits, but that the

Box **5.1**

Community forest management in Nepal

Nepal's Community Forestry Management (CFM) programme is considered one of the most successful of its kind in the world (Bijaya et al., 2016). There are 19,361 Community Forest User Groups (CFUGs), which encompass 2.5 million households, managing 30% (~1.8 million ha) of the nation's forests (Government of Nepal, 2020). A CFUG has the management and use rights of the forest, but the community forest land ownership belongs to the state (Acharya, 2002; Thoms, 2008; Dahal, 2017). CFM has been touted as a means to improve both forest condition and local livelihoods, and has led to significant improvements in forest condition in Nepal (Gautam et al., 2004; Nagendra et al., 2008; Sunam and McCarthy, 2010; Shrestha and McManus, 2013; Bijaya et al., 2016), with forest cover in Nepal increasing from 35% in 1985 to 44% in 2015 (Forest Research and Training Centre, 2019).

Community forest management has also alleviated poverty in Nepal (Oldekop *et al.*, 2019). Villages with CFM reduced the number of households living in poverty more than villages without CFM, between 2000 and 2012 (Oldekop *et al.*, 2019). Yet, benefits of CFM are unequally distributed among CFUG households. A ma-

potential has not yet been realised in most countries. In their recent authoritative review of available evidence on community forests examining 697 published cases of CFM, Hajjar *et al.* (2016) found that 68% of cases that reported on livelihood outcomes indicated that community or household incomes increased after CFM implementation.

Several reviews of community forest management have found the strength of evidence of the potential of CFM to generate welfare improvements as lacking. As Bowler *et al.* (2012) asserted, "poor study design, variable reporting of study methodology or context, and lack of common indicators make evidence synthesis difficult." In the systematic review of CFM referenced above, Hajjar *et al.* (2016) also showed that the available literature is characterised by a predominant focus on South Asian cases, qualitative analyses, and data and analytical gaps preventing generalisable conclusions about observed socio-economic and environmental outcomes of CFM. jority of studies on CFM in Nepal have found that benefits to poor and low caste households are often much less than those to the welloff households in absolute terms (Adhikari et al., 2004; Iversen et al., 2006; Yadav et al., 2015; Bijaya et al., 2016). Poor and marginalised households are often more reliant on forests for livelihoods (Adhikari et al., 2004; Chhetri et al., 2016). Yet they often bear more costs both directly (e.g. contribution of time and labour for patrols) and indirectly (e.g. fewer livestock due to grazing controls) (Sunam and McCarthy, 2010; Parajuli et al., 2015). CFM institutional arrangements often fail to effectively account for heterogeneity in power, socio-economic status and knowledge among forest resource users within CFUGs, allowing rich and elite members to exercise their power over poor and disadvantaged households (Sunam and McCarthy, 2010; Adhikari et al., 2014). In sum, despite the success of CFM in Nepal in improving poverty outcomes, there is still much room for improvement, particularly with respect to equitable distribution of benefits and democratic engagement in decision making with poor and marginalised households.

Despite this, several recent studies have shown that community forests indeed provide material benefits to users (Chhatre and Agrawal, 2009; Persha et al., 2011). In a rigorous analysis of 18,000 community forests in Nepal, Oldekop et al. (2019) show that CFM reduced both poverty and deforestation (Box 5.1). In a similar national-level analysis, Rasolofoson et al. (2017) found that CFM in Madagascar had a small but positive impact on household living standards, particularly for those closer to forests and with more education. Similarly, Santika et al. (2019) show that Indonesian village forests contributed to win-win outcomes and substantial economic benefits to the poor, but that the flow of poverty reduction benefits was linked to higher order variables related to land use classifications and zoning regulations. Studies such as these are beginning to create the knowledge and evidence base necessary to assess quantitatively the contributions of community forest management to poverty reduction at the national rather than the local scale



Interviewing rural people to understand policy effects on their livelihoods and the environment Photo © Conghe Song

5.3.3 Forest concessions

Under concessionary forest governance, central governments or their forest departments provide companies and communities with forest resource (typically timber) extraction rights in commercially valuable, government-owned forests in exchange for a stream of revenues (Agrawal et al., 2008; Bulkan, 2014). The private concession model in forest governance has been in existence at least since the imperial trading period of the early 1700s and was implemented widely during colonial rule in countries such as Burma (now Myanmar), India, Indonesia and different parts of sub-Saharan Africa (Hardin, 2002; Hardin and Bahuchet, 2011). The largest concession areas occur in the cool temperate and boreal forests of Canada and Russia (FAO, 2018). But concessions also affected 20% of tropical forests in the 2000s, and most tropical timber is harvested through concessionary arrangements that vary in details across countries (Asner et al., 2009). Concessions enabled the extraction of 270 million cubic metres of tropical timber (ITTO, 2015) valued at more than USD 7 billion in 2015 (Pearson et al., 2018), with the FAO estimating the export value of timber from tropical countries to be in excess of USD 10 billion annually (Kishor and Damania, 2007).

A variety of logging concession arrangements exist including community concessions in Central

America (Gretzinger, 1998; Taylor, 2010). Private and corporate forest concessions are the dominant form of forest governance in tropical forests in Southeast Asia, parts of the Amazon, and especially in Central and West Africa (World Bank, 2002a), covering at least 75 million hectares of forests (Agrawal et al., 2008; Banerjee et al., 2009; Grut, 2010; Sodikoff, 2012). Commercial logging companies supply the capital and the technical expertise needed to undertake the different forms of planning and selective logging for forestry operations in remote, capital-poor contexts where harvestable timber exists. Governments provide the legitimacy and legal foundations for resource extraction that the companies need to undertake logging operations.

The most basic concession agreements specify the area of concession, the volume of timber extraction, and amounts and cadence of royalty payments to governments by concession holders, usually companies. Over time, concession agreements have evolved in many countries to include provisions for local public goods in the form of employment, schooling and basic healthcare for communities near logging concessions (van Hensbergen, 2016).

The concession form is vulnerable to corruption at all stages of its development and implementation (Kishor and Damania, 2007). Poor enforcement and price variations mean that tropical

country governments lose vast sums in revenues and as value addition to their national income. The World Bank estimates that poor countries lose upwards of USD 5 billion because of illegal logging (World Bank, 2002b). Illegal revenues also fuel armed conflicts and military challenges to incumbent governments, leading to political instability and violence against community members living near concessions (Le Billon, 2001; Davis, 2005; Kishor and Damania, 2007). Sustainability certification (Section 5.5.9) and improved enforcement are among the instruments national and international decision-makers have used to try to improve forest concession outcomes, including: lower levels of logging, sustainable forest management, improved local incomes and increased revenue streams to national governments (Ebeling and Yasué, 2009). National and international forest sector reform efforts (Section 5.4) seek to improve governance, reduce corruption and control illegal logging, but these efforts have achieved only sporadic and patchy success.

Concessions generate substantial income through timber harvesting and trade, particularly for logging companies (Ross, 2001; Medjibe and Putz, 2012; Straumann, 2014). Aggregate estimates of the value generated through concessions are available for the formal sector. But the contributions of concessions to local incomes and poverty alleviation are only documented for specific locations through case studies. Part of the reason is that economic contributions of forest concessions can be direct – through employment, income and service provision, but also indirect – through improved road connectivity, sales of goods and services to concession employees, and agricultural production.

Assessments of contributions of forest concessions to local incomes and poverty alleviation vary by space and in time. In a recent study, Jung et al. (2019) used nationally representative panel data to estimate that the wealth score (based on asset holding) of those living close to forest concessions in Liberia improved by 10% or more compared to the assets of those farther away from concessions, likely as a result of indirect effects on agricultural output. Lescuyer et al. (2012) found a similar effect of forest concessions in two communities in eastern Cameroon. In another study focusing on Gabon and concerning 17 communities around two forest concessions, Iponga et al. (2018) found minimal contributions to livelihoods from the non-timber forest product gathering activities of local residents. In general, it is difficult to escape the conclusion that writings on forest concessions provide only limited evidence of their contributions to poverty reduction even as concessions generate substantial benefits and profits for large logging companies (Scudder *et al.*, 2019).

5.3.4 Protected areas

Protected areas (PAs) are, in theory, clearly defined geographical spaces, recognised, dedicated and managed to achieve the long-term conservation of nature with associated ecosystem services and cultural values (IUCN, 2008). They are a popular policy instrument in the global fight against loss of biodiversity and ecosystem services (Hanauer and Canavire-Bacarreza, 2015).

Protected areas are a global phenomenon covering just under 15% of the world's terrestrial surface and inland waters (UNEP-WCMC and IUCN, 2016), with countries committing to increasing this to 17% by 2020 (Schleicher *et al.*, 2019). The first formally-recognised national park (Yellowstone National Park in the U.S.) was established in 1872. Yet, community-protected areas have existed since ancient times and globally their land surface equals that of official reserves (Muench and Martínez-Ramos, 2016).

The International Union for Conservation of Nature (IUCN) classifies PAs according to their management objectives, ranging from strictly protected nature reserves and national parks, to community conserved areas and those areas allowing sustainable use of natural resources (Dudley, 2008). Depending on this classification, PAs can support poverty reduction by securing the rights of people to land and valuable natural resources, supplying ecosystem services, generating economic benefits including through ecotourism (see Section 5.5.3) and improving infrastructure in remote areas (Andam et al., 2010; Brockington and Wilkie, 2015). However, there is also a large evidence base showing that people have been displaced or denied access to resources by the establishment of parks and reserves, threatening peoples' rights and livelihoods (Brockington and Wilkie, 2015). Indeed, the establishment and management of many national parks are often reflective of forest conservation, a top-down protectionist approach to park management (Bruner et al., 2001). These are envisaged as places where rural livelihoods do not belong (Brockington et al., 2006), human habitation is often excluded through the forced removal of local and Indigenous people (Magome and Murombedzi, 2003) and nature is seen to be preserved as 'wilderness' (Colchester, 2004). Human-induced pressures on PAs and the conflict between biodiversity conservation and the needs of local people are predicted to increase due to numerous factors, including disputes over traditional territorial claims, land-grabbing by the landless or industrial-scale commodity producers, forced evictions, market forces, and a reduction in distance between PAs and human population centres (Brockington and Wilkie, 2015; Oldekop *et al.*, 2016).

Several studies provide strong evidence that PAs can reduce poverty. Using poverty measures based on national census data in Costa Rica and a poverty index in Thailand, one study found that the net impact of ecosystem protection was to alleviate poverty in both nations (Andam et al. 2010). However, not all segments, sub-districts or poor households experienced poverty alleviation from PAs (Andam et al., 2010). Applying a quasi-experimental research design to data collected by Andam et al. (2008; 2010) in Costa Rica and Thailand, another study assessed the heterogeneity of PA impacts on poverty reduction, finding that PAs in areas associated with high poverty did, on average, reduce poverty while also reducing deforestation (Ferraro et al., 2011). However, a quasi-experimental, panel study of three PAs in Cambodia found limited impact on poverty of households within the PAs as compared to their matched controls (Clements and Milner-Gulland, 2015). Another study using matching methods found an overall negative PA impact on household wealth in China (Duan and Wen, 2017).

Several studies provide evidence of trade-offs across the landscape and support for the simple theory that the opportunity cost of land has significant moderating effects on the impacts of PAs (Hanauer and Canavire-Bacarreza, 2015). Using quasi-experimental methods, Hanauer and Canvire-Bacarreza (2015) found that the biophysical characteristics associated with the most avoided deforestation were the characteristics associated with the least poverty alleviation in Bolivian cases; that is, the same characteristics that may have improved the social welfare impacts of study PAs may also limit their conservation effectiveness (e.g. PAs on lands that are highly suitable for agriculture, far from major cities and infrastructure, or where a high percentage of adults are employed in agriculture) (Ferraro et al., 2011). Win-win outcomes were most commonly associated with locations at intermediate distances from major cities (40-80 km) and on land of moderate to poor agricultural potential (Ferraro et al., 2011). Similarly, Sims (2010) found that the largest positive socio-economic impacts from tourism in PAs in Thailand occurred at intermediate distances from major cities. Another study using data from 190,000 households across 34 countries found that households near PAs with tourism had higher wealth levels and a lower likelihood of poverty (by 16%) than similar households living far from PAs (Naidoo *et al.*, 2019). In explaining heterogenous impacts of PAs, a meta-analysis found that PAs that integrated local people as stakeholders and sought to empower them tended to be more effective at achieving both positive conservation and socio-economic outcomes (Oldekop *et al.*, 2016).

While the science of PA mechanisms is still at a fledgling stage (Ferraro and Pressey, 2015; Hanauer and Canavire-Bacarreza, 2015), there is a growing empirical evidence base documenting whether, and by how much, PAs affect the environment and human welfare (Ferraro and Hanauer, 2014; Borner et al., 2020). Though it faces a diversity of challenges, the global network of PAs will likely remain a key option for maintaining and enhancing biodiversity conservation, hence the potential of protected areas for exacerbating or alleviating poverty will need to be monitored. Given recent proposals to protect half of the earth (Dinerstein et al., 2017; 2019) determining impacts is critical (Schleicher et al., 2019), and the use of impact evaluation techniques using rigorous, quantitative methods to infer causality from non-experimental data is becoming the norm in this domain (Ferraro and Pressey, 2015).

5.4 Regulatory Levers

Although rights-based interventions (considered above) are a form of regulatory lever, in this section, we review other levers that are principally oriented around laws, policies, and regulations that determine how forests and trees are managed, used, conserved and/or restored. Regulatory levers tend to be developed and implemented by local, sub-national or national governments. We review decriminalisation and formalisation of informal operations, modification or simplification of regulatory frameworks, log export bans and procurement policies, and examine the evidence that these levers have demonstrably affected poverty.

5.4.1 Decriminalisation and formalisation of informal operations

The informal forestry sector (unincorporated enterprises that may also be unregistered and/or small – Lewis, 2016) is estimated to employ 45-50 million people worldwide, compared to 13 million employed in the formal forestry sector (FAO, 2014). Informality does not necessarily equate to dealing in illegal goods or purposefully evading regulations. Indeed, the vast majority of informal

workers are working poor, for whom existing regulations are simply irrelevant or inappropriate (Kaimowitz, 2003; Lewis, 2016). Informality in the forest sector then encompasses various 'shades' of illegality, ranging from unlicensed collection of forest products or charcoal production to harvesting and trading protected species. Importantly, the degree of illegality can vary significantly depending on national legislation (Weng, 2015). Formalisation in the context of the forest sector has been defined in a number of different ways, often focusing on the formal process of identifying, codifying and registering the rights to access, own or trade land and forest resources (e.g. Meinzen-Dick and Mwangi, 2009; Putzel et al., 2015; Kelly and Peluso, 2015), or on enhancing the degree to which supply networks are controlled by official and explicit rules and institutions (Choi and Hong, 2002; Schure et al., 2013) and the extent to which producers engage with them (Erbaugh et al., 2016).

Formalisation can be a way of allowing the poor to convert their possessions and labour into capital, which in turn can be used to generate added value (e.g. through accessing credit) (de Soto, 2000). In addition, formalisation can benefit producers through enhancing the protection of rights (Chen, 2007), encouraging productive investments (Hirons et al., 2018), reducing incentives for corruption (Zulu and Richardson, 2012), allowing producers to fetch higher prices for products on formal markets, and keeping producers out of law enforcement trouble and having equipment confiscated (Hajjar et al., 2011). Yet, formalisation in the forest sector is rarely implemented with poverty alleviation as the sole or primary objective (Lele et al., 2010; McDermott, 2014; Putzel et al., 2015). Recent years have witnessed a particular interest in curbing informality – often wrongly equated with illegality - as a way to also enhance legality and sustainability.

Few studies in the forest sector in the tropics have focused explicitly on the relationship between formalisation and poverty alleviation. To date, fully formalised, small-scale forest product value chains are not yet mainstream in national or regional economies (Lewis, 2016; Doggart and Meshack, 2017). This is because poor, small-scale value chain actors struggle to comply with costly and complex formal regulations (Kaimowitz, 2012; Obidzinski et al., 2014; see Section 5.4.2) that are often poorly adapted to their needs (Hansen and Treue, 2008). In many instances, various types of formalisation efforts in the forest sector have excluded and marginalised poor small-scale workers (Anderson and Pacheco, 2006), criminalised legitimate but informal livelihoods (Hansen and Treue, 2008; Purnomo *et al.*, 2009; Cerutti *et al.*, 2013; Hirons *et al.*, 2018), reduced incomes (Chen, 2007; Wynberg *et al.*, 2015), limited access rights to key commodities (Anderson *et al.* 2018), as well as increased elite capture and exploitation by more powerful actors (Lele *et al.*, 2010; Ndoye and Awono, 2010; Schure *et al.*, 2013; Weng and Putzel, 2017).

In contrast, Cerutti et al. (2019) found that some small and medium-sized forest enterprises registered within the national forestry licensing scheme (SVLK) in Indonesia reportedly found easier access to formal financial credit than they used to find when they operated in the informal sector, with improved access to the international market (Cerutti et al., 2019). Schure et al. (2013) suggested that taxes generated through formalised and decentralised woodfuel chain governance in Central and West Africa had been reinvested in local social projects, while Hautdider and Gautier (2005) found that woodcutters in Mali benefitted from formalisation through harvesting quotas, formally allocated selling points, and improved oversight. However, implementation of forest commodity regulations is often weak and may incentivise corruption as well as free-riding (Schure et al., 2013). In Mozambique, Jones et al. (2016) found that the lax enforcement of formal regulations in the charcoal sector was in fact critical to enabling the participation of small-scale producers - and especially women - participating on a casual basis.

When costs are perceived to outweigh benefits, Obidzinski *et al.* (2014) found evidence of Indonesian producers opting not to renew SVLK certificates. Recognising the potential social pitfalls of formalisation, recent discourses tend to emphasise various safeguards and complementary measures aimed at more inclusive and socially sustainable formalisation. Such measures may include decentralising land and resource governance (Putzel *et al.*, 2015), promoting cooperatives/producer organisations (Tilahun *et al.*, 2016) or simplifying regulations (Lewis, 2016). These levers and their impacts on poverty are assessed elsewhere in this chapter (see especially Sections 5.3.1, 5.4.2. and 5.5.6).

5.4.2 Modifying or simplifying regulatory frameworks

Among the oft cited barriers preventing communities and smallholders from engaging in the formal forestry sector are overly bureaucratic and technical processes in completing forest management plans, obtaining permits and other legal documents, and complying with burdensome regulations that were largely designed for a forest sector dominated by large companies (Medina et al., 2008). As such, one proposed solution has been to require simplified management plans that are easier to complete by smallholders and communities, with the intention of bringing their forest activities into the formal sector. In having the necessary legal documentation, these actors could theoretically better access markets for their products and fetch higher prices than they would obtain on informal markets. They would also be less at risk of being fined or jailed for illegal activities (Hajjar et al., 2011). This lever would likely be one of several regulatory changes needed to enhance the ability of small-scale forest enterprises to contribute to poverty alleviation (Badini et al., 2018).

While it is not clear how widespread regulatory changes have been, a handful of studies in Latin America and Africa have examined the effects of simplified forest management plans on poverty. Forestry laws modified in the mid-1990s and 2000s in Bolivia and Ecuador allowed for a number of special regulations for small-scale management, simplifying the requirements for obtaining permits or plans so that they were less costly to prepare and implement or, as in the case of Beni province in Bolivia, allowing smallholders to extract timber without a management plan at all if they needed the income to buy basic goods (de Koning, 2011; Pacheco, 2012). Yet, the regulations still proved to be barriers for smallholders, requiring inputs and technologies. Often the benefits from following the law did not outweigh the costs, resulting in many smallholders continuing to participate in informal markets (Pacheco, 2012). In Cameroon, one study found that simplified management plans brought financial benefits to some communities, but the study did not compare the effects of simplified plans relative to non-simplified plans (Bruggeman et al., 2015). In an ex-ante study, Sanogo et al. (2014) predicted simplified management plans in Senegal would only have a small impact on local development when considering small management areas.

In summary, to date no study has attempted to empirically disentangle the effects of simplified management plans on poverty from the effects of other factors (such as tenure reform, market access and other barriers to SMFEs), but many continue to point to the difficulties associated with overly bureaucratic and technical processes to participate in the formal sector.

5.4.3 Log export bans

Export bans (or high export taxes) for unprocessed log timber have been implemented in many countries (predominantly in low- and middle-income countries, but also in some high-income countries) to counter deforestation and environmental degradation associated with the timber trade and/or to induce development of a domestic processing industry. Effectively an export tax and input subsidy to domestic processing (Dean, 1995), these bans were expected to increase employment and value-added in a nation's forestry sector by increasing the overall domestic processing of logs. A number of countries in Southeast Asia, Africa and Latin America enacted such export bans, primarily during the 1970s, 80s, and 90s (Goodland and Daly, 1996), and in some cases, enacted them more than once (Resosudarmo and Yusuf, 2006). The costs and benefits of log export bans (LEBs) on both welfare and environmental resources have been debated extensively (Goodland and Daly, 1996; Kishor et al., 2002; Resosudarmo and Yusuf, 2006; Tumaneng-Diete et al., 2005). While some empirical studies and economic models indicated that export bans have indeed increased domestic processing capacity, exports of secondary processed wood products and employment in the domestic processing sector (reviewed in Goodland and Daly, 1996), a number of models show that the increased employment in the processing sector does not compensate for the number of jobs lost in logging operations following LEB policies (Resosudarmo and Yusuf, 2006). Gillis (1988) estimated that Indonesia lost millions of dollars by banning log exports. We found one study that specifically modelled the effects of an LEB on households in poverty, showing that an LEB in Indonesia would result in decreased incomes across agricultural and rural households, at least in the short run (Resosudarmo and Yusuf, 2006).

5.4.4 Procurement policies

Timber procurement policies aim to ensure that timber is coming from legal and/or sustainable sources. They have been implemented at the domestic level, where governments wish to source timber for public works from legal sources, or internationally, to ensure that timber being imported into a country is sourced legally (such as the EUTR or US Lacey Act). A particularly prominent international intervention in this realm has been the Voluntary Partnership Agreements (VPA) of the EU's Forest Law Enforcement, Governance and Trade Programme, which are bilateral trade agreements between 15 countries and the European Union that include a series of governance reforms in the exporting country to ensure that timber being imported into the EU comes from legal sources. Some argue that the opportunities

for domestic governance reform and opening of the political process that the VPA presents can be used by civil society to ensure pro-poor policy reforms such as strengthening land tenure and access rights for marginalised rural communities and Indigenous peoples (Hobley and Buchy, 2013; Richards and Hobley, 2016; Tegegne et al., 2017). Yet, we did not find evidence suggesting that this pathway has resulted in poverty reduction or enhanced economic opportunities for the forest-reliant poor. Instead, some authors have pointed to potential negative effects of these legality policies on small-scale timber producers, particularly if they are required to bear the cost of implementation or if self-employed people in the informal sector are squeezed out, exacerbating poverty in forest-reliant communities (Eba'a Atyi et al, 2013; Hajjar, 2015).

Alternatively, procurement policies can be used domestically to boost small-scale producers. For example, in Oaxaca, Mexico, the state government's policy regulating the purchasing of furniture for government institutions from community forests has boosted furniture production in three community forest enterprises (Tanaka, 2012).

5.5 Market and Supply Chain Levers

In this section, we review levers that are based on market mechanisms and whose success depends, at least in part, on commodification or commercialisation of trees, forest products or forest ecosystem services. Market and supply chain levers may be developed and implemented by governments, private sector bodies, or NGOs. Participation in such levers is generally voluntary. We review payments for ecosystem services, REDD+, ecotourism, small and medium forest enterprises, market access, forest producer organisations, company-community partnerships, contract production, certification, zero deforestation commitments and boycotts, and examine the evidence that these levers have demonstrably affected poverty.

5.5.1 Payments for ecosystem services

Programmes of payments for ecosystem services (PES) are conservation contracts intended to encourage environmentally favourable activities. The usual structure of these programmes is that willing participants offer to conserve a landscape (e.g. forest conservation), engage in a productive activity with a reduced ecological footprint (e.g. *agroforestry*), or restore ecological services (e.g. *reforestation*). Once these efforts are verified, participants receive payment (Wunder, 2015).

When they yield additional environmental benefits, PES programmes are expected to compensate participants for the value they forego by not carrying out the productive activity in which they were going to engage in the absence of the payments (Engel et al., 2008). Theoretically, people should not accept payments that are less than this minimal opportunity cost. Therefore, it should be the case that PES payments either have no impact on poverty (if they exactly compensate for lost profits) or increase incomes (in the event that they exceed the amount of lost profits). Nonetheless, if households do not fully understand the mechanism or the benefits from deforestation, they could lose out from the transfer scheme. Here we examine whether rigorous studies with large numbers of observations show this to be the case. There are already a number of reviews of the impact of PES programmes on poverty (e.g. Bulte et al., 2008; Lipper et al., 2009; Palmer and Engel, 2009; Alix-Garcia and Wolff, 2014). Much early evidence came from Mexico and China, two countries with large existing PES-type programmes that started in the early 2000s. All estimations documented declines in, or no effect on, poverty.

New evidence from a broader range of countries supports the finding that PES programmes have either a small positive or neutral effect on poverty reduction. Two nationally-representative studies in Mexico showed small poverty reduction impacts that have decreased over time, probably due to the erosion of payment values from inflation. A combination of matching between accepted and rejected applicants and panel data approaches showed that beneficiaries experienced small increases in household assets that were greater in areas with lower deforestation risk (Alix-Garcia et al., 2015). Sims and Alix-Garcia (2017) confirmed a small but significant decrease in poverty in Mexican PES-receiving localities from 2000-2010. Later work comparing beneficiaries to non-beneficiaries from 2011-2014 using a regression discontinuity approach showed zero impact on assets, food expenditures and housing characteristics (Alix-Garcia et al., 2019). A smaller study (eight villages, 261 households) in southern Mexico comparing matched payment recipients and non-recipients (non-applicants) found positive impacts on household assets, but only where payments were large (Jones et al., 2018). In Costa Rica, programme impacts estimated using matching (recipients to non-applicants) have generally revealed neutral effects despite the fact that payments are large (Robalino et al., 2014; Arriagada et al., 2015). The observation that programme recipients continue to re-enrol suggests that participation benefits

Box **5.2**

China's largest payment for ecosystem services programme

Following devastating natural disasters in the early 2000s believed to be caused by soil erosion from croplands on slopes of mountainsides, the Chinese government initiated the largest payments for ecosystem services programme in the world, the Conversion of Cropland to Forest Programme (CCFP) (China State Council, 2000). Also known as the Sloping Land Conversion Programme or the Grain for Green Programme, the CCFP enrolled farming households to convert their croplands on slopes to forests. In return, the government provided grain compensation to the farmers for the forests created on their cropland (Liu et al., 2008), although the grain compensation was later replaced with cash. The secondary goal of the CCFP was poverty alleviation as most of the households who were eligible to participate in the programme live in poor regions of the country. The ecosystem services that the government buys from farmers are soil and water conservation. Farmers initially signed an eight-year agreement with the government for CCFP. After the initial contract ended, the Chinese government renewed the contract for another eight years, but at half of the initial compensation rate. In 2015, the

likely exceed costs, but are not sufficient to induce detectable decreases in poverty.

A number of new studies have emerged in China and Vietnam. Using a selection model to compare participants to non-participants between 2008 and 2014, Phan et al. (2018) found a positive effect of PES on income growth. One caveat is that control households in the sample had significantly higher incomes than recipients in the baseline period. Treacy et al. (2018), using a difference-in-difference model, found a positive effect of the Sloping Lands Conversion Programme (SLCP) on migration in China and no impact on income. Using a matched panel of household data from 1996 to 2010, Liu and Lan (2018) estimated initial positive effects of the SLCP on agricultural productivity but this positive effect declined over time. One study examining the impacts of the SLCP on income between 1999 and 2006 found negative effects via a reduction in crop income (Yang et al., 2018). However, these results should be taken with caution, as the data did

Chinese government started a second round of the CCFP.

A recent report released by the Chinese government (Xinhuan News Agency, 2019) showed that the CCFP converted 13.27 million ha of cropland to forest or grass, enrolling 41 million households (158 million people) since 2001. The average compensation over the programme period is almost CNY 9,000 (USD 1,270) per household. In addition to the direct compensation, CCFP stimulated significant income structure change from farm work to off-farm employment (Song et al., 2014; Rodriguez et al., 2016) because CCFP freed farm labour, stimulating rural out-migration (Zhang et al., 2018). Although the percent of cash compensation did not make up much of the total household income in general, it was a significant cash income for poor households (Song et al., 2014). Additionally, farmers have several rights over the forests grown on their croplands, incentivising them to allow the forests to mature. The programme significantly increased forest coverage in China which is now providing ecosystem services such as carbon sequestration (Liu et al., 2008).

not include any non-recipient group to help identify counterfactual trends, and there was no mechanism to allow causal identification of impacts.

Finally, there has been substantial new work focused on the poverty alleviation effects of PES in Africa. Two quasi-experimental studies in Mozambique estimate neutral to positive effects on household income. Using difference-in-difference techniques to compare project and non-project villages revealed neutral impacts of a long-standing PES project on household income between 2001 and 2008 (Jindal et al., 2012), while a matching analysis of PES versus non-PES households in the same project (but only in the year 2006) showed positive association with cash incomes and consumption expenditures (Hegde and Bull, 2011). The discrepancy between these two could be due to a combination of noise in the recall data used to create the difference in differences, the difference between comparisons of changes versus levels, and/or different survey instruments.

A conservative interpretation would be that payments in Mozambique did no harm to the incomes of participants. Two recent randomised control trials (RCTs) also provide evidence of short-term positive impacts on participant incomes. In Uganda, a PES RCT found small increases in non-food expenditures (Jayachandran *et al.*, 2017). A similar methodology applied in Burkina Faso suggested that the timing of payments may be important (Adjognon *et al.*, 2019); food expenditures of participants in a reforestation RCT increased by 12% and food insecurity decreased for payments timed to coincide with the lean season.

Overall, our understanding of PES anti-poverty impacts continues to evolve. There is no substantial evidence that such programmes hurt participants' incomes, nor is there extensive evidence of substantial positive effects on poverty. In some cases, PES has resulted in win-win outcomes for both the environment and programme participants. The studies examined do not always provide information on why there are positive impacts in some settings and not others. However, the simple framework that began this section would suggest that payment levels may currently be only just sufficient to compensate for the opportunity cost of engaging in PES contracts. Targeting payments to properties that have low opportunity cost but high risk of deforestation could increase poverty alleviation impacts without sacrificing conservation objectives. Finally, the most recent work suggests that where there is annual income variation, timing payments to the moment when incomes are lowest may generate important impacts on poverty by smoothing consumption.

5.5.2 REDD+

Policies, projects and other interventions related to Reducing Emissions from Deforestation and forest Degradation, and fostering conservation, sustainable management of forests and enhancement of forest carbon stocks (REDD+) are among the more prominent attempts to mitigate climate change since 2010 (Parrotta et al., 2012). Belonging to the broad family of payments for ecosystem services interventions, REDD+ initiatives provide monetary compensation in exchange for reductions in terrestrial greenhouse gas emissions in the forest sector. Early REDD+ interventions were primarily at the local scale with performance-linked payments made to local communities and households by international NGOs, voluntary market mechanisms, or in many cases through pilot schemes supported by bilateral donors or multilateral organisations. As countries with forests have developed the capacity to implement REDD+ policies and monitor emissions reduction at the national scale, results-based payments to sub-national jurisdictions and countries are becoming more important (Maniatis *et al.*, 2019; Wunder *et al.*, 2020).

While the primary objective of REDD+ interventions is to reduce emissions, REDD+ may influence poverty via two pathways. The first is through REDD+ payments to governments and local bodies for verified emissions reductions or promises of emissions reductions, and benefit sharing of such payments, with adjustments for costs of adherence to REDD+ objectives (e.g. to monitor and enforce rules designed to limit emissions). The second is through changes in forest benefits to local users and governments as forests recover when users limit extraction of carbon-intensive products from forests to conform to REDD+ objectives. Such indirect effects may be positive in cases where forest recovery occurs and users are able to harvest non-carbon-intensive benefits such as fodder, foods or non-wood products. Restrictions on harvests may also turn out to be costly if they affect extraction of valuable timber or other carbon-intensive forest products such as firewood, or if local resource users are displaced (Beymer-Farris and Bassett, 2012). REDD+ safeguards have been widely adopted to protect against negative impacts on Indigenous and traditional forest-dependent people (UNFCCC, 2011).

We found no comprehensive and rigorous assessments of the effects of REDD+ on poverty or on consumption, incomes, asset building, education, health or other indicators of well-being. Although national level payments for avoided deforestation at scale have begun to flow, there are no analyses of how these payments have been shared with local populations and for what benefits, or how effectively recipient governments have used such payments to address poverty of stakeholders involved in reducing deforestation.

Two studies at the local level did not find evidence for effects of REDD+ on material indicators of well-being or poverty, and suggest that positive effects of REDD+ payments are possible, but have been modest at best (Danielsen *et al.*, 2011; Awono *et al.*, 2014). A study of benefit sharing for REDD+ in Nepal found that direct contributions of REDD+ projects to households' incomes were nominal – from 3.2% of income of poorest households to 0.3% of the income for the less poor households (Shrestha *et al.*, 2017). Payments had a small effect on inequality but also increased social tensions. Local incomes under the N'hambita Community Carbon Project in Mozambique were higher for

project participants, but primarily as a result of wage payments during the tree planting phase of the project. Annual carbon payments for agroforestry contracts were equivalent to about two months of wages (Groom and Palmer, 2012), but their study could not estimate whether incomes increased because of lack of baseline data. The permanence of payments was also unclear. The field experiment conducted by Jayachandran et al. (2017), although not a formal REDD+ intervention, found positive effects of payments on carbon outcomes, but statistically insignificant effects on incomes. Jagger and Rana's (2017) quasi-experimental analysis based on publicly available data for Indonesia found potentially negative effects on well-being in REDD+ sites.

A recent systematic review of REDD+ found 350 local-level projects in implementation across the tropics (Duchelle et al., 2018). According to this review, no national level study of payments was available, and few existing studies of REDD+ projects provide careful causal estimates of carbon or non-carbon outcomes on the ground. The relatively more numerous studies of well-being outcomes "highlight small or insignificant results" (Duchelle et al., 2018). In a review of 41 REDD+ projects across 22 countries that adhered to some reporting standards, Lawlor et al. (2013) found that participants received a wide range of payments (from USD 1 to USD 134 per year) and that contributions to infrastructure and education services were modest. The more important contribution of these projects was to local tenure security. Other studies have suggested that the effects of REDD+ may even be negative by adversely affecting local politics, institutions and livelihoods (Chomba et al., 2016; Shrestha et al. 2017; Milne et al. 2019).

Overall, existing studies of REDD+ tend to focus on institutional structures, implementation procedures, relationships to past forest sector interventions and issues related to emissions reduction. A number of analyses have examined: a) how differing conditionality structures may affect emissions reduction outcomes (Irawan and Tacconi, 2009; Hoang et al., 2013) and b) linkages from the local to the national and supranational levels (Bluffstone et al., 2013; Kashwan and Holahan, 2014). But few studies, as discussed above, provide clear evidence on poverty outcomes of REDD+, in part because of the slow pace of national level implementation of REDD+ and subsequently because of unclear and complex benefit sharing structures. Because performance-based payments are only now beginning to supplant promise - or preparation-based payments for REDD+, it is difficult to attribute indirect REDD+ benefits from forest restoration or preservation of forests to local poverty reduction. Existing evidence on the effectiveness of REDD+ for poverty reduction is limited, but available analyses point to the critical importance of benefit sharing and stakeholder participation if REDD+ is to contribute to improvements in well-being either through direct transfers or indirectly.

5.5.3 Ecotourism

The tourism industry accounts for nearly 10% of the global economy (WTTC, 2019), and ecotourism is the fastest developing sector of the tourism industry. Much of the literature on ecotourism comprises case studies and reflections. These provide a range of perspectives on the value of ecotourism and different estimates of its contributions to national and local incomes. Defences and critiques of ecotourism both suggest that it is a promising route for generating material benefits for those living in proximity to tropical terrestrial and marine biodiversity.

Two common features of ecotourism are consistent across different definitions: it is a low impact form of tourism that helps conserve nature and it generates socio-economic benefits for local populations to help reduce poverty (Blangy and Wood, 1993; Buckley, 1994; Wunder, 2000). Ecotourism can contribute to poverty reduction in four different ways: 1) improvements in employment and wages of those employed; 2) shared visitor fees for forested locations in protected areas; 3) revenues from visitor purchases of local goods and services; and 4) infrastructure development with spillover effects in areas with high numbers of travellers and visitors. Typically, these benefits do not flow directly to the poor, but may be mediated through owners of hotels and restaurants, government agencies that receive protected area fees, and owners of businesses that sell goods and services to ecotourists. Further, ecotourism revenues result not just from the presence of forests and trees in protected areas but also from the presence of wildlife in forests.

Because ecotourism is such an important and growing part of tourism, estimates of its economic contribution, especially for national economies but in some cases for local communities as well, are widely available. One study on ecotourism used visits to protected areas in selected countries to estimate the total number of protected areas' visitors globally (Balmford *et al.*, 2015). Its estimate of ~8 billion visitors was associated with an amount of USD 600 billion per year in direct in-country expenditure and USD 250 billion per year in consumer surplus. These figures resemble other top down
estimates of total ecotourism revenues. WTTC estimates total direct and indirect contributions of tourism to the global economy at approximately USD 8.8 trillion per year and the share of ecotourism to be 10%, but growing at a faster pace compared to tourism. However, available statistics do not break down total spending by proportions for different social or economic groups.

Studies of ecotourism at the local level typically focus on measures of development and poverty-related impacts in terms of generation of local jobs and incomes, and in some cases the creation of new infrastructure (Snyman, 2012; Wishitemi et al., 2015; Chirenje, 2017). Over the past three decades, thousands of case studies of ecotourism suggest that it contributes effectively both to local employment and incomes, but also that these contributions tend to benefit those who are better off and with the capacity to provide hospitality services to visitors. One example is a study of ecotourism around six Panda Reserves in China that found that ecotourism reduced poverty but increased income inequality, particularly for households residing within the reserves (Ma et al., 2019). A number of other studies have similarly examined local effects of ecotourism and point to positive outcomes in relation to livelihoods, socio-economic development and poverty reduction (Simpson, 2012; Yi-Fong, 2012; Snyman, 2017; Lonn et al., 2018).

5.5.4 Small and medium forest enterprises

Small and medium forest enterprises are smallscale forest-based businesses that generate income from a diversity of forest-related activities and products, including timber and fuelwood producers, carpentry shops, NTFP producers and ecotourism (Macqueen, 2008). Mayers et al. (2016) stated that about USD 125-130 billion of gross value-added may be contributed by SMFEs worldwide and 80-90% of all forestry enterprises in many countries are SMFEs. Employing at least 20 million people worldwide (FAO, 2013). SMFEs can play an important role in the reduction of poverty as they generate employment opportunities and spread wealth locally (Kozak, 2007; Tomaselli et al., 2012; Sanchez Badini et al., 2018). Positive evidence of this role comes from, inter alia, Bolivia, Brazil, Burkina Faso, Cameroon, China, The Gambia, Guatemala, Kenya, Mexico, Papua Guinea, Peru, Nepal and South Africa (Macqueen, 2008; Tomaselli et al., 2012; Foundjem-Tita et al., 2018). Among the factors that explain this role in poverty reduction, Kozak (2007) mentions that: i) SMFEs tend to be labour-intensive; ii) they may thrive and grow under favourable conditions; iii) they cater to local and domestic markets; iv) they empower local entrepreneurs; and v) they pursue other objectives that include the distribution



The transformation of forest products into consumer goods can support local economies (Oaxaca, Mexico) Photo © Reem Hajjar

of earnings among stakeholders, the participation in policy dialogues and the contribution to community development. They also support decentralisation, tenure and access rights, and empowerment of vulnerable groups such as women and Indigenous communities (Sanchez Badini et *al.*, 2018). Yet, despite a strong theory of change and many case studies showing their positive contributions to local *prosperity* (Macqueen, 2008; Macqueen et *al.*, 2020), there have been limited impact assessments linking SMFEs directly to poverty reduction.

The difficulty in stating their impact in more generalisable terms is partly due to the diversity of contextual conditions in which they operate that may help or hinder their success (see Chapter 4). In a global literature review, Sanchez Badini et al. (2018) identified 12 critical success factors, essential for creating favourable enabling business environments for SMFEs to thrive and work towards poverty alleviation. These are: a stable and transparent macroeconomic setting; simplified and proportional regulatory frameworks; nuanced approaches to forest law enforcement; tenure security and clarity; devolved management and land use planning rights; appropriate and accessible markets; sufficient natural capital; available and accessible financial capital; sufficient forest management, business management and organisational capacities; and clustering through the creation of networks, associations, and cooperatives.

5.5.5 Market access

Improving market access in the context of poverty alleviation, narrowly defined, refers to interventions that enhance the physical (e.g. roads or infrastructure) and technical (e.g. intermediation networks, means of transportation) conditions for smallholders to access markets for selling their produce and/or to access inputs and services. In a wider sense, it refers to a disparate array of actions that affect the relationships of smallholders in the markets including actions for enhancing their capacities to engage in those markets and accrue benefits from market participation, such as reversing regulatory barriers, reducing market risks and transaction costs, and building capacities (Chamberlin and Jayne, 2013). While improving roads and infrastructure is often associated with development interventions, it may result from private investments (e.g. logging or plantation companies) that indirectly affect market access to local populations making use of those roads (Kleinschroth and Healey, 2017), stimulating as well the emergence of informal intermediation networks (Mejía and Pacheco, 2014).

The theory of change behind actions aimed at improving smallholders' access to markets assumes that enhanced market participation leads to positive impacts on household income and poverty alleviation (IFAD, 2015). Yet, greater market engagement may also increase risks or the ability for smallholders to capture economic rents, which may flow to actors better positioned in the value chain (Pacheco, 2012). For benefits to accrue to smallholders, several factors, processes and conditions shaping smallholders' market engagement have to be reversed or improved. These include technical, economic, policy and regulations, and institutional factors. Yet beyond markets, overall outcomes of market participation concern other conditions that facilitate access to other factors (e.g. technology, infrastructure, finance) (Torero, 2011). It is widely recognised that expanding market opportunities for smallholders, particularly in markets of high-value products, has positive impacts for smallholders (Russell and Franzel, 2004). While there is extensive evidence of how improving smallholders' access to agricultural markets affects livelihoods (IFAD, 2015), the evidence of this link for forests and trees is relatively slim.

Market access interventions can be traced back to rural development projects aimed at building farm-to-market roads, that evolved into integrated rural development programmes in the early 1970s (Ruttan, 1984), yet over time those evolved into programmes for enhancing agricultural competitiveness embracing the different market dimensions, and efforts to mainstream sustainable natural resources management. The latter included initiatives to enhance market access and competitiveness for smallholders to benefit from timber as well as NTFPs extraction, and tree-crops production in agroforestry systems. Market access is a key ingredient of rural development and poverty alleviation projects, and it has been widely embraced, with some variations, particularly across developing countries.

Clear evidence of the impacts that enhancing market access has on alleviating poverty of smallholders in forest landscapes is limited and context-dependent. In Ghana, improving roads may lead to better market integration and higher yields, and improved land use while reducing farm expansion into forests (Acheampong *et al.*, 2018). In Vietnam, enhancing access (through both physical and technical aspects) to markets (mainly international markets for timber and high-value products) may have had positive income benefits for smallholders (Frey *et al.*, 2018). However, some studies show that economic returns to smallholders remained low even after the improvement of road infrastructure, such as in Brazil, suggesting that improving physical connectivity is not sufficient to increase rents of smallholders (Oliveira *et al.*, 2019).

The variation in outcomes suggests the importance of looking at the other factors and conditions explaining such variation. For example, outcomes may be defined by location, access to means of production transaction costs, access to infrastructure, technology, transportation and market/ governmental institutional relationships (Torero, 2011). The poverty impacts of interventions aimed at improving market access, in the context of wider efforts for value chain development, often depend on the households' levels of pre-existing asset endowments (Donovan and Poole, 2013). When specifically considering smallholder forestry and tree-farmers, the most important variables may include clear ownership of trees, reliable markets, sympathetic legal and regulatory frameworks, and availability of technical options (Midgley et al., 2017). Additional ingredients identified by comparative studies when assessing smallholder engagement in markets, particularly in agroforestry, are also access to information and contractual agreements (Russell and Franzel, 2004). In sum, multiple factors, processes and conditions can influence the benefits that smallholders may obtain from improved market participation, some of which are reviewed in this chapter.

The evidence provided here is comprised of different case studies, with a focus on small-scale timber *plantations*, NTFPs and some high-value agroforestry products. There is a need for more systematic assessments across a variety of situations of market engagement in order to further develop the evidence concerning the linkages between market access and poverty reduction, and to determine whether the outcomes in the forestry sector may differ from those in the agricultural sector.

5.5.6 Forest producer organisations

Forest producer organisations (FPOs) are any formal or informal group, association, cooperative or union of forest producers (Tirivayi *et al.*, 2018), with a purpose of producing, processing or marketing goods originating from forests, including timber and wood products as well as commercial NTFPs (Pasiecznik and Savenije, 2015). FPOs vary in size, composition and legal form (deMarsh *et al.*, 2014; Tirivayi *et al.*, 2018), and are found across the world (e.g. Pasiecznik and Savenije, 2015; FAO and Agri-Cord, 2016).

The theory of change linking FPOs to poverty alleviation holds that forest-based producers face a number of challenges that impede their abilities to benefit from economic opportunities provided by forests, including insecure tenure; disorganisation; poor access to markets, services and information; lack of capacity; and exploitation by more powerful actors (Macqueen, 2008). FPOs can help address these challenges by facilitating the aggregation of products; enhanced bargaining power; better access to capital, inputs, technical services and markets; as well as increased political power (deMarsh et al., 2014; Pasiecznik and Savenije, 2015; Hajjar and Kozak, 2017; Tirivayi et al., 2018). In turn, overcoming these challenges is expected to increase members' incomes (FAO and AgriCord, 2016).

A few studies explicitly assess the performance of producer organisations in terms of poverty alleviation specifically in a forest context. In Ethiopia, an econometric analysis of five forest-adjacent communities found that following the devolution of forest rights, local communities gained access to frankincense production and trading by organising themselves in cooperatives (Tilahun et al., 2016). Cash income from frankincense cooperatives resulted in a 3.6% reduction in poverty rates among member households, as well as significantly higher incomes and lower poverty levels than non-members, though the authors also note that membership in cooperatives was biased towards relatively better-off households. In Côte d'Ivoire and Ghana, a study of 453 cacao producers across six sites found forest cooperative members to generate relatively higher incomes from cacao than non-members (Calkins and Ngo, 2010). And in Turkey, a study analyzed socio-economic household survey data from 203 small-scale timber producing villages, and also found cooperative members to have higher incomes in comparison to non-members - though wealthier households were significantly more likely to be members (World Bank, 2017).

A number of largely qualitative case studies across a range of forest commodities indicate that FPOs can contribute significantly to poor members' incomes (e.g. Tiveau, 2008; Pandit *et al.*, 2009; Pasiecznik and Savenije, 2015; Tieguhong and Schure, 2015; Humphries *et al.*, 2020). However, not all of these studies provided clear accounts of the data and methods used to calculate the claimed improvements in income (e.g. Pasiecznik and Savenije 2015; FAO and AgriCord, 2016). Tirivayi *et al.* (2018) also suggested that FPOs can play an important role in providing various social protection services, though noting that their study found no evidence on the effectiveness of those services.



Sharing knowledge to improve the chances of success of a cashew (Anacardium occidentale) plantation in Benin Photo © Dan C. Miller

Positive effects on incomes or poverty alleviation were attributed particularly to improved access to more lucrative markets or buyers (Burke, 2010; Pasiecznik and Savenije, 2015), collective bargaining (Tiveau, 2008; Ahenkan and Boon, 2010) and collective ownership of productive resources (Dammert, 2019), facilitating access to training, technical assistance and credit (Birchall, 2003; Calkins and Ngo, 2010; Pasiecznik and Savenije, 2015; FAO and AgriCord 2016; World Bank, 2017), and targeted development interventions and livelihood-diversification interventions (World Bank, 2017). In Burkina Faso's largely female-dominated shea nut value chain, 76% of surveyed women noted improvements in their financial situation as a result of their participation in shea producer groups (Chen, 2017). Evidence from Latin America found that collective organisation was instrumental to obtaining certification for forest products (Duchelle, 2009; Dana and Mallet 2014), while Mala et al. (2012) found FPOs to be instrumental in strengthening the bargaining power of NTFP producers in Mali vis-à-vis traders. Even in instances where incomes are limited, cooperatives

can provide employment to marginalised groups with limited alternatives (Burke, 2010; Shackleton *et al.*, 2011). Studies have also identified key enabling conditions, including: devolution of forest tenure or forest product user rights to cooperatives (Tilahun *et al.*, 2016; World Bank, 2017), technical and financial assistance from external actors (Brown *et al.*, 2011; FAO and AgriCord, 2016; Humphries *et al.*, 2020), as well as direct relationships between FPOs and international buyers, particularly with respect to niche markets (Elias and Carney, 2005; Lybbert *et al.*, 2011).

However, membership fees and other upfront investments associated with FPOs can effectively work to exclude the poorest community members (Kazoora *et al.*, 2006; Oduro and Osei-Akoto, 2008; Pandit *et al.*, 2009; Atmiş *et al.*, 2010; Shiferaw *et al.*, 2011). Exclusionary institutional arrangements, such as one-member-per-household rules, as well as time and labour requirements associated with participation, can serve to specifically exclude women from participating in FPOs (Stoian *et al.*, 2018). In addition, while Elias and Arora-Jonsson (2017) found the shea union in Burkina Faso to build many members' social capital and strengthen cohesion, social divisions including along lines of gender, age and ethnicity often influenced processes of inclusion and exclusion.

A few studies (e.g. Atmiș et al., 2010; le Polain de Waroux and Lambin, 2013) found that FPO membership had no or limited impacts on poverty alleviation; Markelova et al. (2009) cautioned against generalising from successful case studies since failures tend to receive less attention. Indeed, a number of studies highlighted challenges facing FPOs, including: poor tenure security; complex or weak regulatory environment; poor market access; limited scope and scale of operations; limited investment capital; price fluctuations; conflict and elite capture (Molnar et al., 2007; Macqueen, 2008; le Polain de Waroux and Lambin, 2013; Schure et al., 2013; Tirivayi et al., 2018). Critically, such challenges may affect the financial viability of cooperatives over time (Dossa, 2012), particularly in instances where costs are subsidised by external donors (Humphries et al., 2020). Indeed, members' poverty can constrain economic growth, if FPO members prioritise immediate needs over longterm investments (Atmis et al., 2010).

5.5.7 Company-community partnerships

Forest-related company-community partnerships refer to the full range of formal and informal relationships, agreements, deals between communities and companies with the expectation of realising gains from sharing capacities and risks (Mayers, 2000; Ros-Tonen et al., 2008; Le Tourneau and Greissing, 2010). The partnerships typically include contract production (discussed in Section 5.5.8), joint ventures and equity sharing agreements, farm forestry support, farm forestry crop-sharing, concessions leased from communities, and group certification with company-leased land from farms, among others (Andrew et al., 2000; Mayers and Vermeulen, 2002; Ojwang, 2000). The products from these deals include timber products such as logs and wood fibre, and a variety of NTFPs such as nut oils, rubber, resins, juice pulp, and agricultural products for the cosmetic, food, automobile and pharmaceutical industries. Company-community partnerships are expected to result in the vertical integration of disconnected rural forest enterprises into global supply chains by providing rural producers with better access to markets and capacity (Mayers, 2006; Vermeulen et al., 2008), improving incomes and net returns from land and labour (Brubacher, 1998; Ojwang, 2000; Mayers, 2006).

A number of case studies have described vari-

ous company-community contracts, but few have carefully examined the explicit impacts of these contracts on poverty alleviation (Mayers, 2006). One study found that communities in the Brazilian Amazon engaged in logging contracts with a company saw increases in household incomes relative to communities not participating in such contracts, without compromising NTFP harvests (Menton et al., 2009). The one-off income from timber sales was used for investing in agricultural production or household infrastructure. Other case studies describe benefits from company-community partnerships such as: increased incomes; access to markets and sometimes premium prices; employment opportunities; improving land use options; securing land rights; and upgrading social infrastructure (Le Tourneau and Greissing 2010; Mayers and Vermeulen, 2002; Menton et al., 2009; Morsello et al., 2012; Vermeulen et al., 2008). Out-grower schemes for commercial forest products (further discussed in 5.5.8) have also improved human, physical, and financial capitals of organised smallholder producers in Ghana, Papua New Guinea and South Africa among other places, and improved their resilience to shocks and vulnerabilities (Warner and Bauer, 2002; Mayers, 2006; Ntisiful, 2010). In many cases, however, the inequitable distribution of benefits within communities can deepen social inequity and weaken social cohesion, while power imbalances between partnering communities and companies can increase community dependence on external actors and result in unfair or inequitable distribution of benefits in these partnerships (Mayers, 2006; Ros-Tonen et al., 2008; Menton et al., 2009; Le Tourneau and Greissing, 2010).

By building consensus on partnership aims, governance reforms that secure tenure and land rights for local communities, improving capacity of local communities to negotiate partnerships, equitable risk sharing, long-term commitment to the partnership, ethical business practices, and periodic evaluations, these partnerships can significantly contribute to poverty alleviation and environmental outcomes (Desmond and Race, 2000; Mayers and Vermeulen, 2002).

5.5.8 Contract production

Contract production, a type of company-community partnership, is a form of vertical coordination within value chains in which production is carried out through a fixed-term formal or informal sales agreement between a producer and a processing or marketing company (Little and Watts, 1994). While contractual attributes are highly diverse (Bellemare and Lim, 2018), they often involve commitments by a company to provide inputs, credits, technical support, a guaranteed output market and/or fair offtake prices (Da Silva and Rankin, 2013). Producers in turn commit to fulfilling the company's process or product requirements, delivery schedules and exclusivity terms (Eaton and Shepherd, 2001).

While typically commercially driven, such arrangements are widely viewed by policymakers and development practitioners as a promising tool to overcoming the pervasive market imperfections that perpetuate rural poverty (Meemken and Bellemare, 2019). Companies can more effectively manage the high transaction costs associated with open market sourcing, while small producers gain access to new and often more secure (global) markets and the resources needed to produce marketable surplus (Ton *et al.*, 2018; Meemken and Bellemare, 2019). This has the potential to translate into higher incomes, more resilient livelihoods, competitiveness and total factor productivity gains.

Contract production in developing countries is especially prevalent in the agricultural sector (Grosh, 1994; Oya, 2012). In addition to traditional cash crops such as sugarcane, tobacco, cotton and tea, across much of the tropics, tree crops such as cacao and oil palm are also widely produced under production contracts. In countries such as India, Thailand and South Africa, timber species such as teak, pine, and eucalyptus are also commonly cultivated under such arrangements (Sartorius and Kirsten, 2002; Boulay and Tacconi, 2012). Since many NTFPs suffer from diseconomies of scale, and quantities and qualities can be difficult to control (Pierce et al., 2008), few are harvested and/ or processed under contract. Most documented cases come from the Amazon, typically involving some form of 'community-company partnership agreements' for comparatively high-value NTFPs such as Brazil nut, palm hearts and acai (van Andel, 2007; Morsello et al., 2012).

Because many agribusinesses are often unable to achieve economies of scale through plantation production systems, labour conflicts and restrictive tenure regulations, supplementary smallholder sourcing is often an economic necessity (Vermeulen and Cotula, 2010; Ton *et al.*, 2018). While some critics contend that contract production can be an exploitative and extractive mode of production due to the inherent power imbalances and uneven dependency structures (Little and Watts, 1994; Oya, 2012), several empirical studies employing econometric techniques indicate that contract production has been widely associat-

ed with household income and farm profitability gains (Bolwig et al., 2009; Miyata et al., 2009; Bellemare, 2012; Narayanan, 2014; Girma and Gardebroek, 2015). However, some of these studies can be critiqued due to, inter alia, selection, publication and reporting biases and weak identification (Ton et al., 2018; Bellemare and Bloem, 2018). Other dimensions of welfare are less addressed by this literature, but some studies do point to positive food security, gender equality and subjective well-being effects (Morsello et al., 2012; Dedehouanou et al., 2013; Bellemare and Novak, 2017). In demonstrating that participation may intensify environmentally unsustainable activities and household labour burdens, the study by Morsello et al. (2012) of Brazil nuts in Bolivia and Brazil does point to potential participation trade-offs.

Since most of these studies are based on case studies of specific contracting relations or are confined to specific geographic areas, findings do tend to suffer from a lack of external validity (Meemken and Bellemare, 2019). One of the few studies based on nationally-representative survey data suggests that the oft-cited income effect is likely overstated (Meemken and Bellemare, 2019). Furthermore, with the exception of Morsello et al. (2012) and Girma and Gardebroek (2015), much of the quasi-experimental literature is focused on agricultural (tree) crops. Despite the abundance of studies on timber contract production, such studies are largely qualitative, published as grey literature, reliant on descriptive statistics and/or do not consider counterfactuals (e.g. Cairns, 2000; Desmond and Race, 2000; Mayers and Vermeulen, 2002; Howard, 2005).

There is some evidence that contract production can exacerbate social differentiation. For example, most studies control for, and identify self-selection biases and processes of involuntary exclusion. They observe that contract production participants tend to be more affluent, educated and asset endowed (Miyata et al., 2009; Bellemare, 2012; Narayanan, 2014). The participation of vulnerable and marginalised groups can be constrained by the perceived opportunity costs of allocating finite land and labour resources to the contracted commodity, reduced risk tolerance and companies imposing eligibility criteria to control transaction costs (Bellemare, 2012; Ton et al., 2018). This calls into question whether contract farming - in the absence of additional institutional support and formal checks and balances - adequately serves as a tool for inclusive rural development. Positive spillover/contamination effects may nevertheless materialise through job creation, infrastructure investments, and technology/skill transfers (Ton *et al.*, 2018). However, few studies explicitly assess such externalities. Meemken and Bellemare (2019) found positive spillover effects onto local labour markets, but did not observe any positive welfare impacts on non-participant households.

5.5.9 Certification

Certification is "a procedure by which a third party gives written assurance that a product, process or service is in conformity with certain standards" (ISO, 1996). Certification relevant to forestry can take many forms, but has long been found in the form of voluntary sustainability standards in high forest-risk commodity sectors. This includes the Forestry Stewardship Council (FSC), the Roundtable on Sustainable Palm Oil (RSPO), Rainforest Alliance/UTZ, Fair Trade, and the Sustainable Agriculture Network (SAN), which certify forest-related commodities such as palm oil, timber, pulp and paper products, coffee and cacao, and a range of agricultural crops. Each of these systems has its own sustainability standards that include a wide range of social and environmental criteria that producers have to comply with to become certified. In some countries and sectors, producers and governments have also begun developing mandatory certification systems (e.g. palm oil in Indonesia) and national standards (e.g. the Programme for the Endorsement of Forest Certification (PEFC) (Overdevest, 2010; Schouten and Bitzer, 2015; Pacheco et al., 2018). Public standards have been critiqued as being inferior to, and for undermining, private standards (Overdeveest, 2010; McCarthy, 2012; Hospes, 2014), while others suggest that they help prepare smaller producers for other standards (Higgins and Richards, 2018; Schoneveld et al., 2019).

Certification by smallholders and community-based organisations is widely viewed as an important rural development mechanism. The theory of change holds that adoption of better practices can enhance productivity and resilience, and reduce production risks, while creating opportunities to sell to buyers that can offer improved terms of trade (e.g. price premiums, offtake guarantees, services). These could lead to higher and more stable income for smallholders, thus contributing to poverty reduction.

The evidence is mixed as to whether these benefits materialise in practice. Many certification systems were developed in response to, and accommodating the needs of, corporate producers, and therefore respond poorly to the interests of smaller producers (Brandi *et al.*, 2015). Smaller producers typically face comparatively high compliance barriers and are often not incentivised, or are unable, to adapt their production practices and strategies (Schoneveld *et al.*, 2019; Brandi *et al.*, 2015). They often lack the necessary resources to absorb certification costs, or the capacity to adopt prescribed practices and to manoeuvre the public bureaucracy in order to obtain the needed legal documentation (Brandi *et al.*, 2015; Schoneveld *et al.*, 2019). Furthermore, price premiums rarely suffice and access to alternative, oftentimes informal, market outlets reduce the certification imperative, especially in the timber and palm oil sector (Burivalova *et al.*, 2017).

FSC also certifies forests managed for NTFPs, but certification of NTFPs has proven to be especially problematic due to inter alia low profit margins, underdeveloped markets, and legality issues (Pierce et al., 2008). For NTFPs, much of the existing literature has examined how and under what conditions they can be certified, but not their impacts. However, evidence from Brazil suggests that NTFPs serving large, mature markets can be successfully certified with the right government, donor and civil society support (Guedes Pinto et al., 2008). One study on Brazil nuts suggested the economic impacts of NTFP certification is mixed and highly context-specific (Duchelle et al., 2014). A study exploring the gendered impacts of organic certification of shea nuts similarly provided inconclusive evidence (Kent, 2018). Nevertheless, with the exception of the cacao and coffee sectors, certification rates remain low amongst small producers and community forest management units and enterprises.

Despite the systematic barriers to adoption, a multitude of studies on smallholder cacao certification in Ghana and Ivory Coast demonstrate that tangible social benefits can accrue. The adoption of better practices can increase agronomic capacity, yields and resilience to shocks; thereby positively impacting farm-level profitability, which often translates into improved household income and (financial) well-being (Krain *et al.*, 2011; Paschall and Seville, 2012; Gockowski *et al.*, 2013; Ingram *et al.*, 2014; Astrid Fenger *et al.*, 2017). Since certification typically demands improved farmer organisation, (support for) certification also contributes to the development of more professionalised producer associations (Ingram *et al.*, 2018a).

In the timber sector, the limited evidence on the impacts of community forest management certification suggests that certification is rarely financially attractive, but can help strengthen land tenure and community empowerment (Cerutti *et al.* 2014; Burivalova *et al.*, 2017). Because of high costs and market uncertainties, certification of small-scale tree growing similarly rarely translates into net (economic) benefits (Maraseni *et al.*, 2017; Maryudi *et al.*, 2017; Ling *et al.*, 2018; Flanagan *et al.*, 2019). Nevertheless, there are some documented examples of community-based forest enterprises overcoming financial viability issues when partnering with social movements and government agencies (Macqueen *et al.*, 2015).

The scientific quality of existing studies is mixed. Many of the cacao impact studies are published as grey literature, with unclear methods and analytical rigour (Ingram et al., 2018a). Similarly, few studies on certified community forest management conform with standards for impact assessment (Burivalova et al., 2017) and neither do studies on palm oil, NTFPs or timber. Existing literature also tends to focus heavily on practices, productivity gains and income, but often neglects to critically interrogate other pertinent impacts such as effect on household labour burden, food security and portfolio composition, as well as broader societal effects. Where certification can become an important lever for poverty alleviation, it is usually when smallholders are more actively engaged or their needs considered in the design of standards (Loconto and Dankers, 2014).

5.5.10 Zero deforestation commitments

Supply chain initiatives are commitments made by private sector entities - either individual companies or groups of companies - to adopt more sustainable sourcing policies, in relation to one or more commodities (e.g. timber, soy, palm oil, beef), often by a specific date (Brown and Zarin, 2013; Lambin et al., 2018). They include "aspirational goals by single companies or coalitions of actors, corporate codes of conduct and sustainability standards that, in some cases, are implemented through certification schemes and moratoria" (Lambin et al., 2018). As they relate to forests, many of these supply chain initiatives are frequently referred to as "zero deforestation commitments" (ZDCs). Such ZDCs can be implemented at the property level or across larger jurisdictions and frequently apply to all producers within the adopting-company's supply chain (Meyer and Miller, 2015). As of March 2017, Lambin et al. (2018) noted that at least 447 producers, processors, traders, manufacturers and retailers had made at least 760 public ZDC commitments.

Many ZDCs and other supply chain initiatives contain criteria that relate to social dimensions of sustainability. Zero deforestation commitments frequently include guarantees to improve a company's conduct towards various groups of people,

including Indigenous and other forest-dependent people who live in and around forests used for commodity production; labourers employed by commodity-producing or processing companies; and smallholders who produce commodities and sell them into larger supply chains (Newton and Benzeev, 2018). Therefore, if companies that adopt ZDCs honour their pledges, then poverty may be reduced in one or more ways. For example, if Indigenous and forest-dependent people are properly compensated for their land, or if they are given greater opportunities to retain control or ownership of, or access to, forested lands, they may be better off relative to a scenario in which land is 'grabbed' by commodity-producing companies (Liao et al., 2016). If labourers are fairly compensated for their work, and if children are not exploited in commodity production but are rather free to pursue education, then levels of poverty may in time be reduced among communities of people employed in the commodity sector. And if smallholder commodity producers are fairly paid for their production, and are not excluded from supply chains by costly procedures, then they may be more likely to earn higher incomes. Many of these examples, and many of the environmental and social criteria contained within supply chain initiatives including ZDCs, refer to the elimination of worst-practices. Such initiatives often do not promote or strive for best-practices. But even the elimination of worst-practices might conceivably reduce poverty, if doing so removes barriers and constraints to individuals and communities.

A recent review of the impacts of ZDCs on social outcomes, including poverty, identified no evidence of a relationship between supply chain initiatives and poverty alleviation (Newton and Benzeev, 2018). Larson *et al.* (2018) reported a suggestion by a government official that the stringency of the Indonesian Palm Oil Pledge (a supply chain commitment by five major palm oil companies) may have negatively affected smallholder income by making it more difficult for them to sell their produce. In sum, we found no evidence that supply-chain commitments have reduced poverty or improved human well-being. In part, this could be due to the primacy of the environmental focus of many such pledges.

5.5.11 Boycotts

Consumer boycotts of timber from particular companies, countries or regions have been promoted as a mechanism by which to encourage more sustainable and more responsible timber production. The theory of change states that boy-

cotts will work if a firm's profit is sufficiently reduced to prompt it to adopt changes (Delacote, 2006). However, the likelihood of success of a boycott can be low, as a consequence of: a) the challenges of coordinating consumer behaviour, b) the allure of free-riding behaviour, c) the paradox that those consumers who have the greatest capacity to affect a firm's profits also have the highest opportunity cost of engaging in a boycott, and d) the challenge of only small numbers of consumers being sufficiently concerned (Delacote, 2009). If a boycott does overcome these challenges, the theory of change suggests that by being temporarily excluded from markets, economies and firms that depend on timber production are more likely to adopt more sustainable methods, including by becoming certified - for example, through FSC certification (see Section 5.5.9). In turn, FSC certification is theorised to result in economically just and equitable production standards, which may confer greater benefits on labourers, employees and communities living in and around timber-production areas. Thus, while timber boycotts have principally been motivated by environmental concerns, there is a plausible causal mechanism by which they could lead to social benefits, including poverty reduction. That said, if a boycott persisted for any length of time it could also plausibly exacerbate poverty locally if it were to damage the local market, jobs and income opportunities.

Various organisations have called for boycotts of non-certified timber (Damette and Delacote, 2011) and boycotts are reported to have been successful in catalyzing the cessation of logging old-growth forests and adopting more sustainable practices (Innes, 2006). For example, boycotts organised by the Rainforest Action Network and others have resulted in hundreds of timber retailers adopting FSC certification (Innes, 2006). Indeed, along with broader public concern about deforestation, boycotts are reported to have been responsible for the rise of FSC and other certification programmes (Putz and Viana, 1996).

Any impacts of boycotts on poverty are most likely to be manifested through the adoption of sustainability standards such as FSC certification. As such, we encountered no direct evidence that boycotts have led to measurable poverty reduction or to changes in other measures of human well-being. But to the extent that boycotts are effective in promoting the adoption of sustainability standards, and to the extent that the adoption of sustainability standards in turn leads to poverty reduction, there may be an indirect cause-and-effect connection between boycotts and poverty reduction.

5.6. Forest and Tree Management Levers

5.6.1. Agroforestry

Agroforestry refers to the intentional integration of trees and other woody perennials in crop and livestock systems. Agroforestry practices can improve farmer livelihoods and resilience through diversifying agricultural production and income sources. For example, a large-scale study of five countries in sub-Saharan Africa found that a third of rural smallholder households grow trees, which contribute an estimated 17% of total annual gross income for these households (Miller et al., 2017). Several other levers can be used to support the more optimal integration of trees into farming systems. These include extension programmes, PES, certification schemes, linking producer organisations to out-grower schemes, strengthening seed delivery systems, improving tenure security, and addressing policy and institutional barriers, such as laws that prevent or overly regulate the harvesting of protected forest species on farm.

There have been a few notable impact evaluations of agroforestry interventions that evaluated economic, social and ecosystem services outcomes (Kuntashula and Mungatana, 2013; Bostedt *et al.*, 2016; Coulibaly *et al.*, 2017; Amadu *et al.*, 2020; Hughes *et al.*, 2020). These have shown that extension and training, coupled with the provision of tree germplasm in some cases, led to increased agroforestry adoption. This in turn led to increased yields, household income, food security and dietary diversity. For example, in Malawi, agroforestry adoption contributed to a 20-35% increase in yields, which provided increased income opportunities as well as better food security (Coulibaly *et al.*, 2017; Amadu *et al.*, 2020).

However, evidence on the linkages between agroforestry adoption and such impacts acquired through the use of rigorous impact evaluation methods such as randomised control trials and quality quasi-experimental studies, is extremely limited (Miller *et al.*, 2020). The few studies that have used impact evaluation methods tend to not directly assess poverty outcomes, but instead use proxies such as income, expenditure and food security. Additionally, existing impact evaluations often have a high risk of bias (Miller *et al.*, 2020).

Other studies that use non-randomised regression analysis provide evidence that agroforestry can contribute towards alleviating poverty, but these studies are also limited. In Bangladesh, a participatory agroforestry programme was associated with significant poverty reduction, measured using the headcount index, poverty gap index and Foster-Greer-Thorbecke methods to assess levels of poverty and extreme poverty (Islam *et al.*, 2012). The programme improved the poverty situation of 33% of participating households, reduced the poverty gap of 10% of participating households, and reduced the severity of poverty of 5% of participating households. Agroforestry programmes like this enable farmers to diversify their production and increase their income sources, including through the sale of timber, fuelwood and tree crops, thereby contributing to poverty reduction. Dairy production and returns can also be enhanced among producers who make use of leguminous high protein fodder species (Place *et al.*, 2009).

In other contexts, trees are established to enhance the provisioning of ecosystem services, such as carbon sequestration. In these cases, incentive or market-based programmes (Section 5.5) are used to promote agroforestry (Hegde and Bull, 2011; Pagiola et al., 2016; Haggar et al., 2017). A common example is coffee agroforestry, where coffee trees are grown under the shade of other tree species. Certification schemes allowing farmers to certify that their products were sustainably produced, such as Fairtrade, Organic or Rainforest Alliance coffee certifications, enable a market-based approach for farmers to receive higher prices for their products for practising agroforestry (e.g. shade-grown coffee) and using other sustainable practices. The results of such schemes, however, are mixed, highlighting the importance of context-specific, evidence-based design of such programmes. There is also little evidence of their long-term effectiveness.

There is evidence of a difference in the labour burden between women and men in some agroforestry systems (Kiptot and Franzel, 2012). Typically, women tend to be burdened with much of the labour of planting and maintaining trees and are often excluded from the higher value agroforestry enterprises, such as timber and commodity crops. Women participate in the lower-value enterprises that men avoid, such as collection of fuelwood, fodder, mulch and indigenous fruits and vegetables. These lower-value enterprises can still contribute a significant portion of women's annual income (between USD 7 and USD 2,629 annual revenue from agroforestry products) (Kiptot and Franzel, 2012). Several impact evaluations of different agroforestry programmes found that women have lower participation in and benefits from agroforestry interventions (Place et al., 2005; Hegde and Bull, 2011) due to different endowments, discrimination and exclusion, or inequitable programme design. Women's disproportionate tenure insecurity is another major disincentive that restricts their participation in agroforestry (Quisumbing *et al.*, 2014).

Along with improving incomes through increased yields or incentive provision, agroforestry can enhance resilience and support farmers to adapt to climate change (Verchot et al., 2007; Thorlakson and Neufeldt, 2012; Quandt et al., 2019). For example, agroforestry can diversify the products that farmers sell, providing additional sources of income, particularly in times of need. Similar to livestock, small-scale farmers often sell timber and other products on farm to meet a pressing expenditure need, such as school and hospital fees (Schreckenberg et al., 2002; Place et al., 2009; Kiptot and Franzel, 2011). Some agroforestry practices can also help farmers withstand climate-related shocks through preserving soil moisture during dry spells and by protecting crops from floods (Garrity et al., 2010; Thorlakson and Neufeldt, 2012).

Overall, agroforestry programmes have the potential to contribute towards poverty alleviation, but additional research is necessary to understand how this potential can be better exploited across the planet's heterogenous landscapes and socio-economic contexts.

5.6.2 Forest restoration, reforestation and afforestation

The return or expansion of forest cover can be achieved in many different ways. Forest restoration refers to the return of a near to original forest ecosystem (Lamb and Gilmour, 2003). More commonly used terms refer to ecological restoration, ecosystem restoration or forest landscape restoration (FLR), although a recent review found at least 24 different terms associated with restoration (Mansourian, 2018). Widely used approaches to return trees to a landscape include afforestation (i.e., planting forests on land not classified as forests) and reforestation (i.e., planting trees on land classified as forests) (Stanturf et al., 2014). Forest restoration activities are further differentiated by the amount of human involvement they require. Natural regeneration can improve the ecological function of degraded forests and return converted areas into forest land with minimal or no human involvement (Chazdon and Guariguata, 2016). Alternatively, forest restoration can also occur through human activities that prepare sites, remove unwanted species or individuals and plant trees (Le et al., 2015). Though forest restoration has gained visibility as a cost-effective method for removing atmospheric carbon

(Stanturf et al., 2015; Busch et al., 2019; Bastin et al., 2019; Brancalion et al., 2019), its ability to promote livelihoods and well-being outcomes remains uncertain. Recognising this, FLR was defined in 2000 as an approach with the twin goals of enhancing ecological integrity and human well-being (Mansourian, 2005). A review of an FLR project in Madagascar, for example, found that 1,400 households were able to benefit from alternative income generating activities promoted under the project, including improved rice production, which led to a 2-4 fold increase in production (Mansourian et *al.*, 2018).

Forest restoration, reforestation or afforestation can produce livelihood impacts through direct and indirect benefits (Erbaugh and Oldekop, 2018). Direct benefits refer to what households receive from the implementation of forest restoration activities. They may include payments for tree planting activities, payments for ecosystem services that restored forests provide, the provision of land or resource rights, or the delivery of training to populations proximate to restored forest areas (Adams et al., 2016). Indirect benefits refer to outcomes that occur as a result of having implemented restoration activities, and they are most likely to affect poverty over longer-time horizons. They include benefits from improved ecosystem function (e.g. greater NTFP availability, water quality and availability, soil fertility, carbon sequestration), strengthened resource rights for local populations and improvements in infrastructure (Adams et al., 2016; Erbaugh and Oldekop, 2018). Though these pathways highlight many possible contributions forest restoration can make to livelihoods and well-being, research on forest restoration finds a mixed impact on poverty alleviation and livelihood benefits.

A growing body of evidence demonstrates how forest restoration, reforestation or afforestation provides direct livelihood benefits. A large-scale afforestation programme in China, the Sloping Land Conversion Project (see Box 5.2), provided subsidies for afforestation activities to low-income, rural households. The programme has demonstrated that afforestation programmes can incentivise the intensification of smallholder agriculture and increase off-farm labour earnings (Zhou et al., 2007; Yin et al., 2014). Small-scale projects have also had positive livelihood benefits. A social forestry programme in South Kalimantan increased farm-based income and natural forest cover (Hiratsuka et al., 2019) and farmer-managed natural regeneration in Ghana increased asset ownership and income diversity (Weston et al., 2015). Though these examples show how direct benefits from forest restoration projects and activities can produce livelihood benefits, assessing whether or not they reduce poverty is constrained by two barriers. First, restoration, reforestation or afforestation activities can lead to direct disadvantages, including the reduction of available agricultural land that decreases on-farm income or employment opportunities (Robbins and Harrell, 2014), the reduction of compensation or monetary benefits (Alix-Garcia and Wolff, 2014; Börner et al., 2017) and the elimination of resource access rights (Barr and Sayer, 2012; Galudra et al., 2014; Urgenson et al., 2014). Second, direct economic benefits rarely (if ever) continue ad infinitum. Ensuring long-term benefits from restored forests complement direct and often short-term benefits may be important for lasting poverty reduction from forest restoration.

The ability of forest restoration to contribute indirect benefits that result in poverty alleviation is inconclusive. However, many studies find that restored forests contribute to a diversification of livelihood strategies and increases in income from timber and NTFPs (Aronson et al., 2010; Le et al., 2012; Adams et al., 2016; Erbaugh and Oldekop, 2018; Ota et al., 2018). For example, secondary forest regeneration coupled with the harvest of a local palm fruit in the coastal Atlantic forest region of southeastern Brazil improved farmer income, as well as soil quality and forest structure (de Souza et al., 2016). Farmer-managed natural regeneration (FMNR), a practice where farmers actively manage and protect natural trees and shrubs to encourage an increase in woody vegetation, can also diversify incomes and improve livelihoods. For example, Haglund et al. (2011) estimated that FMNR contributed between USD 17 and 21 million to gross annual income of rural households (or between USD 46 and 56 per capita) in the region of Maradi, Niger, an 18-24% increase in income. Indirect benefits from forest restoration also accrued to households in central China as a result of the Mountain-River-Lake (MRL) Programme in the Poyang Basin. The MRL Programme is associated with lifting 9 million people out of poverty between 1983 and 2008. Though most of this poverty reduction is attributed to remuneration, improved credit from restoration activities, and access to agricultural technology, indirect benefits such as reduced flooding/soil erosion also contributed to reductions in poverty (Huang et al., 2012). These examples are promising, but they do not rely on counterfactual analysis, and so may falsely attribute poverty reduction to restoration activities. Despite their limitations, these examples show that indirect well-being benefits from restored forests often accrue over years or decades. Rights that ensure rural poor communities are able to use and manage restored forests are therefore key to reducing poverty through forest restoration (Nagendra, 2007; Mansourian and Vallauri, 2014).

Forest restoration can also deliver indirect institutional benefits to communities by providing improved clarity surrounding tenure or enhanced resource rights for newly restored forests (Le *et al.*, 2012; Erbaugh and Oldekop, 2018; Fox and Cundill, 2018). However, research more often finds that the success of restoration is predicated upon clear institutions for resource use (Galabuzi *et al.*, 2014; Call *et al.*, 2017; Legesse *et al.*, 2018). Thus, while the direction of benefits from forest restoration to ecological and economic benefits is clear though loosely substantiated, it remains less certain whether forest restoration tends to generate transparent tenure or vice-versa.

The impacts of forest restoration on poverty reduction are determined by interactions between the mechanism of governance used to implement restoration activities (e.g. CFM, PES), the process of implementation, and specific restoration goals. However, three general themes emerge from the relationship between forest restoration and poverty reduction. First, when forest restoration is associated with livelihood impacts, direct benefits often support the intensification of agriculture and/ or the diversification of livelihoods to include more off-farm earnings and a greater number of products. Second, few studies have conclusively substantiated that livelihood benefits from services provided by restored forests meaningfully benefit proximate households to alleviate poverty. However, when poverty alleviation from restored forests seems likely, it occurs over years or decades. And third, forest restoration success and poverty alleviation from restored forests are closely associated with institutions for resource use.

5.7 Summary of Key Findings

We reviewed the evidence that forest-sector policies, programmes and strategies (i.e., levers) have alleviated poverty. We studied 21 different rightsbased, regulatory, market and supply chain, and forest and tree management levers for which we could identify a plausible theory of change of how implementation of that lever might alleviate poverty (Table 5.1).

Overall, while we found substantial, varied, and context-dependent evidence of these levers being associated with mitigating poverty, including by supporting or improving well-being, we found limited evidence of these levers being associated with reducing poverty (i.e., moving people out of poverty). It is worth reiterating, however, that many of these levers were primarily set up for forest conservation or other non-poverty related objectives, rather than with the explicit aim to reduce poverty.

From the studies that specifically examined poverty reduction (i.e. moving people above a certain poverty-level threshold), some of the strongest evidence came from ecotourism, protected areas – particularly those associated with ecotourism (e.g. Naidoo *et al.*, 2019; Ma *et al.*, 2019), community forest management (e.g. Oldekop *et al.*, 2019) and agroforestry (e.g. Islam *et al.*, 2012). Rigorous studies on payments for ecosystem services show small, but statistically significant, decreases in poverty in some cases (e.g. Sims and Alix-Garcia, 2017).

Out of the studies that more generally examined *poverty mitigation* (i.e. increasing income, assets and other aspects of well-being), a multitude of cases showing positive outcomes came from community forest management (e.g. Rasolofoson *et al.*, 2017), forest producer organisations, (e.g., FAO and AgriCord, 2016), SMFEs (Macqueen, 2008), PES (e.g., Adjognon *et al.*, 2019), tree crop contract production (Morsello *et al.*, 2012) and, to a much lesser extent, REDD+ (in terms of its focus on tenure reforms – e.g. Lawlor *et al.*, 2013; Duchelle *et al.*, 2018).

5.7.1 Differentiated impacts

Few studies provided socially disaggregated information on poverty outcomes by showing how the levers included in the review affected different groups. However, a number of studies highlight the importance of social heterogeneity in the context of the assessed levers, including those levers with the strongest evidence of poverty alleviation. The assessed studies generally attributed socially differentiated outcomes, including differentiated opportunities, benefits, and trade-offs, to a combination of underlying material and sociocultural inequalities and the failure of a given lever to sufficiently account for and address those inequalities. For instance, insufficient financial resources may hinder the poorest producers from complying with formal standards (e.g. Obidzinski et al., 2014) or paying the membership fees for producer organisations (e.g. Shiferaw et al., 2011). While ecotourism may reduce poverty, it also risks increasing income inequality between households (Ma et al., 2019). Gender differences (Stoian et al., 2018), variations in ethnicity (Elias and Arora-Jonsson, 2017) or other axes of social differentiation often accentuate exclusionary outcomes. For instance, a number

of studies on ecotourism noted that women were typically relegated to lower-paying, gender-conforming jobs, while more remunerative positions were taken up by men (Gentry, 2007; Tran and Walter, 2014). Women also experienced a disproportionate loss of income due to forest exclosures associated with a PES programme (Tuijnman et al., 2020), while many agroforestry practices increased women's labour burden, often without generating commensurate or accessible benefits (Kiptot and Franzel, 2012). Women's participation and benefits were lower than those of men in PES programmes in Kenya (Kariuki and Birner, 2016), while in a global comparative study on REDD+, women in project sites reported declines in subjective well-being in comparison to male-dominated groups within the same sites and women in control sites (Larson et al., 2018).

5.7.2 Interpretation

An absence of clear and high quality evidence that forest-sector levers have moved people out of poverty does not necessarily constitute evidence that such levers cannot or even have not reduced poverty. Rather, it appears that relatively few researchers have explicitly explored poverty reduction, per se, through forest sector interventions. Many more studies have explored indicators of poverty mitigation, including impacts on income, assets and well-being. As an example of this distinction, there remains little concrete evidence of whether REDD+ has led to poverty reduction, but well-funded and coordinated efforts have systematically characterised REDD+'s contributions to livelihoods in cases globally. More studies are needed that explicitly speak about poverty reduction as an outcome variable of interest, rather than just poverty mitigation, in order to more fully assess forest-sector levers' impacts on poverty alleviation. Additionally, few studies have examined these phenomena at national or regional scales, instead typically examining impacts at the scale of a few communities or similar level administrative jurisdictions. Larger scale studies are needed to enable rigorous assessments of the role of these levers in relation to poverty.

The mixed conclusions on the efficacy of many of the levers is also an attestation to the importance of contextual differences, including the presence of enabling conditions and contemporaneous drivers of change (see Chapters 2 and 4, and Box 4.1 in Chapter 4), in shaping the effects of different levers. For several levers, we did not find generalisable, clear-cut evidence of impacts, positive or negative, given that conditions on the ground vary widely across the globe. But site-specific studies do show that several levers have contributed to poverty mitigation under certain circumstances and in the presence of key enabling factors, including in conjunction with other levers. For example, having clear and secure local tenure rights to land and forest resources is key to the success of SMFEs, CFM, PES, community-company partnerships and agroforestry. Many SMFEs are reliant on tenure reform, market access, forest producer organisations and formalisation policies, to name a few necessary enabling conditions for their success. Market access alone is an insufficient condition to ensure poverty reduction; other factors enhance the effects of market access - among them the presence of forest producer organisations, certification and contract production. Protected areas in Costa Rica and Thailand were most effective at alleviating poverty when ecotourism opportunities were available. As such, the likelihood of success of a number of levers is intertwined with the functional presence of other levers.

Finally, a number of cross-cutting tools that often support programmatic interventions were not discussed in this chapter, but are relevant to the success of many levers. For example, new and enhanced technologies including equipment upgrades, mechanisation and improved germplasm can be important components of SMFEs, CFM, reforestation and agroforestry (Burney et al., 2015; Haase and Davis, 2017; Hansen et al., 2019 see also Chapter 6, Box 6.2). Financial capital, in the form of credit, aid or subsidies, can be essential in implementing many of the reviewed levers (Macqueen et al., 2008; Humphries et al., 2012; Sanchez Badini et al., 2018). Capacity building, including financial literacy, financial inclusion and improved management practices, often accompany interventions that bring new practices and ventures to producers (Pokorny et al., 2010; Hajjar et al., 2011; Elson, 2012). Safeguards such as free, prior and informed consent (FPIC) and participation in intervention design are increasingly recognised in a rights-based discourse as essential components of interventions aiming to improve the lives of forest-reliant people (Lawlor et al., 2013; FAO, 2018). These supporting components of interventions may in and of themselves have poverty impacts, but we did not have the granularity to isolate and assess those outcomes.

We encountered significant variance in the methods used to study different forest-sector levers. The literature on some levers was dominated by econometric analyses (e.g. PES programmes, protected areas) while the literature on other levers was dominated by qualitative or mixed method

case-studies (e.g. timber contract production). Different methodologies offer competing advantages, including the degree to which one can offer reliable conclusions about the contribution of any given lever to poverty alleviation. For example, probably the most rigorous evidence, in terms of being able to isolate and quantify the impact of forest-sector levers on local people, comes from payments for ecosystem services programmes and protected areas analysis. Here, a number of controlled, econometric studies with large sample sizes found that PES programmes on the whole did no harm to participant households, and provided small increases in some cases to household incomes and assets, but also did not find support for a strong role in poverty reduction. Two recent randomised controlled trials found positive impacts on well-being measures (Jayachandran et al., 2017; Adjognon et al., 2019). Studies of protected areas have similarly utilised matching-based, quasi-experimental designs, and national and global datasets to show their positive impacts on poverty reduction, as well as the conditions that increase likelihood of impacts (namely, presence of ecotourism and locations at intermediate distances from major cities; Ferraro et al., 2011; Naidoo et al., 2019). For ecotourism, while not assessed through similarly rigorous study designs, evidence of impact has been tracked through the number of visitors and the benefits they bring in terms of expenditures in local and national economies. Meanwhile, several levers were predominately assessed using case studies in variable contexts (e.g. company-community partnerships, SMFEs). On their own these provide rich information on mechanisms and outcomes, but, in aggregate, the variability in case contexts makes it difficult to assess the specific contributions of the lever to poverty alleviation, and challenging to make any generalised assessments across contexts. The absence of such evidence should not be interpreted as the ineffectiveness of these levers in potentially addressing poverty.

Importantly, this chapter does not evaluate the poverty alleviation impacts of non-forest sector interventions. This includes programmes such as cash transfers, energy substitutions, education and infrastructure initiatives, non-tree related agriculture extension and other levers that are implemented both within but also outside of forested landscapes. Such levers are likely to have substantial impacts on the poverty status of forest-reliant people in rural areas. Indeed, many are more explicitly focused on dimensions of poverty alleviation as their primary objective. In contrast, many of the forest-sector interventions that we reviewed are focused primarily on forest conservation, and include social objectives only as a second-order concern. One example of a non-forest sector lever on poverty alleviation is the national cash-transfer programmes in Brazil that accounted for an average 54% of household income among agricultural households at the forest frontier (Dou *et al.*, 2017). Relatedly, Indonesia's national anti-poverty programme reduced village-level deforestation by 30% by reducing the reliance of rural households both on deforestation as a coping strategy and on forest products as an alternative to market-purchased goods (Ferraro and Simorangkir, 2020). Such programmes and impacts are not captured in this review and should be evaluated separately.

5.8 Knowledge Gaps and Future Research

We identified four ways in which the evidence base for how different forest-sector levers contribute to poverty alleviation could be strengthened. First, where appropriate, research designs could control for varying contextual conditions and isolate the mechanisms and levers in question to help illuminate the role of these levers in poverty alleviation. Second, more studies are needed that look at promising levers' contributions to moving people out of poverty rather than focusing on well-being contributions. Third, little research has examined the long-term poverty alleviation effects of forest-sector interventions. Fourth, our review does not explore the relative economic costs of alternative levers. In sum, a combination of more rigorous and long-term research designs, along with examinations of the cost-effectiveness of different levers, would go a long way to contributing to the design of effective interventions for poverty alleviation.

Future research could also address values and outcomes that are central to other Sustainable Development Goals. For example, reviews could examine how forest sector interventions affect rights, *equity*, *adaptation* and *resilience*, or carbon sequestration. The interactions among these additional outcomes and poverty alleviation is ripe for investigation; more broadly, future research might also ask, "[how] can inclusive, equitable and sustainable forest management contribute to poverty alleviation?"

5.9 Conclusions

Forest-reliant communities are variously (and sometimes simultaneously) affected by rightsbased, regulatory, market and supply chain, and forest and tree management levers within complex socio-environmental landscapes. Different actors, including governments, communities, private sector organisations and NGOs, variously develop, fund and implement these levers. Teasing apart and isolating the effects of any one lever on poverty alleviation is challenging given available evidence. That said, there is evidence to demonstrate that some interventions – including ecotourism and community forest management – can have detectable and significant impacts on poverty reduction, while many studies show that many of the reviewed levers have had positive impacts on poverty mitigation. To add to a rich body of case study research, further studies that explicitly focus on poverty reduction as an outcome of interest and that isolate causal mechanisms, including through quantitative methodologies with robust counterfactuals where appropriate, could help to extend this understanding of how forest-sector policies, programmes and strategies can help to alleviate poverty among the rural poor.

5.10 References

Acharya, K. P. 2002. Twenty-four years of community forestry in Nepal. International Forestry Review 4(2): 149-156.

Acheampong, E. O., Sayer, J. and Macgregor, C. J., 2018. Road improvement enhances smallholder productivity and reduces forest encroachment in Ghana. Environmental Science and Policy 85: 64-71.

Acheampong, E. and Marfo, E., 2011. The impact of tree tenure and access on chainsaw milling in Ghana. *Ghana J. For.* 27: 68–86.

Adams, C., Rodrigues, S. T., Calmon, M. and Kumar, C., 2016. Impacts of large-scale forest restoration on socioeconomic status and local livelihoods: what we know and do not know. *Biotropica* 48(6): 731–744.

Adhikari, B., Falco, S. D. and Lovett, J. C., 2004. Household characteristics and forest dependency: evidence from common property forest management in Nepal. Ecological Economics 48: 245-257.

Adhikari, S., Kingi, T. and Ganesh, S., 2014. Incentives for community participation in the governance and management of common property resources: the case of community forest management in Nepal. *Forest Policy and Economics* 44:1-9.

Adjognon, G. S., van Soest, D. and Guthoff, J., 2019. Reducing Hunger with Payments for Ecosystem Services (PES): Experimental Evidence from Burkina Faso. Washington DC: The World Bank.

Agrawal, A., Chhatre, A. and Hardin, R., 2008. Changing governance of the world's forests. *Science* 320(5882): 1460-1462

Agrawal, A., Hajjar, R., Liao, C., Rasmussen, L. V., and Watkins, C., 2018. Editorial overview: Forest governance interventions for sustainability through information, incentives, and institutions. *Current Opinion in Environmental Sustainability* 32: 1-7.

Ahenkan, A. and Boon, E., 2010. Commercialization of non-timber forest products in Ghana: Processing, packaging and marketing. Food Agric and Env 8: 962-969.

Akinola, A.O. and Wissink, H. (eds.). 2019. Trajectory of Land Reform in Post-Colonial African States. The Quest for Sustainable Development and Utilization. Cham: Springer.

Ali, A., and Behera, B. 2015. Household participation and effects of community forest management on income and poverty levels: Empirical evidence from Bhutan. *Forest Policy and Economics* 61, 20-29.

Alix-Garcia, J. and Wolff, H., 2014. Payment for ecosystem services from forests. Annu. Rev. Resour. Econ. 6(1): 361–380.

Alix-Garcia, J. M., Sims, K. R. E. and Yañez-Pagans, P., 2015. Only one tree from each seed? Environmental effectiveness and poverty alleviation in Mexico's Payments for Ecosystem Services Program. American Economic Journal: Economic Policy 7(4): 1–40. Alix-Garcia, J. M., Sims, K. R. E., Orozco-Olvera, V. H., Costica, L., Medina, J. D. F., Romo-Monroy, S. and Pagiola, S., 2019. Can environmental cash transfers reduce deforestation and improve social outcomes? a regression discontinuity analysis of Mexico's national Program 2011--2014. Washington DC: The World Bank.

Amadu, F. O., Miller, D. C. and McNamara, P. E., 2020. Agroforestry as a pathway to agricultural yield impacts in climate-smart agriculture investments: Evidence from southern Malawi. *Ecological Economics* 167: 106443.

Andam, K. S., Ferraro, P. J., Pfaff, A., Sanchez-Azofeifa, G. A. and Robalino, J. A., 2008. Measuring the effectiveness of protected area networks in reducing deforestation. PNAS 105(42): 16089-16094.

Andam, K.S., Ferraro, P.J., Sims, K.R., Healy, A. and Holland, M.B., 2010. Protected areas reduced poverty in Costa Rica and Thailand. PNAS 107(22): 9996-10001.

Anderson, J., Benjamin, C., Campbell, B. and Tiveau, D. 2006. Forests, poverty and equity in Africa: new perspectives on policy and practice. *International Forestry Review* 8(1): 44-53.

Andersson, K. P. and Pacheco, D., 2006. Turning to forestry for a way out of poverty: is formalizing property rights enough? In: Linking the Formal and Informal Economy: Concepts and Policies, ed. Guha-Khasnobis, B., Kanbur, R. & Ostrom, E. Oxford: Oxford University Press.

Andrew, M., Fabricius, C. and Timmermans, H., 2000. An Overview of Private Sector Community Partnerships in Forestry and other Natural Resources in Eastern Cape. Instruments Sustain. Priv. Sect. For. South Africa Ser. 70.

Antinori, C. and Bray, D. B., 2005. Community forest enterprises as entrepreneurial firms: economic and institutional perspectives from Mexico. World Development 33(9): 1529-1543.

Aronson, J., Blignaut, J. N., Milton, S. J., Le Maitre, D., Esler, K. J., Limouzin, A., Fontaine, C., et al., 2010. Are socioeconomic benefits of restoration adequately quantified? a meta-analysis of recent papers 2000-2008) in restoration ecology and 12 other scientific journals. Restoration Ecology 182: 143–154.

Arriagada, R. A., Sills, E. O., Ferraro, P. J. and Pattanayak, S. K., 2015. Do payments pay off? Evidence from participation in Costa Rica's PES program. PloS One 10(8): e0136809.

Asner, G. P., Rudel, T. K., Mitchell Aide, T., Defries, R. and Emerson, R., 2009. A contemporary assessment of change in humid tropical forests. *Conserv. Biol.* 23: 1386–95

- Astrid Fenger, N., Skovmand Bosselmann, A., Asare, R. and de Neergaard, A., 2017. The impact of certification on the natural and financial capitals of Ghanaian cocoa farmers. Agroecology and Sustainable Food Systems 41(2): 143-166.
- Atmlş, E., Günsen, H. B. and Özden, S., 2010. How can Turkey's forest cooperatives contribute to reducing rural poverty?. Unasylva (English ed.) 61(234/235): 51-53.
- Awono, A., Somorin, O. A., Atyi, R. E. A. and Levang, P., 2014. Tenure and participation in local REDD+ projects: Insights from southern Cameroon. Environmental Science and Policy 35:76-86.
- Balmford, A., Green, J. M. Anderson, M., Beresford, J., Huang, C., Naidoo, R., Walpole, M. et al., 2015. Walk on the wild side: estimating the global magnitude of visits to protected areas. PLoS Biol 132: e1002074.
- Banerjee, O., Macpherson, A. J. and Alavalapati, J., 2009. Toward a policy of sustainable forest management in Brazil: a historical analysis. *The Journal of Environment and Development* 182: 130-153.
- Barr, C. M. and Sayer, J. A., 2012. The political economy of reforestation and forest restoration in Asia-Pacific: Critical issues for REDD+. *Biological Conservation* 154: 9–19.
- Bastin, J. F., Finegold, Y., Garcia, C., Mollicone, D.,
 Rezende, M., Routh, D., Zohner, C.M. and Crowther, T.
 W. 2019. The global tree restoration potential. *Science* 364(6448): 76–79.
- Beall, E., 2012. Smallholders in global bioenergy value chains and certification. Environment and Natural Resources Management Working Paper, 50. Rome: FAO.
- Beauchamp, E. and Ingram, V. 2011. Impacts of community forests on livelihoods in Cameroon: lessons from two case studies. International Forestry Review 13(4): 389-403.
- Bellemare, M. F. and Novak, L. (2017). Contract farming and food security. American Journal of Agricultural Economics 99(2): 357-378.
- Bellemare, M.F. and Bloem, J.R., 2018. Does contract farming improve welfare? A review. World Development 112: 259-271.
- Bellemare, M.F. and Lim, S., 2018. In all shapes and colors: Varieties of contract farming. *Applied Economic Perspectives and Policy* 40(3): 379-401.
- Bellemare, M.F., 2012. As you sow, so shall you reap: The welfare impacts of contract farming. World Development 40(7): 1418-1434.
- Benjaminsen, T.A., Holden, S., Lund, C. and Sjaastad, E. 2008. Formalisation of land rights: Some empirical evidence from Mali, Niger and South Africa. Land Use Policy 26: 28–35.
- Beymer-Farris, B.A. and Bassett, T.J., 2012. The REDD menace: Resurgent protectionism in Tanzania's mangrove forests. *Global Environmental Change* 22(2): 332-341

- Bijaya, G. C. D., Cheng, S., Xu, Z. et al. 2016. Community forestry and livelihood in Nepal: a review. The Journal of Animal & Plant Sciences 26(1): 1-12.
- Bijiaya, G. C. D., Cheng, S., Xu, Z., Bhandari, J., Wang, L. and Liu, X., 2016. Community forestry and livelihood in Nepal: a review. The Journal of Animal and Plant Sciences, 26(1): 1-12.
- Birchall, J., 2003. Rediscovering the cooperative advantage-Poverty reduction through self-help. Geneva: International Labour Organisation.
- Blangy, S. and Wood, M.E., 1993. Developing and implementing ecotourism guidelines for wildlands and neighboring communities. In: Ecotourism: A Guide for Planners and Managers, edited by K. Lindberg and D. E. Hawkins. Bennington VT: The Ecotourism Society.
- Bluffstone, R., Robinson, E. and Guthiga, P., 2013. REDD+ and community-controlled forests in lowincome countries: Any hope for a linkage? *Ecological Economics* 87: 43-52.
- Bolwig, S., Gibbon, P. and Jones, S., 2009. The economics of smallholder organic contract farming in tropical Africa. World Development 37(6):1094-1104.
- Börner, J., Baylis, K., Corbera, E., Ezzine-de-Blas, D.,
 Honey-Rosés, J., Persson, U. M. and Wunder, S., 2017.
 The Effectiveness of Payments for Environmental Services. World Development 96: 359–374.
- Börner, J., Schulz, D., Wunder, S. and Pfaff, A., 2020. The Effectiveness of Forest Conservation Policies and Programs. Annu. Rev. Resour. Econ. 11: 1–20.
- Bose, P., 2011. Forest tenure reform: exclusion of tribal women's rights in semi-arid Rajasthan, India. International Forestry Review 132:220-232.
- Bostedt, G., Hörnell, A. and Nyberg, G. 2016. Agroforestry extension and dietary diversity – an analysis of the importance of fruit and vegetable consumption in West Pokot, Kenya. Food Security 8(1): 271-284.
- Boulay, A. and Tacconi, L., 2012. The drivers of contract eucalypt farming in Thailand. *International Forestry Review* 14(1): 1-12.
- Bowler, D. E., Buyung-Ali, L. M., Healey, J. R., Jones, J. P., Knight, T. M. and Pullin, A. S., 2012. Does community forest management provide global environmental benefits and improve local welfare? *Frontiers in Ecology and the Environment* 10(1): 29-36.
- Brancalion, P. H. S., Niamir, A., Broadbent, E., Crouzeilles, R., Barros, F. S. M., Almeyda Zambrano, A. M., Baccini, A., et al., 2019. Global restoration opportunities in tropical rainforest landscapes. Science Advances 5(7): eaav3223.
- Brandi, C., Cabani, T., Hosang, C., Schirmbeck, S., Westermann, L. and Wiese, H., 2015. Sustainability standards for palm oil: challenges for smallholder certification under the RSPO. Journal of Environment and Development 24(3): 292–314.

Bray, D. B., Antinori, C. and Torres-Rojo, J. M. 2006. The Mexican model of community forest management: The role of agrarian policy, forest policy and entrepreneurial organization. *Forest Policy and Economics* 8(4): 470-484.

Brockington, D., and Wilkie, D., 2015. Protected areas and poverty. Philosophical Transactions of the Royal Society B: Biological Sciences, 370(1681): 20140271.

Brockington, D. and Igoe, J., 2006. Eviction for Conservation: A Global Overview. *Conserv. Soc.* 4: 424–470.

Bromley, D.W., 1992. The Commons, Common Property, and Environmental Policy. *Environmental and Resource Economics* 2: 1-17.

Bromley, D.W. and Cernea, M., 1989. The management of common property natural resources: some conceptual and operational fallacies. World Bank Discussion Papers No. 57. Washington, D.C.: The World Bank.

Brown, S. and Zarin, D., 2013. What does zero deforestation mean. *Science* 342(6160): 805-807.

Brubacher, D., 1998. Aboriginal forestry joint ventures: Elements of an assessment framework. *Forestry Chronicle.* 74: 353–358.

Bruggeman, D., Meyfroidt, P. and Lambin, E.F.,
2015. Production forests as a conservation tool:
Effectiveness of Cameroon's land use zoning policy.
Land use policy 42: 151–164.

Bruner, A. G., Gullison, R. E. and Balmford, A., 2004. Financial costs and shortfalls of managing and expanding protected-area systems in developing countries. BioScience 54(12): 1119-1126.

Buckley, R., 1994. A framework for ecotourism. Annals of Tourism Research 21: 661- 65.

Bulkan, J., 2014. Forest grabbing through forest concession practices: the case of Guyana. *Journal of sustainable forestry* 33(4): 407-434.

Bulte, E. H., Lipper, L., Stringer, R. and Zilberman, D., 2008. Payments for ecosystem services and poverty reduction: concepts, issues and empirical perspectives. Environment and Development Economics 13(3): 245–254.

Burivalova, Z., Hua, F., Koh, L. P., Garcia, C. and Putz, F., 2017. A critical comparison of conventional, certified and community management of tropical forests for timber in terms of environmental, economic and social variables. *Conservation Letters* 10(1): 4-14.

Burke, B. J., 2010. Cooperatives for "fair globalization"? Indigenous people, cooperatives, and corporate social responsibility in the Brazilian Amazon. Latin American Perspectives 37(6): 30-52.

Burney, O., Aldrete, A., Reyes, R. A., Sa, R. and Mexal, J. G., 2015. Mexico: Addressing challenges to reforestation. *Journal of Forestry* 113(4): 404–413. Busch, J., Engelmann, J., Cook-Patton, S. C., Griscom, B. W., Kroeger, T., Possingham, H. and Shyamsundar, P., 2019. Potential for low-cost carbon dioxide removal through tropical reforestation. Nature Climate Change 9(6): 463–466.

Cairns, R. I., 2000. Outgrower timber schemes in KwaZulu-Natal–Do they build sustainable rural livelihoods and what interventions should be made? London: IIED & CSIR.

Calkins, P. and Ngo, A. T., 2010. The impacts of farmer cooperatives on the well-being of cocoa producing villages in Côte d'Ivoire and Ghana. *Canadian Journal* of Development Studies/Revue canadienne d'études du développement 30(3-4): 535-563.

Call, M., Mayer, T., Sellers, S., Ebanks, D., Bertalan, M., Nebie, E. and Gray, C., 2017. Socio-environmental drivers of forest change in rural Uganda. *Land Use Policy* 62: 49–58.

Carter, M.R., 2003. Designing land and property rights reform for poverty alleviation and food security. Paper presented at 29th Session of the Committee on World Food Security, Rome, 12 - 16 May 2003, informal panel on the impact of access to land on improving food security and alleviating poverty.

Cerutti P.O, Lescuyer G, Tsanga R, Kassa S.N, Mapangou P.R, Mendoula, E.E, Missamba-Lola, A.P., et al., 2014. Social impacts of the Forest Stewardship Council certification: An assessment in the Congo basin. Occasional Paper 103. Bogor: CIFOR.

Cerutti, P. O., Goetghebuer, T., Leszczynska, N., Newbery, J., Breyne, J., Dermawan, A., Mauquoy, C., et al., 2019. Collecting Evidence of FLEGT-VPA Impacts for Improved FLEGT Communication - Synthesis report. Rome, Italy, Center for International Forestry Research (CIFOR), Analyses for Economic Decisions (ADE), and Food and Agriculture Organization of the United Nations (FAO).

Cerutti, P. O., Tacconi, L., Lescuyer, G. and Nasi, R., 2013. Cameroon's hidden harvest: commercial chainsaw logging, corruption, and livelihoods. Society & Natural Resources 26(5): 539-553.

Chamberlin, J. and Jayne, T. S., 2013. Unpacking the Meaning of 'Market Access': Evidence from Rural Kenya. World Development 41: 245-264.

Chazdon, R. L. and Guariguata, M. R., 2016. Natural regeneration as a tool for large-scale forest restoration in the tropics: prospects and challenges. *Biotropica* 48(6): 716–730.

Chen, M. A., 2007. Rethinking the informal economy: Linkages with the formal economy and the formal regulatory environment. DESA working paper no 46.

Chen, T., 2017. Impact of the shea nut industry on women's empowerment in Burkina Faso: a multi-dimensional study focusing on the Central, Central-West and Hauts-Bassins regions. Rome : Food and Agriculture Organization of the United Nations (FAO). Chhatre, A. and Agrawal, A., 2008. Forest commons and local enforcement. *Proceedings of the national Academy* of sciences 105(36): 13286-13291.

Chhatre, A. and Agrawal, A., 2009. Trade-offs and synergies between carbon storage and livelihood benefits from forest commons. *Proceedings of the national Academy of sciences* 106(42): 17667-17670.

Chhetri, B. B. K., Asante, P., Yoshimoto, A. 2016. Forest dependence and inequality: an empirical study from community forestry in Kaski, Nepal. FORMATH 15: 33-43.

China State Council, 2000. Opinions of the State Council on Further Implementation of Pilot Projects of Conversion of Cropland to Forest (Grass). State Council 2000, No. 24.

Chirenje, L. I., 2017. Contribution of ecotourism to poverty alleviation in Nyanga, Zimbabwe, Chinese Journal of Population Resources and Environment 15(2): 87-92.

Choi, T. Y. and Hong, Y., 2002. Unveiling the structure of supply networks: case studies in Honda, Acura, and DaimlerChrysler. *Journal of Operations Management* 20(5): 469-493.

Chomba, S., Kariuki, J., Lund, J. F. and Sinclair, F., 2016. Roots of inequity: How the implementation of REDD+ reinforces past injustices. *Land Use Policy* 50: 202-213.

Clements, T. and Milner-Gulland, E.J., 2015. Impact of payments for environmental services and protected areas on local livelihoods and forest conservation in northern Cambodia. *Conserv. Biol.* 29: 78–87.

Colchester, M., 2004. Conservation policy and indigenous peoples. Environ. Sci. Policy 7: 145–153.

Coulibaly, J. Y., Chiputwa, B., Nakelse, T. and Kundhlande, G., 2017. Adoption of agroforestry and the impact on household food security among farmers in Malawi. Agricultural Systems 155: 52-69.

Cronkleton, P., Pulhin, J.M. and Saigal, S., 2012. Comanagement in community forestry: how the partial devolution of management rights creates challenges for forest communities. *Conservation and* Society 10(2): 91-102.

Da Silva, C. A. and Rankin, M., 2013. Contract Farming for inclusive market Access. Rome: FAO.

Dahal, G.R., Pokharel, B.K, Khanal, D.R. and Pokhrel, P.R., 2017. Why Does Tenure Security Matter in Community Forestry? A Critical Reflection from Nepal. Journal of Forest and Livelihood 15(1): 15-26.

Damette, O. and Delacote, P., 2011. Unsustainable timber harvesting, deforestation and the role of certification. *Ecological Economics* 70(6): 1211-1219.

Dammert, J. L., 2019. Exploring inclusivity with the United Nations model in the Peruvian Amazon. In: Exploring inclusive palm oil production. Edited by Jezeer, R. and Pasiecznik, N. Tropenbos International: Wageningen: Tropenbos International. Dana, L. P. and Mallet, J., 2014. An unusual empirical pattern in an indigenous setting: cooperative entrepreneurship among Brazil nut (Bertholletia excelsa) harvesters. *International Journal of Entrepreneurship and Small Business* 222: 137-158.

Danielsen, F., Skutsch, M., Burgess, N. D., Jensen, P. M.,
Andrianandrasana, H., Karky, B., Lewis, R. et al., 2011.
At the heart of REDD+: A role for local people in
monitoring forests? Conservation letters 42: 158-167.

Davis, M., 2005. Forests and conflict in Cambodia. International Forestry Review 72: 161-164.

de Koning, J., 2011. Reshaping institutions: bricolage processes in smallholder forestry in the Amazon. PhD thesis. Wageningen: Wageningen University.

de Soto, H., 2000. The mystery of capital: Why capitalism triumphs in the West and fails everywhere else. Basic Civitas Books.

de Souza, S. E. X. F., Vidal, E., Chagas, G. de F., Elgar, A. T. and Brancalion, P. H. S., 2016. Ecological outcomes and livelihood benefits of community-managed agroforests and second growth forests in Southeast Brazil. Biotropica 48(6): 868–881.

Dean, J.M., 1995. Export Bans, Environment, and Developing Country Welfare. *Review of International Economics* 3: 319–329.

Dedehouanou, S.F., Swinnen, J. and Maertens, M., 2013. Does Contracting Make Farmers Happy? Evidence from Senegal. *Review of Income and Wealth* 59: S138-S160.

Deininger, K., 2003. Land policies for growth and poverty reduction. A World Bank policy research report. Washington, D.C.: World Bank Group.

Delacote, P., 2006. Are consumer boycotts effective? Presented at: 6th IDEI-LERNA Conference on Environmental Resource Economics–Environment, Finance and Corporate Behavior.

Delacote, P., 2009. On the sources of consumer boycotts ineffectiveness. *The Journal of Environment and Development* 18: 306-322.

DeMarsh, P., Boscolo, M., Savenije, H., Grouwels, S., Zapata, J., Campbell, J. and Macqueen, D., 2014. Making change happen. What can governments do to strengthen forest producer organizations. Forest Farm Facility Working Paper. Rome: FAO.

Desmond, H. and Race, D., 2000. Global survey and analytical framework for forestry out-grower arrangements. Final Report submitted to the Food and Agriculture Organization (FAO) of the United Nations. Canberra: ANU Forestry.

Dinerstein, E., Olson, D., Joshi, A., Vynne, C., Burgess, N.D., Wikramanayake, E., Hahn, *et al.*, 2017. An ecoregion-based approach to protecting half the terrestrial realm. *BioScience* 67(6): 534-545.

Dinerstein, E., Vynne, C., Sala, E., Joshi, A.R., Fernando, S., Lovejoy, T.E., Mayorga, J., *et al.*, 2019. A Global Deal for Nature: Guiding principles, milestones, and targets. *Sci. Adv.* 5: 1–18. Doggart, N. and Meshack, C., 2017. The marginalization of sustainable charcoal production in the policies of a modernizing African nation. *Frontiers in Environmental Science* 5: 27.

Donovan, J. and Poole, N., 2013. Asset building in response to value chain development: lessons from taro producers in Nicaragua. *International Journal of Agricultural Sustainability* 11(1): 23-37.

Dossa, Z. A., 2012. Cooperatives: A Development Strategy? An Analysis of Argan Oil Cooperatives in Southwest Morocco. Euricse Working Paper, No 29/12.

Dou, Y., Deadman, P., Robinson, D., Almeida, O., Rivero, S., Vogt, N. and Pinedo-Vasquez, M., 2017. Impacts of cash transfer programs on rural livelihoods: A case study in the Brazilian Amazon Estuary. *Human Ecology* 45(5): 697-710.

Duan, W. and Wen, Y., 2017. Impacts of protected areas on local livelihoods: Evidence of giant panda biosphere reserves in Sichuan Province, China. Land use policy 68: 168–178.

Duchelle, A. E., 2009. Conservation and livelihood development in Brazil nut-producing communities in a tri-national Amazonian frontier. Gainesville: University of Florida.

Duchelle, A. E., Kainer, K. A. and Wadt, L. H., 2014. Is certification associated with better forest management and socioeconomic benefits? A comparative analysis of three certification schemes applied to Brazil nuts in Western Amazonia. Society and Natural Resources 272: 121-139.

Duchelle, A. E., Simonet, G., Sunderlin, W. D. and Wunder, S., 2018. What is REDD+ achieving on the ground? Current Opinion in Environmental Sustainability 32: 134-140.

Dudley, N. (ed.), 2008. Guidelines for Applying Protected Area Management Categories. Gland: IUCN.

Eaton, C. and Shepherd, A., 2001. Contract farming: partnerships for growth (No. 145). Rome: FAO.

Eba'a Atyi, R., Assembe-Mvondo, S., Lescuyer, G. and Cerutti, P., 2013. Impacts of international timber procurement policies on Central Africa's forestry sector: The case of Cameroon. *Forest Policy and Economics* 32: 40–48.

Ebeling, J. and Yasué, M., 2009. The effectiveness of market-based conservation in the tropics: Forest certification in Ecuador and Bolivia. *Journal of* environmental management 902: 1145-1153.

Elias, M. and Arora-Jonsson, S., 2017. Negotiating across difference: Gendered exclusions and cooperation in the shea value chain. *Environment and Planning D:* Society and Space 35(1): 107-125.

Elias, M. and Carney, J., 2005. Shea butter, globalization, and women of Burkina Faso. In: A companion to feminist geography. Edited by Nelson, L. and Seager, J. pp. 93-108. Elson, D., 2012. A guide to investing in locally controlled forestry. In: Growing Forest Partnerships in Association with FAO, IIED, IUCN, the Forests Dialogue and the World Bank. London: IIED.

Engel, S., Pagiola, S. and Wunder, S., 2008. Designing payments for environmental services in theory and practice: An overview of the issues. *Ecological Economics* 65(4): 663-674.

Erbaugh, J. T. and Oldekop, J. A., 2018. Forest landscape restoration for livelihoods and well-being. *Current Opinion in Environmental Sustainability* 32: 76–83.

Erbaugh, J. T., Nurrochmat, D. R. and Purnomo, H., 2017. Regulation, formalization, and smallholder timber production in northern Central Java, Indonesia. *Agroforestry Systems* 91(5): 867-880.

FAO and Agricord, 2016. Forest and Farm Producer Organizations – Operating Systems for the SDGs. Strength in numbers. FAO, Rome 2016.

FAO, 2014. State of the World's Forests 2014. Rome: FAO.

FAO, 2018. Rethinking forest concessions - Improving the allocation of state-owned forests for better economic, social and environmental outcomes. Rome: FAO.

FAO, 2011. Reforming forest tenure. Issues, principles and process. FAO Forestry Paper No. 165. Rome: FAO.

Fenger, A.N., Skovmand Bosselmann, A., Asare, R. and de Neergaard, A., 2017. The impact of certification on the natural and financial capitals of Ghanaian cocoa farmers. Agroecology and Sustainable Food Systems 41(2): 143-166.

Ferraro, P. J. and Simorangkir, R., 2020. Conditional cash transfers to alleviate poverty also reduced deforestation in Indonesia. *Science Advances* 624: eaaz1298.

Ferraro, P. J. and Pressey, R. L., 2015. Measuring the difference made by conservation initiatives: protected areas and their environmental and social impacts. Philosophical transactions of the Royal Society of London. Series B, Biological sciences 370(1681): 20140270.

Ferraro, P. J., Hanauer, M. M. and Sims, K. R., 2011. Conditions associated with protected area success in conservation and poverty reduction. *Proceedings of the National Academy of Sciences* 108(34): 13913-13918.

Ferraro, P.J. and Hanauer, M.M., 2014. Advances in measuring the environmental and social impacts of environmental programs. *Annual Reviews of Environment and Resources* 39: 495-517.

Fisher, R., Barrow, E., de Silva, J., Ingles, A., Shepherd, G. 2012. Improved Access to Forest Resources: Experience in informal tenure reform from IUCN's Livelihoods and Landscapes Strategy. Gland: IUCN.

Flanagan, A. C., Midgley, S. J., Stevens, P. R. and McWhirter, L., 2019. Smallholder tree-farmers and forest certification in Southeast Asia: productivity, risks and policies. Australian Forestry 82(1): 18-28. Forest Research and Training Center (FRTC), 2019. National level forests and land cover analysis of Nepal using Google Earth images. Kathmandu: Ministry of Forests and Environment, Nepal.

Fortman, L., 1985. The tree tenure factor in agroforestry with particular reference to Africa. Agroforestry Systems 2(4): 229–251.

Fox, H. and Cundill, G., 2018. Towards increased community-engaged ecological restoration: A review of current practice and future directions. *Ecological Restoration* 36(3): 208–218.

Frey, G. E., Cubbage, F. W., Ha, T. T. T., Davis, R. R., Carle, J. B., Thon, V. X. and Dzung, N. V., 2018. Financial analysis and comparison of smallholder forest and state forest enterprise plantations in Central Vietnam. International Forestry Review 202: 181-198.

Gabay, M. and Rekola, M. 2019. Background Analytical Study. Forests, peaceful and inclusive societies, reduced inequality, education, and inclusive institutions at all levels. Background study prepared for the fourteenth session of the United Nations Forum on Forests. (Available at: https://www.un.org/esa/forests/wpcontent/uploads/2019/03/UNFF14-BkgdStudy-SDG 4-10-16-March2019.pdf [Accessed on 01/03/20]).

Galabuzi, C., Eilu, G., Mulugo, L., Kakudidi, E., Tabuti, J. R. S. and Sibelet, N., 2014. Strategies for empowering the local people to participate in forest restoration. *Agroforestry Systems* 88(4): 719–734.

Galudra, G., van Noordwijk, M., Agung, P., Suyanto, S. and Pradhan, U., 2014. Migrants, land markets and carbon emissions in Jambi, Indonesia: Land tenure change and the prospect of emission reduction. Mitigation and Adaptation Strategies for Global Change 19(6): 715–731.

Garrity, D. P., Akinnifesi, F. K., Ajayi, O. C., Weldesemayat, S. G., Mowo, J. G., Kalinganire, A., Larwanou, M. et al., 2010. Evergreen Agriculture: a robust approach to sustainable food security in Africa. Food Security 2(3): 197-214.

Gautam, A. P., Shivakoti, G. P. and Webb, E. L., 2004a. A review of forest policies, institutions, and changes in the resource condition in Nepal. *International Forestry Review* 6(2): 136-148.

Gentry, K. M., 2007. Belizean women and tourism work: Opportunity or impediment? Annals of Tourism Research 34(2): 477-496.

Gillis, M., 1988. Indonesia: Public policies, resource management and the tropical forest. In: Public Policies and the Misuse of Forest Resources, edited by R. Repetto and M. Gillis. Cambridge: Cambridge University Press.

Gilmour, D., 2016. Forty years of community-based forestry -A review of its extent and effectiveness. Rome: FAO.

Girma, J. and Gardebroek, C., 2015. The impact of contracts on organic honey producers' incomes in southwestern Ethiopia. *Forest Policy and Economics* 50: 259-268. Gockowski, J., Afari-Sefa, V., Sarpong, D.B., Osei-Asare, Y.B. and Agyeman, N.F., 2013. Improving the productivity and income of Ghanaian cocoa farmers while maintaining environmental services: what role for certification? International Journal of Agricultural Sustainability 11(4): 331-346.

Goodland, R. and Daly, H., 1996. If tropical log export bans are so perverse, why are there so many? *Ecological Economics* 18: 189–196.

Government of Nepal, 2020. Community Forestry. (Available at: http://dof.gov.np/dof_community_ forest_division/community_forestry_dof; Accessed on 15 June 2020)

Gretzinger, S. P., 1998. Community forest concessions: an economic alternative for the Maya Biosphere Reserve in the Petén, Guatemala. In: Timber, tourists and temples: Conservation and development in the Maya forest of Belize, Guatemala and Mexico, edited by R.B. Primack, D. Bray, H.A. Galletti and I. Ponciano. Washington DC and Covelo: Island Press, pp. 111-124.

Groom, B. and Palmer, C., 2012. REDD+ and rural livelihoods. *Biological Conservation* 154: 42-52.

Grosh, B., 1994. Contract farming in Africa: an application of the new institutional economics. *Journal of African economies* 32: 231-261.

Grut, M., 2010. Bidding system for allocating forest concessions in Africa. International Forestry Review 122: 177-180.

Guedes Pinto, L. F., Stanley, P., Gomes, P. C. and Robinson, D., 2008. Experience with NTFP certification in Brazil. *Forests, Trees and Livelihoods* 18(1): 37-54.

Haase, D. and Davis, A., 2017. Developing and supporting quality nursery facilities and staff are necessary to meet global forest and landscape restoration needs. *Reforesta* (4): 69–93.

Haggar, J., Soto, G., Casanoves, F. and Virginio, E. d. M., 2017. Environmental-economic benefits and tradeoffs on sustainably certified coffee farms. *Ecological Indicators* 79: 330-337.

Haglund, E., Ndjeunga, J., Snook, L. and Pasternak, D., 2011. Dry land tree management for improved household livelihoods: Farmer managed natural regeneration in Niger. Journal of Environmental Management 92(7): 1696-1705.

Hajjar, R. and Kozak, R., 2017. The evolution of forest producer associations and their current role in REDD+: Case studies from Quintana Roo, Mexico. *Land Use Policy* 60: 373–383.

Hajjar, R., 2015. Advancing small-scale forestry under FLEGT and REDD in Ghana. Forest Policy and Economics 58: 12–20.

Hajjar, R., McGrath, D. G., Kozak, R. A. and Innes, J. L., 2011. Framing community forestry challenges with a broader lens: Case studies from the Brazilian Amazon. Journal of environmental management 92(9): 2159-2169. Hajjar, R., Oldekop, J.A. Cronkleton, P., Etue, E., Newton, P., Russel, A.J.M.. Tjajadi, J.S., et al., 2016. The data not collected on community forestry. *Conservation Biology* 30(6): 1357-1362.

 Hanauer M.M. and Canavire-Bacarreza, G., 2015.
 Implications of heterogeneous impacts of protected areas on deforestation and poverty. *Phil. Trans. R. Soc.* B 370: 20140272.

Hansen, C. P. and Treue, T., 2008. Assessing illegal logging in Ghana. International Forestry Review 10(4): 573-590.

Hansen, J., Hellin, J., Rosenstock, T., Fisher, E., Cairns, J., Stirling, C., Lamanna, C. et al., 2019. Climate risk management and rural poverty reduction. Agricultural Systems 172: 28–46.

Hardin, R. and Bahuchet, S., 2011. Concessionary politics: property, patronage and political rivalry in Central African forest management. *Current Anthropology* 52: S3.

Hardin, R., 2002. Concessionary Politics in the Congo Basin. WP #6, Institutions and Governance Program. Washington, DC: World Resources Institute.

Hautdidier, B. and Gautier D., 2005. What local benefits does the implementation of rural wood markets in Mali generate? In: African forests between nature and livelihood resources: Interdisciplinary studies in conservation and forest management, edited by M. A. F. Ros-Tonen and A. J. Dietz, 191–220. Ceredigion, UK: Edwin Mellen Press.

Hayes, T. and Persha, L., 2010. Nesting local forestry initiatives: Revisiting community forest management in a REDD+ world. *Forest Policy and Economics* 12(8): 545-553.

Hegde, R. and Bull, G. Q., 2011. Performance of an agroforestry based Payments-for-Environmental-Services project in Mozambique: A household level analysis. Ecological Economics 71: 122–130.

Higgins, V. and Richards, C., 2018. Framing sustainability: Alternative standards schemes for sustainable palm oil and South-South trade. *Journal of Rural Studies* 65: 126-34.

Hiratsuka, M., Nakama, E., Satriadi, T., Fauzi, H., Aryadi, M. and Morikawa, Y., 2019. An approach to achieve sustainable development goals through participatory land and forest conservation: a case study in South Kalimantan Province, Indonesia. Journal of Sustainable Forestry 38(6): 558–571.

Hirons, M., McDermott, C., Asare, R., Morel, A., Robinson, E., Mason, J., Boyd, E., et al., 2018. Illegality and inequity in Ghana's cocoa-forest landscape: How formalization can undermine farmers control and benefits from trees on their farms. Land Use Policy 76: 405-413.

Hoang, M. H., Do, T. H., Pham, M. T., Van Noordwijk, M. and Minang, P. A., 2013. Benefit distribution across scales to reduce emissions from deforestation and forest degradation (REDD+) in Vietnam. Land Use Policy 31: 48-60. Hobley, M., Buchy, M., 2013. How can a VPA contribute to poverty reduction? Barcelona.

Hospes, O., 2014. Marking the success or end of global multi-stakeholder governance? The rise of national sustainability standards in Indonesia and Brazil for palm oil and soy. Agriculture and Human Values 31: 425–437.

Howard, M., 2005. Small-scale timber Production in South Africa: what role in reducing poverty? London: IIED.

Huang, L., Shao, Q. and Liu, J., 2012. Forest restoration to achieve both ecological and economic progress, Poyang Lake basin, China. *Ecological Engineering* 44: 53–60.

Hughes, K., Morgan, S., Baylis, K., Oduol, J., Smith-Dumont, E., Vågen, T.-G. and Kegode, H., 2020.
Assessing the downstream socioeconomic impacts of agroforestry in Kenya. World Development 128: 104835.

Humphries, S., Holmes, T., de Andrade, D. F. C., McGrath, D. and Dantas, J. B., 2020. Searching for win-win forest outcomes: learning-by-doing, financial viability, and income growth for a community-based forest management cooperative in the Brazilian Amazon. World Development 125: 104336.

Humphries, S., Holmes, T.P., Kainer, K., Koury, C.G.G., Cruz, E. and de Miranda Rocha, R., 2012. Are community-based forest enterprises in the tropics financially viable? Case studies from the Brazilian *Amazon. Ecol. Econ.* 77: 62–73.

IFAD, 2015. Revised Approach Paper: Smallholder Access to Markets (SAM). Evaluation Synthesis. Rome: FAO.

Ingram, V., van Rijn, F., Waarts, Y. and Gilhuis, H., 2018. The impacts of cocoa sustainability initiatives in West Africa. *Sustainability* 10(11): 4249.

Ingram, V., Waarts, Y., Ge, L., Van Vugt, S., Wegner, L., Puister-Jansen, L., Ruf, F. and Tanoh, R., 2014. Impact of UTZ certification of cocoa in Ivory Coast. Assessment framework and baseline. Wageningen: LEI Wageningen UR.

Innes, R., 2006. A theory of consumer boycotts under symmetric information and imperfect competition. *The Economic Journal* 116(511): 355-381.

Iponga, D. M., Mikolo-Yobo, C., Lescuyer, G., Assoumou, F. M., Levang, P., Tieguhong, J. C. and Ngoye, A., 2018. The contribution of NTFP-gathering to rural people's livelihoods around two timber concessions in Gabon. Agroforestry systems 92(1): 157-168.

Irawan, S. and Tacconi, L., 2009. Reducing emissions from deforestation and forest degradation (REDD) and decentralized forest management. *International Forestry Review* 11(4): 427-438.

Islam, K. K., Hoogstra, M., Ullah, M. O. and Sato, N., 2012. Economic contribution of participatory agroforestry program to poverty alleviation: a case from Sal forests, Bangladesh. *Journal of Forestry Research* 232: 323-332. ITTO, 2015. Biennial Review and Assessment of the World Timber Situation 2013–2014. Yokohama: International Tropical Timber Organization.

IUCN – International Union for Conservation of Nature, 2008. Protected Areas. (Available at: https://www. iucn.org/theme/protected-areas/about)

Iversen, I., Chhetry,, B., Francis, P., Gurung, M., Kafle, G., Pain, A. and Seeley, J., 2006. High value forests, hidden economics and elite capture: evidence from forest user groups in Nepal's Terai. Ecological Economics 58: 93-107.

Jagger, P. and Rana, P., 2017. Using publicly available social and spatial data to evaluate progress on REDD+ social safeguards in Indonesia. *Environ Sci Policy* 76:59-69.

Jagger, P., Luckert, M. M. K., Duchelle, A. E., Lund, J. F. and Sunderlin, W. D., 2014. Tenure and forest income: Observations from a global study on forests and poverty. World Development 64: S43-S55.

Jayachandran, S., De Laat, J., Lambin, E.F., Stanton, C.Y., Audy, R. and Thomas, N.E., 2017. Cash for carbon: A randomized trial of payments for ecosystem services to reduce deforestation. *Science* 357(6348): 267-273.

Jindal, R., Kerr, J. M. and Carter, S., 2012. Reducing poverty through carbon forestry? Impacts of the N'hambita community carbon project in Mozambique. World Development 40(10): 2123–2135.

Johnson Gaither, C., Yembilah, R. and Samar, S. B., 2019. Tree registration to counter elite capture of forestry benefits in Ghana's Ashanti and Brong Ahafo regions. *Land Use Policy* 85: 340–349.

Jones, D., Ryan, C. M. and Fisher, J., 2016. Charcoal as a diversification strategy: The flexible role of charcoal production in the livelihoods of smallholders in central Mozambique. *Energy for Sustainable Development* 32: 14-21.

Jones, K. W., Brenes, C. L. M., Shinbrot, X. A., López-Báez, W. and Rivera-Castañeda, A., 2018. The influence of cash and technical assistance on household-level outcomes in payments for hydrological services programs in Chiapas, Mexico. Ecosystem Services 31: 208–218.

Jung, S., Liao, C., Agrawal, A. and Brown, D. G., 2019. Evidence on wealth-improving effects of forest concessions in Liberia. *Journal of the Association of Environmental and Resource Economists* 6(5): 961-1000.

Kaimowitz, D., 2003. Forest law enforcement and rural livelihoods. International Forestry Review 5(3): 199-210.

Kariuki, J. and Birner, R., 2016. Are market-based conservation schemes gender-blind? A qualitative study of three cases from Kenya. Society and natural resources 29(4): 432-447.

Kashwan, P. and Holahan, R., 2014. Nested governance for effective REDD+: Institutional and political arguments. International Journal of the Commons 8(2): 554–575. Kazoora, C., Acworth, J., Tondo, C. and Kazungu, B., 2006. Forest-based associations as drivers for sustainable development in Uganda. IIED Small and Medium Forest Enterprise Series No. 15. Edinburgh: IIED.

Kelly, A. B. and Peluso, N. L. 2015. Frontiers of commodification: State lands and their formalization. Society & Natural Resources 28(5): 473-495.

Kent, R., 2018. "Helping" or "appropriating"? Gender relations in shea nut production in Northern Ghana. Society & Natural Resources 31(3): 367-381.

Kiptot E. and Franzel S., 2011. Gender and agroforestry in Africa: are women participating? ICRAF Occasional Paper No. 13. Nairobi: World Agroforestry Centre.

Kiptot, E. and Franzel, S., 2012. Gender and agroforestry in Africa: a review of women's participation. *Agroforestry Systems* 84(1): 35-58.

Kishor, N. and Damania, R., 2007. Crime and justice in the Garden of Eden: improving governance and reducing corruption in the forestry sector. The many faces of corruption. Washington, DC: The World Bank.

Kishor, N., Mani, M. and Constantino, L., 2002. Economic and Environmental Benefits of Eliminating Log Export Bans – The Case of Costa Rica. *IMF Working Paper*. WP/02/6, 555–582.

Kleinschroth, F. and Healey, J. R., 2017. Impacts of logging roads on tropical forests. *Biotropica* 49(5): 620-635.

Kozak, R. A. 2007. Small and medium forest enterprises: instruments of change in the developing world. Washington, DC: Rights and Resources Initiative.

Krain, E., Miljard, E., Konan, E. and Servat, E., 2011. Trade and pro-poor growth: Introducing Rainforest Alliance certification to cocoa production in Côte d'Ivoire. Eschborn: Deutsche Gesellschaft für Internationale Zusammenarbeit.

Kuntashula, E. and Mungatana, E., 2013. Estimating the causal effect of improved fallows on farmer welfare using robust identification strategies in Chongwe, Zambia. Agroforestry Systems 87(6): 1229-1246.

Lamb, D. and Gilmour, D., 2003. Rehabilitation and Restoration of Degraded Forests. Gland and Cambridge: IUCN and WWF.

Lambin, E.F., Gibbs, H.K., Heilmayr, R., Carlson, K.M., Fleck, L.C., Garrett, R.D., de Waroux, Y.L.P., et al., 2018. The role of supply-chain initiatives in reducing deforestation. Nature Climate Change 8(2): 109-116.

Larson, A. M., Solis, D., Duchelle, A. E., Atmadja, S., Resosudarmo, I. A. P., Dokken, T. and Komalasari, M., 2018. Gender lessons for climate initiatives: A comparative study of REDD+ impacts on subjective wellbeing. World Development 108: 86-102.

Lawlor, K., Madeira, E. M., Blockhus, J. and Ganz, D. J., 2013. Community participation and benefits in REDD+: A review of initial outcomes and lessons. *Forests* 42: 296-318. Lawry, S., Samii, C., Hall, R., Leopold, A., Hornby, D. and Mtero, F., 2017. The Impact of Land Property Rights Interventions on Investment and Agricultural Productivity in Developing Countries: a Systematic Review. Journal of Development Effectiveness 9(1):61-81.

Le Billon, P., 2001. The political ecology of war: natural resources and armed conflicts. *Political geography* 20(5): 561-584.

le Polain de Waroux, Y. and Lambin, E. F., 2013. Niche commodities and rural poverty alleviation: Contextualizing the contribution of argan oil to rural livelihoods in Morocco. Annals of the Association of American Geographers 103(3): 589-607.

Le Tourneau, F.M. and Greissing, A., 2010. A quest for sustainability: Brazil nut gatherers of São Francisco do Iratapuru and the Natura Corporation. *Geography Journal*. 176: 334–349.

Le, H. D., Smith, C., Herbohn, J. and Harrison, S., 2012. More than just trees: Assessing reforestation success in tropical developing countries. *Journal of Rural Studies* 28(1): 5–19.

Legesse, B. A., Jefferson-Moore, K. and Thomas, T., 2018. Impacts of land tenure and property rights on reforestation intervention in Ethiopia. *Land Use Policy* 70: 494–499.

Lele, S., Pattanaik, M. and Rai, N. D., 2010. NTFPs in India: Rhetoric and reality. In: Wild Product Governance, edited by S. A. Laird, R. J. McLain and R. Wynberg. London and Washington D.C.: Earthscan, Pp. 85-112.

Lescuyer, G., Mvondo, S. A., Essoungou, J. N., Toison, V., Trébuchon, J. F. and Fauvet, N., 2012. Logging concessions and local livelihoods in Cameroon: from indifference to alliance? *Ecology and Society* 17(1): 7-19.

Lewis, S., 2016. Informality and Inclusive Green Growth: Evidence from the 'Biggest Private Sector' Event. London: International Institute for Environment and Development (IIED).

Liao, C., Jung, S., Brown, D. G. and Agrawal, A., 2016. Insufficient research on land grabbing. *Science* 353(6295): 131.

Ling, S., Smith, H., Xaysavongsa, L. and Laity, R., 2018. The evolution of certified teak grower groups in Luang Prabang, Lao PDR: an action research approach. *Small-Scale Forestry* 17(3): 343-360.

Lipper, L., Sakuyama, T., Stringer, R. and Zilberman, D. (eds.), 2009. Payment for environmental services in agricultural landscapes: Economic policies and poverty reduction in developing countries. Rome: FAO and Springer.

Little, P.D. and Watts, M. (eds.), 1994. Living under contract: contract farming and agrarian transformation in sub-Saharan Africa. Madison: Univ. of Wisconsin Press.

Liu, J., Li, S., Ouyang, Z., Tam, C. and Chen, X., 2008. Ecological and socioeconomic effect of China's policies for ecosystem services. PNAS 105(28): 9477-9482. Liu, Z. and Lan, J., 2018. The effect of the sloping land conversion programme on farm household productivity in rural China. The *Journal of Development Studies* 54(6): 1041–1059.

Loconto, A. and Dankers, C., 2014. Impact of international voluntary standards on smallholder market participation in developing countries: a review of the literature. Rome: Food and Agriculture Organization of the United Nations (FAO).

Lonn, P., Mizoue, N., Ota, T., Kajisa, T. and Yoshida, S., 2018. Evaluating the contribution of communitybased ecotourism (CBET) to household income and livelihood changes: A case study of the chambok CBET program in Cambodia. *Ecological Economics* 151: 62-69.

Lybbert, T. J., Aboudrare, A., Chaloud, D., Magnan, N. and Nash, M., 2011. Booming markets for Moroccan argan oil appear to benefit some rural households while threatening the endemic argan forest. PNAS 108(34): 13963-13968.

Ma, B., Cai, Z., Zheng, J. and Wen, Y., 2019. Conservation, ecotourism, poverty and income inequality–A case study of nature reserves in Qinling, China. World Development 115: 236-244.

Macqueen, D., 2008. Supporting Small Forest Enterprises: A Cross-Sectoral Review of Best Practice. IIED Small and Medium Forestry Enterprise Series No. 23. London: IIED.

Macqueen, D., Bolin, A. and Greijmans, M. (eds.), 2015. Democratising forest business: a compendium of successful locally controlled forest business organisations. London: IIED.

Macqueen, D., Bolin, A., Greijmans, M., Grouwels, S. and Humphries, S. 2020. Innovations towards prosperity emerging in locally controlled forest business models and prospects for scaling up. World Development 125: 104382.

Magome, H. and Murombedzi, J., 2003. Sharing South African National Parks: Community Land and Conservation in a Democratic South Africa. In: Decolonizing Nature – Strategies for Conservation in a Post-Colonial Era, edited by W.M. Adams and M. Mullingan. London: Earthscan Publications Ltd.

Mala, W.A, Tieguhong J.C, Ndoye O., Grouwels S. and Betti J.L., 2012. Collective action and promotion of forest based associations on non-wood forest products in Cameroon. *Development in Practice* 22(8): 1122- 1134.

Maniatis, D., Scriven, J., Jonckheere, I., Laughlin, J. and Todd, K., 2019. Toward REDD+ Implementation. Annual Review of Environment and Resources 44: 373-398.

Mansourian, S. and Vallauri, D., 2014. Restoring forest landscapes: important lessons learnt. *Environmental Management* 53(2): 241-251.

Mansourian, S., 2005. Overview of forest restoration strategies and terms. In: Forest restoration in landscapes : Beyond Planting Trees, edited by S. Mansourian, D. Vallauri, N. Dudley. New York: Springer, pp. 8-13. Mansourian, S., 2018. In the eye of the beholder: reconciling interpretations of forest landscape restoration. Land Degradation & Development 29(9): 2888-2898.

Mansourian, S., Razafimahatratra, A. and Vallauri, D., 2018. Lessons Learnt from 13 Years of Restoration in a Moist Tropical Forest: The Fandriana-Marolambo Landscape in Madagascar. Paris: WWF France.

Maraseni, T. N., Son, H. L., Cockfield, G., Duy, H. V. and Dai Nghia, T., 2017. The financial benefits of forest certification: Case studies of acacia growers and a furniture company in Central Vietnam. *Land Use Policy* 69: 56-63.

Marfo, E., Opuni-Frimpong, E. and Acheampong, E., 2012. Fractured tenure, unaccountable authority and benefit capture: constraints to improving community benefits under climate change mitigation schemes in Ghana. Conserv. Soc. 10(2): 161.

Markelova, H., Meinzen-Dick, R., Hellin, J. and Dohrn, S., 2009. Collective action for smallholder market access. Food Policy 34(1): 1-7.

Maryudi, A., Nawir, A. A., Sekartaji, D. A., Sumardamto, P., Purwanto, R. H., Sadono, R., Suryanto, P. et al., 2017. Smallholder Farmers' Knowledge of Regulations Governing the Sale of Timber and Supply Chains in Gunungkidul District, Indonesia. Small-Scale Forestry 16(1): 119-131.

Mayers, J. and Vermeulen, S., 2002. Company-community forestry partnerships: From raw deals to mutual gains. London: International Institute for Environment and Development.

Mayers, J., 2000. Company-community forestry partnerships: a growing phenomenon. Unasylva 51: 33–41.

Mayers, J., Buckley, L. and Macqueen, D. 2016. Small, but many, is big. Challenges in assessing the collective scale of locally controlled forest-linked production and investment. London: IIED.

Mayers, J. and Vermeulen, S., 2002. Power from the trees: How good forest governance can help reduce poverty, World Summit on Sustainable Development.

McCarthy, J. F., 2012. Certifying in Contested Spaces: private regulation in Indonesian forestry and palm oil. Third World Quarterly 33(10): 1871-1888.

McDougall, C., Jiggins, J., Pandit, B. H., Thapa Magar Rana, S. K. and Leeuwis, C., 2013. Does adaptive collaborative forest governance affect poverty? Participatory action research in Nepal's community forests. Society and Natural Resources 26(11): 1235-1251.

Medina, G., Pokorny, B., Campbell, B., 2008. Favouring local development in the Amazon: lessons from community forest management initiatives, CIFOR Livelihood Brief No.8. Bogor: CIFOR.

Medjibe, V. P. and Putz, F. E., 2012. Cost comparisons of reduced-impact and conventional logging in the tropics. *Journal of Forest Economics* 18(3): 242-256. Meemken, E. M. and Bellemare, M. F., 2019. Smallholder farmers and contract farming in developing countries. PNAS 117 (1): 259-264.

Meinzen-Dick, R., 2009. Property Rights for Poverty Reduction? DESA Working Paper No. 91 ST/ESA/2009/ DWP/91. URL: https://www.un.org/esa/desa/ papers/2009/wp91_2009.pdf [Accessed on 10 June 2020].

Meinzen-Dick, R. and Mwangi, E., 2009. Cutting the web of interests: Pitfalls of formalizing property rights. *Land Use Policy* 26(1): 36-43.

Mejía, E. and Pacheco, P., 2014. Forest use and timber markets in the Ecuadorian Amazon. Bogor: CIFOR.

Menton, M.C.S., Merry, F.D., Lawrence, A. and Brown, N., 2009. Company-community logging contracts in Amazonian settlements: Impacts on livelihoods and NTFP harvests. Ecology and Society 14(1): 39.

Meyer, C. and Miller, D., 2015. Zero deforestation zones: The case for linking deforestation-free supply chain initiatives and jurisdictional REDD+. *Journal of Sustainable Forestry* 34(6-7): 559-580.

Midgley, S. J., Stevens, P. R. and Arnold, R. J., 2017. Hidden assets: Asia's smallholder wood resources and their contribution to supply chains of commercial wood. *Australian Forestry* 80(1): 10-25.

Miller, D. C., Muñoz-Mora, J. C. and Christiaensen, L., 2017. Prevalence, economic contribution and determinants of trees on farms across Sub-Saharan Africa. Forest Policy and Economics 84: 47-61.

Miller, D. C., Ordoñez, P. J., Brown, S. E., Forrest, S., Nava, N. J., Hughes, K. and Baylis, K., 2020. The impacts of agroforestry on agricultural productivity, ecosystem services and human well-being in low-and middleincome countries: An evidence and gap map. *Campbell Systematic Reviews* 16(1): e1066.

Miller, D.C., Rana, P., Nakamura, K., Forrest, S., Cheng, S.H.T., Ahlroth, S.E. and Perge, E.B., 2019. A Systematic Review of the Impact of Forest Property Rights Interventions on Poverty. Washington, D.C.: World Bank Group.

Milne, S., Mahanty, S., To, P., Dressler, W., Kanowski, P. and Thavat, M. 2019. Learning from 'actually existing' REDD+: A synthesis of ethnographic findings. *Conservation & Society* 17(1): 84-95.

Miyata, S., Minot, N. and Hu, D., 2009. Impact of contract farming on income: linking small farmers, packers and supermarkets in China. *World development* 37(11): 1781-1790.

Molnar, A., Liddle, M., Bracer, C., Khare, A., White, A and Bull, J., 2007. Community-based forest enterprises in tropical forest countries: status and potential. Washington D.C.: RRI.

Monterroso, I., Larson, A.M., Mwangi, E., Liswanti, N. and Cruz-Burga, Z., 2019. Mobilizing Change for Women Within Collective Tenure Regimes. Resource Equity. (Available at: https://landwise.resourceequity.org/ records/3125-mobilizing-change-for-women-withincollective-tenure-regimes [Accessed on 6 October 2020]). Morsello, C., Ruiz-Mallén, I., Diaz, M. D. M. and Reyes-Garcia, V., 2012. The effects of processing non-timber forest products and trade partnerships on people's well-being and forest conservation in Amazonian societies. PloS one 7(8): e43055.

Muench, C. and Martínez-Ramos, M., 2016. Can Community-Protected Areas Conserve Biodiversity in Human-Modified Tropical Landscapes? The Case of Terrestrial Mammals in Southern Mexico. *Tropical Conservation Science* 178–202.

Nagendra, H., 2007. Drivers of reforestation in humandominated forests. Proceedings of the National Academy of Sciences 104(39): 15218-15223.

Nagendra, H., Pareeth, S., Sharma, B., Schweik, C.M. and Adhikari, K.R., 2008. Forest fragmentation and regrowth in an institutional mosaic of community, government and private ownership in Nepal. *Landscape Ecology* 23: 41-54.

Naidoo, R., Gerkey, D., Hole, D., Pfaff, A., Ellis, A. M., Golden, C. D., Herrera, D. et al., 2019. Evaluating the impacts of protected areas on human well-being across the developing world. *Science Advances* 5: 1–8.

Narayanan, S., 2014. Profits from participation in high value agriculture: Evidence of heterogeneous benefits in contract farming schemes in Southern India. Food Policy 44:142-157.

Ndoye, O. and Awono, A., 2010. Case Study B Policies for Gnetum spp. Trade in Cameroon: Overcoming Constraints that Reduce Benefits and Discourage Sustainability. In: Wild Product Governance, edited by S. A. Laird.Abingdon and New York: Routledge, pp. 99-104.

Newton, P. and Benzeev, R., 2018. The role of zerodeforestation commitments in protecting and enhancing rural livelihoods. *Current Opinion in Environmental Sustainability* 32: 126-133.

Newton, P., Agrawal, A. and Wollenberg, L., 2013. Enhancing the sustainability of commodity supply chains in tropical forest and agricultural landscapes. *Global Environmental Change* 23 (6): 1761-1772.

Ntsiful, A., 2010. Outgrower oil palm plantations scheme by private companies and poverty reduction in Ghana. Turks and Caicos: St. Clements University.

Obidzinski, K., Dermawan, A., Andrianto, A., Komarudin, H. and Hernawan, D., 2014. The timber legality verification system and the voluntary partnership agreement (VPA) in Indonesia: Challenges for the small-scale forestry sector. *Forest Policy and Economics* 48: 24–32.

Oduro, A. D. and Osei-Akoto, I., 2008. Market participation and rural poverty in Ghana in the era of globalization. *African Development Review* 20(1): 94-114.

Ojwang, A., 2000. Community-company partnerships in Forestry in South Africa. An examination of trends. London and Pretoria: IIED and CSIR. Okuro, S.O., 2011. Rethinking World Bank Driven Land Tenure Reforms in Kenya. In: Strategies Against Poverty: Designs from the North and alternatives from the South, edited by A. Puyana Mutisand S.O. Okuro. Buenos Aires: CLACSO, pp. 273-299

Oldekop, J. A., Holmes, G., Harris, W. E. and Evans, K. L., 2016. A global assessment of the social and conservation outcomes of protected areas. *Conservation Biology* 30(1): 133–141.

Oldekop, J. A., Sims, K. R., Karna, B. K., Whittingham, M. J. and Agrawal, A., 2019. Reductions in deforestation and poverty from decentralized forest management in Nepal. Nature Sustainability 2(5): 421-428.

Oldekop, J.A., Sims, K.R.E., Whittingham, M.J. and Agrawal, A., 2018. An upside to globalization: International outmigration drives reforestation in Nepal. Global Environmental Change 52: 66–74.

Oliveira, A. S., Soares, B. S., Costa, M. A., Lima, L., Garcia, R. A., Rajao, R. and Carvalho-Ribeiro, S. M., 2019. Bringing economic development for whom? An exploratory study of the impact of the Interoceanic Highway on the livelihood of smallholders in the Amazon. Landscape and Urban Planning 188: 171-179.

Ota, L., Herbohn, J., Harrison, S., Gregorio, N. and Engel, V. L., 2018. Smallholder reforestation and livelihoods in the humid tropics: a systematic mapping study. *Agroforestry Systems* 92(6): 1597–1609.

Overdevest, C., 2010. Comparing forest certification schemes: the case of ratcheting standards in the forest sector. Socio-economic Review 8(1): 47-76.

Oya, C., 2012. Contract farming in sub-Saharan Africa: A survey of approaches, debates and issues. *Journal of Agrarian Change* 12(1): 1-33.

Pacheco, P., 2012. Smallholders and Communities in Timber Markets: Conditions Shaping Diverse Forms of Engagement in Tropical Latin America. *Conservation and Society* 102: 114-123.

Pacheco, P., Barry, D., Cronkleton, P. and Larson, A.M. 2009. El papel de las instituciones informales en el uso de los recursos forestales en América Latina. Programa de Bosques y Gobernanza No. 15/2009. Bogor: CIFOR.

Pacheco, P., Schoneveld, G. C., Dermawan, A., Komarudin, H. and Djama, M., 2018. Governing sustainable palm oil supply: Disconnects, complementarities and antagonisms between state regulations and private standards. *Regulation and Governance* 14(3): 568-598.

Pagiola, S., Honey-Rosés, J. and Freire-González, J., 2016. Evaluation of the Permanence of Land Use Change Induced by Payments for Environmental Services in Quindío, Colombia. PLoS One 11(3): e0147829.

Palmer, C. and Engel, S., 2009. Avoided deforestation: Prospects for mitigating climate change. London and New York: Routledge.

Pandit, B. H., Albano, A. and Kumar, C., 2009. Community-based forest enterprises in Nepal: an analysis of their role in increasing income benefits to the poor. Small-scale Forestry 8(4): 447.

- Parajuli, R., Lamichhane, D. and Joshi, O., 2015. Does Nepal's community forestry program improve the rural household economy? A cost-benefit analysis of community forestry user groups in Kaski and Syangja districts of Nepal. *Journal of Forest Research* 20: 475-483.
- Parrotta, J., Wildburger, C. and Mansourian, S., 2012. Understanding relationships between biodiversity, carbon, forests and people: The key to achieving REDD+ objectives. A global assessment report prepared by the Global Forest Expert Panel on Biodiversity, Forest Management, and REDD+. IUFRO World Series, 31. Vienna: IUFRO.

Paschall, M. and Seville, D., 2012. Certified cocoa: Scaling up farmer participation in West Africa. London: International Institute for Environment and Development/Sustainable Food Lab.

Pasiecznik, N. (Ed.), 2015. Effective forest and farm producer organizations. European Tropical Forest Research Network.

Pearson, T. R., Bernal, B., Hagen, S. C., Walker, S. M., Melendy, L. K. and Delgado, G., 2018. Remote assessment of extracted volumes and greenhouse gases from tropical timber harvest. *Environmental Research Letters* 13(6): 065010.

Persha, L., Agrawal, A. and Chhatre, A., 2011. Social and ecological synergy: local rulemaking, forest livelihoods and biodiversity conservation. *Science* 331(6024): 1606-1608.

Phan, T.-H. D., Brouwer, R., Hoang, L. P. and Davidson, M. D., 2018. Do payments for forest ecosystem services generate double dividends? An integrated impact assessment of Vietnam's PES program. *PloS One* 13(8):e0200881.

Pierce, A. R., Shanley, P. and Laird, S. A., 2008. Non-timber forest products and certification: strange bedfellows. *Forests, Trees and Livelihoods* 18(1): 23-35.

Place, F., Adato, M., Hebinck, P. and Omosa, M., 2005. The impact of agroforestry-based soil fertility replenishment practices on the poor in western Kenya (Vol. 142).
Washington DC: International Food Policy Research Institute.

Place, F., Roothaert, R. L., Maina, L., Franzel, S., Sinja, J. and Wanjiku, J., 2009. The impact of fodder trees on milk production and income among smallholder dairy farmers in East Africa and the role of research. Nairobi: World Agroforestry Centre.

Pokorny, B., Sabogal, C. and de Jong, W., 2010. Challenges of community forestry in tropical America. Bois Et Forets Des Tropiques 303: 53–66.

Prosterman, R. L., Mitchell, R. and Hanstad, T. (eds.). 2009. One Billion Rising. Law, Land and the Alleviation of Global Poverty. Leiden University Press.

Purnomo, H., Guizol, P. and Muhtaman, D. R., 2009. Governing the teak furniture business: A global value chain system dynamic modelling approach. Environmental Modelling & Software 24(12): 1391-1401.

- Putz, F. E. and Viana, V., 1996. Biological challenges for certification of tropical timber. *Biotropica* 28(3): 323-330.
- Putzel, L., Kelly, A. B., Cerutti, P. O. and Artati, Y., 2015. Formalization as development in land and natural resource policy. Society & Natural Resources 28(5): 453-472.

Quandt, A., Neufeldt, H. and McCabe, J. T., 2019. Building livelihood resilience: what role does agroforestry play? Climate and Development 11(6): 485-500.

Quisumbing, A. R. and Kumar, N., 2014. Land rights knowledge and conservation in rural Ethiopia: Mind the gender gap. Washington DC: IFPRI.

Rasolofoson, R. A., Ferraro, P. J., Ruta, G., Rasamoelina, M. S., Randriankolona, P. L., Larsen, H. O. and Jones, J. P. G., 2017. Impacts of community forest management on human economic well-being across Madagascar. *Conservation Letters* 10(3): 346–353.

Resosudarmo, B.P. and Yusuf, A.A., 2006. Is the Log Export Ban an Efficient Instrument for Economic Development and Environmental Protection? The Case of Indonesia. Asian Economic Papers 5: 75–104.

Richards, M. and Hobley, M., 2016. Empowering civil society in forest policies and governance: livelihood impact assessment of EU FLEGT Voluntary Partnership Agreements. International Forestry Review. 18: 345–356.

Robalino, J., Sandoval, C., Villalobos, L. and Alpizar, F., 2014. Local effects of payments for environmental services on poverty. Discussion Paper. Washington DC: Resources for the Future (RFF).

Robbins, A. S. T. and Harrell, S., 2014. Paradoxes and challenges for China's forests in the reform era. *China Quarterly* 218(1): 381–403.

Robson, J.P. and Berkes, F., 2011. Exploring some of the myths of land use change: Can rural to urban migration drive declines in biodiversity? *Global Environmental. Change* 21: 844–854.

Rodriguez, L. G., Hogarth, N. J., Zhou, W., Xie, C., Zhang, K and Putzel, L., 2016. China's conversion of cropland to forest program: a systematic review of the environmental and socioeconomic effects. *Environmental Evidence* 5(1): 21.

Ros-Tonen, M.A.F., van Andel, T., Morsello, C., Otsuki, K., Rosendo, S. and Scholz, I., 2008. Forest-related partnerships in Brazilian Amazonia: There is more to sustainable forest management than reduced impact logging. Forest Ecology and Management 256: 1482–1497.

Ross, M. L., 2001. Timber booms and institutional breakdown in Southeast Asia. Cambridge: Cambridge University Press.

RRI (Rights and Resources Initiative), 2015. Who Owns the World's Land? A global baseline of formally recognized indigenous and community land rights. Washington, D.C.: RRI. RRI, 2018. At a Crossroads: Consequential trends in recognition of community-based forest tenure. Washington, D.C.: RRI.

Russell, D. and Franzel, S., 2004. Trees of prosperity: Agroforestry, markets and the African smallholder. Agroforestry Systems 61-2(1): 345-355.

Ruttan, V. W., 1984. Integrated rural development programmes: A historical perspective. World Development 12(4): 393-401.

Sanchez Badini, O., Hajjar, R. and Kozak, R., 2018. Critical success factors for small and medium forest enterprises : A review. Forest Policy and Economics. 94: 35–45.

Sanogo, D., N'Diaye, M., Badji, M. and Beye, S.A.,
2014. Optimisation de l'utilisation des ressources communes dans le bassin arachidier du sénégal: Évaluation ex ante de plans et d'alternatives de gestion durable. Biotechnology, Agronomy, Society and Environment 18(3): 339–352.

Santika, T., Wilson, K. A., Budiharta, S., Kusworo, A., Meijaard, E., Law, E. A., Friedman, R., et al., 2019. Heterogeneous impacts of community forestry on forest conservation and poverty alleviation: Evidence from Indonesia. *People and Nature* 12: 204-219.

Sartorius, K. and Kirsten, J., 2002. Can small-scale farmers be linked to agribusiness? The timber experience. *Agrekon* 41(4): 295-325.

Schleicher, J., Zaehringer, J.G., Fastré, C., Vira, B., Visconti P. and Sandbrook, C., 2019. Protecting half of the planet could directly affect over one billion people. Nature Sustainability 2(12): 1-3.

Schoneveld, G.C., van der Haar, S., Ekowati, D., Andrianto, A., Komarudin, H., Okarda, B., Jelsma, I., et al., 2019. Certification, good agricultural practice and smallholder heterogeneity: Differentiated pathways for resolving compliance gaps in the Indonesian oil palm sector. Global Environmental Change 57: 101933.

Schouten, G. and Bitzer, V., 2015. The emergence of Southern standards in agricultural value chains: A new trend in sustainability governance? *Ecological Economics* 120: 175-184

Schreckenberg, K., Degrande, A., Mbosso, C., Baboulé,
Z.B., Boyd, C., Enyong, L., Kanmegne, J., et al., 2002.
The social and economic importance of Dacryodes edulis (G. Don) HJ Lam in Southern Cameroon.
Forests, Trees and Livelihoods 12(1-2): 15-40.

Schure, J., Ingram, V., Sakho-Jimbira, M. S., Levang, P. and Wiersum, K. F., 2013. Formalisation of charcoal value chains and livelihood outcomes in Central-and West Africa. *Energy for Sustainable Development* 17(2): 95-105.

Scudder, M. G., Baynes, J. and Herbohn, J., 2019. Timber royalty reform to improve the livelihoods of forest resource owners in Papua New Guinea. *Forest policy and economics* 100: 113-119. Shackleton, S., Paumgarten, F., Kassa, H., Husselman, M. and Zida, M., 2011. Opportunities for enhancing poor women's socioeconomic empowerment in the value chains of three African non-timber forest products (NTFPs). International Forestry Review 13(2): 136-151.

Shiferaw, B., Hellin, J. and Muricho, G., 2011. Improving market access and agricultural productivity growth in Africa: what role for producer organizations and collective action institutions? *Food Security* 3(4): 475-489.

Shrestha, K. K. and McManus, P., 2013. The politics of community participation in natural resources management: lessons from community forestry in Nepal. Australian Forestry 71(2): 135-146.

Shrestha, S., Shrestha, U. B. and Bawa, K. S., 2017. Contribution of REDD+ payments to the economy of rural households in Nepal. *Applied Geography* 88: 151-160.

Shyamsundar, P., Ahlroth, S., Kristjanson, P. and Onder, S., 2020. Supporting pathways to prosperity in forest landscapes – A PRIME framework. World Development 125: 104622.

Simpson, M. C., 2012. The impacts of tourism initiatives on rural livelihoods and poverty reduction in South Africa: Mathenjwa and Mqobela. In: Responsible Tourism: Critical Issues for Conservation and Development, edited by A.Spenceley. London: Routledge, pp. 267-294.

Sims, K. R. E. and Alix-Garcia, J. M., 2017. Parks versus PES: Evaluating direct and incentive-based land conservation in Mexico. *Journal of Environmental Economics and Management* 86: 8–28.

Sims, K. R. E., 2010. Conservation and development: Evidence from Thai protected areas. Journal of Environmental Economics and Management 602): 94–114.

Snyman, S. L., 2012. The role of tourism employment in poverty reduction and community perceptions of conservation and tourism in southern Africa. *Journal* of Sustainable Tourism 20(3): 395-416.

Snyman, S., 2017. The role of private sector ecotourism in local socio-economic development in southern Africa. *Journal of Ecotourism* 16(3): 247-268.

Sodikoff, G. M., 2012. Forest and labor in Madagascar: From colonial concession to global biosphere. Bloomington: Indiana University Press.

Song, C., Zhang, Y., Mei, Y., Liu, H., Zhang, Z., Zhang, Q., Zha, T., et al., 2014. Forests Sustainability and China's Sloping Land Conversion Program (SLCP): A Comparison among Three Sites in Anhui, Hubei and Shanxi. Forest Policy and Economics 38: 161-167.

Stanturf, J. A., Palik, B. J. and Dumroese, R. K., 2014. Contemporary forest restoration: A review emphasizing function. Forest Ecology and Management 331: 292–323. Stanturf, J.A., Kant, P., Lillesø, J.P.B., Mansourian, S., Kleine, M., Graudal, L. and Madsen, P., 2015. Forest landscape restoration as a key component of climate change mitigation and adaptation. Vienna: International Union of Forest Research Organizations (IUFRO).

Stoian, D., Meldrum, G., Lamers, H., Blare, T., Elias, M. and Donovan, J., 2018. Exploring a genderresponsive asset-based approach to enhance the transformative potential of value chain development in Guatemala, India and Peru. Paper presented at the CGIAR PIM Workshop "Rural Transformation in the 21st Century: The Challenges of Low-Income, Late-Transforming Countries" held as Pre-Conference Symposium of the International Conference of Agricultural Economists in Vancouver, BC, Canada on July 28, 2018.

Straumann, L., 2014. Money logging: on the trail of the Asian timber mafia. Zurich: Salis Verlag.

Sullivan, R. O., Roth, M., Antwi, Y. A., Ramirez, P. and Sommerville, M., 2018. Land and tree tenure innovations for financing smallholder cocoa rehabilitation in Ghana. Paper prepared for presentation at the "2018 World Bank Conference on Land and Poverty".

Sunam, R. K. and McCarthy, J. F., 2010. Advancing equity in community forestry: recognition of the poor matters. International Forestry Review 12(4): 370-382.

Tanaka, H., 2012. Competitive Strategies in Community Forestry: Leveraging Through Second-tier Collective Enterprises in Mexico and Guatemala. PhD thesis, University of Idaho.

Taylor, P. L., 2010. Conservation, community and culture? New organizational challenges of community forest concessions in the Maya Biosphere Reserve of Guatemala. *Journal of Rural Studies* 262: 173-184.

Tegegne, Y.T., Ramcilovic-Suominen, S., Fobissie, K., Visseren-Hamakers, I.J., Lindner, M., Kanninen, M., 2017. Synergies among social safeguards in FLEGT and REDD+ in Cameroon. Forest Policy and Economics. 75: 1–11.

Thoms, C. A., 2008. Community control of resources and the challenge of improving local livelihoods: a critical examination of community forestry in Nepal. *Geoforum* 39(3): 1452-1465.

Thorlakson, T. and Neufeldt, H., 2012. Reducing subsistence farmers' vulnerability to climate change: evaluating the potential contributions of agroforestry in western Kenya. *Agriculture and Food Security* 1(15): 1-13.

Tilahun, M., Maertens, M., Deckers, J., Muys, B. and Mathijs, E., 2016. Impact of membership in frankincense cooperative firms on rural income and poverty in Tigray, Northern Ethiopia. *Forest Policy and Economics* 62: 95-108. Tilahun, M., Maertens, M., Deckers, J., Muys, B. and Mathijs, E., 2016. Impact of membership in frankincense cooperative firms on rural income and poverty in Tigray, Northern Ethiopia. *Forest Policy and Economics* 62: 95-108.

Tirivayi, N., Nennen, L. and Tesfaye, W. 2018. The role of forest producer organizations in social protection. Forestry Working Paper No. 7. Rome: FAO.

Tiveau D., 2008. Harvesting forests to reduce poverty. CIFOR Annual Report 2008: Thinking Beyond the Canopy. Bogor: CIFOR. Pp. 18-21.

Tomaselli, M.F., Timko, J. and Kozak, R., 2012. The Role of Government in the Development of Small and Medium Forest Enterprises: Case Studies from The Gambia. *Small-scale Forestry* 11: 237–253.

Ton, G., Vellema, W., Desiere, S., Weituschat, S. and D'Haese, M., 2018. Contract farming for improving smallholder incomes: What can we learn from effectiveness studies? World Development 104: 46-64.

Tran, L. and Walter, P., 2014. Ecotourism, gender and development in northern Vietnam. *Annals of Tourism* Research 44: 116-130.

Treacy, P., Jagger, P., Song, C., Zhang, Q. and Bilsborrow, R. E., 2018. Impacts of China's Grain for Green Program on migration and household income. *Environmental Management* 62(3): 489–499.

Tuijnman, W., Bayrak, M. M., Hung, P. X. and Tinh, B. D., 2020. Payments for environmental services, gendered livelihoods and forest management in Vietnam: a feminist political ecology perspective. Journal of Political Ecology 27(1): 317-334.

Tumaneng-Diete, T., Ferguson, I.S. and MacLaren, D., 2005. Log export restrictions and trade policies in the Philippines: Bane or blessing to sustainable forest management? *Forest Policy and Economics* 7: 187–198.

UNEP-WCMC and IUCN, 2016. Protected Planet Report 2016. Cambridge and Gland: UNEP-WCMC and IUCN.

UNFCCC, 2011. Paragraph 71 and Appendix 1 of decision 1/CP.16 https://unfccc.int/resource/docs/2010/ cop16/eng/07a01.pdf

Urgenson, L., Schmidt, A. H., Combs, J., Harrell, S., Hinckley, T., Yang, Q., Ma, Z. *et al.*, 2014. Traditional Livelihoods, Conservation and Meadow Ecology in Jiuzhaigou National Park, Sichuan, China. *Human* Ecology 42(3): 481–491.

Van Andel, T., 2007. A company-community partnership for FSC-certified non-timber forest product harvesting in Brazilian Amazonia: requirements for sustainable exploitation. In: Partnerships in Sustainable Forest Resource Management: Learning from Latin America, edited by M. A. F. Ros-Tonen. Leiden and Boston: Brill, pp. 169-188.

Van Hensbergen, B., 2016. Forest Concessions – Past Present and Future. Forestry and Institutions Working Paper, 36. Rome: FAO.

- Verchot, L. V., Van Noordwijk, M., Kandji, S., Tomich, T., Ong, C., Albrecht, A., Mackensen, J., et al., 2007.
 Climate change: linking adaptation and mitigation through agroforestry. Mitigation and Adaptation Strategies for Global Change 12(5): 901-918.
- Vermeulen, S. and Cotula, L., 2010. Making the most of agricultural investment: A survey of business models that provide opportunities for smallholders. London: IIED.
- Vermeulen, S., Nawir, A.A., Mayers, J., 2008. Rural poverty reduction through business partnerships? Examples of experience from the forestry sector. Environment, Development and Sustainability 10: 1–18.
- Warner, R. and Bauer, M., 2002. Mama Lus Frut scheme: an assessment of poverty reduction. Canberra: Center for International Economics.
- Warning, M. and Key, N., 2002. The social performance and distributional consequences of contract farming: An equilibrium analysis of the Arachide de Bouche program in Senegal. *World Development* 302: 255-263.
- Weng, X., 2015. The rural informal economy: Understanding drivers and livelihood impacts in agriculture, timber and mining. IIED Working Paper. London: IIED.
- Weng, X. and Putzel, L., 2017. Rural livelihoods and informal timber in Cameroon. Forest governance, small-scale loggers and Chinese traders in the bubinga (Guibourtia spp.) trade. Bogor: CIFOR.
- Werner, W. and Kruger, B., 2007. Redistributive Land Reform and Poverty Reduction in Namibia. A working paper for the research project on 'Livelihoods after Land Reform'. Institute for Poverty for Land and Agrarian Studies (PLAAS) / Institute of Development Studies, South Africa. (available at: https://assets.publishing.service.gov.uk/ media/57a08bd440f0b652dd000f02/60332_Namibia_ Country_Paper.pdf [accessed on 10 June 2020]).
- Weston, P., Hong, R., Kaboré, C. and Kull, C. A., 2015.
 Farmer-Managed Natural Regeneration Enhances
 Rural Livelihoods in Dryland West Africa.
 Environmental Management 55(6): 1402–1417.
- Wishitemi, B. E., Momanyi, S. O., Ombati, B. G. and Okello, M. M., 2015. The link between poverty, environment and ecotourism development in areas adjacent to Maasai Mara and Amboseli protected areas, Kenya. *Tourism Management Perspectives* 16: 306-317.
- World Bank, 2002a. Forest Concession Policies and Revenue Systems: Country experience and policy changes for sustainable tropical forestry. Washington, D.C. : World Bank.
- World Bank, 2002b. Building Institutions for Markets. World Development Report 2002. Washington, D.C. : World Bank.
- World Bank, 2017. Poverty, forest dependence and migration in the forest communities of Turkey: evidence and policy impact analysis. PROFOR. Washington, D.C. : World Bank.

- WTTC (World Travel and Tourism Council), 2019. https:// www.wttc.org/about/media-centre/press-releases/ press-releases/2019/travel-tourism-continuesstrong-growth-above-global-gdp/. Accessed on 26 February, 2020.
- Wunder, S., 2000. Ecotourism and economic incentives An empirical approach. *Ecological Economics* 32(3): 465–479
- Wunder, S., 2015. Revisiting the concept of payments for environmental services. *Ecological Economics* 117: 234-243.
- Wunder, S., Duchelle, A. E., Sassi de, C., Sills, E.O.,
 Simonet, G. and Sunderlin, W. D. 2020. REDD+ in
 Theory and Practice: How Lessons From Local
 Projects Can Inform Jurisdictional Approaches. Front.
 For. Glob. Change 3(11): 1-17.
- Xinhua News Agency, 2019. The practice of the harmonious coexistence between human and nature – A systematic review of the achievements of the 20-year implementation of the Conversion of Cropland to Forest Program, September 9, 2019.
- Yadav, B. D., Bigsby, H. and MacDonald, I., 2015. How can poor and disadvantaged households get an opportunity to become a leader in community forestry in Nepal? *Forest Policy and Economics* 52: 27-38.
- Yang, H., Yang, W., Zhang, J., Connor, T. and Liu, J., 2018. Revealing pathways from payments for ecosystem services to socioeconomic outcomes. *Science Advances* 4(3): eaao6652.
- Yi-fong, C. 2012. The indigenous ecotourism and social development in Taroko National Park area and San-Chan tribe, Taiwan. *GeoJournal* 77(6): 805-815.
- Yin, R., Liu, C., Zhao, M., Yao, S. and Liu, H. 2014. The implementation and impacts of China's largest payment for ecosystem services program as revealed by longitudinal household data. *Land Use Policy* 40: 45–55.
- Zhang, Q., Bilsborrow, R., Song, C. and Huang, Q., 2018.
 Determinants of out-migration in rural China:
 Effects of payment for ecosystem services. *Population* and Environment 40(2): 182-203.
- Zhou, S., Yin, Y., Xu, W., Ji, Z., Caldwell, I. and Ren, J., 2007. The costs and benefits of reforestation in Liping County, Guizhou Province. China. Journal of Environmental Management 85(3): 722–735.
- Zulu, L. C. and Richardson, R. B., 2013. Charcoal, livelihoods, and poverty reduction: Evidence from sub-Saharan Africa. *Energy for Sustainable Development* 17(2): 127-137.



Chapter 6

Global Forces of Change: Implications for Alleviating Poverty and Sustaining Forests

Coordinating Lead Authors: Priya Shyamsundar

Lead Authors: Jennifer Zavaleta Cheek, Laura Vang Rasmussen, Daniel C. Miller and Johan Oldekop Contributing Authors: Laura Aileen Sauls, Kira A. Sullivan-Wiley, James T. Erbaugh, P.P. Krishnapriya, Pablo Pacheco, George Schoneveld, Conghe Song and Joleen Timko

TABLE OF CONTENTS

6.1 Introduction	
6.2 Growth in Commodity Markets	
6.3 Climate Change	
6.4 The Shifting Landscape of Forest Sector Financing	
6.5 Technological Change and Interconnectivity	
6.6 Global Socio-Political Movements	
6.7 The Spread of Infectious Diseases	
6.8 Conclusions: Global Forces of Change and Poverty Alleviation	
6.9 References	

Abstract

This chapter examines six major global forces likely to influence forests and tree-based systems and considers their implications for poverty. The global trends discussed include: (i) growth in commodity markets, (ii) climatic impacts mediated through changes in forests, (iii) trends in private and public financing, (iv) technological advances and interconnectivity, (v) global socio-political movements, and (vi) emerging infectious diseases. These trends bring both opportunities and risks to the forest-reliant poor. In a business-as-usual scenario, the cumulative risks posed by global forces, in conjunction with limited rights, resources and skills required to prosper from global changes, are likely to place poor and transient poor households under additional stress. Global forces will also affect the non-poor, who may be able to better adapt to these changes. The chapter concludes with a discussion on how levers to better manage forests, combined with supportive conditions, can contribute to a different and more prosperous future for forests and people.

6.1 Introduction

Poverty⁸ dynamics reflect the continuously shifting world that the world's poor face. While change is constant, some global forces may be particularly influential and important to understand. Global forces of change can be opaque at local or regional scales, but can create enormous shocks, impeding gains made in alleviating poverty or threatening the transient poor. These forces can also create uncertainty regarding the efficacy of existing policy levers (see Chapter 5) for alleviating forest poverty. On the other hand, they may also generate socio-economic opportunities that help move people out of poverty. Thus, it is useful to probe critical global forces of change to anticipate or forecast likely scenarios in *forest* poverty dynamics.

The COVID 19 pandemic has put a spotlight on global forces of change and their local costs. Forest fragmentation and the increase in zoonotic diseases, including COVID-19, underscore the need to attend to the linkages among forest cover change in distant places, health and sustainable development (Di Marco et al., 2020; Dobson et al., 2020). More broadly, these connected changes exemplify how global mechanisms can directly and indirectly affect communities living in, near or otherwise relying on forests. This chapter focuses on a subset of six major global changes that are likely to have wide-ranging influence on forest-poverty dynamics in the years ahead: (i) growth in commodity markets; (ii) climatic impacts mediated through changes in forests; (iii) trends in private and public financing; (iv) technological advances and interconnectivity; (v) global socio-political movements; and (vi) emerging infectious diseases (EIDs).

The global changes we examine in this chap-

ter emerged from expert discussions through GFEP meetings during 2019-2020 and a review of the literature on global changes and forest interlinkages (Eakin et al., 2014; Liu et al., 2015; Oldekop et al., 2017; Watts et al., 2019; World Economic Forum, 2020). The identified global forces generally met three criteria: they are largely exogenously driven, but influence forests and tree-based systems in multiple regions; they are dynamic, reflecting shifting geo-political conditions; and they are likely to influence forest-poverty dynamics. Other trends such as urbanisation (Jiang and O'Neill, 2017), economic globalisation (Lambin and Meyfroidt, 2011) and demographic changes were excluded because of the rural focus of this analysis. However, we do reference some of these other trends in the discussion on forest commodity markets, and in the context of migration as a response to climate change (Cattaneo et al., 2019).

In the sections that follow, we describe each global change and discuss its implications for forest-poverty dynamics based on available literature. The chapter concludes with an acknowledgement of the limitations of our analyses and a critical evaluation of conditions and strategies that may be needed to sustain forests and alleviate poverty, given global changes.

6.2 Growth in Commodity Markets

6.2.1 Introduction

Demand for commodities produced in the tropics plays an important role in *deforestation* and *degradation* (Curtis *et al.*, 2018; Seymour and Harris, 2019). Global commodity markets are likely to continue to contribute to land cover change and the levers

⁸ Throughout this assessment report, all terms that are defined in the glossary are introduced for the first time in a chapter using italics.



Source: Lawson et al., 2014

launched to decouple market growth and forest loss are critical for influencing *poverty alleviation*.

6.2.2 Commodity demand and deforestation

Demand for four commodities – wood products, beef, soy, and palm oil – has significantly modified tropical forests (Newton *et al.*, 2013; Persson *et al.*, 2014; Curtis *et al.*, 2018; Henders *et al.*, 2018). Other commodities, such as coffee (Philpott *et al.*, 2008; Tadesse *et al.*, 2014), cacao, cassava (Gockowski and Sonwa, 2011) and illicit coca production (Armenteras *et al.*, 2013) also play an important role in varied geographies.

Demand for beef has had the biggest impact on deforestation (Henders *et al.*, 2015; World Economic Forum, 2020). In South America, 71% of deforested area during 1990-2005 was used for pasture creation (De Sy *et al.*, 2015). Brazil, China, the European Union (EU) and the United States (US) are

major beef producers (Brack *et al.*, 2016), with Brazilian cattle production having the largest forest footprint (Henders *et al.*, 2015; Pendrill *et al.*, 2019). Soy, with cultivation in South America increasing by over 170% during 1990-2010 (WWF, 2014), also contributes to significant land use changes (le Polain de Waroux *et al.*, 2019). Brazil and the US are responsible for most global soybean exports (Gale, *et al.*, 2019), which are used mainly for meeting demand for animal feed (Brack *et al.*, 2016).

Indonesia and Malaysia produce about 80-90% of all palm oil (Henders *et al.*, 2015; Brack *et al.*, 2016). During 2000-16, oil palm cultivation contributed to 23% of deforestation in Indonesia, with the deforestation rate declining by 2014-16 to under 15% (Austin *et al.*, 2019).

Tropical timber is mainly produced in Brazil, Indonesia, Malaysia and Papua New Guinea (Persson *et al.*, 2014). During 2000-16, timber *plantations* contributed to 14% of deforestation in Indonesia, with the rate peaking in 2010-12 (Austin *et al.*, 2019). Arguably, timber plantations play multiple roles: they can reduce pressure on natural forests (Bowyer *et al.*, 2005; Ainembabazi and Angelsen, 2014); contribute to restoring degraded lands (Bowyer *et al.*, 2005); and improve smallholders' *livelihoods* (Khamzina *et al.*, 2012; Roshetko *et al.*, 2013). In general, however, evidence on plantation-led improvements in livelihoods and poverty is both limited and mixed (Malkamäki *et al.*, 2018; Santika *et al.*, 2019).

Commodity growth is likely to continue. Beef production in developing countries is projected to be 17% higher in 2028 relative to the average of 2016 and 2018 (OECD and FAO, 2019). China's beef imports, in particular, have increased by 160% from the levels in 2011 and are projected to rise by another 55% by 2026 (World Economic Forum, 2017; 2020), with production responding in Argentina, Brazil, China, Mexico, Pakistan, South Africa, and the US. Soybean production is projected to increase to 390 million tonnes by 2050, increasing land under soy cultivation by some 30% between 2005-07 and 2050 (Alexandratos and Bruinsma, 2012). A smaller time-period projection suggests a 33% increase in soybean exports between 2028-29 and 2016-17 (Gale et al., 2019), with growth in export demand led by China and supply led by Brazil and the US. Global palm oil production is also

expected to increase by 1.8% per year between 2016-18 and 2028, with the production of palm oil in Indonesia and Malaysia likely to be affected by stricter environmental and agricultural policies (OECD and FAO, 2019). Demand for wood products is projected to increase three-fold (by volume) by 2050 relative to 2010 (WWF, 2012).

6.2.3 Decoupling commodity growth and deforestation

As demand for tropical commodities at the forest frontier continues to grow, there is pressure to decouple this growth from deforestation. Ten strategies rise to the top to achieve net zero deforestation commitments (ZDCs) rates, while maintaining or increasing production of commodities at the forest frontier (World Economic Forum, 2020). The strategies emphasise building on core principles of sustainable intensification (certification, pilot scaling), increased funding for sustainability (demand and financing) and improved governance (property rights, jurisdictional approaches, illegality/enforcement). These mechanisms (Table 6.1) are likely to impact the forest-proximate poor engaged in agricultural activities. Many governments, donor countries and non-profit and corporate agencies are aligning themselves around the strategies identified in Table 6.1. However, the



The transformation of forests into arid land endangers the long-term provision of critical resources (W National Park, Benin) Photo © Dan C. Miller
Strategies for decoupling deforestation from commodity growth and poverty implications*			
STRATEGIC ACTION	POSSIBLE MECHANISMS	POTENTIAL POVERTY IMPLICATIONS	
Eliminate illegality from supply chains	 Align forest laws and codes Improve data and enforcement Strengthen voluntary commitments 	 Secure property rights (+) Aligning laws may reduce smallholder agency (-) Strengthen Indigenous and community management (+) Compliance (-) 	
Certify palm oil	 Increase demand for certified products Support sustainable palm oil supply chains Improve/expand certification 	 High certification costs (-) Better bargaining power with companies and/market access (+) 	
Scale up sustainable intensification of cattle	 Increase adoption Enforce to reduced extensification Increase credit and finance for intensification 	 Productivity increases could help income (+) Co-benefits (+) Corporate preferential treatment (+) Tied to participation (?) 	
Increase smallholder yields for palm oil and cocoa	 Best management practices, input increases and replanting for oil palm Public and private credit, training, organisation and aggregation 	 Productivity increases help as smallholders produce around 40% of global palm oil (+) (WEF, 2017) Average cacao bean yields (global) stable or decreased in last two decades (?) 	
Sustainable production of soy	Build on the Amazon soy moratorium to strengthen public-private agreementsPlant soy on land already cleared for cattle	 Integration into global markets (+) Reduced land grabbing (+) 	
Accelerate jurisdictional program- mes	 Jurisdiction-level certification of ZDC commodities and sustainable intensification Public-private investment in enforcement, registration, planning and programmes for smallholders Include political support, private sector cooperation, collaborative planning, transparency and finance 	 Integration into global markets (+) Reduced land grabbing (+) 	
Address land conflicts, tenure security and land rights	 Combine legal protection with land registries, especially for Indigenous and community lands Include conflict resolution mechanisms 	 Reduced land grabbing (+) Secure access, esp. to Indigenous communities (+) Gender differences in tenure (-) Unresolved conflicts (-) 	
Mobilise demand for ZDC commodities	 Sustainable sourcing of soy and palm oil in consumer markets (China/India) Local companies brought into global discussions 	 Increase in demand for sustainably sourced products leads to small holder integration into markets (+) Increase in land grabbing and loss of market access to smallholders (-) 	

Table 6.1

Redirect finance to deforestation-free supply chainsDirect capital/finance to ZDC supply chainsUnequal distributions of capital and finance (-)Direct impact investing using public-private partnership platformsDirect impact investing using public-private partnership platformsDirect impact investing using insurance, cooperatives, training, insurance, certification, impact investment (+)	ACTION	POSSIBLE MECHANISMS	IMPLICATIONS
	Redirect finance to deforestation-free supply chains	 Direct capital/finance to ZDC supply chains Increase smallholder access to finance Direct impact investing using public-private partnership platforms 	 Unequal distributions of capital and finance (-) Support for smallholders – finance, cooperatives, training, insurance, certification, impact investment (+)
 Improve quality and availability of data/alerts Expand tools like the 'High Carbon Stock Approach' Improve geospatial information on land tenure, concessions, and licences Agreed definitions ("forest", "ZDC", and "high conservation area") Maps can legitimise claims by people in power and Indigenous and local communities (-/+) Tools that prioritise are only as good as the information fed to them (+/-) 	Improve quality and availability of deforestation and supply chain data	 Increase near real-time deforestation data/alerts Expand tools like the 'High Carbon Stock Approach' Improve geospatial information on land tenure, concessions, and licences Agreed definitions ("forest", "ZDC", and "high conservation area") 	 Maps can legitimise claims by people in power and Indigenous and local communities (-/+) Tools that prioritise are only as good as the information fed to them (+/-)

* Chapter 5 mainly reviews forest-related market levers of change, while the focus here is on agriculture-related actions affecting agricultural commodities primarily sourced at the forest frontier.

Source: Adapted from priorities list for 'decoupling' discussed by the World Economic Forum. The poverty implications are based on a review of material presented in the same document (World Economic Forum 2017).

extent to which decoupling of commodity growth and deforestation occurs will, at least partly, depend on how well the mechanisms in Table 6.1 are implemented. In addition, the potential poverty outcomes detailed in the Table will likely vary, based on a confluence of local conditions (see Chapter 4).

6.2.4 Implications for forest poverty dynamics

Land use change can offer opportunities to smallholders who are able to take advantage of global markets. Evidence from Paraguay, for instance, suggests that farmers and Indigenous communities have improved incomes and livelihoods through soy cultivation (Cardozo et al., 2016). Likewise, there is evidence that oil palm cultivation has increased smallholder incomes and rural employment in Asia, reducing poverty rates (Qaim et al., 2020) and benefitting nonfarm employment and income (Bou Dib et al., 2018). In Brazil, soybean cultivation supports some 2.5 formal sector jobs outside of agriculture per square kilometre of production (Richards et al., 2015).

Along with income and jobs, commodity markets also bring new risks. Box 6.1, drawing from Ingram (2014) and Ingram et al. (2017), illustrates the case of bush mango trade in West Africa which benefits many households. However, market supply chains ignore longer term considerations, and gains can be eroded if local institutions are unable to adjust rapidly to match market growth.



Shea butter fruit (Vitellaria paradoxa) recently collected in the forest Photo © Daniel C. Miller

Box 6.1

Understanding the short-term and longer term influence of global markets on forest use and local livelihoods by examining the bush mango supply chain

Global demand from distant markets is increasingly driving the profitability of many timber and non-timber forest products (NTFPs), with varied short and long term outcomes. In the case of bush mangoes, growing in several countries in Africa, global markets contribute to increased income for many poor households, while also increasing incentives to over-harvest, undermining long-term sustainability.

The bush mango (*Irvingia* spp.) is a popular commodity that is both traded and used domestically. The ground kernels of bush mangoes are used as a condiment and sauce thickener; and the kernels are increasingly processed in Europe and the US as an important ingredient in weight-loss aids, health supplements and cosmetics (Ingram, 2014). The seeds are used for cooking oil, the juice is used in cooking and wine, the pulp as a dye and the timber is used for construction. The growth in demand for bush mangoes has contributed to improving livelihoods. In Southern Cameroon for example, 5,200 people are directly employed

Market growth can favour some stakeholders while dislodging others. In Brazil and Indonesia, lack of legal clarity on rights has made households vulnerable to land grabbing (FOE, 2008; Gabay and Alam, 2017). In Malaysia, expansion of oil palm cultivation brought in foreign workers and contributed to wage suppression (Abdullah et al. 2011). The Brazilian state has sought to counter smallholder displacement by resettling landless farmers from poor regions and connecting them directly to soy companies. However, rugged geography, small scale operations and high production costs have limited smallholder partnerships with companies (Lima et al. 2011). In general, certain characteristics of smallholder operations and high barriers to entry may result in their inability to capture market surplus.

Land speculation fuelled by demand for commodities often contributes to rural conflict, as has been the case in Brazil for example (Nepstad *et al.*, 2006; Nepstad and Stickler, 2008). Likewise in Indonesia, expansion of oil palm has led to community-firm conflicts, exacerbated by weak governance, in the bush mango value chain, contributing on average to 31% of total household incomes (Ingram *et al.*, 2017). For stakeholders located further away from the forest, the income gains are even more pronounced – contributing an average of 48% of exporters' annual income and 57% of retailers'.

Despite bush mangoes making substantial livelihood contributions, the value chain is increasingly becoming unsustainable because of: a) lack of consistent regulatory control and enforcement; b) continuing high demand; c) low levels of cultivation driven by alternate uses of agricultural land; d) declining wild resources evidenced by harvesters travelling longer distances; and e) clearance of the species' natural forest habitat. Customary tenure that seems to allow access to trees within common forests on a 'first-come, first-served' basis is not compatible with growing demand. Some combination of technical advice, formal and customary rule changes and improved monitoring is required to sustain this market.

ambiguous contracts with firms and the failure of firms to meet obligations (Rist *et al.*, 2010). Poverty alleviation requires strengthening of formal and informal governance arrangements that favour poor households and ensure resource use is sustainable (see Box 6.1).

Reforms to commodity markets (see Table 6.1) can change social and environmental outcomes. For instance, about 35-40% of palm oil is produced by small landholders (World Economic Forum, 2017). Thus, strengthening credit, training and technology access can significantly increase productivity and reduce deforestation (World Economic Forum, 2017; 2020). Additionally, buttressing rights can reduce land grabbing, with potential positive environmental outcomes, and jurisdictional approaches may be able to increase transparent decision-making. Smallholders will also need to be supported through institutional innovations that enable them to pool resources and increase market shares (Poole and de Frece, 2010). Additional research on the social dimensions of different supply chain reform initiatives would help identify whether companies are adopting voluntary practices (Thorlakson *et al.*, 2018) and how they may mitigate poverty (Newton and Benzeev, 2018).

6.3 Climate Change

6.3.1 Introduction

Climate change increases risks to the lives and livelihoods of the *forest-reliant* poor and affects the forest *ecosystem services* on which they depend. These changes will worsen with deforestation, itself a major contributor to climate change (Lawrence and Vandecar, 2015; World Resources Institute, 2018), creating a positive feedback loop (Staal *et al.*, 2020). On the other hand, improved *forest management* offers potential opportunities to mitigate climate change (Griscom *et al.*, 2017) and enable forest and tree dependent households to better adapt to climatic variability (Pramova *et al.*, 2012; Mbow *et al.*, 2014).

6.3.2 Climate change, forests and feedback loops

Climate induced changes have varied effects on forests which also differ according to forest type and region. Changes in temperature, carbon dioxide and precipitation can increase the length of tree growing seasons (Walther et al., 2002), alter the distribution of terrestrial vegetation (Bertin, 2008), shift species' geographic ranges (Hansen et al., 2001; Ortega et al., 2019), influence productivity (Vitasse et al., 2009), increase the risk and intensity of natural disasters such as drought, fires, flooding and insect outbreaks (Seppälä et al., 2009; Staal et al., 2020), and affect biodiversity and ecosystem services (Seymour and Busch, 2016), among others. Two key climate effects - increased heat stress and increased frequency, intensity and/or amount of heavy precipitation (Mora et al., 2017; Coffel et al., 2018; Olsson, et al., 2019; Prevedello et al., 2019) - are likely to have direct and forest-induced effects on the lives and livelihoods of forest-proximate communities. Increased flooding (Zhu et al., 2010; Neumann et al., 2015), drought frequency and severity (Sheffield et al., 2012; Steinkamp and Hickler, 2015; Staal et al., 2020), intensified cyclones (Walsh, et al., 2016; Bacmeister et al., 2018) and increased coastal erosion (Alongi, 2015; Johnson et al., 2015; Harley et al., 2017) threaten existing forests and tree-based goods and services. In many locations extreme droughts are predicted to increase the number, intensity, length and severity of forest fires (Jolly et al., 2015; Abatzoglou and Williams, 2016; Knorr et al., 2016; Taufik et al., 2017).

The release of sequestered forest carbon resulting from deforestation, degradation and forest fires produces a significant amount of greenhouse gases, creating a positive feedback loop that accelerates climate change. Averaged over 2015 - 2017, global loss of tropical forests contributed about 4.8 billion tonnes of carbon dioxide per year (or about 8-10% of annual human emissions of carbon dioxide) (World Resources Institute, 2018). Illustratively, on many days, the 127,000 fires that raged across Indonesia in 2015 generated more greenhouse gas emissions per day than daily average emissions from the entire economy of the United States that year (Harris et al., 2015; Seymour and Busch, 2016). Conversely, improved forest management can also act as a natural climate solution (Griscom et al., 2017). Reductions in deforestation and increasing reforestation in the tropics can cost-effectively provide some 10-20.9% of the reductions in emissions between 2020-2030 required to meet the Paris 2°C warming goal (Busch et al., 2019).

6.3.3 Implications for forest-poverty dynamics

Climate change directly threatens the forest-reliant poor by destroying assets, impeding livelihoods and reducing ecosystem services (Hallegatte et al., 2015; IPCC, 2018). As forests degrade, those that depend on forests and trees for income and subsistence may have to travel further or even migrate to maintain their livelihoods. Air pollution, water contamination, psycho-social harm and visibility impairment from wild fires can seriously harm human health (Fowler, 2003). Floods, tropical storms, degraded landscapes and landslides in forest landscapes can also lead to loss of human life, livestock and dwellings (Das and Vincent, 2009; Samir, 2013). Even though we do not know where the exact impacts of climate change will be felt most since other factors also impact on where different tree species will be more or less abundant, drier and hotter areas will most likely have negative impacts on livelihoods (Seymour and Busch 2016, Olsson et al., 2019).

Increased drought and loss of evapotranspiration from forests pose threats to agriculture (Lawrence and Vandecar, 2015). Rain-dependent, small-scale farmers who are amongst the world's poorest are increasingly vulnerable to income loss and food insecurity (Damania *et al.*, 2017). Rainfed agriculture, accounting for up to 95% of cropped land in sub-Saharan Africa and 60% in South Asia and producing most of the world's staple grains (Wani et al., 2009), will likely be affected by increased drought (Cooper et al., 2009). Increasing temperatures, exacerbated by forest loss (Cohn et al., 2019), can also make outdoor labour more hazardous (Suter et al., 2019) and increase mortality (Mora et al., 2017).

Migration will likely be an important adaptation response to climate-induced extreme future events (Cattaneo et al., 2019). Migration, as a global trend, is difficult to estimate, but the number of international migrants appears to have fluctuated between 2.7 and 3.3 percent of the world population between 1950 and 2017 (de Haas et al., 2019). Data from 1970 onwards show a steady climb in the number of international migrants from 85 million people (2.3% of the global population) in 1970 to 272 million migrants (3.5% of the global population) in 2019 (International Organization for Migration, 2020). While climate change-induced migration is expected to increase in the future (Marchiori et al., 2012; Missirian and Schlenker, 2017), peoples' movements will vary depending on the speed of climate events (slow or rapid onset), other available adaptation opportunities, household access resources, etc. (Cattaneo et al., 2019). Notably, Rigaud et al., (2018), taking into account demographic, socio-economic and climate scenarios, estimated that there will likely be 143 million 'within country' climate migrants by 2050. There are, however, many uncertainties in projecting future migration, making this a rich area for further research (Cattaneo et al., 2019).

Improved forest management can also play an important role in climate change adaptation. Forest-based climate mitigation and adaptation (e.g. reforestation and restoration initiatives) can provide new income opportunities for forest-dependent communities (Ota et al., 2020). While tree planting projects can have negative impacts on livelihoods if they compete for land used by the poor (Seppälä et al., 2009; Smith et al., 2019), they can also help secure land tenure (Guillerme et al., 2011; Ota et al., 2020), reduce soil erosion (Korkanç 2014) and associated flooding (Yin and Li, 2001) and landslides (Pradhan et al., 2012), and potentially increase yields and reduce the risks associated with farming (Maas et al., 2013; Brown et al., 2018). The local (Prevedello et al., 2019) and regional (Cohn et al., 2019) heat-reducing effects of increased tree cover will be particularly important in tropical areas, which are expected to experience higher temperatures (Mora et al., 2017; Coffel et al., 2018). The choice of tree species and location of tree planting will have direct repercussions on poverty alleviation with, for example, some species more likely to provide income through the sale of non-timber forest products such as shea butter from shea nuts (see Chapter 3, Box 3.7). Chapter 5 discusses different levers such as REDD+ and *payments for ecosystem services* (PES) that can be used for improving forest management and conclude that the evidence on their poverty outcomes is somewhat unclear. That said, the body of research on PES is larger and suggests that these schemes are likely to reduce poverty slightly or be poverty neutral as they seem to cover the opportunity costs of conservation.

6.4 The Shifting Landscape of Forest Sector Financing

6.4.1 Introduction

Forest-related financing can be an important driver of poverty alleviation in forested landscapes. The forest sector is generally financed through budgetary allocations from domestic governments, international aid and, increasingly, private sources. The overall and relative amounts from different sources is changing with implications for *poverty reduction*.

6.4.2 Overseas development assistance

Overseas development assistance (ODA) for forestry generally supports forest protection and improvement, rural economic development and forest-related climate mitigation and adaptation (Environmental Defence Fund and Forest Trends, 2018). Building on previous work (Agrawal et al., 2013), we undertook a review of multi-lateral international forestry aid. Data for this analysis came from the Organisation for Economic Cooperation and Development (OECD), World Bank, Global Environment Facility (GEF), Asian Development Bank, African Development Bank and Inter-American Development Bank databases. Forest-related aid projects in these databases were identified using the following keywords in titles or descriptions: "forest," "agroforestry," "deforestation," or "tree". "Restoration" was not included because the associated projects extended well beyond forests, but forest restoration projects were otherwise included in the dataset. Data included forest-related climate financing (e.g. REDD+) to the extent that this was identified through the search terms. To eliminate duplicates, we removed projects with the same name, country and/or committed amount.

Our review suggests that nearly USD 7 billion in international and bilateral aid was allocated to

forest projects during 2014-2017. Average annual forest aid was roughly USD 1.7 billion, a reduction from USD 3.5 billion per year between 2000 and 2013. Not all data were available for 2018 and 2019, but trends at the World Bank and Inter-American Development Bank indicate that funding decreased during these years relative to 2017 by 77% and 53%, respectively. These figures suggest that international forestry aid may be showing a stagnant or declining trend. According to OECD, forest sector aid over the last five years represents 1% of official development aid across all sectors (USD 177.18 billion) (OECD, 2020).



Source: Authors' assessment based on data from: Organisation for Economic Cooperation and Development (OECD), World Bank, Global Environment Facility (GEF), Asian Development Bank, African Development Bank and Inter-American Development Bank

The geographic allocation of international forest aid was uneven during the period reviewed. Most funding was directed to countries in Asia (USD 3.1 billion), with almost half as much to the Americas (USD 1.4 billion), and about a third as much to Africa (USD 1 billion). Forest-related aid projects cover a range of approaches, including *afforestation/reforestation*, payments for ecosystem services, alternative livelihood provision for forest-proximate people, consolidation of national parks, sustainable forest management and agroforestry, among others.

6.4.3 Growth in Private Investments

While systematic data on private financing of the forestry sector is limited and available evidence suggests that private financing falls significantly short of estimated needs (Castren *et al.*, 2014), private forestry impact investments are steadily in-

creasing (Bass *et al.*, 2019; Ginn, 2020). Impact investments are carried out by companies or funds with the explicit goal of generating both private and social returns (Ginn, 2020). Impact investors include a range of investment firms, high networth individuals and ordinary, unaccredited individuals (e.g., through the Mosaic platform, people can invest as little as USD 25 in community solar projects (Global Impact Investing Network website; Rodin and Brandenburg, 2014)).

The Global Impact Investing Network (GIIN) estimates that over 1,720 organisations managed USD 715 billion in impact investment assets globally in 2019 (Hand et al., Sunderji, 2020), with 37 funds in Canada, Oceania and the US managing at least USD 9.4 billion directly in forestry and related assets (Bass et al., 2019). These funds represent investments that provide returns from forest products, including timber and forest carbon offsets, while also achieving environmental and social co-benefits (often pursued in part through the use of Forest Stewardship Council and Sustainable Forestry Initiative certifications). The majority of these funds focus on investments in Australia, Canada and the United States, though six of the 37 funds invest in projects in sub-Saharan Africa, Latin America and Southeast Asia (Bass et al., 2019). One standout example is the European based Livelihoods Carbon Funds (not included in the GIIN analysis above), which invested EUR 40 million in forest-reliant livelihood projects in the developing world and plans on increasing investment in a second phase to EUR 100 million (Livelihoods Carbon Funds, 2020). Accurately identifying how much of the investment is directed toward forests is, however, difficult, both because these investments have dual- or triple-bottom lines and may be categorised within larger, less specific categories like conservation projects (Hand et al., 2020), and because reliable data on foreign and domestic private investments in forestry are limited (Castren et al., 2014).

Impact investing adds some of its greatest value when financing enterprises that have few alternative sources of funding and are likely to have long lags before seeing returns (Brest and Burn, 2013). Financing of nascent forestry enterprises and locally-owned small and medium forest enterprises (SMFEs) (see Chapter 5) can generate employment and spread *prosperity* to local forest-dependent communities (Kozak, 2007; MacQueen, 2008; Sanchez *et al.*, 2018; Macqueen *et al.*, 2020). Financing is also mounting in the agroforestry space, with some investors (e.g., Technoserve, 2020) combining business advice with capital investments. Growth in impact investments is further fuelled through non-profit organisation-investor partnerships (for instance, the Nature Conservancy – Ginn, 2020) and its increasing recognition by multilateral organisations.

6.4.4 Carbon offsets financing through the voluntary market

Private sector interest in carbon offsets may be on the rise, offering some opportunities and potential pitfalls for addressing the forest-poverty relationship. The voluntary carbon offsets market typically brings private sector funding to Forest and other Land Use (FLU) projects related to afforestation/ reforestation, avoided deforestation (e.g. REDD+), landscape management and agroforestry.



Forest fires endanger rural livelihoods and well-being (Galicia, Spain) Photo © Nelson Grima

Figure 6.3



Source: Data drawn from multiple reports, Forest Trends, State of Voluntary Carbon Market reports (2007-2019)

In order to understand how voluntary carbon market investments have changed over time, we drew on annual data reported in each Forest Trends' State of the Voluntary Carbon Market report from when they began tracking markets in 2006 through 2019 (2006 data reported in the 2007 report, and 2017 and 2018 data reported in the 2019 report). Market value (in millions USD) and total transacted volume of offsets (metric tonnes CO_2 equivalent – MtCO₂e) were drawn from total values from the 2019 report (Forest Trends Ecosystems Marketplace, 2019) for all years from 2006 through 2018. Transacted volume of forest and land use (FLU) offsets (in $MtCO_2e$) were drawn from annual reports, which represent minimum estimates due to delays in reporting from surveyed respondents, while "other volume" was calculated as the remainder of total volume minus FLU volume. As such, the percentage of total offsets in each year from FLU presented in Figure 6.3 is likely an underestimate for each year.

As Figure 6.3 shows, the voluntary forest carbon market is still small (~ USD 300 million of financing for all offsets in 2018). However, FLU offsets increased by 264% between 2016 and 2018 (relative to a 21% growth in other offsets) (Forest Trends, 2020). The trend in FLU projects may reflect increased media reports on Natural Climate Solutions (Forest Trends Ecosystems Marketplace, 2019). Interest by several large corporations in natural climate solutions may further fuel this trend.

6.4.5 Implications for forest-poverty dynamics

Meticulous data-driven evaluations of the poverty outcomes of overseas forestry aid remain rare. A review of impact evaluation documents on the development impact website (DIME, 2020) of the World Bank, for instance, suggests that there are very few systematic evaluations of forest specific development aid. Public forestry interventions face the same challenges related to outcome attribution confronted by other development sectors. In addition, individual forest-focused public projects often have multiple objectives, making it difficult to identify poverty reduction outcomes that may be tied to one of many components. Large forestry programmes, with multiple investments over many years, can be reasonably assessed using a broad array of indicators and qualitative discussions, see for instance an evaluation of the Programme on Forests (Wells and Altman, 2011), but care needs to be taken to adequately address potential confounding factors (Ferraro, 2009).

An appraisal of the forestry portfolio of the largest donor in the sector, the World Bank, sheds some light on the poverty implications of forestry aid. Shyamsundar et al. (2020) show that the World Bank's forestry investments (worth over USD 1 billion and completed between 2002-2015) are mainly in middle-income countries, with low income countries accounting for only 10% of projects. Still, a majority of projects in the World Bank's forestry portfolio included poverty related components such as technical support and training to improve community forestry and/or smallholder plantations, support for nurseries and small-scale forest businesses, strengthening forest rights etc. (Shyamsundar et al., 2020). In general, the limited evidence on poverty outcomes of forestry aid points to the need for careful evaluations, an area for further research. That said, poverty alleviation through forest-related aid is unlikely to increase given the stagnancy in overseas forestry assistance identified in Section 6.4.2.

Private sector financing in the form of impact investments usually complements public or private non-profit financing (see examples in Ginn, 2020). Consideration of co-benefits, including pover-

ty reduction, by the carbon offsets market appears to be increasing. Over the 2016-2018 period, the volume of projects certified using a combination of Voluntary Carbon Standard (VCS) and Climate, Community and Biodiversity (CCB) offsets, both of which pay attention to social outcomes, increased by 325%, while total offsets only increased by 53%, indicating market preference for projects with social and biodiversity co-benefits (Forest Trends Ecosystems Marketplace, 2019). Currently 126 projects are under consideration with the CCB standard in *low- and middle-income countries* (Verra Registry Database, 2019).

Even though private sector forestry financing is steadily increasing, both public and private investments remain small relative to, for example, the financing required for large scale restoration (Castren et al., 2014). Growing global social movements and attention to Indigenous communities and local peoples (Sauls, 2020) may nudge forest-related private investments toward social outcomes. However, to prevent worst case scenarios of land grabbing and corporate 'greenwashing', increases in impact investments would need to be matched by an equal empowerment of local communities and monitoring of social outcomes. Furthermore, to ensure that the voluntary carbon offsets market meets social objectives, standards need to pay better attention to issues such as income predictability, transaction costs and meaningful local participation, especially for and among smallholders (De La Fuente and Hajjar, 2013; Melo et al., 2014).

6.5 Technological Change and Interconnectivity

6.5.1 Introduction

Pathways that link forests and livelihoods are moderated and mediated by material technology. These technologies affect forest-livelihood relationships by, for example, improving information available on forest resources, providing more accurate and scalable methods for forest monitoring and connecting forest proximate peoples.

6.5.2 The rise of geospatial data

Changes in bio-geophysical data availability, remote sensors and computational speed have improved the ability to monitor and study forest-livelihood relationships. Rapid advances over the past thirty years in computing and internet technology have contributed to publicly available, high-resolution earth observation data,

enabling reliable and replicable assessments of global land cover change (Loveland and Dwyer, 2012). Since 2008, a variety of products and new remote sensors have enhanced forest monitoring and evaluation, with high-resolution tree-cover maps produced from Landsat data beginning in 2000 (Hansen et al., 2013) being the most widely utilised. The Hansen et al. (2013) dataset has since been used to identify a variety of forest-related trends such as intact primary forest landscapes (Potapov et al., 2017) and drivers of tree-cover loss (Curtis et al., 2018). Public and private satellites that provide higher-resolution imagery, LIDAR technology piloted on unmanned aerial vehicles (UAVs), and the use of artificial intelligence (AI) to analyse data, compose the newest frontier of spatial data and analysis. Satellite imagery at resolutions greater than 30 m per pixel help identify fine spatial patterns and changes in forest structure (Steven et al., 2003; Kayitakire et al., 2006). LIDAR imagery enables three-dimensional analyses of canopy height and density, facilitating assessments of tree diversity and forest structure (Ferraz et al., 2016; Ganivet and Bloomberg, 2019). Real-time forest monitoring provides information on where and when forest cover change occurs, improving ability to track forest loss hotspots and forest fires (Davies et al., 2009; Wheeler et al., 2014; Hansen et al., 2016). Enhanced availability of spatial data also generates novel ways to monitor forest ecosystem services. For example, Global Forest Watch provides a variety of publicly available spatial data on forest cover to aid in the analysis of when, where and why forest landscapes change (Global Forest Watch, 2014). Combining data from different sensors, which enables the estimation of above-ground forest carbon (Asner et al., 2010; Le Toan et al., 2011), with spatial data on the extent of forest cover, can help identify a suite of carbon-related services and other ecosystem services (Martínez-Harms et al., 2016).

6.5.4 New technowlogies that affect monitoring and evaluation of forest products

Several new monitoring technologies improve the ability to monitor and regulate products provisioned by forest systems (Marvin et al., 2016). These technologies include acoustic sensors and UAVs equipped with alternate sensors and biomonitoring. Acoustic sensors record and process sound to provide information about forest systems and biodiversity. Combining spatial data with bioacoustics data can test the assumption that forest conservation and regeneration lead to biodiversity conservation. Standard usage of UAVs for conservation include the use of high-resolution imagery and image classification to spot and count wildlife, map land cover and promote real-time monitoring of protected areas (Wich and Koh, 2018; Iacona et al., 2019). UAVs are also able to detect poachers and poaching in real-time (Kamminga et al., 2018). Forensic science uses visual, chemical and genetic techniques to determine the origin of a wood samples (Dormontt et al., 2015). Combining rapidly advancing forensic technology to ascertain the origin of wood samples promises to reduce illegal logging and strengthen legally sourced supply chains (Tnah et al., 2010; Sasaki et al., 2016). Box 6.2 provides an example of how forest communities in Guatemala have been able to use UAVs to serve their needs.



Tree-based systems can help rural people meet their daily needs (Viñales Valley, Cuba) Photo © Daniel C. Miller

Box **6.2**

Global norms and technology have shaped community forest management in Guatemala

The case of the Association of Forest Communities of Petén (ACOFOP) in northern Guatemala highlights the role of technology in increasing community decision-making over forest resources. Since 1995, ACOFOP has promoted a community forest concessions model that allows local communities in and around the Maya Biosphere Reserve the right to sustainably harvest timber and non-timber forest products like the xate palm, three species of the Chamaedorea genus and ramón (Brosimum alicastrum) nut. Through strengthened local governance, collaboration with international and local non-governmental organisations, community-led technical support and consistent funding over the start-up period, ACOFOP's member communities have achieved success in reducing deforestation, virtually stopping forest fires and increasing community well-being, through individual income and the expansion of community health and education services (Stoian et al., 2018).

Two technical factors contribute to ACOFOP's success: i) enhanced affordability and accessibility of geospatial data, software and emerging technologies; and ii) the rise of hand-held

Increasingly, private-sector actors are also using enhanced technology to provide goods and services that will affect forest-poverty pathways. In 2017, Microsoft began a USD 50 million project to spur innovation and application of AI for the monitoring, modelling and management of environmental resources (Joppa, 2017). This project partners with organisations to use AI to better interpret and analyze data on ecological systems.

6.5.5 Social media, networks and the rise in user-group connectivity

Mobile phone accessibility can enhance access to market information, facilitate peer-to-peer learning, and ease logistics planning for smallholders. However, the impact of these innovations for forest-dependent communities is unclear (Duncombe, 2016), with mobile phones, thus far, playing a limited role in directly contributing to increased income (Sife *et al.*, 2010; Baird and Hartter, 2017). communication technologies, particularly mobile phones with access to social media (Sauls, 2020). Since 2016, the international NGO Rainforest Foundation has trained community forest stewards in the use of UAVs and opensource GIS technologies to support their efforts to monitor forest fires and deforestation along their concession borders. With these tools in hand, in addition to now easy-to-access data from sources like NASA's Fire Information for Resource Management System, ACOFOP has been able to respond quickly to fire threats. Increased mobile phone ownership and broad access to social media also play a role in turning these successes into narratives that can reach a broad public, helping ACOFOP gain allies nationally and internationally. These allies in turn play an important role in advocating for the ACOFOP model. The two technology trends are increasingly important for ACOFOP's current struggle to renew community concessions, which will expire over the 2021-2025 period. As of January 2020, the first community organisation to receive its concession, Carmelita, had just received approval of its request for renewal.

That said, mobile phones as a vehicle for access to social media show much promise. Social media engagement enables knowledge exchange, network building and political claims-making for communities whose livelihoods depend significantly on forest resources (see Box 6.3 on the use of social media by Guardians of the Forest Alliance). It also serves as an additional way to interface with official accounts about forest resources and governance. Indigenous and forest community groups connect through applications such as Facebook and Twitter in order to develop alliances in favour of community forest rights, pursue 'boomerang effects' (galvanising international attention and support) that pressure national governments to respect or support local management of forests and to share news about specific phenomena (Keck and Sikkink, 1998; Sauls, 2020). Mobile phones and more accessible photovideo technology also enable forest communities to capture and share their own narratives (e.g. see "If not us, then who", 2020) (Mitchell-Walthour, 2020). Members who engage in international trainings or exchanges use social media to maintain networks, which serve to disseminate best practices around sustainable forestry and effective advocacy (Bebbington *et al.*, 2018a; Sauls, 2020).

<u>Box 6.3</u>

Indigenous and local communities and the boomerang effect of social media

In 2019, members of the Guardians of the Forest Alliance (composed of Indigenous and forest community groups from the Amazon, Central America and Mexico, the Congo Basin and Indonesia) used social media (primarily Facebook) to share news ranging from the case of a municipal (non-Indigenous) government chopping down peri-urban trees in the Honduran Muskitia to first-person alerts about major forest fires in the Brazilian Amazon. Groups from disparate regions shared and re-Tweeted news using tags such as "#landrightsnow", "#GuardiansoftheForest", "#AmazonFire" and versions in local languages, amplifying the voices of specific forest peoples and the Alliance's common cause of enhancing Indigenous and community forest rights, especially in the face of climate change. The interconnection between Indigenous groups, bolstered through social media connections, has enabled a consolidation of core discourses and claims, creating a cohesive agenda on Indigenous forest rights at the international level, even while local conditions vary greatly (Sauls, 2020). For forest groups, social media provides an explicit tool for 'boomerang effect' politics as well as a platform for mutual support, advice and visibility (Virtanen, 2015).

6.5.6 Implications for forest poverty dynamics

Spatial data provides a way to identify and monitor forest resources that is replicable, reliable and low-cost. In addition, use of spatial data can be helpful in designing mechanisms such as payments for ecosystem services (Mitchell *et al.*, 2017; Curtis *et al.*, 2018; World Bank Ecofys, 2018), which may bring forest-reliant communities additional financing, strengthen the argument for community-based forest management (Blackman *et al.*, 2017; Santika *et al.*, 2019) and facilitate the mapping and monitoring of Indigenous territories, thereby reducing risks from external threats (Paneque-Gálvez *et al.*, 2017). In general, however, enhanced availability of material technologies improves the detection of forest cover change, but does not provide immediate solutions to address the complexities of forest-related policies, land tenure, monitoring and enforcement challenges (Gaveau *et al.*, 2017; Erbaugh and Nurrochmat, 2019).

Through advances in acoustic sensors, data from UAVs and forensic timber science promise to improve forest management, use and regulation. Such technologies provide new opportunities for employment for technologically literate forest proximate people, advance opportunities for research on forest areas, strengthen sustainable supply chains and can contribute to community-based management or co-management of forest areas (Bellfield *et al.*, 2015; Marvin *et al.*, 2016; Iacona *et al.*, 2019). They may also help curtail human-wildlife conflict, a major cause of crop raiding, injury and even death in many tropical forests (Nyhus, 2016).

Improved technologies commonly reinforce regulations that limit extra-legal benefits enjoyed by forest proximate people, with governments tracking illegal expansion of commodity agriculture or of local populations into state forests (Musinsky *et al.*, 2018). Thus, technology-assisted monitoring can reduce benefits to forest proximate people who do not hold secure forest management or access rights.

Advanced technology generally remains the purview of experts (states, researchers and NGOs). This exclusivity means that many communities rarely use it in their own interests. Thus, unless capacity building is included in rolling out such technologies, the ability of rural communities to use these tools to support and enhance their own livelihoods may be limited. New Apps such as TIMBY (This Is My Backyard) that allow communities to monitor illegal logging (The Goldman Environmental Prize) and technology-wielding associations such as ACOFOP in Guatemala (Box 6.2) provide counter examples. Even in the case of ACOFOP, however, the deployment of LIDAR, contributing to the 'discovery' of additional Mayan ruins, has catalysed new interest in displacing forest communities (Devine, 2018).

Social media connects forest proximate people and appears to be increasingly important as a tool for sharing information, spurring innovation and garnering attention. Democratising the use of material technology and the interpretation of data is a critical next step in empowering communities to manage forest resources and directly alleviate poverty.

6.6 Global Socio-Political Movements

6.6.1 Introduction

Climate and biodiversity loss have transitioned from topics primarily discussed by biophysical scientists to issues of widespread public concern backed by international efforts to synthesise scientific evidence (Díaz *et al.*, 2018; IPBES, 2019; IPCC, 2019). A range of global social movements, including the spread of protests against *inequality* and racism and in support of Indigenous peoples' rights and climate change action, and countervailing forces, including anti-environmental populism, shape contemporary politics around forests and the forest-reliant poor.

6.6.2 Changing political and civic landscapes

The last few years have seen changes in global discourses, as well as political priorities relating to forests and the environment. In several countries, governments are rejecting climate mitigation policies and other associated environmental priorities that require leaders to use short-term political capital for long-term economic and environmental gains. The United States' withdrawal from the 2015 Paris Agreement exemplifies this trend; however, it finds resonance in the re-positioning of several other governments, including countries such as Brazil, with significant forests upon which minority people rely (Fearnside, 2018; McCarthy, 2019).

The changing political landscape portends potential conflicts between priorities at national and international scales, particularly real and perceived trade-offs between conservation and forest-based economic development. It has brought renewed attention to frontier resources and the opportunities they offer (Rasmussen and Lund, 2018) contributing, in some cases, to a call for reducing environmental and social protections for the peoples and ecosystems in frontiers (de la Vega-Leinert and Schönenberg, 2020). An ongoing, and potentially deepening, commitment to extractivism as the basis of development, especially in contexts of resurgent populism on both the right and left, may pose a renewed threat to forests and their inhabitants as political shifts erode human and environmental rights (Bebbington et al., 2018a; 2018b; 2018c).

The contemporary moment also features a major counter-current to national anti-environmental political shifts. Public awareness of the threat posed to human well-being from environmental change and the nexus between unjust practices and governance structures, and unsustainable practices is at an all-time high (Lee et al., 2015; Fagan and Huang, 2019). The climate youth movement, for example, has become increasingly popular and inspired by youth activists such as Greta Thunberg, Autumn Peltier, Bruno Rodriguez, and many others. In September of 2019 alone, there were over 2,500 events scheduled in over 150 countries to sound the alarm about the climate crisis (Tollefson, 2019). Other movements such as the Extinction Rebellion call for nonviolent civil disobedience to compel governments to act before biodiversity loss and rising temperatures reach a tipping point. Forests - and other 'natural climate solutions' (Griscom et al., 2017) - are central to the demands made by these movements, which are occurring simultaneously with other mass movements demanding political accountability (e.g. Brazil, Chile, France, Hong Kong and the US) and the fusing of environmental and social justice concerns (e.g. Green New Deal in the US or the European Green Deal) (Wright, 2019).

6.6.3 Indigenous rights and social justice movements

Since the 1980s, Indigenous Peoples and forest communities' historic rights over global forest lands have slowly gained recognition. Indigenous Peoples and local communities are estimated to have legal rights to over 15.3% of forestland in the 58 most forested countries in the world (Rights and Resources Initiative, 2018), although the actual figure is likely to be much higher. The growing recognition of Indigenous and community rights over forests reflects shifts in development and environmental conservation theory, as well as self-identified and well organised forest-dependent communities staking their ancestral claims to land and resources. When faced with local protest while implementing reforms to decentralise forest governance and regularise landholding, major development donors, such as the World Bank, increasingly supported formalised, collective land rights arrangements (Jackson and Warren, 2005; Bryan, 2012; Anthias and Radcliffe, 2015). Since the 1990s, the failure of exclusionary models to achieve biodiversity and forest conservation has also led to the inclusion of Indigenous and local communities via extractive reserves, Indigenous and Afro-descendent territories, and



Deforestation can affect the livelihoods of millions of people. Two men are preparing lunch in the forests of Lao PDR. Photo © Nelson Grima

co-management arrangements. Whether rights to forests support ecological and social well-being depends partially on the institutions and practices of Indigenous Peoples and local communities, especially as they interact with dominant economic forces and external institutions (Robinson *et al.* 2014; Bebbington *et al.*, 2018a). It also depends on the degree to which governments continue to respect and uphold the rights of forest groups or use force, increasingly prevalent, to suppress civic action to achieve social and environmental *equity* (Middeldorp and Le Billon, 2019; Scheidel *et al.*, 2020).

In the middle of the COVID-19 pandemic, 2020 has also seen a reckoning related to discrimination, especially anti-Black racism, and injustice based on racial, ethnic, and religious identity in many countries around the world. Building from previous protest waves in the UK, South Africa, Brazil, and the US in particular, this emerging international movement has raised awareness of institutionalised racism across many countries and sectors, including the environment (Finney, 2014; Knudsen and Andersen, 2019; Miles, 2019). While these protests have, on the one hand, highlighted how marginalised groups are harmed by policies of the state, they have also underscored how excluding diverse voices in professionalised forestry, conservation and development organisations may

lead to an undervaluation of the lived experiences of minority groups and their experiences in nature (Finney, 2014; Kloek *et al.*, 2017; Hays, 2019). The current movement is already spurring reflection on how forest conservation and management might become more inclusive, including by directly grappling with legacies of colonialism and dispossession that have disproportionately affected Indigenous people and people of colour (Mollett and Kepe, 2018).

6.6.4 Implications for forest-poverty dynamics

New environmental movements, including youth protests, are pushing governments toward action on climate change, often rallying around visible threats like forest fires. These efforts, reinforced by global dissatisfaction with increasing inequality, and layered onto ongoing Indigenous rights' movements, often view social and environmental justice as paired goals. This combined set of priorities could substantively address poverty in forested areas; however, whether attention translates into action – and whether actions to address climate change and forest loss are inclusive of the needs of forest-proximate and dependent communities – depends on broader political conditions. The concurrent rise of governments that are promoting extractivism as a mode of economic development, exemplified in 2020 by Brazil and the US, provides a direct challenge to environmental movements. The anti-environmental perspectives held by these governments, and other business leaders, may sacrifice forests and the environment for economic growth, but with potentially limited poverty alleviation impacts.

Many of the world's Indigenous communities depend on forests for their livelihoods, making land rights and secure access to forests a priority for these groups (Rights and Resources Initiative, 2018). Additional major challenges to the social well-being of forest-reliant peoples relate directly to political conditions, including extractivism, the roll back of social protection, illicit activities (such as illegal mining, poaching and narco-trafficking) and corruption across scales (Devine, 2014; McSweeney et al., 2014; Yagoub, 2014; Tollefson, 2016). While the causal link between Indigenous and community forest rights and poverty alleviation is mixed and systematic data are lacking, case study-based evidence (identified in previous chapters) suggests that income and community-provided social services increase with greater control of forest resources (Bocci et al., 2018). The many institutional and contextual factors that can influence community-forest relations, as well as the range of land rights typologies, definitions of poverty and methodologies for assessment, make comparative analyses difficult (Fisher et al., 2008; Hajjar et al., 2016; Cheng et al., 2019).

How the world's forests and tree-based land uses serve as a sustainable means for people to move out of poverty will at least partly be shaped by interactions between, and the combined consequences of, global anti- and pro-environmental forces. Emerging social trends against racism and inequality may also lead to a more pro-poor environmentalism, which could bring additional support to Indigenous communities and local peoples.

6.7 The Spread of Infectious Diseases

6.7.1 Introduction

Infectious diseases are an important cause of global morbidity and mortality, responsible for some 10 million deaths or 1/5th of all deaths worldwide in 2016 (Hay *et al.*, 2017). The past two decades have seen a rise in emerging infectious diseases (EIDs), such as Ebola, SARS, MERS, the novel Coronavirus (COVID-19) and others, which is likely to continue (Centers for Disease Control, 2020; Allen *et al.*, 2017). Some 70% of EIDs originate from interactions among wild and/or domestic animals and humans (Morse *et al.*, 2012).

6.7.2 Zoonoses and anthropogenic changes

Research over the past several decades has documented the importance of forest loss and increasing forest edge for established vector-borne diseases such as dengue and malaria (Husnina *et al.*, 2019; MacDonald and Mordecai, 2019; Chaves *et al.*, 2020). However, zoonoses, or diseases that spread from vertebrate animals to humans (WHO, 2020a), have received relatively less widespread global attention until COVID-19 (Di Marco *et al.*, 2020).

Anthropogenic changes, including deforestation and expansion of agricultural land that increase contact between humans and wildlife, intensification of livestock production near wildlife areas, and increases in hunting and trading of wildlife all contribute to zoonoses (Allen et al., 2017; Dobson et al., 2020). Deforestation and biodiversity disruption can create new breeding habitats for disease vectors by changing the ecological conditions that regulate predator-prey relationships and make wildlife more vulnerable to disease (Pongsiri et al., 2009; Keesing et al., 2010). Climatic changes, such as increases in temperature and changes in precipitation patterns in forested areas, can also change the geographic range, population density, prevalence of infection by zoonotic pathogens and the pathogen load in individual hosts and vectors (Mills et al., 2010). Trade in wildlife and wet markets contribute to zoonoses by increasing contact between animals and humans (Bell et al., 2004; Wolfe et al., 2005; Dobson et al., 2020). This may have been the case with the COVID-19 virus (Sohrabi et al., 2020), though there are some prevailing uncertainties (Zhang and Holmes, 2020), particularly because the emergence and spread of zoonoses can take different complex pathways (Epstein et al., 2006; Altizer et al., 2011).

6.7.3 Implications for forest-poverty dynamics

The reduction of infectious disease is directly tied to SDG 3 on health, which aims to "ensure healthy lives and promote wellbeing for all at all ages." Health pandemics also bring enormous economic disruptions, eroding capacity to meet SDG 1.

The Ebola outbreak (which had killed more than 11,000 people in West Africa by 2016 (Centers for Disease Control, Ebola, 2019) may have contributed to a 12% reduction in the combined GDP of Guinea, Liberia and Sierra Leone relative to pre-Ebola expectations, and changes in economic transactions across several other countries in Africa (World Bank, 2015). COVID-19 as of 5 October 2020 had affected over 35 million people, resulting in over one million deaths worldwide (WHO, 2020b). Varying estimates suggest that the global economy may contract by 3% (I.M.F., 2020) to 5% (World Bank, 2020) in 2020. Assuming a 5% contraction of the global economy, rural populations in extreme poverty are expected to increase by 15% globally (Laborde *et al.*, 2020). Notably, in many parts of the world, COVID-19 is occurring where the background rate of malaria, dengue and other infectious diseases already take a huge health toll (Lorenz *et al.*, 2020; Saavedra-Velasco *et al.*, 2020).

The pathways through which the welfare effects of COVID-19 will manifest are identified in

Figure 6.4, with implications for many other EIDs. For the forest-reliant poor, changes in labour and non-labour income are likely, especially if their work is tied to forest-related tourism or disrupted global forestry supply chains. The implications of the complete stoppage of ecotourism, identified as a critical lever for poverty reduction in Chapter 5, is particularly dire. Household consumption can decline as dependents increase, with urban members returning, laws against bushmeat hunting are strengthened and, public services drop (CIFOR, 2020; Shyamsundar, 2020; World Bank, 2020). In remote Indigenous territories, EIDs can pose a serious existential threat because of limited access to immediate health care and ability to reduce spread, once exposed (Conde, 2020; Taylor, 2020).



Source: Adapted from Fig. 1 in World Bank (2020): Poverty and Distributional Impacts of COVID-19: Potential Channels of Impact and Mitigating Policies and TNC Internal Brief on Rural Communities (2020)

There are potential positive feedback loops between health shocks and rural poverty (Rohr et al., 2019). Forests tend to act as a safety net when rural communities face covariate shocks (Wunder et al., 2014). Thus, pandemic-related economic shocks may lead rural communities to increase their extraction from forests, contributing to deforestation and degradation, with additional indirect negative effects on household welfare (Shyamsundar, 2020). Initial findings from Madagascar, for instance, note an increase in fires and forest clearing near protected areas as people who normally relied on tourism income prepare to invest more in agriculture (Eklund et al., 2020). In addition, macro-policy responses to the economic contractions resulting from COVID-19 may lead to a reduction in ODA, including funding for forests, and incentivise governments to loosen regulations around forest protection, potentially undermining subsistence and forest-based income and forest rights (Bebbington et al., 2018b; Gonzales, 2020; Vila Benites and Bebbington, 2020).

Policies to deal with risks associated with zoonoses have so far been largely reactive, focusing on disease investigation and vaccine development. Moving forward, options for public investments such as the OneHealth approach, which seek to integrate ecological and human health considerations, offer an integrated model to tackle zoonoses (Centers for Disease Control, 2020; Di Marco et al., 2020). Given the enormous costs and welfare implications of the COVID-19 pandemic, it would be cost-effective to invest in 'preventive' policies (Dobson et al., 2020). These may include a range of strategies: reduction in forest fragmentation and livestock and agricultural production in proximity to wildlife, increase in forest buffer areas, rural health clinics, and wildlife trade restrictions, improved wildlife and livestock disease surveillance etc. (Bloomfield et al., 2020; Di Marco et al., 2020; Dobson et al., 2020). To the extent that the forest-reliant poor may be involved in the pathways leading to zoonotic epidemics, public investments that strengthen food supply chains, provide alternatives to illegal wildlife use and trade, and reduce unmanaged encroachment of natural areas may offer triple win opportunities.

6.8 Conclusions: Global Forces of Change and Poverty Alleviation

This chapter discussed six major global forces that are likely to influence forest-poverty dynamics. While these forces are not entirely new, they are growing in significance. It is likely that their impact on the forest-reliant poor will increase because of the precarity and *vulnerability* of the poor. However, these global forces also present opportunities for poverty alleviation. Changes in forest cover and global and local responses to these changes will influence how forest-poverty dynamics are affected. These considerations are discussed below and illustrated in Figure 6.5 (see also Table 6.2).

As discussed in Chapter 2, forests play an important role in poverty alleviation. They can help maintain peoples' well-being by supporting subsistence needs, act as a safety net by helping to reduce risk by smoothing consumption, and can be a source of prosperity if people can add value to forest resources. Forests can also have negative impacts on well-being, such as through land use and wildlife conflicts that can push people further into poverty. As illustrated in Figure 6.5, the simultaneous occurrence of multiple global forces has implications for forest-poverty dynamics. For example, as previously discussed, climate change and EIDs pose health and economic risks to the forest-reliant poor that may together push larger numbers of households into transient or extreme poverty. At the same time, some of these global forces may prove to be powerful counter-measures. For example, public finance can help people move out of poverty (e.g. through capacity building or access to credit) and support well-being by strengthening ecosystem services (e.g. reforesting watersheds to improve water quality and reduce flooding). Private financing can increase cash income flows (through payments for carbon, for example), enabling smallholders to build assets to move out of poverty and better adapt to climate change. Advanced technology, wielded well, can clarify rights, reduce land conflicts and help poor communities access markets. Interconnectivity is also a powerful tool for social networking, helping Indigenous communities, for instance, to maintain their traditional uses of forests. However, rapidly emerging technologies can also increase poverty in cases where technical forest monitoring reduces subsistence forest uses.

The futuristic review undertaken in this chapter suggests that global changes offer both opportunities and risks to the forest-reliant poor. However, as Figure 6.6 shows, in a 'business as usual' scenario, the multiple risks posed by global changes to the forest-reliant poor may well overshadow any opportunities for poverty alleviation. This is largely because poor households have limited capacity or resources to take advantage of new opportunities. Nevertheless, strategies exist to reduce risks and improve conditions that would allow the poor households to be pushed out of poverty by these



Note: ES = Ecosystem services, EIDs = Emerging infectious diseases

global forces (see Table 6.2). Cross-sectoral strategies, such as OneHealth, that transcend the silos of health, biodiversity conservation and poverty alleviation, for instance, may mitigate risks and lead to alternative models of development for forest landscapes.

Figure 6.6. identifies a potential 'improved' future scenario for the forest-reliant poor with measures undertaken to reduce global risks and strengthen capacity to manage risks and opportunities. Specific measures may include:

- financing of commodity supply chain reforms (strengthening transparency, training, skills and resources for smallholders to access global value chains);
- technologies that work for the poor (including those that enable monitoring of investments and commodity flows);

- strengthened land rights, particularly of Indigenous peoples;
- OneHealth actions (buffer areas between agriculture and livestock production and forests, wildstock and human disease surveillance, alternatives to wildlife trade);
- investments in climate adaptation that reduce exposure to natural disasters and stabilise ecosystem services; and
- global social movements that boost the voice of forest-reliant peoples.

Many of these strategies can work in concert with, and build the enabling environment for promising levers of change, such as community forest management, ecotourism, agroforestry, and small and medium forest enterprises, discussed in Chapter 5.



Many of the global changes discussed in this chapter act as shocks to households - they manifest as negative health impacts, land losses, landuse conflicts, loss of resource access and political support, among others. However, they may also open up new opportunities, contributing to income and employment and enhancing remote people's ability to connect to the rest of the world. Empowering the forest-reliant poor with strengthened self-governance and technical skills represents an opportunity. Democratising the use of technology and data interpretation will further help to empower forest communities. Capacity building and support contribute to facilitating the use of these technologies and to build the resilience of the poor in the face of climate change impacts. The role of private financing in forest management and its increasing recognition of the importance of social

and environmental outcomes can also be a welcome development.

There are large gaps in knowledge related to both global trends and forest-poverty dynamics. The written literature on varying effects of climate change on the forest proximate poor or of market supply chains on human welfare, for instance, is limited. Thus, our analyses related to global changes depend, in part, on historical evidence or conceptual theories of change. Furthermore, the analysis in this chapter does not address meta trends such as urbanisation or economic globalisation that have indirect, but potentially large, though uncertain, effects on the forest-reliant poor. It is also unable to do justice to the uncertainties related to available projections in future global changes. For instance, while at least one major study projects the likelihood of 143 million 'internal' climate

migrants by 2050 (Rigaud *et al.*, 2018), many uncertainties regarding migration projections prevail (Cattaneo *et al.*, 2019). Climate-induced local and global migration are rich areas for future research.

The net effects of global forces of change on forest-poverty dynamics will vary across local contexts based on the enabling conditions discussed in Chapter 4, and the effects of global forces on the levers identified in Chapter 5 warrant further study. Research into measures across sectors that account for the combined strength of these (and other) global forces may serve to improve outcomes on forest-poverty dynamics and lead to alternative models of development for forest landscapes.



Agroforestry systems provide a wide range of direct and indirect ecological and socioeconomic benefits Photo © Daniel C. Miller

					Table 6.2
		Global force	s: First order effects		
GLOBAL FORCES	FIRST ORDER DIRECT EFFI	ECTS			
	Cash Income	Contributions to Subsistence Income	Ecosystem Services	Non-material Well-being	Critical Insights
Commodity growth	 + Increased incomes for farmers + New employment opportunities for rural labourers + Increase in income, capital accumulation, and higher expenditures on food, health, education, and durable consumer goods in smallholder farm households + Benefit non-farm households through gains in employment and income + Supports formal sector jobs outside of agriculture - Wage income may decrease if outsiders migrate into region - Land speculation may make it costly for smallholders to acquire land assets 	 Land grabbing and land use conflicts can reduce subsistence income Increased yields and income may occur through agroforestry 	 Deforestation associated with commodity markets may reduce ecosystem services (e.g. clean water), increase disease vectors (e.g. malaria) + Deforestation may also reduce some negative externalities (e.g. human-wildlife conflicts) 	 Land use conflicts may reduce well-being, particularly of Indigenous communities 	 Governance and security of tenure critical for reducing land grabbing and deforestation Commodity markets have multiplier economic effects that go beyond local communities; labour movements may suppress wages in local areas Market and price risks prevail Characteristics of smallholders (small scale, limited knowledge and networks, power imbalances when partnering with large corporations, lack of access to credit etc.) may make it difficult for poor to gain from commodity markets
Climate change	 Assets lost to natural disasters could limit earning potential +/- Unclear how shifts in tree composition/range may change abundance of species of value 	 Changes in forest composition may increase time costs of forest product collection 	 Loss of forests from phenologi- cal changes, natural disasters, and pest outbreaks would reduce the availability and quality of ecosystem services 	 Increased disease and insect outbreaks may expose people to zoonotic diseases 	The impact of changes in range and composition may or may not be positive based on wether forest biodiversity of value becomes more or less abundant

 Increased disturbance through natural disasters will only have negative or neutral effects on the poor Forest loss and degradation from climate change will reduce the availability and quality of ecosystem services, which has negative impacts for the forest-dependent With careful design that recognises local rights and implementation with appropriate safeguards, forest-related climate mitigation strategies can be pro-poor 	 COVID-19 and the economic downturn may reduce financing for the forestry sector As environmental organisations and movements gain more international traction, there is pressure to invest in Indigenous communities' ownership and access to land and their poverty reduction
	 + Forestry projects that build capacity and human capital often focus on improving management of community forest and forest access. + Some impact investment firms combine financing with business and technical training
	 + Public investments in forest management will likely enhance ecosystem services +/- Private investments in forest management may improve some services, but may also negatively affect biodiversity (and associated ecosystem services) depending on what types of trees and activities are supported
 Increased climatic shocks from wildfires, floods, and other natural disasters may reduce the amount and quality of forest products 	 Projects that seek to reforest and reduce deforestation may reduce subsistence uses if forests become more strictly managed for conservation or carbon
 Exposure to insect and pest spread in forest can have negative consequences for crops adjacent to forest new economic opportunities linked to REDD+ 	 + Pro-poor private sector investments will likely increase cash income + Some public investments in forestry couple poverty reduction goals with income generating projects (e.g. small-scale forest businesses and nurseries) + Public programmes such as REDD+ are designed specifically to increase incomes to conserve forests + Voluntary carbon offset markets may grow to provide more cash income
	ublic and private inancing for forests

GLOBAL FORCES	FIRST ORDER DIRECT EFFE	icTS			
	Cash Income	Contributions to Subsistence Income	Ecosystem Services	Non-material Well-being	Critical Insights
Technology and Interconnectivity	 + Improvements in monitoring accuracy and availability improve monitoring for forestry and NTFPs + Technological improvements create new opportunities to connect buyers/sellers of forest products - Improvements in monitoring reduce informal/illegal cash income 	 Improvements in monitoring may reduce forest manage- ment, use and access where rights are contested 	+ More consistent and temporally extensive monitoring/data enable better characterisation of ecosystem services and their losses	 + Greater access to information enhances decision-making and autonomy + Connectivity and social media strengthens rights, draws attention to injustice - Increasingly easy surveillance can challenge local rights and decision-making 	 Who has access and the ability to use new forest management technology determines whether it is used to surveil or empower forest proximate people Indigenous people and local communities often rely on experts to access technological advances Connectivity through social media and virtual marketplaces can generate livelihood benefits Technological capacity building coupled with rights for forest management promise long-term benefits for poverty reduction
Global Social and Political Movements	 +/- Environmental movements pushing for forest/climate action could enhance or decrease forest-based income +/- Indigenous and community land rights impact on cash income depend on model employed (e.g. community forest concessions/REDD+ may be positive, while Indigenous territories rights may be neutral) 	 Populist, extractivist political conditions push the agricultural frontier further into forests, reducing access to subsistence goods and services Stronger indigenous forest rights should lead to greater ease of access to subsistence zones and products 	 Populist, extractivist political conditions push the agricultural frontier further into forests, reducing ecosystem services (e.g. provisioning, regulating) + Greater environmental protections from new environmental movements lead to climate action that preserves vital services 	 + New social/land rights enhance autonomy, self-determination and protect Indigenous socio-cultural systems Populist, nationalist movements undermine previous forest rights and social gains, destabilising marginalised groups' access to resources and rights 	 Because these are political trends, their impacts will vary based on specific local conditions Securing Indigenous territorial rights may or may not lead to greater integration into markets, depending on other factors

 Poverty impacts more likely in urban areas relative to rural areas Indirect effects on rural communities through reductions in consumption (as a result of returning migrants, reduction in remittances, supply chain and public service bottlenecks) Direct losses in income and employment in some critical forest dependent sectors such as tourism The role of macro-economic and political forces may mediate some of these effects
 Emotional, psychological, physical well-being reduced due to disease burden and stresses associated with economic woes
 Where cash income is lost and communities must resort to forest services for survival, additional stress may reduce ecosystem health
 Morbidity/mortality effects + Forest products and services may serve as a safety net for communities who have lost formal employment and cash income
 Economic lock down leads to urban-to-rural migration, loss of employment and cash income for millions Forest-related tourism sector specific employment and income losses are likely to be significant Morbidity/mortality effects
Emerging Infectious Diseases

6.9 References

Abatzoglou, J.T. and Williams, A.P., 2016. Impact of anthropogenic climate change on wildfire across western US forests. PNAS 113(42): 11770-11775.

Abdullah, R., Ismail, A. and Rahman, A.K., 2011. Labour requirements in the Malaysian oil palm industry in 2010. Oil Palm Industry Economic Journal 11(2): 1-12.

Agrawal, A., Cashore, B., Hardin, R., Shepherd, G., Benson, C. and Miller, D., 2013. Economic contributions of forests. Background Paper, Prep United Nations Forum For Tenth Sess 8–19 April 2013. New York: UNFF.

Ainembabazi, J.H. and Angelsen, A., 2014. Do commercial forest plantations reduce pressure on natural forests? Evidence from forest policy reforms in Uganda. Forest Policy and Economics 40: 48-56.

Alexandratos, N. and Bruinsma, J., 2012. World agriculture towards 2030/2050: the 2012 revision. Available at: https://ageconsearch.umn.edu/record/288998/.

 Allen, T., Murray, K.A., Zambrana-Torrelio, C., Morse, S.S., Rondinini, C., Di Marco, M., Breit, N., et al., P., 2017.
 Global hotspots and correlates of emerging zoonotic diseases. Nature communications 8(1): 1-10.

Alongi, D.M., 2015. The impact of climate change on mangrove forests. *Current Climate Change Reports* 1(1): 30-39.

Altizer, S., Bartel, R. and Han, B.A., 2011. Animal migration and infectious disease risk. *Science* 331(6015): 296-302.

Anthias, P. and Radcliffe, S.A., 2015. The ethnoenvironmental fix and its limits: Indigenous land titling and the production of not-quite-neoliberal natures in Bolivia. *Geoforum* 64: 257-269.

Armenteras, D., Rodriguez, N. and Retana, J., 2013. Landscape dynamics in northwestern Amazonia: an assessment of pastures, fire and illicit crops as drivers of tropical deforestation. *PLoS One* 8(1): e54310.

Austin, K.G., Schwantes, A., Gu, Y. and Kasibhatla, P.S., 2019. What causes deforestation in Indonesia?. Environmental Research Letters 14(2): 024007.

Bacmeister, J.T., Reed, K.A., Hannay, C., Lawrence, P., Bates, S., Truesdale, J.E., Rosenbloom, N. and Levy, M., 2018. Projected changes in tropical cyclone activity under future warming scenarios using a highresolution climate model. *Climatic Change* 146(3-4): 547-560.

Badini, O.S., Hajjar, R. and Kozak, R., 2018. Critical success factors for small and medium forest enterprises: A review. Forest Policy and Economics 94: 35-45.

Baird, T.D. and Hartter, J., 2017. Livelihood diversification, mobile phones and information diversity in Northern Tanzania. Land Use Policy 67: 460-471. Bass, R., Murphy, P. and Dithrich, H., 2019. Scaling Impact Investing in Forestry. New York: GIIN.

Bebbington, A.J., Sauls, L.A., Rosa, H., Fash, B. and Bebbington, D.H., 2018a. Conflicts over extractivist policy and the forest frontier in Central America. European Review of Latin American and Caribbean Studies/Revista Europea de Estudios Latinoamericanos y del Caribe (106): 103-132.

Bebbington, A.J., Bebbington, D.H., Sauls, L.A., Rogan, J., Agrawal, S., Gamboa, C., Imhof, A., et al., 2018b. Resource extraction and infrastructure threaten forest cover and community rights. PNAS 115(52): 13164-13173.

Bebbington, D.H., Verdum, R., Gamboa, C. and Bebbington, A.J., 2018c. The Infrastructure-Extractives-Resource Governance Complex in the Pan-Amazon. European Review of Latin American and Caribbean Studies/Revista Europea de Estudios Latinoamericanos y del Caribe (106): 183-208.

Bell, D., Roberton, S. and Hunter, P.R., 2004. Animal origins of SARS coronavirus: possible links with the international trade in small carnivores. Philosophical Transactions of the Royal Society of London. Series B: Biological Sciences 359(1447): 1107-1114.

Bellfield, H., Sabogal, D., Goodman, L. and Leggett, M., 2015. Case study report: Community-based monitoring systems for REDD+ in Guyana. *Forests* 6(1): 133-156.

Bertin, R.I., 2008. Plant phenology and distribution in relation to recent climate change. *The Journal of the Torrey Botanical Society* 135(1): 126-146.

Blackman, A., Corral, L., Lima, E.S. and Asner, G.P., 2017. Titling indigenous communities protects forests in the Peruvian Amazon. PNAS 114(16): 4123-4128.

Bloomfield, L.S., McIntosh, T.L. and Lambin, E.F., 2020. Habitat fragmentation, livelihood behaviors, and contact between people and nonhuman primates in Africa. *Landscape Ecology* 35(4): 985-1000.

Bocci, C., Fortmann, L., Sohngen, B. and Milian, B., 2018. The impact of community forest concessions on income: an analysis of communities in the Maya Biosphere Reserve. World Development 107: 10-21.

Bou Dib, J.B., Krishna, V.V., Alamsyah, Z. and Qaim, M., 2018. Land-use change and livelihoods of non-farm households: The role of income from employment in oil palm and rubber in rural Indonesia. *Land Use Policy* 76: 828-838.

Bowyer, J.L., Howe, D.J., Guillery, P. and Fernholz, K., 2005. Fast-growth Tree Plantations for Wood Production: Environmental Threat Or a Means of 'Saving' Natural Forests. Dovetail Partners Inc., White Bear Lake, MN.

Brack, D., Glover, A. and Wellesley, L., 2016. Agricultural Commodity Supply Chains. London: Chatham House. Brest, P. and Born, K., 2013. When can impact investing create real impact. Stanford Social Innovation Review 11(4): 22-31.

Brown, S.E., Miller, D.C., Ordonez, P.J. and Baylis, K., 2018. Evidence for the impacts of agroforestry on agricultural productivity, ecosystem services, and human well-being in high-income countries: a systematic map protocol. *Environmental Evidence* 7(1): .24.

Bryan, J., 2012. Rethinking territory: social justice and neoliberalism in Latin America's territorial turn. *Geography Compass* 6(4): 215-226.

Busch, J., Engelmann, J., Cook-Patton, S.C., Griscom, B.W., Kroeger, T., Possingham, H. and Shyamsundar, P., 2019. Potential for low-cost carbon dioxide removal through tropical reforestation. Nature Climate Change 9(6): 463-466.

Cardozo, M.L., Salas, D., Ferreira, I., Mereles, T. and Rodríguez, L., 2016. Soy expansion and the absent state: indigenous and peasant livelihood options in eastern Paraguay. *Journal of Latin American Geography* 15(3): 87-104.

Castrén, T., Katila, M., Lindroos, K. and Salmi, J., 2014. Private financing for sustainable forest management and forest products in developing countries: Trends and drivers. Washington, DC: Program on Forests (PROFOR). Available at: https://www.profor.info/ sites/profor.info/files/publication/PROFOR Private Finance_08 20.pdf.

Cattaneo, C., Beine, M., Fröhlich, C.J., Kniveton, D., Martinez-Zarzoso, I., Mastrorillo, M., Millock, K., Piguet, E. and Schraven, B., 2019. Human migration in the era of climate change. *Review of Environmental Economics and Policy* 13(2): 189-206.

Chaves, L.S.M., Fry, J., Malik, A., Geschke, A., Sallum, M.A.M. and Lenzen, M., 2020. Global consumption and international trade in deforestation-associated commodities could influence malaria risk. *Nature communications* 11(1): 1-10.

Cheng, S.H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Rana, P., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. Environmental Evidence 8(1): 3.

Coffel, E.D., Horton, R.M. and de Sherbinin, A., 2017. Temperature and humidity based projections of a rapid rise in global heat stress exposure during the 21st century. Environmental Research Letters 13(1): 014001.

Cohn, A.S., Bhattarai, N., Campolo, J., Crompton, O., Dralle, D., Duncan, J. and Thompson, S., 2019. Forest loss in Brazil increases maximum temperatures within 50 km. Environmental Research Letters 14(8): 084047.

Conde, M., 2020. Brazil in the time of coronavirus. Geopolitica(s) 239–249. Cooper, P., Rao, K.P.C., Singh, P., Dimes, J., Traore, P.C.S., Rao, K., Dixit, P. and Twomlow, S.J., 2009. Farming with current and future climate risk: Advancing a'Hypothesis of Hope'for rainfed agriculture in the semi-arid tropics. *Journal of SAT Agricultural Research* 7: 1-19.

Curtis, P.G., Slay, C.M., Harris, N.L., Tyukavina, A. and Hansen, M.C., 2018. Classifying drivers of global forest loss. *Science* 361(6407): 1108-1111.

Damania, R., Desbureaux, S., Hyland, M., Islam, A., Moore, S., Rodella, A.S., Russ, J. and Zaveri, E., 2017. Uncharted waters: The new economics of water scarcity and variability. Washington D.C.: The World Bank.

Das, S. and Vincent, J.R., 2009. Mangroves protected villages and reduced death toll during Indian super cyclone. PNAS 106(18): 7357-7360.

Davies, D.K., Ilavajhala, S., Wong, M.M. and Justice, C.O., 2008. Fire information for resource management system: archiving and distributing MODIS active fire data. *IEEE Transactions on Geoscience and Remote Sensing* 47(1): 72-79.

Devine, J., 2014. Counterinsurgency ecotourism in Guatemala's Maya biosphere reserve. Environment and Planning D: Society and Space 32(6): 984-1001.

Devine, J.A., 2018. Community forest concessionaires: resisting green grabs and producing political subjects in Guatemala. *The Journal of Peasant Studies* 45(3): 565-584.

Díaz, S., Pascual, U., Stenseke, M., Martín-López,
B., Watson, R.T., Molnár, Z., Hill, R., et al., 2018.
Assessing nature's contributions to people. Science 359(6373): 270-272.

Dobson, A.P., Pimm, S.L., Hannah, L., Kaufman, L., Ahumada, J.A., Ando, A.W., Bernstein, A., et al., 2020. Ecology and economics for pandemic prevention. Science 369(6502): 379-381.

Dormontt, E.E., Boner, M., Braun, B., Breulmann, G.,
Degen, B., Espinoza, E., Gardner, S., et al., 2015.
Forensic timber identification: It's time to integrate disciplines to combat illegal logging. Biological Conservation 191: 790-798.

Duncombe, R., 2016. Mobile phones for agricultural and rural development: A literature review and suggestions for future research. *The European Journal* of Development Research 28(2): 213-235.

Eakin, H., DeFries, R., Kerr, S., Lambin, E.F., Liu, J.,
Marcotullio, P.J., Messerli, P., et al., 2014. Significance of telecoupling for exploration of land-use change.
In: Rethinking global land use in an urban era, edited by K.C. Seto and A. Reenberg (pp. 141-161). Cambridge MA: MIT Press.

- Eklund, J., Jokinen, A. and Toivonen, T., 2020. COVID-19 causes a threat to protected areas in the Global South – evidence from Madagascar. DIGGEOLAB Blog, University of Helsinki. (Available at: <u>https://blogs. helsinki.fi/digital-geography/2020/06/01/covidmadagascar-protectedareas/</u> [accessed on 1 September 2020]).
- Environmental Defense Fund and Forest Trends, 2018. Mapping Forest Finance. A Landscape of Available Sources of Finance for REDD+ and Climate Action in Forests. New York, Washington D.C. and Gland: Environmental Defense Fund, Forest Trends and IUCN.
- Epstein, J.H., Field, H.E., Luby, S., Pulliam, J.R. and Daszak, P., 2006. Nipah virus: impact, origins, and causes of emergence. *Current Infectious Disease Reports* 8(1): 59-65.
- Erbaugh, J.T. and Nurrochmat, D.R., 2019. Paradigm shift and business as usual through policy layering: Forest-related policy change in Indonesia (1999-2016). Land use policy 86: 136-146.
- Fagan, M. and Huang, C., 2019. A look at how people around the world view climate change – Pew Research Center Spring 2018 Global Attitudes Survey: 1–8.
 (Available at: http://www.pewresearch.org/facttank/2018/10/19/5-charts-on-global-views-of-china/ [accessed on 1 September 2020]).
- Fearnside, P.M., 2018. Why Brazil's new president poses an unprecedented threat to the Amazon. Yale Environment 360(08).
- Ferraro, P.J., 2009. Counterfactual thinking and impact evaluation in environmental policy. New directions for evaluation 2009(122): 75-84.
- Ferraz, A., Saatchi, S., Mallet, C. and Meyer, V., 2016. Lidar detection of individual tree size in tropical forests. *Remote Sensing of Environment* 183: 318-333.
- Finney, C., 2014. Black faces, white spaces: Reimagining the relationship of African Americans to the great outdoors. Chapel Hill: UNC Press Books.
- Fisher, R., Maginnis, S., Jackson, W., Barrow, E. and Jeanrenaud, S., 2008. Linking Conservation and Poverty Reduction: landscapes, people and power. Abingdon: Routledge.
- FOE, 2008. Losing ground: the human rights of oil palm plantation in Indonesia. London, Edinburgh and Bogor: Friends of the Earth, LifeMosaic and Sawit Watch.
- Forest Trends Ecosystems Marketplace, 2019. Financing Emission Reductions for the Future: State of Voluntary Carbon Markets 2019. Washington D.C.: Forest Trends. (Available at: https://www.forest-trends.org/wpcontent/uploads/2019/12/SOVCM2019.pdf. [accessed on 1 September 2020]).
- Fowler, C.T., 2003. Human health impacts of forest fires in the southern United States: a literature review. *Journal of Ecological Anthropology* 7(1): 39-63.

- Gabay, M. and Alam, M., 2017. Community forestry and its mitigation potential in the Anthropocene: The importance of land tenure governance and the threat of privatization. *Forest Policy and Economics* 79: 26-35.
- Gale, F., Valdes, C. and Ash, M., 2019. Interdependence of China, United States, and Brazil in Soybean Trade. New York: US Department of Agriculture's Economic Research Service (ERS) Report: 1-48.
- Ganivet, E. and Bloomberg, M., 2019. Towards rapid assessments of tree species diversity and structure in fragmented tropical forests: A review of perspectives offered by remotely-sensed and fieldbased data. *Forest ecology and management* 432: 40-53.
- Gaveau, D.L., Pirard, R., Salim, M.A., Tonoto, P., Yaen, H., Parks, S.A. and Carmenta, R., 2017. Overlapping land claims limit the use of satellites to monitor no-deforestation commitments and no-burning compliance. *Conservation Letters* 10(2): 257-264.
- Ginn, W. J., 2020. Valuing Nature: A Handbook for Impact Investing. Washington D.C.: Island Press.
- Gockowski, J. and Sonwa, D., 2011. Cocoa intensification scenarios and their predicted impact on CO₂ emissions, biodiversity conservation, and rural livelihoods in the Guinea rain forest of West Africa. Environmental management 48(2): 307-321.
- Gonzales, J., 2020. Brazil Minister Advises Using COVID-19 to Distract from Amazon Deregulation. *Mongabay*. (Available at: <u>https://news.mongabay</u>. <u>com/2020/05/brazil-minister-advises-using-covid-</u> <u>19-to-distract-from-amazon-deregulation/</u> [accessed on 1 September 2020]).
- Griscom, B.W., Adams, J., Ellis, P.W., Houghton, R.A., Lomax, G., Miteva, D.A., Schlesinger, W.H., et al., 2017. Natural climate solutions. *Proceedings of the National Academy of Sciences* 114(44): 11645-11650.
- Guillerme, S., Kumar, B.M., Menon, A., Hinnewinkel, C., Maire, E. and Santhoshkumar, A.V., 2011. Impacts of public policies and farmer preferences on agroforestry practices in Kerala, India. *Environmental management* 48(2): 351-364.
- De Haas, H., Czaika, M., Flahaux, M.L., Mahendra, E., Natter, K., Vezzoli, S. and Villares-Varela, M., 2019. International migration: Trends, determinants, and policy effects. *Population and Development Review* 45(4): 885-922.
- Hajjar, R., Oldekop, J.A., Cronkleton, P., Etue, E., Newton, P., Russel, A.J., Tjajadi, J.S., et al., 2016. The data not collected on community forestry. *Conservation Biology* 30(6): 1357-1362.
- Hallegatte, S., Bangalore, M., Bonzanigo, L., Fay, M., Kane, T., Narloch, U., Rozenberg, J., et al., 2015. Shock waves: managing the impacts of climate change on poverty. Washington D.C.: The World Bank.

Hand, D., Dithrich, H. and Sunderji, S., 2020. Annual Impact Investor Survey 2020. New York: GIIN.

Hansen, A.J., Neilson, R.P., Dale, V.H., Flather, C.H.,
Iverson, L.R., Currie, D.J., Shafer, S., et al., 2001.
Global change in forests: responses of species,
communities, and biomes: interactions between
climate change and land use are projected to cause
large shifts in biodiversity. *BioScience* 51(9): 765-779.

Hansen, M.C., Potapov, P.V., Moore, R., Hancher, M., Turubanova, S.A., Tyukavina, A., Thau, D., et al., 2013.
High-resolution global maps of 21st-century forest cover change. *Science* 342(6160): 850-853.

Hansen, M.C., Krylov, A., Tyukavina, A., Potapov, P.V., Turubanova, S., Zutta, B., Ifo, S., et al., 2016. Humid tropical forest disturbance alerts using Landsat data. Environmental Research Letters 11(3): 034008.

- Harley, M.D., Turner, I.L., Kinsela, M.A., Middleton, J.H., Mumford, P.J., Splinter, K.D., Phillips, M.S., et al., 2017. Extreme coastal erosion enhanced by anomalous extratropical storm wave direction. Scientific reports 7(1): 1-9.
- Harris, N., Minnemeyer, S., Sizer, N., Mann, S.A. and Payne, A.O., 2015. With Latest Fires Crisis, Indonesia Surpasses Russia as World's Fourth-Largest Emitter. World Research Institute. (Available at: <u>http://www. wri.org/blog/2015/10/latest-fires-crisis-indonesiasurpasses-russia-world's-fourth-largest-emitter</u> [accessed on 1 September 2020]).
- Hay, S. I., Abajobir, A.A., Abate, K.A., Abbafati, C., Abbas, K.M., Abd-Allah, F., Suliankatchi Abdulkader R., et al., 2017. Global, regional, and national disabilityadjusted life-years (DALYs) for 333 diseases and injuries and healthy life expectancy (HALE) for 195 countries and territories, 1990-2016: A systematic analysis for the Global Burden of Disease Study 2016. The Lancet. 390(10100): 1260–1344.
- Hays, C.M., 2019. The 'Park' as Racial Practice: Constructing Whiteness on Safari in Tanzania. Environmental Values 28(2): 141-170.
- Henders, S., Ostwald, M., Verendel, V. and Ibisch, P., 2018. Do national strategies under the UN biodiversity and climate conventions address agricultural commodity consumption as deforestation driver? *Land Use Policy* 70: 580-590.
- Henders, S., Persson, U.M. and Kastner, T., 2015. Trading forests: land-use change and carbon emissions embodied in production and exports of forest-risk commodities. Environmental Research Letters 10(12): 125012.
- Husnina, Z., Clements, A.C. and Wangdi, K., 2019.
 Forest cover and climate as potential drivers for dengue fever in Sumatra and Kalimantan 2006-2016: a spatiotemporal analysis. Tropical Medicine & International Health 24(7): 888-898.

- IMF, 2020. World Economic Outlook. Washington DC: International Monetary Fund. (Available at: <u>https://www.imf.org/en/Publications/WEO/</u> <u>Issues/2020/04/14/weo-april-2020</u> [accessed on 1 September 2020]).
- Iacona, G., Ramachandra, A., McGowan, J., Davies,
 A., Joppa, L., Koh, L.P., Fegraus, E., Game, E.,
 Guillera-Arroita, G., Harcourt, R. and Indraswari,
 K., 2019. Identifying technology solutions to bring conservation into the innovation era. Frontiers in Ecology and the Environment 17(10): 591-598.
- Ingram, V., Ewane, M., Ndumbe, L.N. and Awono, A., 2017. Challenges to governing sustainable forest food: Irvingia spp. from southern Cameroon. Forest Policy and Economics 84: 29-37.
- Ingram, VJ., 2014. Win-wins in forest product value chains?: how governance impacts the sustainability of livelihoods based on non-timber forest products from Cameroon. PhD Thesis. Amsterdam: University of Amsterdam.
- International Organization for Migration, 2020. World Migration Report 2020, Chapter 2. Geneva: IOM.
- IPBES, 2019. Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. (S. Díaz, J. Settele, E. S. Brondízio E.S., H. T. Ngo, M. Guèze, J. Aga). Bonn: IPBES Secretariat.
- IPCC, 2018. Summary for Policymakers. In: Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty [edited by Masson-Delmotte, V., P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P.R. Shukla, A. Pirani, W. Moufouma-Okia, C. Péan, R. Pidcock, S. Connors, J.B.R. Matthews, Y. Chen, X. Zhou, M.I. Gomis, E. Lonnoy, T. Maycock, M. Tignor, and T. Waterfield]. Geneva: IPCC:
- IPCC, 2019. Summary for Policymakers. In: Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems [edited by P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmotte, H.- O. Pörtner, D. C. Roberts, P. Zhai, R. Slade, S. Connors, R. van Diemen, M. Ferrat, E. Haughey, S. Luz, S. Neogi, M. Pathak, J. Petzold, J. Portugal Pereira, P. Vyas, E. Huntley, K. Kissick, M. Belkacemi, J. Malley]. Geneva: IPCC.
- Jackson, J.E. and Warren, K.B., 2005. Indigenous movements in Latin America, 1992–2004: controversies, ironies, new directions. Annu. Rev. Anthropol. 34: 549-573.
- Jiang, L. and O'Neill, B.C., 2017. Global urbanization projections for the Shared Socioeconomic Pathways. Global Environmental Change 2: 193-199.

Johnson, J.M., Moore, L.J., Ells, K., Murray, A.B., Adams, P.N., MacKenzie III, R.A. and Jaeger, J.M., 2015. Recent shifts in coastline change and shoreline stabilization linked to storm climate change. *Earth Surface Processes and Landforms* 40(5): 569-585.

Jolly, W.M., Cochrane, M.A., Freeborn, P.H., Holden, Z.A., Brown, T.J., Williamson, G.J. and Bowman, D.M., 2015. Climate-induced variations in global wildfire danger from 1979 to 2013. Nature communications 6(1): 1-11.

Joppa, L. N., 2017. The case for technology investments in the environment. *Nature* 552(7685): 325–328.

Kamminga, J., Ayele, E., Meratnia, N. and Havinga, P., 2018. Poaching detection technologies – a survey. Sensors 18(5): 1474.

Kayitakire, F., Hamel, C. and Defourny, P., 2006. Retrieving forest structure variables based on image texture analysis and IKONOS-2 imagery. *Remote* sensing of environment 102(3-4): 390-401.

Keck, M. E. and Sikkink, K., 1998. Activists beyond Borders: Advocacy Networks in International Politics. Ithaca, NY: Cornell University Press.

Keesing, F., Belden, L.K., Daszak, P., Dobson, A., Harvell,
C.D., Holt, R.D., Hudson, P., Jolles, A., Jones, K.E.,
Mitchell, C.E. and Myers, S.S., 2010. Impacts of
biodiversity on the emergence and transmission of
infectious diseases. Nature 468(7324): 647-652.

Khamzina, A., Lamers, J.P. and Vlek, P.L., 2012.
Conversion of degraded cropland to tree plantations for ecosystem and livelihood benefits. In: Cotton, water, salts and soums, edited by C. Martius, I.
Rudenko, J.P.A. Lamers and P.L.G. Vlek (pp. 235-248).
Dordrecht: Springer.

Kloek, M.E., Elands, B.H. and Schouten, M.G., 2017. Race/Ethnicity in Visual Imagery of Dutch Nature Conservation Organizations. Society & Natural Resources 30(9): 1033-1048.

Knorr, K., Jiang, L. and Arneth, A., 2016. Climate, CO₂ and demographic impacts on global wildfire emissions. Biogeosciences 13(1): 267.

Knudsen, B.T. and Andersen, C., 2019. Affective politics and colonial heritage, Rhodes Must Fall at UCT and Oxford. International Journal of Heritage Studies 25(3): 239-258.

Korkanç, S.Y., 2014. Effects of afforestation on soil organic carbon and other soil properties. Catena 123: 62-69.

Kozak, R.A., 2007. Small and medium forest enterprises: instruments of change in the developing world. Washington, D.C.: Rights and Resources Initiative.

Kreft, H. and Jetz, W., 2007. Global patterns and determinants of vascular plant diversity. PNAS 104(14): 5925-5930. De La Fuente, T. and Hajjar, R., 2013. Do current forest carbon standards include adequate requirements to ensure indigenous peoples' rights in REDD projects? International Forestry Review 15(4): 427-441.

de la Vega-Leinert, A. C. and Schönenberg, R., 2020. Transdisciplinary perspectives on current transformations at extractive and agrarian, frontiers in Latin America. *Journal of Land Use Science* 15(2–3): 99–107.

Laborde, D., Martin, W. and Vos, R., 2020. Poverty and food insecurity could grow dramatically as COVID-19 spreads. Washington, D.C.: International Food Policy Research Institute (IFPRI),

Lambin, E.F. and Meyfroidt, P., 2011. Global land use change, economic globalization, and the looming land scarcity. PNAS 108(9): 3465-3472.

Lawrence, D. and Vandecar, K., 2014. The impact of tropical deforestation on climate and links to agricultural productivity. *Nature Climate Change* 5: 27-36.

Lawson, S. 2014. Consumer Goods and Deforestation: An Analysis of the Extent and Nature of Illegality in Forest Conversion for Agriculture and Timber Plantations. Available at: https://www.forest-trends.org/wpcontent/uploads/imported/for168-consumer-goodsand-deforestation-letter-14-0916-hr-no-crops_webpdf. [accessed on 1 September 2020]).

Lee, T.M., Markowitz, E.M., Howe, P.D., Ko, C.Y. and Leiserowitz, A.A., 2015. Predictors of public climate change awareness and risk perception around the world. Nature climate change 5(11): 1014-1020.

Lima, M., Skutsch, M. and de Medeiros Costa, G., 2011. Deforestation and the social impacts of soy for biodiesel: perspectives of farmers in the South Brazilian Amazon. *Ecology and Society* 16(4).

Liu, L., Oza, S., Hogan, D., Perin, J., Rudan, I., Lawn, J.E., Cousens, S., Mathers, C. and Black, R.E., 2015. Global, regional, and national causes of child mortality in 2000–13, with projections to inform post-2015 priorities: an updated systematic analysis. *The Lancet* 385(9966): 430-440.

Lorenz, C., Azevedo, T. S. and Chiaravalloti-Neto, F., 2020. COVID-19 and dengue fever: A dangerous combination for the health system in Brazil. *Travel Medicine and Infectious Disease* 35: 101659.

Loveland, T.R. and Dwyer, J.L., 2012. Landsat: Building a strong future. *Remote Sensing of Environment* 122: 22-29.

Maas, B., Clough, Y. and Tscharntke, T., 2013. Bats and birds increase crop yield in tropical agroforestry landscapes. *Ecology letters* 16(12): 1480-1487.

MacDonald, A.J. and Mordecai, E.A., 2019. Amazon deforestation drives malaria transmission, and malaria burden reduces forest clearing. PNAS 116(44): 22212-22218.

- Macqueen, D., Bolin, A., Greijmans, M., Grouwels, S. and Humphries, S., 2020. Innovations towards prosperity emerging in locally controlled forest business models and prospects for scaling up. World Development 125: 104382.
- Macqueen, DJ., 2008. Forest Connect: reducing poverty and deforestation through support to community forest enterprises. *International forestry review* 10(4): 670-675.
- Malkamäki, A., D'Amato, D., Hogarth, N.J., Kanninen, M., Pirard, R., Toppinen, A. and Zhou, W., 2018. A systematic review of the socio-economic impacts of large-scale tree plantations, worldwide. *Global Environmental Change* 53: 90-103.
- Marchiori, L., Maystadt, J.-F. and Schumacher, I., 2012. The impact of weather anomalies on migration in sub-Saharan Africa. *Journal of Environmental Economics and Management* 63(3): 355–374.
- Di Marco, M., Baker, M.L., Daszak, P., De Barro, P., Eskew, E.A., Godde, C.M., Harwood, T.D., *et al.*, 2020. Opinion: Sustainable development must account for pandemic risk. PNAS 117(8): 3888-3892.
- Martínez-Harms, M.J., Quijas, S., Merenlender, A.M. and Balvanera, P., 2016. Enhancing ecosystem services maps combining field and environmental data. *Ecosystem services* 22: 32-40.
- Marvin, D.C., Koh, L.P., Lynam, A.J., Wich, S., Davies, A.B., Krishnamurthy, R., Stokes, E., Starkey, R. and Asner, G.P., 2016. Integrating technologies for scalable ecology and conservation. Global Ecology and Conservation 7: 262-275.
- Mbow, C., Smith, P., Skole, D., Duguma, L. and Bustamante, M., 2014. Achieving mitigation and adaptation to climate change through sustainable agroforestry practices in Africa. *Current Opinion in Environmental Sustainability 6*: 8-14.
- McCarthy, J., 2019. Authoritarianism, Populism, and the Environment: Comparative Experiences, Insights, and Perspectives. Annals of the American Association of Geographers 109(2): 301–313.
- McSweeney, K., Nielsen, E.A., Taylor, M.J., Wrathall, D.J., Pearson, Z., Wang, O. and Plumb, S.T., 2014. Drug policy as conservation policy: narco-deforestation. *Science* 343(6170): 489-490.
- Melo, I., Turnhout, E. and Arts, B., 2014. Integrating multiple benefits in market-based climate mitigation schemes: The case of the Climate, Community and Biodiversity certification scheme. Environmental Science & Policy 35: 49-56.
- Middeldorp, N. and Le Billon, P., 2019. Deadly environmental governance: authoritarianism, ecopopulism, and the repression of environmental and land defenders. Annals of the American Association of Geographers 109(2): 324-337.

- Miles, T., 2019. The Afro-Brazilian Struggle for Visibility in Ceará: In Fortaleza, the Black movement has scored important victories in shedding light on the depths of racism in Brazil. But will it be able to transform awareness into structural change?. NACLA Report on the Americas 51(4): 394-400.
- Mills, J.N., Gage, K.L. and Khan, A.S., 2010. Potential influence of climate change on vector-borne and zoonotic diseases: a review and proposed research plan. *Environmental health perspectives* 118(11): 1507-1514.
- Missirian, A. and Schlenker, W., 2017. Asylum applications respond to temperature fluctuations. *Science*, 358(6370): 1610-1614.
- Mitchell-Walthour, G., 2020. Afro-Brazilian Women YouTubers' Use of African-American Media Representations to Promote Social Justice in Brazil. Journal of African American Studies 24(1): 149-163.
- Mitchell, A.L., Rosenqvist, A. and Mora, B., 2017. Current remote sensing approaches to monitoring forest degradation in support of countries measurement, reporting and verification (MRV) systems for REDD+. *Carbon Balance and Management* 12(1): 9.
- Mollett, S. and Kepe, T., 2018. Introduction: Land Rights, Biodiversity Conservation and Justice. In: Land Rights, Biodiversity Conservation and Justice: Rethinking Parks and People, edited by S. Mollett and T. Kepe London: Routledge, p. 13.
- Mora, C., Dousset, B., Caldwell, I.R., Powell, F.E., Geronimo, R.C., Bielecki, C.R., Counsell, C.W., et al., 2017. Global risk of deadly heat. Nature climate change 7(7): 501-506.
- Morse, S.S., Mazet, J.A., Woolhouse, M., Parrish, C.R., Carroll, D., Karesh, W.B., Zambrana-Torrelio, C., et al., 2012. Prediction and prevention of the next pandemic zoonosis. *The Lancet* 380(9857): 1956-1965.
- Musinsky, J., Tabor, K., Cano, C.A., Ledezma, J.C., Mendoza, E., Rasolohery, A. and Sajudin, E.R., 2018. Conservation impacts of a near real-time forest monitoring and alert system for the tropics. *Remote Sensing in Ecology and Conservation* 4(3): 189-196.
- Nepstad, D.C. and Stickler, C.M., 2008. Managing the tropical agriculture revolution. *Journal of Sustainable Forestry* 27(1-2): 43-56.
- Nepstad, D.C., Stickler, C.M. and Almeida, O.T., 2006. Globalization of the Amazon soy and beef industries: opportunities for conservation. *Conservation biology* 20(6): 1595-1603.
- Neumann, B., Vafeidis, A.T., Zimmermann, J. and Nicholls, R.J., 2015. Future coastal population growth and exposure to sea-level rise and coastal flooding-a global assessment. *PloS one* 10(3), p.e0118571.
- Newton, P., Agrawal, A. and Wollenberg, L., 2013. Enhancing the sustainability of commodity supply chains in tropical forest and agricultural landscapes. Global Environmental Change 10(3): e0118571.

Newton, P. and Benzeev, R., 2018. The role of zerodeforestation commitments in protecting and enhancing rural livelihoods. *Current Opinion in Environmental Sustainability* 32: 126-133.

Nyhus, P.J., 2016. Human–wildlife conflict and coexistence. Annual Review of Environment and Resources 41.

OECD and FAO, 2019. OECD-FAO Agricultural Outlook 2019-2028. Paris and Rome: OECD and FAO.

Oldekop, J.A., Rasmussen, L.V., Agrawal, A., Bebbington, A., Bengston, D.N., Blackman, A., Brooks, S., et al., 2019. Key Emerging Issues for Forests and Livelihoods in the 21st Century-June 2019. Forest Collaborative Research. 21.

Olsson, L., H. Barbosa, S. Bhadwal, A. Cowie, K. Delusca,
D. Flores-Renteria, K. Hermans, E., et al., (ed.),
2019. Climate Change and Land: an IPCC special report
on climate change, desertification, land degradation,
sustainable land management, food security, and
greenhouse gas fluxes in terrestrial ecosystems: 345-436.
Geneva: IPCC.

Ortega, J.C., Machado, N., Diniz-Filho, J.A.F., Rangel, T.F., Araújo, M.B., Loyola, R. and Bini, L.M., 2019. Meta-analyzing the likely cross-species responses to climate change. *Ecology and evolution* 9(19): 11136-11144.

Ota, L., Herbohn, J., Gregorio, N. and Harrison, S., 2020. Reforestation and smallholder livelihoods in the humid tropics. *Land Use Policy* 92: 104455.

Paneque-Gálvez, J., Vargas-Ramírez, N., Napoletano, B.M. and Cummings, A., 2017. Grassroots innovation using drones for indigenous mapping and monitoring. Land 6(4): 86.

Pendrill, F., Persson, U.M., Godar, J., Kastner, T., Moran, D., Schmidt, S. and Wood, R., 2019. Agricultural and forestry trade drives large share of tropical deforestation emissions. *Global environmental change* 56: 1-10.

Persson, U.M., Henders, S. and Cederberg, C., 2014. A method for calculating a land-use change carbon footprint (LUC-CFP) for agricultural commodities– applications to Brazilian beef and soy, Indonesian palm oil. *Global change biology* 20(11): 3482-3491.

Philpott, S.M., Arendt, W.J., Armbrecht, I., Bichier,
P., Diestch, T.V., Gordon, C., Greenberg, R., et al.,
2008. Biodiversity Loss in Latin American Coffee
Landscapes: Review of the Evidence on Ants, Birds,
and Trees. Conservation Biology 22(5):1093–1105.

Le Polain de Waroux, Y., Garrett, R.D., Graesser, J., Nolte, C., White, C. and Lambin, E.F., 2019. The restructuring of South American soy and beef production and trade under changing environmental regulations. World Development 121: 188-202. Pongsiri, M.J., Roman, J., Ezenwa, V.O., Goldberg, T.L., Koren, H.S., Newbold, S.C., Ostfeld, R.S., Pattanayak, S.K. and Salkeld, D.J., 2009. Biodiversity loss affects global disease ecology. *Bioscience* 59(11): 945-954.

Poole, N. and de Frece, A., 2010. A Review of Existing Organisational Forms of Smallholder Farmers' Associations and their Contractual Relationships with other Market Participants in the East and Southern African ACP Region, AAACP Paper Series. Rome: FAO.

Potapov, P., Hansen, M.C., Laestadius, L., Turubanova, S., Yaroshenko, A., Thies, C., Smith, W., et al., 2017. The last frontiers of wilderness: Tracking loss of intact forest landscapes from 2000 to 2013. Science advances 3(1): p.e1600821.

Pradhan, B., Chaudhari, A., Adinarayana, J. and Buchroithner, M.F., 2012. Soil erosion assessment and its correlation with landslide events using remote sensing data and GIS: a case study at Penang Island, Malaysia. Environmental monitoring and assessment 184(2): 715-727.

Pramova, E., Locatelli, B., Djoudi, H. and Somorin, O.A., 2012. Forests and trees for social adaptation to climate variability and change. Wiley Interdisciplinary Reviews: Climate Change 3(6): 581-596.

Prevedello, J.A., Winck, G.R., Weber, M.M., Nichols, E. and Sinervo, B., 2019. Impacts of forestation and deforestation on local temperature across the globe. *PloS one* 14(3): e0213368.

Qaim, M., Sibhatu, K.T., Siregar, H. and Grass, I., 2020. Environmental, Economic, and Social Consequences of the Oil Palm Boom. Annual Review of Resource Economics 12.

Rasmussen, M.B. and Lund, C., 2018. Reconfiguring Frontier Spaces: The territorialization of resource control. World Development 101: 388-399.

Richards, P., Pellegrina, H., VanWey, L. and Spera, S., 2015. Soybean development: The impact of a decade of agricultural change on urban and economic growth in Mato Grosso, Brazil. PLoS one 10(4): e0122510.

Rigaud, K.K., de Sherbinin, A., Jones, B., Bergmann, J., Clement, V., Ober, K., Schewe, J., et al., 2018. Groundswell: Preparing for internal climate migration. Washington, D.C.: The World Bank.

Rights and Resources Initiative, 2018. At a Crossroads: Consequential Trends in Recognition of Community-Based Forest Tenure from 2002-2017. Washington, D.C.: Rights and Resources Initiative.

Rist, L., Feintrenie, L. and Levang, P., 2010. The livelihood impacts of oil palm: smallholders in Indonesia. *Biodiversity and conservation* 19(4): 1009-1024.

Robinson, B.E., Holland, M.B. and Naughton-Treves, L., 2014. Does secure land tenure save forests? A metaanalysis of the relationship between land tenure and tropical deforestation. *Global Environmental Change* 29: 281-293. Rodin, J. and Brandenburg, M., 2014. The Power of Impact Investing : Putting Markets to Work for Profit and Global Good. Philadelphia, PA: Wharton School Press.

Rohr, J.R., Barrett, C.B., Civitello, D.J., Craft, M.E., Delius, B., DeLeo, G.A., Hudson, P.J., et al., 2019. Emerging human infectious diseases and the links to global food production. Nature Sustainability 2(6): 445-456.

- Roshetko, J.M., Rohadi, D., Perdana, A., Sabastian, G., Nuryartono, N., Pramono, A.A., Widyani, N., et al., 2013. Teak agroforestry systems for livelihood enhancement, industrial timber production, and environmental rehabilitation. *Forests, Trees and Livelihoods* 22(4): 241-256.
- Saavedra-Velasco, M., Chiara-Chilet, C., Pichardo-Rodriguez, R., Grandez-Urbina, A. and Inga-Berrospi,
 F., 2020. Coinfección entre dengue y COVID-19: Necesidad de abordaje en zonas endémicas. Revista de la Facultad de Ciencias Médicas de Córdoba 77(1).
- Samir, K. C., 2013. Community vulnerability to floods and landslides in Nepal. Ecology and society 18(1).
- Santika, T., Wilson, K.A., Budiharta, S., Law, E.A., Poh, T.M., Ancrenaz, M., Struebig, M.J. and Meijaard, E., 2019. Does oil palm agriculture help alleviate poverty? A multidimensional counterfactual assessment of oil palm development in Indonesia. World Development, 120: 105-117.
- Sasaki, N., Asner, G.P., Pan, Y., Knorr, W., Durst, P.B., Ma, H.O., Abe, I., et al., 2016. Sustainable forest management of tropical forests can reduce carbon emissions and stabilize timber production. Frontiers in Environmental Science. 4: 50.
- Sauls, L.A., 2019. Becoming fundable? Converting climate justice claims into climate finance in Mesoamerica's forests. *Climatic Change*: 1-19.
- Scheidel, A., Del Bene, D., Liu, J., Navas, G., Mingorría, S., Demaria, F., Avila, S., et al., 2020. Environmental conflicts and defenders: A global overview. Global Environmental Change 63: 102104.
- Seppälä, R., Buck, A. and Katila, P., 2009. Adaptation of Forests and People to Climate Change - A Global Assessment Report. Vienna: IUFRO.
- Seymour, F. and Busch, J., 2016. Why Forests? Why Now?: The Science, Economics, and Politics of Tropical Forests and Climate Change. Washington D.C.: Brookings Institution Press.
- Seymour, F. and Harris, N.L., 2019. Reducing tropical deforestation. *Science* 365(6455): 756-757.
- Sheffield, J., Wood, E.F. and Roderick, M.L., 2012. Little change in global drought over the past 60 years. Nature 491(7424): 435-438.
- Shyamsundar, P., 2020. Conservation for the last mile, Perspectives, The Nature Conservancy. (Available at: https://www.nature.org/en-us/what-we-do/ourinsights/perspectives/rural-communities-pandemicconservation/. [accessed on 1 September 2020]).

- Shyamsundar, P., Ahlroth, S., Kristjanson, P. and Onder, S., 2020. Supporting pathways to prosperity in forest landscapes – A PRIME framework. World Development, 125: 104622.
- Sife, A.S., Kiondo, E. and Lyimo-Macha, J.G., 2010. Contribution of mobile phones to rural livelihoods and poverty reduction in Morogoro region, Tanzania. The Electronic Journal of Information Systems in Developing Countries 42(1): 1-15.
- Smith, P., Adams, J., Beerling, D.J., Beringer, T., Calvin, K.V., Fuss, S., Griscom, B., et al., 2019. Landmanagement options for greenhouse gas removal and their impacts on ecosystem services and the sustainable development goals. Annual Review of Environment and Resources 44.
- Staal, A., Flores, B.M., Aguiar, A.P.D., Bosmans, J.H., Fetzer, I. and Tuinenburg, O.A., 2020. Feedback between drought and deforestation in the Amazon. Environmental Research Letters 15(4): 044024.
- Steinkamp, J. and Hickler, T., 2015. Is drought-induced forest dieback globally increasing? *Journal of Ecology* 103(1): 31-43.
- Steven, M.D., Malthus, T.J., Baret, F., Xu, H. and Chopping, M.J., 2003. Intercalibration of vegetation indices from different sensor systems. *Remote Sensing of Environment* 88(4): 412-422.
- Suter, M.K., Miller, K.A., Anggraeni, I., Ebi, K.L., Game, E.T., Krenz, J., Masuda, Y.J., et al., 2019. Association between work in deforested, compared to forested, areas and human heat strain: an experimental study in a rural tropical environment. Environmental research letters 14(8): 084012
- De Sy, V., Herold, M., Achard, F., Beuchle, R., Clevers, J.G.P.W., Lindquist, E. and Verchot, L., 2015. Land use patterns and related carbon losses following deforestation in South America. *Environmental Research Letters* 10(12): 124004.
- Tadesse, G., Zavaleta, E. and Shennan, C., 2014. Coffee landscapes as refugia for native woody biodiversity as forest loss continues in southwest Ethiopia. Biological Conservation 169: 384-391.
- Taufik, M., Torfs, P.J.F., Uijlenhoet, R., Jones, P.D., Murdiyarso, D. and Van Lanen, H.A.J. 2017.
 Amplification of wildfire area burnt by hydrological drought in the humid tropics. *Nature Climate Change* 7: 428-431.
- Taylor, L., 2020. Coronavirus in the Amazon. New Scientist 246(3288): 10.
- Thorlakson, T., de Zegher, J.F. and Lambin, E.F., 2018. Companies' contribution to sustainability through global supply chains. *Proceedings of the National Academy of Sciences* 115(9): 2072-2077.
- Tnah, L.H., Lee, S.L., Ng, K.K.S., Faridah, Q.Z. and Faridah-Hanum, I., 2010. Forensic DNA profiling of tropical timber species in Peninsular Malaysia. Forest ecology and management 259(8): 1436-1446.

Le Toan, T., Quegan, S., Davidson, M.W.J., Balzter, H., Paillou, P., Papathanassiou, K., Plummer, S., Rocca, F., Saatchi, S., Shugart, H. and Ulander, L., 2011. The BIOMASS mission: Mapping global forest biomass to better understand the terrestrial carbon cycle. *Remote sensing of environment* 115(11): 2850-2860.

Tollefson, J., 2016. Political upheaval threatens Brazil's environmental protections. *Nature* 539(7628).

Tollefson, J., 2019. The hard truths of climate change – by the numbers. *Nature* 573: 324-327.

 Vila Benites, G. and Bebbington, A., 2020. Political Settlements and the Governance of COVID-19: Mining, Risk, and Territorial Control in Peru. *Journal* of Latin American Geography 19(3): 215-223.

Virtanen, P.K., 2015. Indigenous social media practices in Southwestern Amazonia. AlterNative: An International Journal of Indigenous Peoples 11(4): 350-362.

Vitasse, Y., Porté, A.J., Kremer, A., Michalet, R. and Delzon, S., 2009. Responses of canopy duration to temperature changes in four temperate tree species: relative contributions of spring and autumn leaf phenology. *Oecologia* 161(1): 187-198.

Walsh, K.J., McBride, J.L., Klotzbach, P.J., Balachandran, S., Camargo, S.J., Holland, G., Knutson, T.R., et al., 2016. Tropical cyclones and climate change. Wiley Interdisciplinary Reviews: Climate Change 7(1): 65-89.

Walther, G.R., Post, E., Convey, P., Menzel, A., Parmesan, C., Beebee, T.J., Fromentin, J.M., et al., 2002. Ecological responses to recent climate change. Nature 416(6879): 389-395.

Wani, S.P., Sreedevi, T.K., Rockström, J. and Ramakrishna, Y.S., 2009. Rainfed agriculture – past trends and future prospects. In: Rainfed agriculture: Unlocking the potential, edited by S. P. Wani, J. Rockström, and T. Oweis. pp.1-35.

Watts, N., Amann, M., Arnell, N., Ayeb-Karlsson, S., Belesova, K., Boykoff, M., Byass, P., Cai, W., Campbell-Lendrum, D., Capstick, S. and Chambers, J., 2019. The 2019 report of The Lancet Countdown on health and climate change: ensuring that the health of a child born today is not defined by a changing climate. The Lancet 394(10211): 1836-1878.

Wells, M.P., Alderman, C. and Altman, S., 2011. An Evaluation of the Impacts of Selected Activities Supported by the Program on Forests in Central America (No. 70513: 1-48). Washington D.C.: The World Bank.

Wheeler, D., Hammer, D., Kraft, R. and Steele, A., 2014. Satellite-based forest clearing detection in the Brazilian Amazon: FORMA, DETER, and PRODES. Washington D.C.: WRI.

Wich, S. A. and Koh, L. P., 2018. Conservation drones: Mapping and monitoring biodiversity. Oxford: Oxford University Press. Wolfe, N.D., Daszak, P., Kilpatrick, A.M. and Burke, D.S., 2005. Bushmeat hunting, deforestation, and prediction of zoonotic disease. *Emerging infectious diseases* 11(12): 1822.

World Bank, 2015. The Economic Impact of Ebola on Sub-Saharan Africa: Updated Estimates for 2015. Washington D.C: World Bank.

World Bank, 2020. Global Economic Prospects, June 2020. Washington D.C.: World Bank.

World Bank Ecofys, 2018. State and trends of carbon pricing 2018. Washington D.C.: World Bank.

World Economic Forum, 2017. Commodities and Forests Agenda 2020: Ten priorities to remove tropical deforestation from commodity supply chains. Report published by the Tropical Forest Alliance 2020 based on research by Climate Focus within the framework of the Tropical Forest Alliance. Davos: WEF.

World Economic Forum, 2020. The Global Risks Report, 2020. Davos: WEF (in partnership with Marsh & McLennan and Zurich Insurance Group).

World Resources Institute, 2018. By the Numbers: The Value of Tropical Forests in the Climate Change Equation. Washington D.C.: WRI.

Wright, R., 2019. The story of 2019: Protests in every corner of the globe. The New Yorker. (Available at: <u>https://www.newyorker.com/news/our-columnists/</u> <u>the-story-of-2019-protests-in-every-corner-of-theglobe</u>. [accessed on 1 September 2020]).

Wunder, S., Börner, J., Shively, G. and Wyman, M., 2014. Safety nets, gap filling and forests: a globalcomparative perspective. World Development 64: S29-S42.

WWF, 2012. Living Forests Report, Chapter 4: forests and wood products. Gland: WWF.

WWF, 2014. The Growth of Soy: Impacts and Solutions. Gland: WWF.

Yagoub, M., 2014. Doubling of Mexico Mining Losses Sign of Growing Criminal Involvement, InSight Crime (blog. Available at: https://www.insightcrime.org/news/ brief/doubling-of-mexico-mining-losses-sign-ofgrowing-criminal-involvement/ [accessed on 1 September 2020]).

Yin, H. and Li, C., 2001. Human impact on floods and flood disasters on the Yangtze River. *Geomorphology* 41(2-3): 105-109.

Zhang, Y. Z. and Holmes, E. C., 2020. A Genomic Perspective on the Origin and Emergence of SARS-CoV-2. Cell 181(2): 223–227.

Zhu, X., Linham, M.M. and Nicholls, R. J., 2010. Technologies for climate change adaptation: coastal erosion and flooding. Roskilde: UNEP Riso Centre on Energy, Climate and Sustainable Development.

Websites

- Centers for Disease Control and Prevention, 2020. Available at: https://www.cdc.gov/onehealth/basics/ index.html. [accessed on 10 June 2020]
- Centers for Disease Control. Ebola, 2019. Available at: https://www.cdc.gov/vhf/ebola/. [accessed on 5 August 2020]
- CIFOR, 2020. Forest News. COVID-19-led ban on wild meat could take protein off the table for millions of forest dwellers. Available at: https://forestsnews. cifor.org/64855/covid-19-led-ban-on-wild-meatcould-take-protein-off-the-table-for-millions-of-forest-dwellers?fnl=en. [accessed on 5 August 2020]
- DIME, 2020. Available at: https://www.worldbank.org/en/ research/dime. [accessed on 5 August 2020]
- Forest Trends, 2020. State of Voluntary Carbon Markets. Available at: https://www.forest-trends.org/publications/?filter=state+of+voluntary+carbon+market. [accessed on 10 June 2020]
- Global Impact Investing Network, 2020. Available at: https://thegiin.org/. Accessed on [15 February 2020]
- Global Forest Watch, 2014. World Resources Institute. www.globalforestwatch.org. [accessed on 7 August 2020]
- Goldman Environmental Prize, 2020. https://www.goldmanprize.org/blog/an-app-to-combat-illegal-logging. [accessed on 5 August 2020]
- If Not Us then Who?, 2020. Available at: https://ifnotusthenwho.me. [accessed on 12 August 2020]
- Livelihood Carbon Funds, 2020. Available at: https:// www.livelihoods.eu/LCF. [accessed on 7 August 2020]
- OECD, 2020. https://stats.oecd.org. [accessed on 13 February 2020]
- Technoserve, 2020. Available at: https://www.technoserve.org/our-work/#how-we-work. [accessed on 7 August 2020]
- Verra Registry Database, 2020. Available at: https://registry.verra.org/app/search/CCB . All projects in Africa, Asia, Latin America, and Oceania including low- and middle-income countries; Excluding projects with statuses Withdrawn, Validation expired, and Verification expired. [accessed on 7 August 2020]
- WHO, 2020a. Available at: https://www.who.int/topics/ zoonoses/en. [accessed on 1 September 2020]
- WHO, 2020b. Available at: https://www.who.int/emergencies/diseases/novel-coronavirus-2019. [accessed on 5 October]



Chapter 7

Conclusion: Forests, Trees and the Eradication of Poverty

Coordinating Lead Authors: Daniel C. Miller

Lead Authors: Stephanie Mansourian, Jennifer Zavaleta Cheek, Mónica Gabay, Reem Hajjar, Pamela A. Jagger, Judith Kamoto, Peter Newton, Johan A. Oldekop, Onja H. Razafindratsima, Priya Shyamsundar and Christoph Wildburger

TABLE OF CONTENTS

7.1.Overview	216
7.2 Key Messages and Implications for Decision-Makers	216
7.3 Knowledge Gaps and Research Priorities	221
7.4 Transformation toward Sustainability and the Eradication of Poverty	224
7.5 References	225

7.1 Overview

This Global Forest Expert Panel has assessed the scientific evidence on how forests and tree-based systems⁹ affect poverty. Our focus has been on the dynamic relationship between forests and poverty, seeking to understand the complexities inherent to this relationship. We considered the multiple dimensions of poverty and the extent to which forests and trees can contribute to its alleviation and, ultimately, eradication (Chapters 2 and 3). In Chapter 4, we explored contextual dimensions that frame forest-poverty dynamics and in Chapter 5 we identified policy and management measures ('levers') that influence how this relationship may evolve. By exploring some of the major global changes now shaping our world (Chapter 6), we also identified key opportunities and challenges for forests and tree-based systems in addressing poverty.

In this chapter, we summarise the key findings that have emerged from our assessment and discuss their implications for decision-makers (Table 7.1). We also identify critical gaps in knowledge (Table 7.2) and then conclude by reflecting on the future for forests, trees and poverty in light of contemporary challenges and opportunities created by the 2020 global pandemic, and major social and political changes taking place in countries around the world. The findings and discussion distilled in this chapter can help inform relevant national and international policy processes, including efforts to achieve the first goal of the 2030 Agenda for Sustainable Development: "End poverty in all its forms everywhere" (UN, 2020). They are also relevant to civil society actors, donor organisations, the private sector and the research community.

7.2 Key Messages and Implications for Decision-Makers

Key Message 1 Forests and trees are critical to global efforts to end poverty

Forests and tree-based systems are essential to global efforts to alleviate and ultimately eradicate poverty as embodied in the United Nations Sustainable Development Goals. Globally, more than 1.6 billion people live within 5km of a forest, more than two-thirds of whom reside in *lowand middle-income countries* (LMICs; Newton *et al.*, 2020). This population includes approximately 250 million of the world's extreme poor, who often rely directly on the goods and services that forests and trees provide. For forest-adjacent communities in many tropical countries, forests contribute 20-25% of household income – roughly the same amount as agriculture (Angelsen *et al.*, 2014). Trees outside forests managed in a range of *agroforestry* systems also contribute substantially to income and other aspects of *human well-being* in contexts around the world (Waldron *et al.*, 2017; Miller *et al.*, 2020).

In addition to the current contribution of forests and tree-based systems to global *poverty alleviation* efforts, their continued existence and health is crucial to retaining future options to support human well-being. Simply put, the loss of forests and trees threatens our planet and our ability to end poverty - and undermines efforts to ensure that poverty does not re-emerge.

The role of forests and tree-based systems in achieving Sustainable Development Goal 1 is multifaceted. Forest and tree-related resources are especially important in supporting the well-being of people in rural communities and in allowing them to manage risk. However, these resources can also help people move out of poverty in some circumstances, including directly through the sale of forest and tree products, and indirectly through the enhancement of soil fertility, water regulation and the provision of other *ecosystem services* supporting food production and other *livelihoods* needs.

Human well-being (and its converse, poverty) is now widely understood as a complex, multidimensional state that goes beyond income measures (Alkire and Santos, 2014; UN, 2015; World Bank, 2018). This broader understanding of poverty means forests and trees are important in addressing SDG 1 beyond income-based measures. Forests and tree-based systems provide both tangible and intangible inputs to household well-being, including the material aspects of people's lives such as energy, health, housing, income and nutrition and non-material aspects such as community relations and trust and those relating to culture and spirituality. These manifold contributions often occur outside formal markets and so are excluded from national income accounts. As a result they are frequently overlooked in development policy discussions, a missed opportunity at best and a reason for disastrous policy outcomes at worst (Scott, 1998; Seymour and Busch, 2016).

Forests and trees also play a crucial role in risk management such that the poor do not sink even
deeper into poverty and the non-poor avoid impoverishment. They perform this function by smoothing consumption and income across seasons and years through the provision of food, fodder, fuel and other products that may be consumed at home or sold. This role for forests and trees is especially relevant to the rural poor because they often do not have access to other forms of insurance and rely on livelihood activities that are subject to external shocks such as crop-raiding or variable weather (Noack *et al.*, 2019). Risk management is becoming even more critical in the face of the growing impact of climate change and other global shocks such as the COVID-19 pandemic.

The extent to which forests and trees can spur movement out of poverty and the conditions under which this role is possible, including in relation to other poverty alleviation measures, are not yet sufficiently researched (see Chapters 2 and 3, and below). Understanding the pathways that guide these forest-poverty dynamics is necessary for future interventions that consider forests as a tool for poverty alleviation.

Key Message 2 Benefits from forests and trees to human well-being are unevenly distributed

Although forests and tree-based landscapes are generally important to poverty alleviation, the benefits they bring are not distributed evenly and are shaped by many factors that vary according to time, geography, and social, economic and political contexts. These factors may constrain or enable the ability of forests and trees to alleviate poverty in a way that is effective, just and sustainable. Understanding these differences ensures that the role of forests in poverty alleviation can be better tailored to a given situation. Unless such distinctions are taken into account, forests and trees risk being ignored as assets that can contribute to poverty alleviation, thereby increasing the risk that they will be further degraded or destroved.

There are fundamental temporal dimensions to the relationship between forests and poverty alleviation. Trees take a long time to grow and yield benefits for people. Yet, mature forests can quickly be lost through logging or fire. There are tradeoffs between short-term livelihood imperatives and the long timescales intrinsic to forest growth. Trade-offs also exist related to the provision of different goods and services over time (e.g. the choice of tree species may determine whether short term soil benefits for agriculture are provided or longterm climate benefits). Trade-offs may also exist between near-term poverty alleviation goals and eradicating poverty over the long term, which implies a need for sustainable management so that forests and trees are able to provide ecosystem goods and services into the future to help prevent the recurrence of poverty. Who defines these tradeoffs and the mechanisms through which they are adjudicated are fundamentally political decisions.

Although there is evidence that forests support poor people to improve their well-being or mitigate risks, the role of forests to move people permanently out of poverty is much less well documented. For example, evidence to date on *non-timber forest products* (NTFPs) suggests that they can contribute to poverty alleviation, but the impacts have largely been at small scales and questions remain about whether they can persist over time.

The contribution of forests and tree-based landscapes also varies across space. Forests and trees matter differently in different places and scales within and across countries. For example, the role of forests in energy for heating and building materials is increasingly prioritised as 'green' in Europe, but in Africa and many LMICs, discussion of this aspect of forests often focuses on negative impacts on human health and the environment. This variation in the real and perceived role of forests differs widely within countries as well. While the overall contribution of forests or trees on farms to household incomes may be relatively small at a national scale, it can be very significant to households in more forested regions of countries. There are also important instances where the converse is true. In many forest and wildlife-rich countries in Africa, timber and tourism are major contributors to national accounts but the benefits may not accrue at the local level - and, worse, local communities may bear the cost of these activities through environmental degradation and restricted access to forest protected areas (see Key Message 4 below). In general, the evidence suggests that the poor are rarely able to capture the bulk of benefits from forests even as forest and trees are vital to them in terms of subsistence and food security.

This inequality extends to relations between the wealthy, mostly northern countries and LMICs. Historically, the timber sector contributed to the economic development of many countries in northern Europe and North America that are now among the world's wealthiest nations. These same countries continue to benefit from the trade in timber harvested in the LMICs. Power imbalances within and between countries constrain the ability of the poor to leverage forest resources to exit poverty. Indeed, investment by major financial actors based in high income countries is a major driver of *deforestation* in LMICs, further fuelling inequalities (Galaz *et al.*, 2018). Markets for commodities such as timber, beef, soy and palm oil will continue to grow and, in principle, could offer potential for poverty alleviation at the forest agricultural frontier. Available evidence, however, suggests that income flows tend to favour owners of land and capital, including large corporations, without reaching the poor – or, worse, coming at the expense of their livelihoods through dispossession. Further, production of these commodities may lead to deforestation, limiting future contributions of forests.

Beyond time and geography, context also matters to the distribution of benefits and costs from forests and tree-based landscapes. Gender, race¹⁰ and economic class, among other social differences, shape forest and tree use. So, too, does political context, with processes and outcomes varying in different national political systems. Political institutions and land use policies have been shown to be key factors when forests and trees produce negative externalities, such as crop-raiding, livestock predation and infectious disease transmission from wildlife to humans. More generally, governance factors play a decisive role in forest-poverty dynamics and the effectiveness of forest and treebased poverty alleviation interventions (see Key Message 5).

Key Message 3 Forests and trees can contribute to the well-being of the poor as they face profound global changes

Our world is in the throes of profound global challenges, which are affecting the most vulnerable and poorest members of society in the harshest ways. More extreme and frequent weather events associated with climate change, widening inequality, concentration of political power in fewer hands and the spread of infectious diseases, among others, exacerbate an already tenuous situation for the poor. Given these threats, forests and trees can be a lifeline. While forests and trees do not offer a 'silver bullet' for securing or stabilising well-being, the poor have been able to harness forest goods and services to manage and mitigate risk, particularly where market access and public service provision is limited.

The multiple risks posed by global changes to the *forest-reliant* poor threaten opportunities for poverty alleviation as poor households have limited capacity or resources to tackle new challenges or take advantage of new opportunities. Forests and trees may play an important role in strategies to reduce risks and improve conditions that would allow poor households to move out of poverty. As such, they merit special attention from policymakers concerned with poverty alleviation.

Several avenues exist for decision-makers to enhance use of forests and trees for risk management. Financing reforms in the commodity supply chain can help to strengthen transparency and develop capacity for smallholders to access global value chains, thus redressing much of the inequality inherent to the current systems. Investments in climate adaptation can contribute to reducing exposure to natural disasters and stabilising ecosystem services for the benefit of the rural poor. Strengthening rights to land and political participation particularly of indigenous peoples – is a priority in many parts of the globe. There are also opportunities for integrated actions that can address health and environmental change such as buffer areas between agriculture and livestock production and forests, wildlife and human disease surveillance, and alternatives to wildlife trade.

Key message 4

Uninformed and poorly aligned forest and land use policies and programmes may lead to excessive costs being borne by the poor

Forest and land use policies and programmes can impose particular costs on the rural poor who may rely on forests and trees to support their well-being and mitigate risk. Exploitation of these natural assets can have direct repercussions on the livelihoods of the poor. For example, timber extraction and forest-based tourism are major contributors to national economies in many countries, but local communities may bear the cost of these activities through degradation of their environment and restricted access to valuable forest area. Measures such as tenure reforms may help the rural poor to secure access to forest resources. However, such reforms have also been found to increase conflicts and tension in cases where market forces have supported elite capture of forest resources. In some instances, more stringent procurement policies may have served to squeeze out smallholders, thus leading to an increase in their poverty levels. Finally, while tree planting may bring many benefits, it may also affect the rural poor by reducing available agricultural land.

As illustrated in the examples above, the evidence suggests that the poor are rarely able to cap-

¹⁰ This term is used here as per the Universal Declaration of Human Rights.

ture the bulk of benefits from forests even as forest and trees are vital to them in terms of subsistence and food security. That the impacts of many forest and tree-based interventions have been unequal may help to explain why they have had limited effects in alleviating poverty.

Key Message 5

Policy and management measures exist to enable forests and trees to effectively address poverty goals even as there is no 'one size fits all' solution

Nobel Prize-winning economist Elinor Ostrom (2007) famously stated that "there are no panaceas" to address the world's sustainability challenges. We confirm this observation for the role of forests and trees in poverty alleviation, finding that numerous natural resource and forest sector policies, programmes, technologies and strategies contribute to addressing poverty. For example, strong evidence exists of a role for forests in poverty alleviation with respect to ecotourism, protected areas - particularly those with ecotourism opportunities – community forest management and agroforestry. Together, such interventions can help to maximise benefits while minimising costs to the poor and other segments of society. Yet, until the forest sector is recognised as a sector that plays a positive role in poverty alleviation, these 'levers' will remain under-utilised.

The review carried out through this GFEP revealed much more evidence on the effect of different forest policy levers on poverty mitigation (i.e., supporting or improving well-being) than on *poverty eradication*. Evidence on the contribution of interventions related to community forest management, forest producer organisations, small and medium scale forest enterprises, *payments for ecosystem services*, and tree crop production was especially positive. Evidence also suggests that interventions supporting stronger tenure and property rights over forest and tree resources are especially important for addressing different dimensions of poverty and support the effectiveness of other levers.

Overall though, the evidence for the impact of the various forest and tree-related levers was mixed, highlighting the importance of contextual differences. Further, many studies show that several levers have contributed to poverty alleviation under specific circumstances and in the presence of key enabling factors, including in conjunction with other levers. This implies that multiple interventions both within and outside the forest sector are needed in tandem, with individual measures alone often likely to struggle to effectively alleviate poverty (Shyamsundar *et al.*, 2020).

Multi-pronged approaches need to pay special attention to inequalities. A number of reviewed studies highlighted the importance of social heterogeneity in the context of the levers that most strongly favour poverty alleviation. Socially differentiated outcomes, including variable opportunities, benefits and trade-offs, result from a combination of underlying material and sociocultural inequalities, and the failure of a given lever to sufficiently account for and address those inequalities. Decision-makers must therefore embrace complexity and carefully consider the context when designing, funding and implementing policies and programmes related to forests and treebased systems. Learning lessons from prior interventions in contexts of interest may be especially instructive. Special attention is needed to consider those who bear the cost or may be left behind in certain policy choices (see Key Message 4).



Tree-based landscapes are sources of socio-economic and ecological benefits (Kenya) Photo © Jennifer Zavaleta Cheek

Table **7.1**

Summary table of conclusions and implications for decision-makers		
KEY MESSAGE	IMPLICATIONS FOR DECISION-MAKERS	CHAPTER REFERENCE
1. Forests and trees are critical to global efforts to end poverty	Recognise the important contribution that forests and trees do and can make to the well-being of hundreds of millions of people around the world, particularly those in rural contexts. The role of forests and tree-based systems is especially crucial for ensuring progress in reducing poverty and in stopping its long-term persistence.	Chapters 2, 3 and 6
2. Benefits from forests and trees to human well-being are unevenly distributed	Differential benefits of forest management must be recognised and steps taken to mitigate patterns of inequality. In particular, forest-related poverty interven- tions should be carefully designed to try to ensure more equitable impacts across social groups and geographic areas. There is a particular need to consider inequalities at multiple scales, including within communities, within countries and across countries. Priority should be given to the forest-reliant poor in such policies, including strengthened self-governance and technical skills.	Chapters 2-5
3. Forests and trees can contribute to the well-being of the poor as they face profound global changes	Recognise forests and trees as a valuable yet often overlooked asset for the poor. Prioritise policies to conserve and sustainably manage forests so they can benefit the poor. Supporting more secure property rights for the forest-reliant poor is especially important.	Chapters 3, 5 and 6
4. Uninformed and poorly aligned forest and land use policies and programmes may lead to excessive costs being borne by the poor	The same policies and programmes that can contribute to poverty alleviation may, if poorly designed or mis-aligned with other measures, exacerbate poverty. Decision-makers need to ensure that policies and programmes in the land use and forest sectors explicitly consider poverty alleviation. Alignment between sectoral policies and programmes can improve the application of multiple levers in tandem, thus increasing their efficiency in alleviating poverty.	Chapters 3-6
5. Policy and management measures exist to enable forests and trees to effectively address poverty goals even as there is no 'one size fits all' solution	There are several measures in the forest sector that can contribute to optimising the role that forests and tree- based systems play in alleviating poverty. Different measures will apply in different contexts and may need to be adapted over time. The combination of multiple measures is likely to yield better results.	Chapters 5 and 6

7.3 Knowledge Gaps and Research Priorities

This report has addressed the need for a global synthesis on contributions of forests and treebased landscapes to alleviating poverty. It has summarised the best available scientific evidence on forests, trees, and poverty dynamics and outcomes. In so doing, it reveals several important gaps in knowledge. Based on these gaps, we highlight five major research priorities that require urgent attention if forests and tree-based systems are to realise their potential in the struggle to end poverty.

Research priority 1 Examine forest-poverty dynamics, especially over the long term

Knowledge of how, and the extent to which, forests and trees can help the poor to permanently escape poverty – and those who are not poor to avoid slipping into poverty – stands as a particularly acute research need.

The literature on the relationship between forests, trees and poverty has focused on cross-sectional analyses or snapshots in time. Relatively few studies examine the dynamics of this relationship, particularly over long time periods (e.g. beyond at least 5 years) and across different contexts and spatial scales. Studies examining forests or trees as a primary pathway out of poverty or, conversely, as generating significant negative externalities are especially rare, limiting our ability to draw general conclusions about forest-poverty dynamics. This gap may reflect a lack of attention to both the influence of international investment and trade on the allocation of benefits from forests and the influence of rules governing access to forests on the global distribution of prosperity. More research on the relationship between forests and inequality within and across settings is therefore needed.

New data sources, including satellite imagery of land use and land cover change and spatially explicit socio-economic datasets, present promising opportunities to address current knowledge gaps on forest-poverty dynamics. Research using these and other data is crucial to allow decision-makers to compare the poverty alleviation aspects of forests and trees with their other benefits, such as *biodiversity* conservation and climate change *mitigation*, and to consider trade-offs between different land use policies. Such research can also inform policymakers of the potential for forests and trees to provide goods and services, manage risk and provide a pathway out of poverty compared with other levers for poverty alleviation.

Research priority 2 Expand the evidence base to cover under-represented geographies and contexts

Major geographic limits exist in our knowledge of the contribution of forests and tree-based landscapes to poverty alleviation. Nearly half of the current evidence base on forests-poverty linkages comes from just five countries: Bangladesh, Brazil, China, India and Nepal (Cheng *et al.*, 2019). Evidence on agroforestry and poverty also exhibits geographical bias (Miller *et al.*, 2020). Our understanding of forest-livelihood linkages in Europe, North America, and West and Central Africa are especially limited.

Three other dimensions of geography stand out as particularly important to advancing our knowledge of forest-poverty dynamics. First, the majority of our knowledge is based on studies of forests and poverty in tropical forest ecosystems. Woodlands, dryland and boreal forests have received less attention in scholarly and policy debates. Second, forests and trees are not only important in rural contexts, but also in urban landscapes, which are rapidly expanding. Attention to forest poverty dynamics in urban settings is critical given demographic trends towards urbanisation in many LMICs. Finally, there is a need to investigate the contribution of forests and trees to poverty alleviation at spatial scales beyond the site or community level giving more attention to landscape, sub-regional, national and regional scales. Moreover, multi-scale analysis is critically important for understanding spillovers and the aggregate effects of policies and programmes.

Beyond new empirical work to address these knowledge gaps, we highlight the general need to update data collection done in recent systematic maps and reviews on this topic (e.g. Cheng *et al.*, 2019; Miller *et al.*, 2020). Ideally, cost-effective means of ensuring regular updates to data repositories can be developed and implemented so that decision-makers have access to the latest relevant evidence.

Research priority 3 Bring to light the 'hidden dimensions' of forest contributions to poverty alleviation

The role that forests and trees play in supporting human well-being and addressing poverty is often indirect, intangible or otherwise not captured in markets or official statistics. For example, there is a great deal of domestic and international trade in NTFPs, but statistics to quantify the economic contribution, value addition and supply chains related to these goods are largely absent. Further, trade in timber and non-timber forest products that is either illegal or moving into markets through informal channels means that the full economic contribution of forests is not recorded in government statistics used for designing policy. Beyond marketed goods provided by forests and trees, there is growing evidence that forests and trees on farms provide ecosystem services that increase agricultural productivity, a benefit rarely entering into policy discussions or national accounts.

There is a need to both improve the evidence base on this topic and to better communicate current knowledge about these positive, yet often hidden, dimensions of forests and tree-based systems.

Research priority 4 Assess the relative effectiveness of different forest-related policy and management interventions for poverty alleviation

This report consolidates current evidence on the effectiveness of different forest- and tree-related policies and programmes in addressing the challenge of poverty. However, major gaps in our knowledge remain. Foremost among these is knowledge of which kinds of levers are likely to be most effective in which contexts and why. We know that, regardless of context, forest policy effectiveness hinges on factors such as responsive macro-institutional frameworks, collaborative processes, clear performance indicators and monitoring systems, and adaptive management and learning (Agrawal et al., 2018). Yet, we lack clear guidance on which, among the range of policy choices, from incentive-based mechanisms like payments for ecosystem services to regulatory mechanisms like strict protected areas, may be most appropriate for a given situation. Research is beginning to address this topic (e.g. Sims and Alix-Garcia, 2017), but more work is needed to build evidence, advance theory and inform policy.

Policy and programme development is increasingly evidence-based, raising demand for research that analyses causal impact. The broad field of environmental and natural resource policy continues to lag behind other policy fields (e.g. education, health, social policy) in building a rigorous evidence base that sheds light on the impact and cost-effectiveness of various policy options (Ferraro and Pattanayak, 2006; Börner et al., 2020). Rigorous studies on the poverty impacts of for-

est (Cheng et al., 2019) and agroforestry (Miller et al., 2020) interventions remain relatively uncommon, with randomised control trials especially rare. Studies that use methods with high potential for establishing causal linkages between interventions and outcomes for forests and poverty typically include baseline or pre-intervention and endline data, collection of data over time, and collection of data for control or comparison groups. Impact evaluation studies that draw on well informed theories of change, use qualitative and quantitative data, give focused attention to heterogeneous impacts across different social groups, geographies and contexts, and provide a cost-effectiveness analysis, are of highest value to decision-makers.

Research priority 5 Identify key barriers to more equitable, just and sustainable use of forests and trees, and ways to overcome them

This report has highlighted that forests and treebased systems provide many benefits to different social groups, but that benefits and costs are frequently unequally distributed. There is an urgent need to rigorously synthesise available evidence and then conduct new empirical investigation into barriers to a more just, equitable and sustainable distribution of benefits and costs. Doing so will require analysis of the political economy of forest and land use policies within and across countries. More generally, research must seek to illuminate how policies that contribute to deforestation and forest degradation, limiting the poverty alleviation potential of forests and trees persist, or are prioritised. Research focused on uncovering the underlying causes preventing the full potential of forests and trees to contribute to poverty alleviation is called for together with efforts to understand how these causes may be effectively addressed to support better outcomes, especially for financially poor and politically and socially marginalised populations.



Forests are intimately linked to the culture and daily life of millions of people, including this Kayapó boy in the Brazilian Amazon Photo © Peter Newton

	Table 7.2	
Research priorities and key questions		
RESEARCH PRIORITY	KEY QUESTIONS	
1. Examine forest-poverty dynamics, especially over the long term	 Under what circumstances can forests and tree-based systems provide a pathway out of poverty? How do different types of forests and tree-based systems affect poverty outcomes over time? And across different spatial scales? How do these outcomes compare with other outcomes, like biodiversity conservation and climate change mitigation? What are the trade-offs and synergies over time and across space for these different outcomes? 	
2. Expand the evidence base to cover under-represented geographies and contexts	 What contributions do forests and tree-based systems make to poverty in comparatively understudied regions like Europe, North America and West Africa? What is their contribution in urban contexts? What insights might we gain from investigating this topic in both low- and middle-income countries and high-income countries? What cost-effective means can be developed to ensure an updated, easily accessible evidence base for decision-makers, researchers and the public? 	
3. Bring to light the 'hidden dimensions' of forest contri- butions to poverty alleviation	 What is the contribution of NTFPs to poverty alleviation? Can more accurate estimates of the contribution of informal and illegal trade in timber and other forest products be made? What is the contribution of forests and trees on farms to agriculture? How do intact forests affect human health? 	
4. Assess the relative effectiveness of different forest-related policy and management interventions for poverty alleviation	 Which interventions are most effective in alleviating poverty, in which contexts? What trade-offs do they imply? How do different factors mediate and moderate impacts? What explains heterogeneity of impacts? How can costs and benefits of different policy options be more equitably distributed? 	
5. Identify key barriers to more equitable, just and sustainable use of forests and trees, and ways to overcome them	 How do forest and tree-related market supply chains affect the poorest? What are the barriers to more equitable, just and sustainable distribution of the benefits and costs of forests and tree-based systems? What are the opportunities for overcoming them and how? 	

7.4 Transformation toward Sustainability and the Eradication of Poverty

Forests and tree-based systems have an important role to play in contributing to one of the greatest challenges of our time: alleviating poverty in a way that is just and sustainable over long periods. Yet, forests and trees are often overlooked in policy debates and action relating to poverty in all its various dimensions. The findings of this report suggest that the relative neglect of these natural resources must change to make enduring progress toward the ambitious targets articulated under SDG 1. It is now time for forests and tree-based systems to take a more central role, particularly in relation to poverty indicators that go beyond the minimum international poverty line. As governments around the world seek to develop responses to the COVID-19 pandemic and retool their economies in the face of climate change and widening inequality, forests and trees are a foundational asset. This report clarifies the multifaceted ways in which this is so.

The importance of forests and tree-based systems to rural livelihoods and poverty alleviation means that their conservation, sustainable management and restoration are paramount. Eliminating extreme poverty while preventing people from falling into poverty will require sustainable management of the resources and services that forests and tree-based systems provide. Development policies intended to alleviate poverty must therefore consider their effects on forests and tree-based systems. Such policies should seek to mitigate damage to forests and trees while taking advantage of the opportunities that forests and tree-based systems present for advancing poverty alleviation goals.

This time of tumult and change presents major challenges for poverty and sustainability goals but it also opens up opportunities to revitalise existing and develop new approaches to make progress toward the SDGs. In this context, forests and treebased systems are one important building block to support transformational change. Responses to the COVID-19 pandemic, in particular, will shape options for forests to sustainably contribute to poverty alleviation and human well-being, as well as deliver other benefits such as biodiversity conservation, climate change mitigation and adaptation, and reduction of threats from infectious disease.

Future major shocks such as regional or global pandemics and natural disasters are to be expected, and integrating natural assets such as forests into an enduring response to these shocks should be a priority. Such a response will require transformational change that includes a marked increase in our understanding and capacity to manage forests and tree-based systems for the benefit of all so that they may contribute to the global challenge of eradicating poverty.

7.5 References

Agrawal, A., Hajjar, R., Liao, C., Rasmussen, L.V. and Watkins, C., 2018. Editorial overview: Forest governance interventions for sustainability through information, incentives, and institutions. *Current Opinion in Environmental Sustainability* 32: 1-7.

Alkire, S. and Santos, M. E., 2014. Measuring Acute Poverty in the Developing World: Robustness and Scope of the Multidimensional Poverty Index. World Development 59: 251-274.

Angelsen, A., Jagger, P., Babigumira, R., Belcher, B., Hogarth, N. J., Bauch, S., Börner, J., et al., 2014.
Environmental income and rural livelihoods: A global-comparative analysis. World Development 64 (Supplement 1): S12-S28.

Börner, J., Schulz, D., Wunder, S. and Pfaff, A., 2020. The Effectiveness of Forest Conservation Policies and Programs. Annual Review of Resource Economics 12.

Cheng, S.H., MacLeod, K., Ahlroth, S., Onder, S., Perge, E., Shyamsundar, P., Rana, P., et al., 2019. A systematic map of evidence on the contribution of forests to poverty alleviation. *Environmental Evidence* 8(1): 3.

Ferraro, P.J. and Pattanayak, S.K., 2006. Money for nothing? A call for empirical evaluation of biodiversity conservation investments. PLoS Biol. 4(4): e105.

Galaz, V., Crona, B., Dauriach, A., Scholtens, B. and Steffen, W., 2018. Finance and the Earth system– Exploring the links between financial actors and non-linear changes in the climate system. *Global Environmental Change* 53: 296-302.

Miller, D. C. and Hajjar, R., 2020. Forests as pathways to prosperity: Empirical insights and conceptual advances. World Development 125: 104647.

Miller, D.C., Ordoñez, P.J., Brown, S.E., Forrest, S., Nava, N.J., Hughes, K. and Baylis, K., 2020. The impacts of agroforestry on agricultural productivity, ecosystem services, and human well-being in low- and middleincome countries: An evidence and gap map. *Campbell Systematic Reviews* 16(1): 1-35.

Newton, P., Kinzer, A., Miller, D.C., Oldekop, J.A. and Agrawal, A., 2020. The number and spatial distribution of forest-proximate people globally. *One Earth* 3(3): 363-370.

Ostrom, E., 2007. A Diagnostic Approach for Going Beyond Panaceas. PNAS 104(39): 15181–15187.

Scott, J.C., 1998. Seeing like a State. New Haven: Yale University Press.

Seymour, F. and Busch, J., 2016. Why Forests? Why Now?: The Science, Economics, and Politics of Tropical Forests and Climate Change. Washington D.C.: Center for Global Development.

Shyamsundar, P., Ahlroth, S., Kristjanson, P. and Onder, S., 2020. Supporting pathways to prosperity in forest landscapes – A PRIME framework. World Development 125: 104622. Sims, K.R.E. and Alix-Garcia, J.M., 2017. Parks versus PES: evaluating direct and incentive-based land conservation in Mexico. *Journal of Environmental Economics and Management* 86: 8–28.

United Nations. 2015. Sustainable development goals. New York: United Nations. (see: https:// sustainabledevelopment.un.org/sdgs, [accessed on 1 September 2020]).

Waldron, A., Garrity, D., Malhi, Y., Girardin, C., Miller, D.C. and Seddon, N. 2017. Agroforestry can enhance food security while meeting other Sustainable Development Goals. *Tropical Conservation Science* 10: 1-6.

World Bank, 2018. Poverty and Shared Prosperity 2018: Piecing Together the Poverty Puzzle. Washington D.C.: World Bank.



Appendix I Glossary of Terms and Definitions Used in the Assessment Report

Absolute poverty: A measure of poverty based on a fixed amount of money needed to meet basic needs such as food, clothing, and shelter (Haughton and Khandker, 2009).

Adaptation (climate): Adjustment in natural or human systems in response to actual or expected climatic stimuli or their effects, which moderates harm or exploits beneficial opportunities (Seppälä *et al.*, 2009)

Afforestation: Establishment of forest through planting and/or deliberate seeding on land that, until then, was not classified as forest (FAO, 2010). According to the definition used by the UNFCCC, afforestation can take place on land that has not been covered by forest for at least 50 years (see also '*Reforestation*').

Agroforestry: A collective name for land use systems and practices in which woody perennials are deliberately integrated with crops and/or animals on the same land management unit. The integration can be either in a spatial mixture or in a temporal sequence (World Agroforestry Centre, 2017).

Alleviation: See 'Poverty alleviation'

Biodiversity (Biological diversity): The variability among living organisms from all sources including, inter alia, terrestrial, marine and other aquatic ecosystems and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems. (CBD, Article 2).

Capabilities: The substantive freedoms a person enjoys to lead the kind of life he or she has reason to value (Sen, 1999). This concept merges the idea of freedom to do and achieve with functionings understood as what a given individual may value doing or being.

Deforestation: The conversion of forest to another land use or the long-term reduction of the tree canopy cover below the minimum 10% threshold (FAO, 2010). Deforestation implies the long-term or permanent loss of forest cover and implies transformation into another land use. Such a loss can only be caused and maintained by a continued human-induced or natural perturbation. Deforestation includes areas of forest converted to agriculture, pasture, water reservoirs and urban areas. The term specifically *excludes* areas where the trees have been removed as a result of harvesting or logging, and where the forest is expected to regenerate naturally or with the aid of silvicultural measures. Deforestation also includes areas where, for example, the impact of disturbance, overutilisation or changing environmental conditions affects the forest to an extent that it cannot sustain a tree cover above the 10% threshold (FAO, 2001).

Ecological restoration: Ecological restoration is the process of assisting the recovery of an ecosystem that has been degraded, damaged, or destroyed (SER, 2004).

Ecological resilience: The ability of a system to absorb impacts before a threshold is reached where the system changes into a different state (Gunderson, 2000).

Ecosystem: A dynamic complex of plant, animal and micro-organism communities and their non-living environment interacting as a functional unit (CBD).

Ecosystem services: Ecological processes or functions having monetary or non-monetary value to individuals or society at large. There are: (i) supporting services such as productivity or biodiversity maintenance, (ii) provisioning services such as food, fibre, or fish, (iii) regulating services such as climate regulation or carbon sequestration, and (iv) cultural services such as tourism or spiritual and aesthetic appreciation (MEA, 2005).

Equity (also its opposite: 'Inequity'): Refers to how *capabilities* (e.g. access to health, education and good nutrition) are distributed within a certain group of individuals (Mora and Muro, 2018). **Inequity** is the unequal distribution of capabilities (Sen, 1999).

Equality (also its opposite: 'Inequality'): The 'sameness' of a distribution of attributes, such as income or consumption, across a whole population (i.e. the state of being equal) (Haughton and Khandker, 2009; Harris and Nisbett, 2018). A popular measure of (in)equality is the Gini coefficient, which ranges from 0 (perfect equality) to 1 (perfect inequality), but is typically in the range of 0.3 to 0.5 for per capita expenditures.

Food security: A situation that exists when all people, at all times, have physical, social and economic access to sufficient, safe and nutritious food that meets their dietary needs and food preferences for an active and healthy life. Based on this definition, four food security dimensions can be identified: food availability, economic and physical access to food, food utilization and stability over time (FAO *et al.*, 2014). **Food insecurity**, by contrast, exists when people lack secure access to sufficient amounts of safe and nutritious food for normal growth and development and an active and healthy life. Food insecurity may be chronic, seasonal or transitory (FAO *et al.*, 2014).

Forest: Land spanning more than 0.5 hectares with trees higher than 5 metres and a canopy cover of more than 10 percent, or trees able to reach these thresholds *in situ*. It does not include land that is predominantly under agricultural or urban land use (FAO, 2010). Forests include both *natural forests* (sensu CPF, 2005) and *planted forests* (sensu FAO, see below). It also includes areas temporarily unstocked, e.g. after disturbance, that are expected to revert back to forest.

Forest-based poverty alleviation: Use of resources from forests and trees for the purpose of lessening deprivation of well-being on either a temporary or lasting basis (Sunderlin *et al.*, 2005).

Forest-based livelihood: For the <u>purposes of this report</u>, defined as "Deriving all or part of one's livelihood from the use of resources from forests and trees".

Forest-dependent (-reliant) people: For the <u>purposes of this report</u>, defined as "People that have a direct relationship with forests and trees, and live within or adjacent to forested areas, and rely on them for their subsistence and/or income".

Forest-proximate people: Refers to people who live in and around forests. The term captures the spatial relationship between people and forests without additional assumptions about the nature of the relationship between them as implied by the related term 'forest-dependent' people. (Newton *et al.*, 2020).

Forest ecosystem: A forest ecosystem can be defined at a range of scales. It is a dynamic complex of plant, animal and micro-organism communities and their abiotic environment interacting as a functional unit, where trees are a key component of the system. Humans, with their cultural, economic and environmental needs are an integral part of many forest ecosystems. (CBD: http://www.cbd.int/forest/definitions.shtml).

Forest degradation: The reduction of the capacity of a forest to provide goods and services. A degraded forest delivers a reduced supply of goods and services from a given site and maintains only limited biological diversity. It has lost the structure, function, species composition and/or productivity normally associated with the natural forest type expected at that site (ITTO, 2002).

Forest fragmentation (also 'habitat fragmentation'): Any process that results in the conversion of formerly continuous forest into patches of forest separated by non-forested lands. (CBD: http://www.cbd.int/forest/ definitions.shtml).

Forest landscape restoration: A planned process that aims to regain ecological integrity and enhance human well-being in deforested or degraded landscapes (WWF and IUCN, 2000; Mansourian *et al.*, 2005).

Forest management: The processes of planning and implementing practices for the stewardship and use of forests and other wooded land aimed at achieving specific environmental, economic, social and/or cultural objectives. Includes management at all scales such as normative, strategic, tactical and operational level management (FAO, 2004).

Forest restoration (see also 'Ecological restoration' and 'Forest landscape restoration'): The process of restoring a forest to its original state before degradation (same functions, same structure, same composition) (CPF, 2005).

Forests and tree-based systems: For the <u>purposes of this report</u>, this includes the spectrum from natural old-growth forests, to those managed to optimise resource yields, to the broad spectrum of agroforestry practices and to single-species tree crop management.

Governance: Interactive processes through which society, the economy, and the environment are steered towards collectively negotiated objectives (Ansell and Torfing, 2016). The concept includes both formal and informal rules and the public, private, and civil society actors that make and implement them (Hydén and Mease, 2004).

Habitat fragmentation: See 'Forest fragmentation'

High-Income Countries (HIC): A group of countries classified as high income based on gross national income per capita estimates using the World Bank Atlas method (World Bank, 2020). High-income economies are currently defined as those with a GNI per capita of USD 12,536 or more in 2019 (see also *Low- and Middle-Income Countries*).

Human well-being (also 'well-being'): A multidimensional concept capturing diverse ideas about what constitutes a 'good life' (McKinnon *et al.*, 2016). It can be defined simply as a positive physical, social and mental state (Summers *et al.*, 2012). Human well-being comprises the objective material circumstances of people's lives such as health, housing and income, social aspects such as community relations and trust, and a subjective dimension relating to how individuals view their own circumstances (OECD, 2017).

Livelihood: The assets (natural, physical, human, financial and social capital), activities and access to them (mediated by institutional and social relations) that together determine how an individual or household makes a living (Scoones, 1998). This definition emphasises means rather than outcomes of making a living whereas poverty is typically an outcome measure of livelihood performance (Sunderlin *et al.*, 2005).

Low- and Middle-Income Countries (LMIC; see also 'High -Income Countries'): A group of countries classified as low-income or middle-income based on gross national income per capita estimates using the World Bank Atlas method (World Bank, 2020). Low-income economies are currently defined as those with a GNI per capita of USD 1,035 or less in 2019. Middle-income countries consist of two groups: lower middle-income economies with a GNI per capita between USD 1,036 and USD 4,045 and upper middle-income countries with a GNI per capita between USD 4,046 and USD 12,535. In this report, classifications of countries by income is the preferred form to identify different country groupings, though other terms such as 'Global South' are sometimes used.

Mitigation (climate): An anthropogenic intervention to reduce the anthropogenic forcing of the climate system; it includes strategies to reduce greenhouse gas sources and emissions and enhancing greenhouse gas sinks (IPCC, 2007).

Non-timber forest products (NTFP): All biological materials other than timber, which are extracted from forests for human use. In addition to trees, forest products are derived from all plants, fungi and animals (including fish) for which the forest ecosystem provides habitat (IUFRO, 2005).

Nutrition: The consequence of the intake of food and the utilisation of nutrients by the body (CFS, 2012).

Nutrition security: A situation that exists when secure access to an appropriately nutritious diet is coupled with a sanitary environment, adequate health services and care, in order to ensure a healthy and active life for all household members. Nutrition security differs from food security in that it also considers the aspects of adequate caring practices, health and hygiene in addition to dietary adequacy (FAO *et al.*, 2014).

Pathway out of poverty: People are often considered to have moved out of poverty when they surpass a certain level of income or consumption (World Bank, 2018) or move beyond a certain locally defined state of poverty (Krishna, 2011). Pathways out of poverty are the means or mechanisms by which people surpass such locally, nationally or internationally defined poverty thresholds.

Payments for Ecosystem (or Environmental) Services (PES): A type of economic incentive offered for those that manage ecosystems (including agricultural lands) to improve the flow of environmental services that they provide. More formally, PES are "Voluntary transactions between service users and service providers that are conditional on agreed rules of natural resource management for generating offsite services" (Wunder, 2015). These transactions (viz., payments) incentives can be provided by anyone at local, regional, and global scales who benefits from the environmental services provided.

Planted forest: Forest predominantly composed of trees established through planting and/or deliberate seeding (FAO, 2010). Includes forests resulting from afforestation, reforestation and some forms of forest restoration (see also 'Afforestation', 'Ecological restoration', 'Forest restoration', 'Forest landscape restoration', 'Reforestation').

Plantation forests (also 'forest plantation'): Planted forests that have been established and are (intensively) managed for commercial production of wood and non-wood forest products, or to provide a specific environmental service (e.g. erosion control, landslide stabilisation, windbreaks, etc.) (Carle and Holmgren, 2003).

Policy instruments (measures): Set of techniques used by governmental and other authorities to try to ensure support and effect social change (Bemelmans-Videc *et al.*, 2010). Such instruments are designed to shape human behaviour and define people's legal rights through regulatory (e.g. prescriptions, proscriptions), financial (e.g. subsidies, taxation) and informational (e.g. education) means. A set of procedural policy instruments act to affect the policy process indirectly through institutional or organisational means (IUFRO, 2005).

Poverty: Deprivation or disadvantage that prevents an individual or group from attaining a certain level of well-being and participating fully in society (Smeeding, 2016; World Bank, 2001). This definition encompasses not only commonly used income or consumption measures of poverty but also a range of non-monetary attributes that directly affect people's capabilities and overall well-being and allow human capabilities to go unrealised (Sen, 1999; Alkire, 2002; World Bank, 2018).

Poverty alleviation: A lessening of deprivation or disadvantage such that well-being is improved. This lessening may include movement above a certain income or consumption threshold, such as international or country-specific poverty lines (termed **'poverty reduction'** or **'poverty elimination'**). It may also include a lessening in the degree of poverty experienced or avoiding falling into poverty (termed **'poverty mitigation'**) (World Bank, 2001; Sunderlin *et al.*, 2005).

Poverty eradication (also 'elimination'): The complete or near absence of people or households in poverty, indicated by the international poverty line or other measures. Implies permanent movement out of poverty by addressing the root causes of why people are impoverished (UN, 2020).

Poverty line: The minimum level of welfare before a person is no longer deemed to be 'poor' (Ravallion, 1998). This level of welfare is often defined in monetary terms, but can also be defined in other ways (e.g. caloric intake, happiness, etc.). There can be different poverty lines for different levels and measures of poverty, internationally and within countries which set their own poverty lines. As of October 2015, the **international poverty line is** USD 1.90 per day (World Bank, 2018).

Poverty reduction: Lessening the number of people who are considered to be poor. This reduction is typically measured as movement above a given poverty line defined in monetary terms (World Bank, 2001), but it could be tracked using measures of other dimensions of poverty. Often understood as a situation in which the poverty rate falls, more or less permanently, based on economic growth (World Bank, 2018).

Poverty trap: Any self-reinforcing mechanism that causes poverty to persist (Azariadis and Stachurski, 2005).

Primary forest: Naturally regenerated forest of native species, where there are no clearly visible indications of human activities [including commercial logging] and the ecological processes are not significantly disturbed (FAO, 2010).

Prosperity: The opposite of poverty. This concept includes multiple dimensions of individual human well-being but is also relational, implying concern for other people and species, now and in the future (Miller and Hajjar, 2020). It is similar to Sen's (1993) notion of 'capabilities for flourishing' with the addition that it includes explicit consideration of environmental sustainability "to flourish as human beings – within the ecological limits of a finite planet" (Jackson, 2009). It also encompasses views about equitable economic growth such as the World Bank's goal of "boosting shared prosperity in a sustainable manner" within each of its partner countries (World Bank, 2018).

Reforestation: Re-establishment of forest through planting and/or deliberate seeding on land classified as forest after a temporary period (<10 years) during which there was less than 10% canopy cover due to human-induced or natural perturbations (adapted from FAO, 2010). According to the definition used by the UNFCCC, reforestation can occur on land that was forested but that has been converted to non-forested land.

Relative poverty: A measure of poverty in relation to the economic status of other members of society using a cutoff point (Haughton and Khandker, 2009).

Resilience: The ability of a social or ecological system to absorb disturbances while retaining the same basic structure and ways of functioning, the capacity for self-organisation, and the capacity to adapt to stress and change (IPCC, 2007). See also *Ecological resilience*.

Restoration: See 'Ecological restoration'; 'Forest restoration'; 'Forest landscape restoration'

Secondary forest: Forests regenerating largely through natural processes after significant removal or disturbance of the original forest vegetation by human or natural causes at a single point in time or over an extended period, and displaying a major difference in forest structure and/or canopy species composition with respect to pristine primary forests (FAO, 2003).

Shifting cultivation: Also referred to as slash-and-burn cultivation or swidden agriculture. A land use system that employs a natural or improved fallow phase, which is longer than the cultivation phase of annual crops, sufficiently long to be dominated by woody vegetation, and cleared by means of fire (Mertz *et al.*, 2009)

Sustainable forest management: A dynamic and evolving concept, aims to maintain and enhance the economic, social and environmental values of all types of forests, for the benefit of present and future generations. The seven thematic elements of sustainable forest management are: (a) extent of forest resources; (b) forest biological diversity; (c) forest health and vitality; (d) productive functions of forest resources; (e) protective functions of forest resources; (f) socio-economic functions of forests; and (g) legal, policy and institutional framework. The thematic elements are drawn from the criteria identified by existing criteria and indicators processes, as a reference framework for sustainable forest management (UN, 2007).

Sustainable Livelihood: A livelihood that can cope with and recover from stresses and shocks, maintain or enhance its *capabilities* and assets, and provide net benefits to other livelihoods locally and more widely, both now and in the future, while not undermining the natural resource base (Chambers and Conway, 1991; Scoones, 1998).

Swidden cultivation: See 'Shifting cultivation'

Tenure: Systems of tenure define and regulate how people, communities and others gain access to land, fisheries and forests. These tenure systems determine who can use which resources, for how long and under what conditions. The systems may be based on written policies and laws, as well as on unwritten customs and practices (FAO, 2012).

Vulnerability (to poverty): The risk of a given person or group falling into poverty in the future, even if they are not necessarily poor now. This risk is often associated with the effects of 'shocks' such as a drought, a drop in farm prices or a financial crisis. Vulnerability is a key dimension of well-being since it affects individual behaviour in terms of investment, production patterns, and coping strategies, and in terms of the perceptions of their own situation (Haughton, and Khandker, 2009).

Well-being: See 'Human well-being'

References

- Alkire, S., 2002. Dimensions of Human Development. World Development 30: 181-205.
- Ansell, C., and Torfing, J., 2016. Introduction: Theories of governance. In: Handbook on theories of governance, edited by C. Ansell, C. and J. Torfing. Cheltenham and Northampton, MA: Edward Elgar Publishing, pp. 1-19.
- Azariadis, C. and Stachurski, J., 2005. Chapter 5 Poverty Traps. In: Handbook of Economic Growth, edited by: S. Bowles, S. Durlauf and K. Hoff. Princeton: Princeton University Press, pp. 295–384.
- Bemelmans-Videc, M.L., Rist, R.C. and Vedung, E.O. eds., 2010. Carrots, sticks, and sermons: Policy instruments and their evaluation. New Brunswick, NJ: Transaction Publishers.
- Carle, J. and Holmgren, P., 2003. Definitions related to planted forests. Working Paper 79, Forest Resources Assessment Programme, Forestry Department. Rome: Food and Agriculture Organization of the United Nations.
- CBD, 1992. Convention on Biological Diversity, Art. 2. Montreal: UNEP.
- CFS (Committe on World Food Security), 2012. Coming To Terms With Terminology. Rome: FAO.
- Chambers, R. and Conway, G., 1992. Sustainable rural livelihoods: practical concepts for the 21st century. IDS Discussion Paper No. 296. Sussex, UK: Institute for Development Studies.
- CPF, 2005. Third Expert Meeting on Harmonizing Forestrelated Definitions for Use by Various Stakeholders. <u>17-19</u> January 2005, Rome: FAO.
- FAO, 2012. Voluntary Guidelines on the Responsible Governance on Tenure of Land, Fisheries and Forests in the Context of National Food Security. Rome: FAO.
- FAO, 2010. Global Forest Resources Assessment. Forestry Paper 163. Rome: Food and Agriculture Organization of the United Nations.
- FAO, 2004. Global Forest Resources Assessment Update 2005 Terms and Definitions. Final version. Rome: Food and Agriculture Organization of the United Nations.
- FAO, 2003. Proceedings of the Workshop on Tropical Secondary Forest Management in Africa: Reality and Perspectives, Nairobi, Kenya, 9-13 December 2002. Forestry Department. Rome: Food and Agriculture Organization of the United Nations.
- FAO, 2001. Global Forest Resources Assessment FRA 2000 – Main report. Rome: Food and Agriculture Organization of the United Nations.
- FAO, IFAD and WFP, 2014. The State of Food Insecurity in the World 2014. Strengthening the enabling environment for food security and nutrition. Rome: FAO.
- Gunderson, L., 2000. Ecological resilience: in theory and application. Ann. Rev. Ecol. Syst. 31: 425-439.
- Harris, J. and Nisbett, N., 2018. Equity in social and

development studies research: insights for nutrition. In: Advancing equity, equality and nondiscrimination in food systems: Pathways to reform. United Nations System Standing Committee on Nutrition. https:// www.unscn.org/uploads/web/news/UNSCN-News43-WEB.pdf

- Haughton, J., and Khandker, S. R.. 2009. Handbook on poverty and inequality (English). Washington, DC: World Bank (available at: http://documents. worldbank.org/curated/en/488081468157174849/ Handbook-on-poverty-and-inequality [Accessed on 24 September 2020]).
- Hydén, G., and Mease, K., 2004. Making sense of governance: empirical evidence from sixteen developing countries. Boulder and London: Lynne Rienner Publishers.
- IPCC, 2007. Climate Change 2007: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Appendix 1: Glossary. Parry, M.L., Canziani, O.F., Palutikof, J.P., van der Linden, P.J. and Hanson, C.E. (eds.). Cambridge: Cambridge University Press. 976 p.
- ITTO, 2002. Guidelines for the Restoration, Management and Rehabilitation of Degraded and Secondary Tropical Forests. Policy Development Series No. 13. Yokohama: ITTO.
- IUFRO, 2005. Multilingual Pocket Glossary of Forest Terms and Definitions, Compiled on the occasion of the XXII IUFRO. World Congress August 2005, Brisbane, Australia. Vienna: IUFRO.
- Jackson, T., 2009. Prosperity without growth: Economics for a finite planet. London: Earthscan.
- Krishna, A., 2011. One illness away: Why people become poor and how they escape poverty. New York: Oxford University Press.
- Mansourian, S., Vallauri, D. and Dudley, N., 2005. Forest Restoration in Landscapes: Beyond Planting Trees. New York: Springer.
- McKinnon, M. C., Cheng, S. H., Dupre, S., Edmond, J., Garside, R., Glew, L., et al., 2016. What are the effects of nature conservation on human well-being? A systematic map of empirical evidence from developing countries. Environmental Evidence 5: 1–25.
- MEA (Millennium Ecosystem Assessment), 2005. Ecosystems and human well-being: synthesis. Washington DC: Island Press.
- Miller, D. C. and Hajjar, R., 2020. Forests as pathways to prosperity: Empirical insights and conceptual advances. *World Development* 125: 104647.Mora, A. and Muro, P., 2018. Political Economies: Inequality and malnutrition. In Advancing equity, equality and nondiscrimination in food systems: Pathways to reform. United Nations System Standing Committee on Nutrition. (available at: https://www.unscn. org/uploads/web/news/UNSCN-News43-WEB.pdf. [Accessed on 24 September 2020]).

- Newton, P., Kinzer, A., Miller, D.C., Oldekop, J.A. and Agrawal, A., 2020. The number and spatial distribution of forest-proximate people globally. *One Earth* 3(3): 363-370.
- OECD, 2017. How's life? 2017: Measuring well-being. Paris: OECD Publishing.
- Ravallion, M., 1998. Poverty lines in theory and practice. Washington, D.C.: The World Bank.
- Scoones, I., 1998. Sustainable Rural Livelihoods. A Framework for Analysis. IDS Working Paper No. 72. Sussex: Institute of Development Studies.
- Sen, A., 1993. Capability and Well-being. In: The Quality of Life, edited by M. Nussbaum and A. Sen. Oxford: Clarendon Press, pp. 30–53.
- Sen, A., 1999. Development As Freedom. New York: Anchor Books.
- Seppälä,R., Buck, A. and Katila, P. (eds.), 2009. Adaptation of Forests and People to Climate Change. A Global Assessment Report. IUFRO World Series Volume 22. Vienna: IUFRO.
- SER (Society for Ecological Restoration) Science & Policy Working Group, 2004. The SER International Primer on Ecological Restoration. Washington D.C.: Society for Ecological Restoration.
- Smeeding, T. M. 2016. Poverty Measurement. In: The Oxford Handbook of the Social Science of Poverty, edited by D. Brady and L. M. Burton. New York: Oxford University Press.
- Summers, J. K., Smith, L. M., Case, J. L. and Linthurst, R. A., 2012. A review of the elements of human well-being with an emphasis on the contribution of ecosystem services. Ambio 41: 327–340.

- Sunderlin, W.D., Angelsen, A., Belcher, B., Burgers, P., Nasi, R., Santoso, L. and Wunder, S., 2005. Livelihoods, forests, and conservation in developing countries: an overview. World development 33(9): 1383-1402.
- UN, 2007. Non-legally binding instrument on all types of forests. Note by the Secretariat 17 October 2007. 10 p.
- UN, 2020. Poverty Eradication | Poverty Eradication. (n.d.). (from: https://www.un.org/development/desa/ socialperspectiveondevelopment/issues/povertyeradication.html [accessed on: 7 May 2020]).
- UNEP, 2008. [Internet site]. What is Indigenous Knowledge? (Available at: http://www.unep.org/IK/default. asp?id=Home [accessed on 24 September 2020]).
- World Agroforestry Centre, 2017. Corporate Strategy 2017-2020 – Transforming lives and landscapes with trees. Nairobi: ICRAF.
- World Bank, 2001. World development report 2000/2001: Attacking poverty. Oxford & New York: Oxford University Press.
- World Bank, 2018. Poverty and Shared Prosperity 2018: Piecing Together the Poverty Puzzle, Washington, D.C.: World Bank.
- World Bank, 2020. World Bank Country and Lending Groups. (available at: https://datahelpdesk.worldbank. org/knowledgebase/articles/906519-world-bankcountry-and-lending-groups. [Accessed on 24 September 2020]).
- Wunder, S., 2015. Revisiting the concept of payments for environmental services. *Ecological economics* 117: 234-243.
- WWF and IUCN, 2000. Minutes from the Workshop on Forests Reborn in Segovia. Gland: WWF and IUCN.

Appendix II List of Panel Members, Authors and Reviewers

Panel Members and Authors

Arun Agrawal University of Michigan Ann Arbor, Michigan, USA E-mail: arunagra@umich.edu

Jennifer Alix-Garcia

Oregon State University Corvallis, Oregon, USA E-mail: jennifer.alix-garcia@oregonstate.edu

Maria Brockhaus University of Helsinki Helsinki, Finland E-mail: maria.brockhaus@helsinki.fi

Sarah E. Brown University of Illinois at Urbana-Champaign Urbana, Illinois, USA E-mail: saraheb3@illinois.edu

James T. Erbaugh

Dartmouth College Hanover, New Hampshire, USA E-mail: J.T.Erbaugh@dartmouth.edu

Dikshya Devkota University of Illinois at Urbana-Champaign Urbana, Illinois, USA Email: Dikshya.Devkota39@gmail.com

Mónica Gabay National University of San Martin Buenos Aires, Argentina E-mail: monagabay@yahoo.com

Reem Hajjar Oregon State University Corvallis, Oregon, USA E-mail: reem.hajjar@oregonstate.edu

Karl Hughes World Agroforestry (ICRAF) Nairobi, Kenya E-mail: K.Hughes@cgiar.org David Humphreys

The Open University Milton Keynes, UK E-mail: david.humphreys@open.ac.uk

Markus Ihalainen

Center for International Forestry Research (CIFOR) Dakar, Senegal E-mail: M.Ihalainen@cgiar.org

Pamela A. Jagger

University of Michigan Ann Arbor, Michigan, USA E-mail: pjagger@umich.edu

Gillian Kabwe The Copperbelt University Kitwe, Zambia E-mail: gillian.kabwe@cbu.ac.zm

Judith Kamoto Lilongwe University of Agriculture and Natural Resources Lilongwe, Malawi E-mail: jkamoto@luanar.ac.mw

P.P. Krishnapriya Duke University Durham, North Carolina, USA E-mail: krishnapriya.perumbillissery@duke.edu

Patricia M. Kristjanson Consultant, World Bank and CGIAR Washington, D.C., USA E-mail: pkristjanson@worldbank.org

Patrice Bigombe Logo University of Yaoundé II Soa, Cameroon E-mail: patricebigombe@hotmail.com

Stephanie Mansourian Consultant, Environment and Development Gingins, Switzerland E-mail: smansourian@infomaniak.ch Samuel Mawutor Oregon State University Corvallis, Oregon, USA E-mail: samuel.mawutor@oregonstate.edu

Blaise-Pascal Ntirumenyerwa Mihigo

University of Kinshasa Kinshasa, Democratic Republic of Congo E-mail: blaisepascalmihigo@gmail.com

Daniel C. Miller University of Illinois at Urbana-Champaign Urbana, Illinois, USA E-mail: dcmiller@illinois.edu

Doris N. Mutta The African Forest Forum (AFF) Nairobi, Kenya E-mail: D.Mutta@cgiar.org

Peter Newton University of Colorado Boulder Boulder, Colorado, USA E-mail: peter.newton@colorado.edu

Johan A. Oldekop University of Manchester Manchester, UK E-mail: Johan.Oldekop@manchester.ac.uk

Symphorien Ongolo University of Göttingen Göttingen, Germany E-mail: songolo@uni-goettingen.de

Pablo Pacheco WWF and CIFOR Washington, D.C., USA E-mail: p.pacheco@cgiar.org

Nitin D. Rai Ashoka Trust for Research in Ecology and the Environment (ATREE) Bangalore, India E-mail: nitinrai@atree.org

Laura Vang Rasmussen University of Copenhagen Copenhagen, Denmark E-mail: lr@ign.ku.dk

Onja H. Razafındratsima South Dakota State University Brookings, South Dakota, USA E-mail: onja.razafındratsima@sdstate.edu **Casey Ryan** University of Edinburgh Edinburgh, UK E-mail: casey.ryan@ed.ac.uk

Laura Aileen Sauls University of Sheffield Sheffield, UK E-mail: l.a.sauls@sheffield.ac.uk

George Schoneveld Center for International Forestry Research (CIFOR) Nairobi, Kenya E-mail: G.Schoneveld@cgiar.org

Priya Shyamsundar The Nature Conservancy Arlington, Virginia, USA E-mail: priya.shyamsundar@tnc.org

Erin O. Sills North Carolina State University Raleigh, North Carolina, USA E-mail: sills@ncsu.edu

Conghe Song University of North Carolina at Chapel Hill Chapel Hill, North Carolina, USA E-mail: csong@email.unc.edu

Dietmar Stoian World Agroforestry (ICRAF) Bonn, Germany E-mail: d.stoian@cgiar.org

Kira A. Sullivan-Wiley Boston University Boston, Massachusetts, USA E-mail: kswiley@bu.edu

Terry Sunderland University of British Columbia Vancouver, British Columbia, Canada E-mail: terry.sunderland@ubc.ca

Joleen Timko Sylvae Research & Consulting Vancouver, British Columbia, Canada E-mail: joleen.timko@me.com

Christoph Wildburger International Union of Forest Research Organizations (IUFRO) Vienna, Austria E-mail: wildburger@iufro.org **Jianmin Xiao** Chinese Academy of Forestry Beijing, China E-mail: xiaojm@caf.ac.cn

Jennifer Zavaleta Cheek

University of Illinois at Urbana-Champaign Urbana, Illinois, USA E-mail: jzcheek@illinois.edu Yue Zhao University of Göttingen Göttingen, Germany E-mail: yzhao1@uni-goettingen.de



2nd meeting of the GFEP Expert Panel on Forests and Poverty in Nairobi, Kenya Photo © World Agroforestry (ICRAF)

Reviewers

Sofia Ahlroth World Bank Group, PROFOR Washington, D.C., USA E-mail: sahlroth@worldbank.org

William L. Ascher Claremont McKenna College Claremont, California, USA E-Mail: william.ascher@cmc.edu

David Conte Food and Agriculture Organization (FAO) Rome, Italy E-Mail: David.Conte@fao.org

Julia P. G. Jones Bangor University Gwynedd, Wales, UK E-mail: julia.jones@bangor.ac.uk

Thais Linhares-Juvenal Food and Agriculture Organization (FAO) Rome, Italy E-mail: Thais.LinharesJuvenal@fao.org

Adrian Martin University of East Anglia Norwich, UK E-mail: adrian.martin@uea.ac.uk Alexandre Meybeck

Center for International Forestry Research (CIFOR-ICRAF) Rome, Italy E-mail: A.Meybeck@cgiar.org

Sheona Shackleton University of Cape Town Cape Town, South Africa E-mail: sheona.shackleton@uct.ac.za

Carsten Smith-Hall University of Copenhagen Copenhagen, Denmark

E-mail: cso@ifro.ku.dk

Andrew Taber Food and Agriculture Organization (FAO) Rome, Italy E-mail: Andrew.Taber@fao.org

Bhaskar Vira University of Cambridge Cambridge, UK E-mail: bv101@cam.ac.uk

GFEP Donors







 Federal Ministry Republic of Austria Agriculture, Regions and Tourism

CPF Members



a

















IUFRO Headquarters

Marxergasse 2 1030 Vienna, Austria Tel + 43 (0)1 877 01 51-0 Fax +43 (0)1 877 01 51-50 Email office@iufro.org www.iufro.org